## **INFORMATION TO USERS**

This manuscript has been reproduced from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality 6" x 9" black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

Bell & Howell Information and Learning 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 800-521-0600

**I MI**<sup>®</sup>

•

# THE SERIAL VERB CONSTRUCTION PARAMETER

Osamuyimen Thompson Stewart

Department of Linguistics McGill University, Montréal

A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements of the degree of

DOCTOR OF PHILOSOPHY

© Osamuyimen Thompson Stewart March 1998



# National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your file Votre reference

Our file Notre reférence

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission. L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-44600-X



Trust in the Lord with all your heart, And Do not lean on your own understanding. In all your ways acknowledge Him, And He will make your paths straight.

Proverbs 3: 5-6

#### ABSTRACT

This thesis investigates Serial Verb Constructions (SVCs) where two or more finite verbs along with their complements occur in a single clause without any form of coordination or subordination. Two basic questions are addressed: (a) what types of SVCs are there, and how are they to be distinguished from other similar constructions ? (b) what is the parameter that allows a language like Èdó to have SVCs, and not English or French ?

It is argued that true SVCs are those in which the verbs share internal as well as external arguments. Based on a battery of syntactic tests, it is proposed that there are two kinds of SVCs with distinct syntactic structures: resultative and consequential. This is contrary to the unified approach in previous works such as Baker (1989) and Collins (1997). It is argued that resultative SVCs are constrained to two verbs, the second of which is typically unaccusative, and they assign their internal theta roles to a single object-true internal argument sharing. Consequential SVCs are less constrained, and involve sequences of transitive verbs, with internal argument sharing realized via an empty category, *pro*, as the object of the second verb. Both kinds of SVCs contain two functional heads: an E(vent) head that binds the events denoted by the verbs which it dominates, and a Voice head that licenses the Agent of the events expressed by those verbs.

Some other constructions that have been classified as SVCs turn out to involve two separate clauses, each with their own E(vent) and Voice heads: covert coordinations, modal-aspectual verb constructions, and instrumental constructions. A syntactic structure for each of these non-SVCs is proposed.

Based on Pollock's (1989) approach to verb raising and the checking theory of Chomsky (1993, 1995), it is argued that SVCs can occur in languages where Tense (or other Infl categories) does not need to be checked. The parameter is as follows: non-SVC languages are those in which Infl must check features with the verb { English, French, Igbo, Chinese etc.}, versus SVC languages where it doesn't { Èdó, Yoruba, Ewe, Akan etc.}

#### Résumé

Cette thèse examine les constructions à verbes en série (SVC) où deux verbes fléchis ou plus, accompagnés de leurs compléments, se trouvent dans un seul syntagme sans aucun élément de coordination ou de subordination. Deux questions principales sont traitées: (a) quelles sortes de SVC y a-t-il, et comment les distinguer d'autres constructions pareilles? (b) quel paramètre permet des SVCs dans une langue telle que l'èdó mais pas en anglais ou en français?

Je propose que les vraies SVC sont celles dans lesquelles les verbes partagent des arguments internes et externes. À l'aide d'un ensemble de tests syntaxiques, je montre qu'il existe deux sortes de SVC avec des structures syntaxiques distinctes: résultatives et conséquent. Ceci va contre les approches unifées telles Baker (1989) et Collins (1997). Je propose que les SVCs résultatives sont contraintes par deux verbes, le deuxième étant typiquement inaccusatif, et qu'elles assignent leurs rôles théta internes à un seul objet, ce qui est un vrai partage de l'argument interne. Les SVC conséquent sont moins contraintes et comprennent des séries de verbes transitifs, où le partage de l'argument interne se réalise par l'existence d'une catégorie vide, *pro*, qui sert d'agent au deuxième verbe. Les deux sortes de SVC contiennent deux têtes fonctionelles: une tête É(vénement) qui lie les événements exprimés par ces verbes et un tête Voix qui autorise l'Agent des événements exprimés par les verbes en question.

D'autres constructions qu'on a classifiées commes des SVC finissent par comprendre deux syntagmes différents, chacun ayant ses propres têtes É(vénement) et Voix: des coordinations indirectes ('covert') des constructions verbales modalesaspectuelles, et des constructions instrumentales. Une structure syntaxique pour chacune des ces non SVCs est proposée.

En me basant sur l'approche de Pollock (1989) concernant la montée des verbes et la théorie de vérification de Chomsky ('checking theory', 1993, 1995), je propose que les SVCs peuvent exister dans des langues où Temps (ou d'autres catégories inflectionnelles) n'a pas besoin d'être vérifié. Le paramètre est le suivant: les langues non SVC sont celles où INFL doit vérifier ses traits avec le verbe (le français, l'igbo, le chinois, l'anglais, etc.); les langues SVC sont celles où ceci ne se fait pas (l'èdó, le yoruba, l'ewe, l'akan, etc.)

#### Acknowledgments

I have been truly blessed to have been surrounded by so many good and warm people during the course of my study. My academic advisor and friend, Mark Baker, is the one person that I would really love to begin this thanksgiving parade with, not because of the time-honored tradition of thanking your advisor first, but because Mark has left an impression on me and my family; your friendship and advice over the last three and a half years have been like a traveler's map, the pilgrim's staff, the pilot's compass, the soldier's sword, and the Christian's character. A large part of my linguistic education has been accomplished while meeting with Mark on a weekly basis; he is responsible for any steps that I have made away from my tendency to oversimplify issues or give up when the data gets messy. In light of Mark's commitment to finding micro and macro parameters in language, it goes without saying that the good of this thesis is the direct result of his suggestions and insights. A glimpse of his rare combination of intelligence with maturity comes from his non-prejudiced reaction to some of the roughest prose that I have churned out. At such times, he has always pushed me to ask the harder question, make stronger prediction, and of course he could always point out two or three other languages with related phenomena. Moreover, I must not fail to acknowledge Linda, Kate, Nicholas, and little Julia who are all part of The Baker heritage that have been so close and warm to my family over the years.

My very special thanks to Lisa deMena Travis who nurtured me in my first year at McGill and has constantly been supporting me ever since. I find your ability of on-the -spot assessment of any analysis or proposal astounding, and I have benefited immensely from this gift. You have very lovely kids, Reuben and Hannah, who have showed so much love and kindness to my daughter.

Nevertheless, I also must acknowledge all the Professors in the department of linguistics at McGill from whom I have benefited in terms of learning and, or whose smile or hello during difficult times have brought me cheer: Glyne Piggott, Michel Paradis (by the way, the offer of a Winter TAship is gratefully acknowledged), Myrna Gopnik, Lydia White, Heather Goad, Nigel Duffield (who always had time to entertain my concerns), Brendan Gillon (guess who loves formal semantics now), Eva Kehayia (for letting me be the TA that I can be), and Nicole Domingue. Special thanks also to Lise Vinnette and Linda Suen for putting the human touch to the business of administration.

Jeff Gruber started it all several years ago as my syntax teacher and no doubt some of the ideas from twelve years ago have become polished in different parts of this thesis. I must also acknowledge the following people for being part of my foundation days in linguistics in Benin and Cambridge: Rebecca Agheyisi, Tim Omoruyi, Betram Okolo, Eddy Erhagbe (and his family), Matt Aikhionbare, E. Imasuen, Grace Masagbor, Victor Omozuwa, Airen Amayo, Greg Nwoye, M.C.K. Uwajeh, Ron Shaefer, Ota Ogie, Alex Tait, P.H Matthews, T. Moore, and Prof. Raymond Elaho.

Outside of the McGill community, I have enjoyed very good academic interactions with Chris Collins, Victor Manfredi (Life Senior Fellow), Rose-Marie Déchaine, Sam Mchombo, Peter Ihionu, Vicki Carstens, John Bowers, Jonathan Bobalijk, Rich Campbell, Yiwola Awoyale, Olasope Oyelaran, and Claire Lefebvre.

I would also like to acknowledge some students with whom I have fellowshiped socially and academically: everyone in the syntax project (past and present), Mengistu & Amanda Amberber, Martine Kessler, Sharon Rose, Roumyana Slabakova (thanks for loaning me a copy of your thesis, it was of great help), Silvina Montrul, Ileana Paul, Phillipe Prevost, Toni Pi, Mikael Vinka, Hiro Hosoi, Mika Kizu, Hidekazu Tanaka, Takako Kawasaki, Hitoshi Akahane Joyce Garavito, Jose Bonneau, Miwako Uesaka, Steve Sheppard, Jeff Steel (who helped me with the French translation of the abstract of this thesis and some last minute editing), Vanessa Taler, Asya Pereltsvaig, Nishiyama Kunio, Desiré Hongues, and Evan Mellander.

I gratefully acknowledge funding from the McGill Faculty of Graduate Studies and Research for the award of a Major Fellowship (Max E. Binz), the SSHRCC grant 410-950970 to Mark Baker and the FCAR of Quebec grant 94ER0578 to Baker, Travis and Lefebvre. In addition, I also acknowledge Travel awards from the Post Graduate Students Society (PGSS) and the McGill Faculty of Graduate Studies and Research (Alma Mater Student Travel grant).

My life truly illustrates the African parable; it takes a whole village to raise a child, and so I would like to thank the following family members who, in being themselves, have helped to make me who I am today: my dad Pa Daniel Stewart-Orumwense (1912-1982), my mum Ayo (òghòghò) Stewart-Orumwense, Lemmy Osagie Stewart (1949-1995), Harrison Stewart, Kenneth Stewart, Dr. and Mrs. Orobator, Dr. and Mrs. Edigin, Osaretin and Osarugue Orumwense (both of whom have showed me and my family God's love), Esosa Stewart, Nosa Stewart, Mimi, Nosa, Tony, Efosa (the Osemwengies) and Mrs. Kate Osemwengie (a very loving mother-in-law). I must also acknowledge those friends who have been like brothers and sisters to me over the years: Jite Eferakorho (to whom I pass the baton), Joe Ehiwe, Andrew Ehiwe, Ralph and Cynthia Ogom, Tony Ereyi, Raphael and Winifred Ereyi, Ejike Eze, Vincent and Mabel Osaghae, Mr. and Mrs. U.L. Irowa, Mrs. Mabel Guobadia and the Guobadias, Peter Ediagbonya, Mrs. E. Atekha, Juliet Iserhienrhien, Edwin Obazuaye, Solomon and Gloria Obano, Tom and Carol Airhiavbere, Jane Airhiavbere, Georgina Murphy, Mohammed Okunseri, Owen Bello, Uxden Ihimekpen, Tesmos Woghiren, Dennis Ediae, Chris Inneh, Amen and Osayomore Eguakun, Tony Uponi, Osasumwen Osaghae, Dickson Oriakhi, Bennedict Osula, Osamu Edebiri, Colleen Ehiwe, Babson Salami and family, Augustine Ogunsuyi, Chikwendu Ibebunjo, Emily Feoshi, my pastor Don Mann with whom I share the D.R.E.A.M, my head Usher Kevin Gundry-White (and the Evangel Pentecostal Church community).

I also want to specially thank the Obá of Benin, Omo nObá Érédiá!wá for his encouragement and support.

My final acknowledgment comes with a short story. In the week prior to handing in this thesis, both my daughter and my wife fell ill, and consequently it seemed like my whole world was crashing. However, this episode brought out something that we will all ever treasure: sacrifice, compromise, and love can conquer all. I can never fully express my thanks to my wife Frances, and the best daughter in the world, Isoken (which means satisfaction) except to say that I love you two. Both of you complement and complete the person in me; this one is for you. However, by all means the greatest gratitude goes to my Lord and Savior Jesus Christ who has saved me, and showed me that at my moments of brokenness, He will always wrap His hands around me. To you O Lord, I give all the praise and Glory. Whatever of this work shall stand, shall stand to His Glory.

#### Note on Orthography and Tones

The orthography employed in the writing of Èdó in this dissertation is generally consistent with the system recommended for the language in 1974 by the Ad hoc Mid-West Language Committee. There are, however, a few modifications that have been introduced: (a) The use of mw for the labio-dental nasal approximant (cf. Agheyisi 1986). Thus, we can form near minimal pairs involving mw in  $\partial mwan$  "here" and its oral counterpart vb in  $\partial vba$  "there".

(b) Nasal vowel is consistently transcribed by an 'n' after the (nasal) vowel (cf. Stewart 1992) to represent nasality (a key feature of Èdó phonology) in the orthography. For example,  $\dot{e}mw\dot{a}$  is written as  $\dot{e}mw\dot{a}n$ , and  $\dot{a}m\dot{e}$  "water" is written as  $\dot{a}m\dot{e}n$ .

(c) For typographical convenience, the two mid lax vowels are represented by underlying [ <u>e</u>, <u>o</u>] rather than underdots (except in the spelling of Èdó where the capitalized È is to be understood as <u>È</u>).

(d) I will fully tone-mark all the lexical items and the sentences, thus departing from the tradition in which tone marking is confined exclusively to those forms which might remain ambiguous without the indication of tone (cf. Report of the Seminar on Èdó Orthography in 1974, Agheyisi 1986, etc.).

(e) Only two tones are marked namely, " ´ " for high tone and " ` " for low tone; intermediate or mid tone is left unmarked. In most cases, such mid or intermediate tone is realized as a downstep on the following tone (low or high). Downstep is represented in the orthography by an exclamation mark [!] (cf. Agheyisi 1986, Melzian 1937). Thus, for example, the word *ógheghè* "edible berries" in Agheyisi (1986) will be written as *óghé!ghè* which illustrates a downstepped low tone.

(f) I adopt the standard assumption that there is a distinction between lexical and grammatical tones (cf. Agheyisi (1986, 1990), Omoruyi (1991), etc.). All nouns, adjectives etc. have [fixed] <u>lexical</u> tones, while only verbs which are inherently toneless (maybe with default tones) bear <u>grammatical</u> tones (tense and possibly aspect, Amayo

1975, Omoruyi 1991, etc.). Put simply, a low tone on a monosyllabic verb indicates nonpast tense, while a high tone expresses past tense. However, it seems that there is a further distinction that can be made based on the syllable structure of the verb (cf. Wescott 1963).<sup>1</sup> Basically, I will adopt the following spelling conventions for verbs:

- (i) One syllable verb with a single mora, e.g. bo 'build'
- (ii) One syllable verb with two moras, e.g. boo 'comfort, console'
- (iii) Two syllable verb with two moras, e.g. kpòló 'sweep'

## The **Èdó** Alphabet

а	b	d	e	<u>e</u>	f	g	gb	gh	h	i
k	kh	kp	1	m	mw	n	ο	<u>o</u>	р	٢
rh	rr	S	t	u	v	vb	w	у	Z	
			A	BBRE	VIATIO	NS				
Fut.				Futu	re tense i	morphe	me			
IMP				Impe	erative (A	(spect)				
INFL				Infle	ction (su	bjunctiv	ve)			
INCF	)			Ince	ptive mar	ker				
Iter				Iterat	tive mark	ær				
Foc.				Focu	is marker	r				
Cop.				Сорі	ula					
cl.				Clitic	2					
subj.	cl.			Subje	ect clitic					
obj. c	:1.			Obje	ct clitic					
neg.				Nega	ition mor	pheme				
Comp	<b>)</b> .			Com	plementi	zer				
PRG.				Prog	ressive (	Aspect)	)			

 $<sup>^{1}</sup>$  I must admit that the issue of tense tones on verbs is a complicated one which deserves more attention than the scope of this thesis allows (cf. Amayo 1975, 1976).

# **Table of Contents**

Page

Title page		0
Dedication		i
Abstract		ü
Résumé	***************************************	iii
Acknowledgn	nents	iv
Note on Ortho	ography and Tones	viii
Chapter 1:	In Search of Serial Verb Constructions	1
1.1	Introduction	1
1.2	Previous Research	5
	1.2.1 Early Research	5
	1.2.2 Principles and Parameters Approaches	8
1.3	On the Definition of SVCs	13
	1.3.1 Interpretation/Verb Sequencing Constraints	14
1.4	Conclusion	19
1.5	Organization of the thesis	19
1.6	Significance of the Study	21
	<b>C</b>	
Chapter 2:	Resultative vs. Consequential SVCs, and CCs	23
2.1	Introduction	23
2.2	Distribution and Licensing of Manner Adverbs	23
	2.2.1 I-type Adverb before First Verb	30
	2.2.2 I-type Adverb before Second Verb	33
	2.2.3 N-type Adverb after First Verb	47
	2.2.4 N-type Adverb after Second Verb	52
2.3	Distribution of Adjuncts: Locative PP	56
	2.3.1 Locative PP after First Verb	57
	2.3.2 Locative PP after Second Verb	59
2.4	Consequences of Adverb and PP Placements	62
2.5	Object Sharing	64
	2.5.1 Evidence for Empty Category: 'tòbórè particle'	65
	2.5.2 Null pro Licensing: unaccusativity vs. transitivity	69

2.6	Subject NP				
	2.6.1 Interpretation and Analysis of 'Subject' and 'Agent'				
	2.6.2 Distribution of Subject Pronoun				
	2.6.3 Subject-oriented Interpretation of 'tobóre'				
2.7	Phonology-Syntax Interface				
	2.7.1 E head and Relative Tones				
	2.7.2 Wh-extraction and Relative Tones				
2.8	Resultative SVCs, Aspect, and Temporal Adverbs				
	2.8.1 Differences in Aspectual Properties				
	2.8.2 Process-Activity Restriction on First Verb				
	2.8.3 Event Delimiter				
	2.8.4 Temporal Adverb				
2.9	Conclusion				
Chapter 3:	Predicate Cleft and Event Quantification in SVCs				
3.1	Introduction				
3.2	Èdó Predicate Cleft Construction				
	3.2.1 Evidence for Category Conversion				
3.3	The Status of the Derived Nominal				
	3.3.1 Semantic Interpretation				
	3.3.2 Cognate Object as Event Argument				
3.4	The Syntax of Predicate Clefts				
	3.4.1 On Event Quantification				
3.5	Predicate Cleft and SVCs				
	3.5.1 Resultative SVCs				
	3.5.2 Consequential SVCs and CCs				
3.6	Conclusion				
Chapter 4:	Double Objects and Object Sharing in SVCs				
4.1	Introduction				
4.2	SVCs and DOCs				
	4.2.1 DOCs and the Associative Construction				
	4.2.2 DOCs in Resultative and Consequential SVCs				
	4.2.3 More on DOCs and SVCs				

4.3	On the Syntax of DOCs			
	4.3.1 Asymmetry between Theme and Goal Objects			
	4.3.2 Double Objects and the ECP			
	4.3.3 Underlying versus Derived Objects			
4.4	Conclusion			
Chapter 5:	Cross-Linguistic Extensions of Analysis			
5.1	Introduction			
5.2	On Structurally Ambiguous SVCs (Yoruba, Ewe, Edó)			
5.3	On "Splitting Verbs' as Resultative SVC (Yoruba, Èdó)			
5.4	Igbo			
	5.4.1 Resultative V-V Compounds			
	5.4.2 Inherent Verb Complements			
	5.4.3 Verb Inflection and Multi-event Constructions			
5.5	Chinese			
	5.5.1 Resultative V-V Compound			
	5.5.2 Multi-event Construction			
5.6	Conclusion			
Chapter 6:	Serial Verb Construction Parameter			
6.1	Introduction			
6.2	SVCs and Secondary Predicates			
6.3	Towards the SVC Parameter: AP vs. VP Resultatives			
6.4	The Serial Verb Parameter			
	6.4.1 The Bare Stem Condition			
	6.4.2 Verb Raising-to-Infl			
	6.4.2.1 Re: Igbo Resultative V-V Compounds			
	6.4.2.2 English AP Resultative SP			
	6.4.2.3 Consequential SVCs vs. CCs			
	6.4.2.4 Consequences: Analysis of Negation			
6.5	'Rule R' and Tense Matching in SVCs			
6.6	Conclusion			

Chapter 7:	Re-analysis of "Serial Verb Constructions" 2			249	
7.1	Introd	uction			
7.2	2 Modal-aspectual Verb Construction			250	
	7.2.1	7.2.1 Apparent Contrast with SVCs: Infl vs. E projection			
	7.2.2	A separate Infl projection before Second Verb 2			
	7.2.3	yá as Evidenc	e for Embedded Clause	254	
	7.2.4	Modal-aspect	ual Verb Restriction	262	
7.3	Restructuring in Modal-aspectual Verb Construction				
	7.3.1 Evidence for Infinitival Complement				
	7.3.2	Evidence for	Restructuring	267	
		7.3.2.1	Verb Raising and Object Cleft	268	
		7.3.2.2	Predicate Cleft	273	
		7.3.2.3	The Class of Restructuring Verbs	280	
	7.3.3	Additional Sy	ntactic tests for Restructured Clause	282	
		7.3.3.1	There are two Subject Positions	283	
		7.3.3.2	I-type Adverb before First Verb	286	
		7.3.3.3	Iterative versus Imperfective	288	
		7.3.3.4	N-type Adverb as VP modifier	292	
		7.3.3.5	Locative PP	294	
7.4	On Instrumental Constructions			295	
	7.4.1	Motivating Su	bject Control	298	
	7.4.2	Infl before Se	cond Verb	299	
	7.4.3	PRO Subject	in Embedded Clause	303	
		7.4.3.1	'tòbórè' Particle	303	
		7.4.3.2	Purposive Clauses	305	
	7.4.4	On the Syntax	of 'Use'-type Instrumental Construction	307	
		7.4.4.1	I-type Adverb	310	
		7.4.4.2	N-type Adverb	311	
		7.4.4.3	Locative PP	313	
		7.4.4.5	Predicate Cleft	313	
		7.4.4.6	Verb Raising	316	
7.5	Conclu	usion		318	

Chapter 8:	SVC Consequences	320
8.1	Introduction	320
8.2	On Resultatives vs. Consequential SVCs, and CCs	320
8.3	On the Serial Verb Construction Parameter	325
	8.3.1 True SVCs vs. other Surface Verb Constructions	328
8.4	Conclusion	329
<b>References</b>		330

#### Chapter One

## In Search of Serial Verb Constructions

#### 1.1 Introduction

The serial verb construction (SVC) is one of the better studied phenomena associated with West African linguistics (cf. Christaller 1875, Westerman 1930, Ansre 1966, Bamgbose 1974, Sebba 1987, Baker 1989, 1991, Lefebvre 1991, Awoyale 1987, 1988, Manfredi 1991, Déchaine 1993, Collins 1997, Campbell 1989, Joseph and Zwicky 1990, Bodomo 1993, Cormack and Neil 1994, etc.).<sup>1</sup> However, in spite of over one century of grammatical analysis the SVC is still an ill-defined and often misinterpreted phenomenon. The following are typical examples from the Èdó language of the sort of things that are often called SVCs:<sup>2</sup>

- a. Òzó dé èvbàré rhié nè Ìfuèkò
  Ozo buy food give to Ifueko
  'Ozo bought the food and gave it to Ifueko.'
  - Úyì hìá lé èvbàré
    Uyi try cook food
    'Uyi tried and cooked the food.'
  - Èsósà kòkó àdésúwà mòsé
    Esosa raise Adesuwa be.beautiful
    'Esosa raised Adesuwa to be beautiful.'

<sup>&</sup>lt;sup>1</sup> These references are not intended to be an exhaustive list of all papers on SVCs, but it is a sampling that attempts to reflect the diversity and range of work on so-called SVCs. Other relevant references are mentioned in the text.

 $<sup>^2</sup>$  Èdó is famous world-wide for its Art works and for having one of the most enduring and greatest Monarchies south of the Sahara (AD 900 to the present day) (cf. Egharevba 1954, Igbafe 1979 etc.). It is a close neighbor geographically and genetically to one of the most discussed languages in the serial verb literature, Yoruba. Together with Igbo, Ewe, and Akan, they all belong to the Kwa group of the Niger-Congo Family (Greenberg 1963) also classified as Benue-Kwa (Elugbe and Williamson 1977, Bennett and Sterk 1977). More specifically, Edó is spoken in Edó State in the midwestern part of Nigeria by over a million and a half speakers (census 1991). In the older literature, it is also referred to as Benin or Bini. The label Edó is itself saddled with different interpretations both in its lay (ethnic) and technical (linguistic) uses. For example, Egharevba (1954) claims that the indigenous name for Benin City (the capital of the Edós) given it by one of its rulers (Óbá Éwuàré, 1440-1473) is Edó. However, according to Agheyisi (1986), '... further ambiguity was introduced .... into the reference of the term Edó as a linguistic label, when linguists decided to use the designation for the group of historically related languages and dialects spoken in various communities within and around the former Benin kingdom. To avoid this confusion, Edó is now used for the single language spoken by the Èdó people excluding its close neighbors, while the term Edoid now refers to the group consisting of Edó and these neighbors (Elugbe 1979). Within this special group Edó is classified as a North-Central Edoid language and it is an SVO language.

- d. Isoken yá ábé fián émió!wó
  Isoken take knife cut meat
  'Isoken used the knife to cut the meat.'
- e. Ènó!sá rhié ùkéké gb<u>é</u>n èbé Enosa take pen write book 'Enosa wrote a note with a pen.'
- f. Òzó lé èvbàré ré
  Ozo cook food eat
  'Ozo cooked the food and ate it.'
- g. Àbi<u>é</u>!yúwà hìín èrhán kpàán àlìmó
  Abieyuwa climb tree pluck orange
  'Abieyuwa climbed the tree and plucked an orange.'

These so-called SVCs are often given sub-labels which indicate the general semantic meanings that the verbs convey. For example, benefactive (1a), manner (1b), result (1c), instrumental (1d), purposive or instrument (1e), and concomitant/simultaneous/sequential (1f,g). The sentences in (1) roughly illustrate the range of SVCs that have been discussed in many other languages.<sup>3</sup>

There are four factors that seem to impede the development of an adequate theory of verb serialization, despite the considerable attention it has gotten. First, and perhaps the biggest problem with the term SVC, is the fact that in the course of the past century there is no systematic restrictive notion of serial verb phenomena nor of the parameter that allows some languages to have this sort of construction and not others. What obtains really are

<sup>&</sup>lt;sup>3</sup> The SVC phenomenon loosely defined as in the text has been acsribed to comparatively superficially similar constructions in a wide variety of languages. In this regard, according to Sebba (1987), there are references in the West African group to 'serial verbs' or possibly similar phenomena in Welmers (1973) for Efik, Junkun, Yoruba (see also Awobuluyi 1973, Bamgbose 1974), Nupe (cf. Hyman 1971), Fante and Akan (cf. Christaller 1875, Stewart 1963) Yatye (cf. Stahlke 1970), Ijo (cf. Williamson 1963), Kru (Givon 1975), Ewe (cf. Westermann 1930, Ansre 1966) Ga (cf. Lord 1973) Fon (cf. Lord 1973). It is crucial to observe the fact that there is no mention of Igbo ( a core member of the West African Kwa group), and this is one issue that will be addressed in this thesis. In the East Asian group, references include Mandarin Chinese (cf. Li and Thompson 1973), Vietnamese, Thai, and Mon-Khmer (cf. Schiller 1991). Serial verbs have also been attested in the Caribbean Creoles such as Jamaican (cf. Bailey 1966), Sranan and Saramaccan (cf. Sebba 1987), Papiamentu (cf. Bendix 1972), Haitian (cf. Wingerd 1977). Furthermore, references to SVC languages also include New Guinean languages (Foley and Olson 1985) and Central American languages (cf. Craig and Hale 1988). Given the trend to find SVCs in a wide variety of languages, I do not assume that this is the entire range of languages but this array serves to provide a picture of the scope of languages for which there is the need to clearly and systematically work out syntactic tests that can be used to define SVCs.

descriptive definitions of SVC that lists some basic properties such as: (a) two or more verbs and their arguments co-occur without any conjunction, (b) these verbs must share the same subject, and (sometimes) the same object, (c) there is usually a single tense/aspect specification for the verbs. Consequently, in the absence of any systematic definition or set of restrictive criteria for determining SVCs there are claims ranging from it being a universally available phenomenon (cf. Déchaine 1993) to it being parameterized (Baker 1989, Collins 1997 etc.). The weakness of these claims is that there is no assurance that these superficial criteria pick out a natural class of structures. For the most part, clear, concise, systematic, and replicable tests for deciding what serial verbs are have not been worked out so far.

Second, it has been assumed that there is very little morphological inflection in many of the languages that have the serial verb phenomena. As a result, many analyses of so-called SVCs fail to systematically provide obvious language-internal evidence for verbal status. Thus, we cannot tell for sure if the verbal sequence called SVCs under the usual description actually lacks any marker of coordination or subordination. This observation can be illustrated with the sentence in (1b) in which the first verb *hiá* 'try' is a Control verb in English but there is no obvious morphological inflection in (1b) that suggests the same thing in Èdó. Furthermore, it can be difficult to tell if the things that are claimed to be verbal are indeed verbs in such languages. This is compounded by the fact that it is difficult to find non-syntactic criteria that distinguishes verbs from say adverbs or particles (cf. Ansre 1966, Bamgbose 1974, Awobuluyi 1973, Agheyisi 1986 etc.).

Third, based on the lack of overt morphology it has been difficult to differentiate SVCs from constructions in which there are sequences of surface verbs that involve clausal embedding, like Control constructions or causatives etc. The indication that there is more to the so-called SVCs than meets the eye comes from the fact that there are differences in the linear order of the verbs and their arguments, as summarized in (2).

(2)	а.	NP V NP V	(compare 1c and 1f)
	b.	NP V NP V NP	(compare 1a, 1e, and 1g)
	c.	NP (V?) NP V NP	(compare 1d)
	d.	NP V V <sup>4</sup>	
	e.	NP V V NP	(compare 1b)

Since there are no markers of coordination or subordination between the verbs in the different templates in (2), the question is which one is underlyingly SVC.

Fourth, and rinally, there is a recent move that appears to disregard the descriptive definition of SVCs as constructions with two or more verbs without any marker of subordination and coordination. Thus, in languages such as Japanese (Nishiyama 1995), Korean (Lee 1993), Marathi (Pandharipande 1990), Gullah (Mufwene 1990), the significance of certain particles with clear conjunctive meanings have been down-played in the wave of analyzing SVCs cross-linguistically. This observation can be illustrated by the following sentences:

- (3) a. John-ga hammer-o tot-te Bill-o nagut-ta (Japanese, Nishiyama=8a)
  -Nom -Acc take-TE -Acc hit-Past
  'John took the hammer and hit Bill.'
  - b. John-wa boosi-o nui-de Mary-ni aisatusi-ta ( "=54)
    -Top hat-Acc take-TE -Dat greet-Past
    'John took off his hat and greeted Mary.'
  - c. ku-nun koki-lul kwe-e mek-ess-ta (Korean, Lee=18) he-Top meat-Acc broil-L eat-Past.Dec. 'He broiled the meat and ate it.'
  - d. ku-nun kang-ul heyemchi-e kenn-ess-ta ("=19) he-Top river-Acc swim-L cross-Past-Dec. 'He swam across the river.'

<sup>&</sup>lt;sup>4</sup> Common examples of this order include sequences of double-unaccusative verbs such as 'fall-break' or 'push-fall' (see the discussion of the sentences in 11 below).

In Japanese, one of the functions of the TE morpheme has been analyzed as conjunction (Uesaka 1996) while in Korean the -e morpheme has been argued to be INFL (Choe 1988) or Comp (Lee 1976), Yang 1976). However, such meanings are often ignored or reinterpreted by those who analyze sentences like (3) as SVCs, for example Nishiyama (1995), Lee (1993), Mufwene (1990) amongst others. Thus, overt conjunctions, covert coordinations (parataxis), and sentences involving particles that imply clausal embedding such as purposives or instruments are treated as SVCs, and this makes it difficult to have a restrictive definition of the notion of SVCs.

In the light of the foregoing discussion, the aims of this thesis are two-fold; (a) to provide clear and systematic tests based on robust empirical evidence that can distinguish the various kinds of SVCs from one another and from other constructions that appear on the surface to be sequences of verbs, (b) to establish and formulate the correct parameter that allows a language such as Èdó to have SVCs but not a language like English.

#### **1.2** Previous Research

As a way to situate the focus of this thesis, I will only provide a brief general review of the way in which the literature on SVCs has developed and what sorts of questions about the phenomenon have been addressed at different times.<sup>5</sup> More particular literature will be discussed as the occasion arises throughout the dissertation. In order to keep the review simple and focused, I propose to divide the discussion into two phases as follows.

## 1.2.1 Early Research

The earliest known description in West Africa of the serial verb phenomena is Christaller (1875) which observed that it is possible for "two or more verbs, not connected by conjunctions to have the same subject". Furthermore, it was proposed that these

<sup>&</sup>lt;sup>5</sup> Sebba (1987) has a thorough and near exhaustive review of the literature on SVCs.

sentences fall into two types: (a) <u>Essential combinations</u>, are those in which one verb is the principal, and another is an auxiliary verb, supplying, as it were, an adverb of time or manner, or forming or introducing a complement or adjunct; or the second verb is supplemental, forming a part of a verbal phrase. In these cases, the eventualities expressed by the two verbs are simultaneous and in an internal or inseparable relation or connection (Sebba 1987:6). On the other hand, (b) <u>Accidental combinations</u> are those where two or more predicates (verbs with or without complements or adjuncts) express different successive actions, or a state simultaneous with another state or action, but happen to have the same subject.

Since Christaller's grammar described the Twi language that is spoken by the Asantes and Fantes of West Africa, the first general outside impression was that the serial verb phenomena is localized to the languages of West Africa and especially to those of the Kwa group (cf. Westerman 1930). These early investigators were not concerned with grammatical-theoretical issues such as why does a sentence have more than one verb or what form are such verbs in; their primary concern was in writing pedagogical grammars that could facilitate interaction between the natives and the foreigners.

However, matters changed with the dawn of the generative enterprise in Chomsky (1957) which formalized the description of a sentence in terms of the notion of a set of Phrase Structure (PS) rules that characterize linguistic competence. The basic idea, then, and even till now, is that a sentence has one main verb or to express the same thing in another way, a clause has only one finite verb. This definition of a sentence or clause made the serial verb phenomena look like some kind of 'oddity' that did not fall within the traditional Euro-centered approach of this framework at the time.

I believe that a shift occurred in the analyses of the serial verb phenomena based on Chomsky (1965). By this time, it was widely accepted that transformations could create structures and then delete portions of them when certain conditions were met. Thus, for those who were interested in the analysis of SVCs, it was no longer vital to pursue the issue of what allows a sentence to have two or more putative finite verbs that occur without an overt conjunction or subordinator in the phrase structure of some languages unlike English. In essence, by the transformational apparatus such verbal strings could very easily be formalized in terms of VP linearization or adjunction, i.e., coordination or subordination ( Schachter 1974, but see Stahlke 1974, for criticism). The question of what allows two or more verbs per sentence in serial verb languages was assumed to be a non-issue by the transformational approach.

Stewart (1963) who is credited with the first generative analysis, introduced an interesting angle to the serial verb puzzle. He was concerned with how to account for the fact that there are missing subjects and objects when two transitive verbs occur. Here, the massive power of transformations to delete recoverable parts of structure came in handy.

However, it became obvious that there was no simple solution to the analysis of missing subjects and objects and in fact that there was no escaping the question of what allows two or more verbs to appear in a clause. Consequently, the analyses of the functional (grammatical) status of the verbs in series as well as the relations between them assumed primary focus. Ansre (1966) observed that some verb-like elements do not have a full set of verbal properties even though they occupy the position that verbs would normally occupy. This launched a new era in the analyses of the serial verb phenomenon in which the primary goal was to establish the different kinds of serial verbs based on the functional status of the verbs as well as the relations between them (cf. Bamgbose 1974, Awobuluyi 1973, Agheyisi 1986b etc.). At this point, people started to realize that the serial verb phenomenon was not restricted only to the languages of West Africa but can also be found in East Asian languages (Li and Thompson 1973, Lu 1977 etc.) and some Creoles (Williams 1976, Jansen, Muysken, and Koopman 1978 etc.). On the whole, functionally based classifications came up with two to seventeen different types of serial verbs, depending on the researcher.

Subsequently, a corresponding trend was begun by those opposed to the syntactic analyses, and serial verbs were analyzed and distinguished in terms of generative semantics and/or the semantics of the verbs (cf. Stahlke 1974, George 1975, 1976, Oyelaran 1982 etc.). Now, fourteen to twenty different types were recognized, but in practice the semantic typology never replaced syntactic analysis. According to Oyelaran (1982, fn. 3) 'semantic classification can be refined *ad infinitum* without ever reducing serial constructions to a minimal number of syntactic primes'. This quotation sets the tone for the goals associated with phase two of the analysis of SVCs. This is in light of the gains in the Principles and Parameters framework (Chomsky 1981) in which one of the basic goals was to investigate remote languages and constructions as realizations of UG (universal grammar) with a view to sorting out parameters of variation.

#### **1.2.2.** Principles and Parameters Approaches

The central goal of generative analyses of SVCs since the early 1980s has been how to account for a more restricted notion of what can be an SVC? and to find out the parameters of variation. Each analysis makes a different proposal based on what it assumes to be the relevant core of so-called SVCs rather than systematically providing tests for what is a SVC. In order to provide a standard for comparison with my analysis and proposals in this thesis, I will now briefly summarize some of the relevant points and assumptions from other analyses done within the Principles and Parameters theory as they relate to different aspects of the traditional definition of SVCs.

On Subjects, all analyses of SVCs have assumed, based on the traditional notion, that there is a single one for each construction in (2). Thus, the only variation that can be observed is in terms of theoretically-based claims about the position in which the subject is generated: Specifier of TP, or IP as in (4b) or VP (internal subject hypothesis) as in (5). However, the analysis of Objects has been more controversial for those who claim that the core of true SVCs are those that involve object sharing.

Two general approaches can be identified. The first is 'true' internal argument sharing and this approach assumes that there is a single structural object NP that is assigned the internal theta roles of both verbs. Therefore, the Èdó sentence in (4a), which illustrates object sharing SVC, would have a structure like (4b) (cf. Baker 1989, 1991):



The basic claim that underlies Baker's (1989, 1991) structure in (4b) is that true internal object sharing arises because both verbs are within the projection of a single maximal projection (they are syntactic co-heads) and this allows them to assign their internal theta roles to a single structural position occupied by the object.<sup>6</sup>

The co-headed approach to object sharing contrasts with a second and now popular group which advocates that object sharing is mediated by an empty category. The analysis implied by this assumption is that there is no true internal argument sharing in the sense of Baker (1989,1991) but rather there are two separate projections of VP, each with its own object argument: an overt one associated with the first verb and a null one linked with the second verb. The null object is then coreferential with the first object. There are, however, different assumptions about the nature of the empty category in the second VP: it could be the trace of A-movement or an A-bar trace of a null operator (cf. Carstens 1988, Law and

<sup>&</sup>lt;sup>6</sup> Lefebvre (1991) proposes that SVCs of the kind in (1d,e) are also co-headed from a lexico-semantic point of view, but these SVCs are co-headed in terms of meaning representation, and are inserted in separate VP projections in the phrase structure.

Veenstra 1992, Campbell 1996, etc.), or a controlled *pro* (Collins 1997, but see Baker and Stewart 1997b). Concerning the former proposal, I assume Collins (1997) convincing arguments that the empty category in these SVCs is neither the trace of NP-movement nor an A-bar trace/ null operator.<sup>7</sup> Thus, based on Collins (1997) the representation for a sentence like (4a) is given in (5).



According to the structure in (5), there is really no true argument sharing as in Baker (1989) since in principle each verb assigns a theta role to a separate object NP and the fact that each verb heads a separate VP projection. Object sharing in this case refers to the control relationship between the overt object of the first verb and the null object of the second verb.

Déchaine (1993) has an account that is similar to Collins (1997) in terms of the double VP structure. However, the difference is that there is some notion of headedness that is associated with the verbs based on the analysis of bivalent projections. Therefore, the different kinds of SVCs would vary in terms of which verb is head. The explicit difference between Déchaine (1993) and Collins (1997), Baker (1989) is that her approach

<sup>&</sup>lt;sup>7</sup> Collins (1997) argues convincingly that the empty category cannot be an A-trace because the structure it derives would violate one generalization about A-movement--the Chain condition of Chomsky and Lasnik (1993)--which is that the tail of an A-Chain cannot be assigned Case, contrary to fact in Ewe where the post-position yi can assign Case. Furthermore, Collins argues that the empty cannot be the trace of A-bar movement because none of the wh-diagnostics can be observed. For example, the empty category in SVC is not an unbounded trace like the one in successive cyclic movement.

recognizes the fact that there are different kinds of SVCs while the latter adopts a unified approach.

Thus, one important similarity between Baker (1989, 1991) and Collins (1997) is that both give a unified analysis of transitive plus result SVCs like (4a) and transitive plus transitive SVCs like (1f) repeated here as (6).

(6) a. Òzó lé èvbàré ré
 Ozo cook food eat
 'Ozo cooked the food and ate it.'

For Baker, (6a) has the same structure as (4a) in which there is a single object without any empty category. This is illustrated in (6b).



Similarly, Collins (1997) assumes that the sentence in (6a) has the same structure and analysis as that in (4a) and so (6a) would have a structural representation as in (7) in which object sharing is mediated by an empty category, *pro*.



11

I should point out that one of the major results of this thesis will be to show that this is a spurious unification of the transitive plus result and transitive plus transitive SVCs. I will show that the two have quite different structures.

Turning now to the issue of what counts as SVCs (i.e., classification) the consensus seems to be that there should be a distinction between a class of 'true' SVCs and a class of covert coordinations (Sebba 1987, Déchaine 1993, Baker 1989, Collins 1997 etc.) the latter having essentially the same structure as VP coordinations in English, except there is no and between the VPs in this case. However, what exactly constitutes covert coordinations (CCs) or 'true' SVCs tends to vary very freely. Concerning the former, clear syntactic tests that will distinguish CC structures are lacking.<sup>8</sup>

In discussing the issue of what counts as 'true' SVCs there are those who have assumed that there is a discernible core which may be primarily the object sharing kinds (1a, 1c,d, and e) and marginally (1b,t) (cf. Baker 1989, 1991, Campbell 1996, Collins 1997 etc.). On the other hand, there are those whose analysis imply that primarily (only) the instrumental or manner kinds (1d and 1e) constitute 'true' SVCs (cf. Lefebvre 1991, Li 1991 etc.) . In part, I believe that the differences in choice of what is SVC reflects theoretical approaches to, and assumptions within, the Principles and Parameters theory. There are a few analyses that are in accordance with my proposed analysis such as Déchaine (1993:202) who propose a typology of four relevant kinds of SVCs based on the contrasts between Haitian, Igbo, and Yoruba . According to this account, Haitian has just the <u>dative</u> or <u>benefactive</u> type of SVC which is similar to the Èdó sentence (1a), while on the surface, at least, Igbo is said to have <u>instrumental</u> and <u>multi-event</u> or <u>sequential</u> kinds and these are similar to (1d) and (1f) respectively. Finally, Yoruba is said to have all four kinds of SVCs which also includes <u>result</u> as illustrated by (1c) in Èdó. The question, then,

<sup>&</sup>lt;sup>8</sup> The general idea that CCs are like VP conjunctions in English is seriously challenged by the fact that Coordinate Structure Constraint (CSC) effects as discussed in Ross (1967) are variable in CCs (see discussion of this point in p. 88, fn. 41 of Chapter two below).

is what are SVCs? The answer to this also illustrates another important result of this dissertation because I intend to show over a range of empirical and conceptual evidence that there are just two kinds of SVCs which must be distinguished from CCs. It is hoped that the discussion of this all-important distinction will clarify some of the overlaps and ambiguities from previous classifications.

One direct consequence of previous classifications of so-called SVCs is that it allows each account to make some specific proposals about the core question that SVC data poses for linguistic theory, namely, what is the parameter that allows such languages to have two or more putative finite verbs within a single clause? The proposals have ranged from the idea that it is the nature of INFL, e.g., that it can license multiple verbs (cf. Baker 1989, Déchaine 1993, Campbell 1996, Collins 1997), through a phrase structure parameter that allows co-headed VP structure (cf. Baker 1989, 1991) to SVCs being analogous to secondary predication (cf. Larson 1991). I suppose that each of these proposals have their legitimate merits, but none has been worked out in much detail. Apart from the discussion in Déchaine (1993), none of the proposed parameters has been amply demonstrated to apply to a range of languages, nor shown to actually derive from or directly map on to other well known principles of grammar (e.g. verb movement to Infl, theta/event role assignments, etc.) that are available, in principle, to all languages.

#### **1.3** On the definition of SVCs

In light of the foregoing review of the literature on SVCs, it would seem that the initial task to perform in the analysis of this phenomenon is to find a way to distinguish in a principled way between the various surface so-called SVCs summarized in (2) and repeated here as (8).

(8) a.

- NP V NP V
- NP V NP V NP b.
- NP (V?) NP V NP c.
- NP V V d.
- NP V V NP e.

In an echo of Christaller (1875), I propose to restrict the term SVCs to just two kinds; one in which the verbs exhibit functional asymmetry (e.g. as verb modifier, or co-heads), and another in which the verbs are simply a sequence of actions. Furthermore, following Stewart (1963) I assume that what is crucial in order to be classified as a true SVC is that there be a missing subject and object for one of the verbs.<sup>9</sup> In other words, SVCs may be provisionally defined as a single clause in which two or more finite verbs occur without any marker of coordination or subordination, sharing a single structural (and semantic) subject and a single object.

This definition rules out, for example, (8e) as a true SVC string since this order would not allow for the object of the second verb to be shared. It also implies that in many instances, 'true' SVCs would have the linear order represented in (8a,d), while those in (8b,c) may be ambiguous between 'true' SVCs like (1a) and covert coordinations like (1g), both with surface NP V NP V NP structures.

#### **1.3.1 Interpretation and Verb Sequencing Constraints**

In this section, I will provide the initial empirical evidence in support of the definition of SVCs given above. This is based on the nature of restrictions on possible verb combinations and how such combinations force particular interpretations of the sentences. I will limit the number of verbs to two at this point in order to keep the discussion fairly

<sup>&</sup>lt;sup>9</sup> This reflects the major empirical claim that has been defended in some analyses as 'Internal argument sharing in SVCs' (cf. Déchaine 1986, Foley and Olson 1985, Baker 1989, and Collins 1997)

simple (see section 2.8 for discussion of restrictions on number of verbs). Consider the following sentences illustrating possible verb sequences:<sup>10</sup>

(9)	а.	Òzó kòkó àdésúwà mòsé trans Ozo raise Adesuwa be.beautiful Ozo raised Adesuwa to be beautiful.'	sitive+stative
	b.	*Òzó kòkó àdésúwà vbi <u>è</u> trans Ozo raise Adesuwa sleep '*Ozo raised Adesuwa to sleep.' OK as 'Ozo raised Adesuwa and he (Òzó)	sitive+unergative slept.'
	c.	Òzó sùá ágá dé Ozo push chair fall 'Ozo pushed the chair down.'	transitive+unaccusative
	d.	*Òzó sùá úyì só Ozo push Uyi cry '*Ozo caused Uyi to cry.' OK as 'Ozo pushed Uyi and he (Ozo) crie	transitive+unergative
(10)	a.	Òzó lé èvbàré rhié nè úyì Ozo cook food give to Uyi 'Ozo cooked the food and gave it to Uyi.'	transitive+transitive
	b.	*Òzó lé èvbàré vbi <u>é</u> Ozo cook food sleep 'Ozo cooked the food and slept.'	transitive+unergative
	c.	*Òzó tòtàá lé èvbàré Ozo sit cook food 'Ozo sat down and cooked the food.'	unergative+ transitive
	d.	*Òzò dé rhàá <u>òmó</u> Ozo fall grab child	unaccusative+transitive
	e.	Òzó gbé úzò khi <u>é</u> n Ozo kill antelope sell 'Ozo killed the antelope and sold it.'	transitive+transitive (=7a)

These sentences in (9-10) are representative samples of what appears to be fairly general observations. In general, observe that they illustrate the linear orders in (8a) and (8b) ( I

 $<sup>^{10}</sup>$  In these sentences, a \* before a sentence is meant to express the fact that the sentence is bad on the reading where the two verbs are within one prosodic unit (no pause or intonation break before the second verb). This is an important point because when there is a prosodic break between the verbs speakers usually indicate this by a pause and the interpretation associated with such sentence will be one of coordination (hence covert coordination) like (1f).

will come back to the discussion of (8d) shortly).<sup>11</sup> I propose that there are two basic kinds of SVCs : (a) <u>resultative</u>, (b) <u>consequential</u>.

Resultative SVCs are ones in which the action of the first verb brings about the result that is denoted by the second verb. This implies that there is a strict cause-effect relationship between the verbs and there is no time lapse between the eventualities that they express. Furthermore, there is the implication that the verbs in the resultative SVC are semantically ordered pairs such that the first verb must denote a cause or process and the second verb must define some kind of state or result. One consequence of this is that in an SVC expressing result, if the first verb is transitive the second verb will almost always be unaccusative but not unergative.

With this characterization of the resultative SVC in mind, observe that in (9a) the second verb is a stative verb and it expresses the result of the action denoted by the first verb (see Baker and Stewart (1997a) for discussion which suggest that stative verbs are unaccusatives in Èdó). (9a) with a stative second verb contrasts very sharply with (9b), in which the second verb is an unergative. In particular, it is important to note that (9b) is ungrammatical only if a resultative SVC reading is intended: otherwise the sentence can imply a sequence of events, Ozo raised Adesuwa and he (Ozo) slept, which does not involve object sharing and so is not SVC (according to my criteria set out above). This reading of two separate and distinct events is what I shall as characteristic of covert coordinations. Similar observations hold for the contrast between (9c) and (9d) in which a standard unaccusative verb is the second verb in the former while the second verb is an unergative in the latter. As you can observe, a resultative reading with object sharing is acceptable for (9c) but only a coordination of events reading is possible for (9d). Thus, these restrictions and interpretations of the verbs show that in the resultative SVC, the

<sup>&</sup>lt;sup>11</sup> I will put aside (8c) and (8e) for now by pointing out that the question mark in (8c) refers to whether it is a verb or not (see Chapter seven for discussion of these two orders as reflecting non-SVC structures).

second verb is typically an unaccusative and that both verbs share subject and object arguments.

Consequential SVCs, on the other hand, are those in which the verbs express a natural sequence of events and they are temporarily ordered in a precedence-consequence iconic relation (Gruber 1992a,b). Thus, (10a) means that Ozo cooked the food and gave it to uyi, but it is possible that he could have sold it instead. Whereas in the resultative SVC there is a direct link between the action picked out by the first verb and the result which I have described in terms of cause-effect. In contrast, in the consequential SVC the action of the second verb is not a result directly caused by that of the first verb, but rather a more indirect consequence, the second step of an overall plan on the part of the agent. It is part of this (semantic) consequence relationship that the object of the second verb must be the same as the object of the first verb. Unlike the resultative, the second verb of a consequential SVC must always be transitive and this is true also for the first verb. Thus, in (10a) where both verbs are transitive they share the same subject and object. However, in (10b) where the second verb is an unergative verb, there cannot be object sharing and in fact, this sentence is interpreted as a sequence of events, Ozo cooked the food and then he (Ozo) slept. Once again, we observe that covert coordinations involve the conjunction of two separate and distinct events expressed freely by verbs without any restrictions as can be observed in both resultative and consequential SVCs. Furthermore, in (10c) in which the first verb is unergative there is also no object sharing reading, while an unaccusative first verb is simply ungrammatical (10d). (10d) is in a sharp contrast with (10e) in which there are two transitive verbs and the sentence is grammatical.

Consequently, I conclude that consequential SVCs are those in which two transitive verbs occur, sharing the same subject and object. Notice that the linear order (8a) and (8b) can occur as consequential SVCs; however based on the restriction that both verbs be transitive it follows that the linear order in (8d) can never be consequential SVC. This pattern is not ruled out for resultative SVCs, however, as shown in (11):

(11)	a.	Òzó dé wú Ozo fall die 'Ozo fell to his death.'	unaccusative+ unaccusative
	b.	*Ozo dé sàán Ozo fall jump 'Ozo fall and then he jumped.'	unaccusative+unergative
	c.	Òzó sàán kpàá Ozo jump leave 'Ozo jumped out.'	unergative+unaccusative
	d.	*Òzó sàán tín Ozo jump fly 'Ozo jumped and flew.'	unergative+unergative

In (11a) we observe that two unaccusative verbs can occur as a resultative SVC such that the first verb expresses a cause the result of which is Ozo's death. Thus, Ozo is the internal object of both verbs as well as the surface subject. However, (11b) with an unergative second verb is ungrammatical on the intended resultative reading where Ozo was about to fall and then he jumped, in other words the falling event could not have caused him to jump. This sentence only has the coordination of event reading, Ozo fell and he also (independently) jumped. The same contrast between unaccusative and unergative second verbs is illustrated in (11c) and (11d) where a resultative reading (in which the sole argument is shared) is acceptable when the second verb is unaccusative (11c) but not when it is unergative (11d). Thus, I conclude that the linear order in (8d) can be a true SVC only if the second verb is unaccusative and both verbs share the same argument.

Two basic questions emerge from this intuitive semantic distinction between resultative SVCs, consequential SVCs, and covert coordinations (CCs). (a) Why aren't examples like (1f) and (1g) structurally ambiguous between a covert coordination structure and a consequential SVC structure? (b) What prevents the sequence of two unaccusative verbs from being a consequential SVC. These and other related questions will form the basis for the fine-grain distinction between the two kinds of SVCs and covert coordinations in subsequent Chapters, especially in Chapter two.

18
## 1.4 Conclusion

This chapter proposed that there are two kinds of SVCs-- resultative and consequential-- and they must share the same subject and object. It is argued that while the second verb of the resultative SVC is typically unaccusative, both verbs of the consequential SVC are always transitive. Furthermore, it was proposed that covert coordinations are not SVCs in that there are two separate objects for the verbs and presumably two subjects, one of them being deleted under identity.<sup>12</sup> A salient confirmation of this distinction between SVCs and CCs is based on the fact speakers process true SVCs as one prosodic unit whereas a pause is usually present in varying degrees before the second verb in covert coordinations.

On the basis of these observations, let us proceed to the analysis of the substantive issues of this thesis as outlined in section 1.1 above, as well as those raised in the review of related literature.

# **1.5** Organization of the thesis

<u>Chapter two</u> presents most of the empirical core of this thesis by studying the behavior of different kinds of adverbs in SVCs which is used to illustrate the distinction between resultative SVCs and consequential SVCs, and covert coordinations. This distribution of adverbs is also taken to reveal the presence of certain lexical and functional projections in their structures. The discussion of the relevant tests will be presented in two related parts. The first part is based on the licensing of word-level categories like INFLtype (henceforth I-type) adverb. The rationale behind this test is based on the expectation that the licensing of word-level categories (I-type adverbs and pre-verbs) will provide insight into the functional structure of the clause.

 $<sup>^{12}</sup>$  In a way, this is consistent with the observation in Déchaine (1993:236) that coordinate clauses always need an overt argument with each verb, either a pronominal or a full DP. Thus, I assume *a priori* that if transitive verbs occur as covert coordinations where there are two separate events and a pause before the second conjunct, then they must each have overt objects. I will come back to the conceptual underpinning of this assumption in section 2.5.1 of Chapter two when I discuss the issue of empty category in SVCs.

The second part of this general adverb-based tests adopts standard syntactic analysis of phrasal-level categories to account for the distribution of Noun-type (henceforth N-type) adverbs and locative PPs. The distribution of these word-level and phrasal-level categories present interesting structural challenges such as that relating to double VP shells and Larsonian-style verb movement, (Larson 1988). In addition, I also investigate the issue of object sharing-- how they are shared syntactically and the interaction of these object NPs with lexical and functional projections. The core of object sharing analysis vis-a-vis null objects will be examined in the light of Baker (1989), Rizzi (1986), Collins (1997), and Baker and Stewart (1997b). I will also examine the general assumption that there is a single subject for SVCs, and based on the analysis of this property I will further distinguish between 'true' SVCs and covert coordinations. Finally, I will discuss the aspectual properties associated with the syntax of certain events such as resultatives (cf. Pustejovsky 1991, Levin and Rappaport 1995, Tenny 1987 etc.) and show that they are quite different from consequential SVCs which do not display any well-defined aspectual features.

<u>Chapter three</u> deals with the predicate cleft construction in Édó based on the basic analysis of cognate objects. The goal is to provide a structurally-based account of the notions "single" versus "double event" that have been loosely associated with SVCs, and also how the licensing of predicate clefts interacts with the distribution of adverb to justify the distinction between resultative and consequential SVCs, and amongst surface verb constructions.

<u>Chapter four</u> deals specifically with double objects constructions--and how they interact--or fail to--with resultative and consequential SVCs. DOCs present data on the basis of which I will examine the fine structure of the proposed lexical and functional projections (where the latter exist).

<u>Chapter five</u> deals with cross-linguistic ramifications of the distinction between resultative and consequential SVCs. The languages include Yoruba where the primary interest is to show that my analysis accounts for certain cases that have been assumed to be problematic in the SVC literature. In addition, I examine Ìgbo, a language on which there is a lot of debate about whether there are actually SVCs, and what the relationship is to the rather large number of resultative V-V compounds. I will argue that resultative SVCs in Èdó show up as resultative V-V compounds in Igbo in a principled way. Finally, I will look at Chinese and show that it is similar to Igbo in many respects based on the proposed distinction between resultative and consequential SVCs. Thus, I will show that predictably Èdó resultative SVCs surface as resultative V-V compounds in Chinese.

<u>Chapter six</u> faces directly the question of what is the parameter that allows a language like  $\dot{E}do$  to serialize but not a language like English. I will show that there are parameters that license serial verbs in  $\dot{E}do$  and AP secondary predicates in a language like English on a fairly comparable basis. As in other works, the relationship between INFL and V will be central to the parameter. However, new facts from verb raising in the sense of Pollock (1989) will shed light on how best to state this.

<u>Chapter seven</u> shows how the various syntactic tests developed in the thesis can be used to clearly distinguish control sentences and causatives from true SVCs. Here, I argue against a traditional view that treats these constructions as manner or instrumental SVCs.

<u>Chapter eight</u> presents the conclusions from this thesis and summarizes its empirical discoveries: the distinction between true SVCs and the various other constructions, and the parameter that distinguishes a SVC language from a non-SVC language.

## 1.6 Significance of the Study

There are at least five areas in which this thesis contributes to knowledge. First, most of the empirical basis for the distinction between resultative and consequential SVCs are genuinely new discoveries in the analyses of SVCs. Second, one theoretical implication of the distinction between resultative and consequential SVCs is that the unified approach adopted in Baker (1989) and Collins (1997) must be rejected. Third, the methodology

adopted illustrates the elegance of investigating different aspects of one language where it becomes possible to see the interplay of micro parameters (cf. Baker 1996) between different constructions which can then be used to make comparisons cross-linguistically. Fourth, this thesis is the first generative analysis of the grammar of Èdó, on which there has been hitherto very little work (cf. Agheyisi 1990, Omoregie 1983, Amayo 1975). Consequently, this thesis is significant in the general task of describing less-studied languages by providing a systematic study of the syntax of Èdó, *inter alia*. Finally, in light of the detailed analysis of SVCs in Èdó, this thesis provides a great deal of important empirical facts toward the analysis of verb serialization cross-linguistically.

#### Chapter 2

## Resultative vs. Consequential SVCs, and Covert Coordinations

## 2.1 Introduction

In Chapter one, I reviewed some of the problems that are related to having a concise definition of SVCs. In particular, I observed that the SVC phenomenon appears to be unconstrained under those analyses that identify SVC types simply on the basis of intuitions about what the combination of verbs express (cf. Oyelaran 1982). The other side of this problem is that some analyses recognize types/classes of SVCs but give a unified syntactic analysis anyway (Baker 1989, Collins 1997). These kinds of classifications and the analyses that underlie them fail in some respects to capture cross-linguistic empirical generalizations, for example why resultative SVCs consistently show up as resultative V-V compounds in Igbo and Chinese.

The goal of this chapter is to provide a battery of syntactic tests on the basis of which we can distinguish two different kinds of SVCs from covert coordinations (CCs). The basic claim to be defended is that SVCs are those constructions in which a single E head quantifies over verbs, and the verbs combine under a single Voice that licenses the subject (and Agent) that sets about a plan of one macro-event which may be resultative or consequential. Adopting a bottom-up approach, the tests to be discussed in this chapter are based on the behavior of different kinds of adverbs which will provide underlying evidence for the internal structure of the clause (lexical and functional projections).

# 2.2 Distribution and Licensing of Manner Adverbs

There are quite a number of recent theories that deal with the syntactic licensing of an adjunct category like adverb, two of which are relevant to the issues discussed in this chapter. For example, Travis (1988) proposes head (feature) licensing for certain adverbs, and Cinque (1997) argues that the hierarchy of adverbs follows from the distribution of functional projections. More generally, I follow Parsons (1990) and assume an account in which adverbs are predicates of events that are denoted by verbs. This approach is based on Davidson (1967), who proposes that verbs explicitly stand for kinds of events, so that a sentence containing such a verb states implicitly that there is an event of that sort. Consequently, I assume that the presence of an adverb indicates that there is an event which it is predicated of, i.e., an adverb modifies (is predicated of) an event that is denoted by the verb.<sup>1</sup> Let us now illustrate these assumptions about the licensing of adverbs, against the background of manner adverbs in Èdó; first in simple sentences, and ultimately in SVCs.

Stewart (1996) proposes that there are two kinds of manner adverbs in Èdó and argues further that the two kinds of adverbs occur in mutually exclusive environments. The contrast is illustrated in (1) and (2):

(1)	a.	Òzó gìégìé kòkò <u>ògó</u> (*gìégìé) Ozo quickly gather bottle (*quickly) 'Ozo is quickly gathering the bottles (*quickly).'	non-past tense
	b.	Òzó gi <u>é</u> !gi <u>é</u> kó!kó <u>ògó</u> (*gi <u>é</u> !gi <u>é</u> ) Ozo quickly gather bottle (*quickly) 'Ozo quickly gathered the bottles (*quickly).'	past tense
(2)	а.	Òzó (* <u>ègìégìé)</u> kòkò <u>ògó ègìégìé</u> Ozo (*quickly) gather bottle quickly 'Ozo (*quickly) is gathering the bottles quickly.'	non-past tense
	b.	Òzó (* <u>ègié</u> !gì <u>é)</u> kòkó <u>ògó ègìégìé</u> Ozo (*quickly) gather bottle quickly 'Ozo (*quickly) gathered the bottles quickly.'	past tense

Adverbs of the kind in (1) can only occur to the left of a verb and occupy the position between the subject and the verb, but never in sentence final position. The other type of adverb illustrated in (2) can only occur at clause boundaries, in this case in a sentence final

<sup>&</sup>lt;sup>1</sup> Although it is quite possible that not all adverbs are event related for example 'Frankly, I like broccoli' (Manfredi, p.c), but I will ignore this difference for the most part since there are predominantly only manner adverbs in Edő.

position but never between the subject and the verb.<sup>2</sup> Observe further that while the adverb in (1) may vary in terms of tones, the one in (2) does not. In addition, there is a non-trivial morphological difference between these two: the adverb of the kind in (1) begins with a consonant, while that in (2) begins with a vowel. This morphological difference matches up very nicely with the difference between nouns and verbs or INFL words in the Èdó language: all nouns must begin with a vowel and all other lexical categories must begin with a consonant (cf. Agheyisi 1990, Amayo 1976, Omoruyi 1986a/b etc.).

Consequently, for descriptive clarity I propose to classify the adverb in (1) as an INFL-type adverb (henceforth I-type adverb) because it exhibits signs of being linked with tense both by its position and by varying for tense tones like verbs (see Stewart 1996 for discussion). Some other examples of I-type adverb include; gèlé (truly), rhèrhé (early), fèkó (slowly, carefully), zèégìé (quickly) etc. The adverb in (2) I will call a Noun-type adverb (henceforth N-type adverb) since it shares the morphological trait of having vowel initial segment with nouns, and like nouns it does not tonally inflect for tense. Some examples of the N-type adverb include, ègìégìé 'quickness', èzèégìé 'quickness' etc. However, despite these differences I assume that both I-type and N-type adverbs are predicates of events.

As the contrast between (1) and (2) shows, there are syntactic conditions that regulate the distribution of both I-type and N-type adverbs. An I-type adverb can only occur as an adjunct to a syntactic position that is tense-related, i.e., a syntactic position where tense features can be checked (as reflected by tense tone copying). Based on the sentences in (1), one possible candidate to which the adverb can adjoin appears to be the verb. This is despite the fact that there is a T(ense) position between the subject and the verb which may be a potential candidate. However, we can confirm the fact that the adverb

 $<sup>^2</sup>$  This kind of adverb may also occur in sentence initial position and it can also be fronted (topicalized) like nouns in the language, but I will not discuss the sentence initial position any further since it is not relevant in determining the internal structure of SVCs (but see p. 106 section 3.2.1)

does not adjoin to T from the sentences in (3) where T is overtly occupied by the future tense morpheme, ghai:

(3)	а.	Òzó ghá gi <u>é</u> !gi <u>é</u> kó!kó <u>ògó</u> Ozo FUT quickly gather bottle 'Ozo will quickly gather the bottles.'	non-past tense
	b.	*Òzó gi <u>é</u> !gi <u>é</u> ghá kó!kó <u>ògó</u>	non-past tense

Ozo quickly FUT gather bottle 'Ozo quickly will gather the bottles.'

When T is overtly filled by the future tense morpheme as in (3a), the I-type adverb occurs between it and the verb.<sup>3</sup> However, when the I-type adverb occurs to the right of T the sentence is ungrammatical (3b). Now, Kayne (1994) and others argue that there is no rightward adjunction to heads. Given this, the contrast between (3a) and (3b) implies that the I-type adverb is not generated in T leaving us then with the option that I-type adverbs adjoin to the left of the verb. While this proposal is a valid account of I-type adverb licensing, there is another possibility, based on the idea proposed in Travis (1994, forthcoming) that TP obligatorily dominates EP (Event Phrase). Under this structure, the I-type adverb can then left-adjoin to the head of the EP; it inflects for tense because EP is a tense-related functional projection (Travis 1994).

The empirical evidence in support of the idea that I-type adverb adjoins to the left of the head of the functional projection EP comes from the distribution of iterative morpheme relative to the presence of the I-type adverb. This is illustrated in (3c).

c. Òzó ghá gié!gié ghá kó!kó <u>ògó</u> non-past tense
 Ozo FUT quickly Iter gather bottle
 'Ozo will quickly gather the bottles repeatedly '

Observe that the adverb occurs between Tense head, that is occupied by the future tense morpheme  $gh\dot{a}$ , and another homophonous morpheme  $gh\dot{a}$  (imperfective) that is in a

<sup>&</sup>lt;sup>3</sup> Observe that both the verb and the I-type adverb now have a different tonal pattern from non-past sentence in (1). I will come back to this issue in section 2.7.1 (see also 2.7.2)

position before the verb.<sup>4</sup> Therefore, based on (3c) we confirm that the I-type adverb does not left-adjoin to the verb since the iterative morpheme can occur between the adverb and the verb. Following Jackendoff (1990:29), I assume that iteration quantifies over an event and so I propose that in (3c) the iterative morpheme is generated in the head of EP. Therefore, we derive an adequate account of the position of the I-type adverb by saying that it is licensed as a left-adjunct to the head of the functional projection EP.<sup>5</sup> The structural representation is given in (4) for the sentence (3c).



There is an interesting wrinkle to the structure in (4) which however does not undermine my proposed account of I-type adverbs but is worth mentioning here in anticipation of certain facts about *restructuring* to be discussed in chapter seven. This is the observation that under certain tenses namely, when T and EP spell-out both the features of tense and iteration, then the I-type adverb can occur as a left adjunct to the VP. The relevant examples are given in (5):

(5)	a.	Òzó ghá!á gi <u>é</u> ! Ozo past Iter quic 'Ozo used to quickl	gi <u>é</u> kó!kó kly gather y gather the	<u>ògó</u> bottle bottles (;	repeatedly).	past tense
	b.	*Òzó ghá!á giệ Ozo past Iter qui	gi <u>é</u> ghá ckly ITer	kó!kó gather	<u>ògó</u> bottle	past tense

<sup>&</sup>lt;sup>4</sup> See section 7.2.1 for discussion of the two ghás

<sup>&</sup>lt;sup>5</sup> This also carries a further implication that both the I-type adverb and the iterative morpheme pick out events. I should point out that EP is the equivalent of "outer" AspP in Travis (1994) and so they are notational variants.

c. \*Òzó gi<u>é</u>!gi<u>é</u> ghá!á kó!kó <u>ògó</u> past tense Ozo quickly past Iter, gather bottle

In (5a), we observe that the I-type adverb occurs to the left of the verb after Tense. Observe, however, that the morpheme in Tense is more complex than that in (3c). While the one in (3c) is a simple form, the one in (5a) can be decomposed into 'ghá' (like 3c) and a vowel 'á'. Thus, in (5a) the downstep (floating) tone within the word indicates that it is a complex derived word. I propose that what we have in (5a) is the result of raising the head of EP into Tense and this brings about a situation in which the I-type adverb adjoins to the left of the verb. We can confirm this analysis based on the ungrammaticality of (5b) where we observe that the position of the raised head cannot be filled. Therefore, I conclude that the I-type adverb may adjoin to the left of the head of EP when it is itself not part of a complex head, otherwise it may then adjoin to the left of the verb which is also tense-related since it can bear tense features (e.g., tonally inflect for tense).

In the case of the N-type adverb as in (2), I propose that it adjoins to the right of a VP. This implies that right-adjunction is acceptable on phrases but not heads (cf. Baker 1996) As such the N-type adverb can be used to mark the right edge of a VP, where the VP projection is defined under standard assumptions as the minimal constituent of [V NP] (along with possible additional complements). In essence, therefore, I assume that the distribution of the N-type adverb would correspond in a straightforward manner with VP projections and this can then be used as a test to determine whether a verb and its complement project a phrase. This adjunction possibility of N-type adverb is illustrated in (6).<sup>6</sup> This structure also shows the difference in terms of licensing between the two kinds of adverbs.<sup>7</sup>

<sup>&</sup>lt;sup>6</sup> It should also be noted that there is nothing in my theory of N-type adverb licensing that prevents it from being right-adjoined to the EP in (6). Although I allow this as a possibility, I am not sure if there is any meaning difference between the two positions (VP/EP) as has been argued for English VP-adjoined and PredP-adjoined adverbs (cf. Bowers, 1993). However, some indication of something similar can be observed in SVCs (p. 49, section 2.2.4)

<sup>&</sup>lt;sup>7</sup> In order to keep the representation simple, I do not include the fact that the I-type adverb can sometimes adjoin to the left of the verb in the context of the complex head bearing ghá!á and in fact I do not use this



One pertinent issue for this theory of adverb licensing is to determine whether an adverb can occur between the verb and its argument. For example, in both French and Icelandic an appropriate adverb can occur between the verb and its argument (cf. Pollock (1989), Laenzlinger (1994), Travis (1988)). However, the sequence of [V Adv NP] is ungrammatical in English (see Stowell (1981), Koizumi (1993), Pesetsky (1989) for discussions). Thus, the relevant question pertaining to the structure in (6) is whether Èdó is like French or English in terms of adverb placement, in other words can an adverb of any type come between a verb and its complement? Consider the sentences in (7):

(7)	a.	*Òzó Ozo	kó!kó gather	gi <u>é</u> !gi <u>é</u> quickly	<u>ògó</u> bottle
	b.	*Òzó Ozo	kòkó gather	<u>ègiégié</u> quickly	<u>ògó</u> bottle

Based on the ungrammaticality of the sentences in (7), we can conclude that Edó is not like French or Icelandic because neither the I-type nor the N-type adverb can occur between a verb and its NP complement.<sup>8</sup> Now, given these background assumptions concerning the

(6)

tense marker as part of the general discussion, it was introduced simply to flag the analysis of EP restructuring in Chapter seven. The structure in (6) predicts that both the I-type and the N-type adverbs can co-occur and this is true (cf. Stewart 1996).

<sup>&</sup>lt;sup>8</sup> I will come back to this issue when I discuss the facts of verb raising in chapter six. Also, in section 3.3 I will extend the discussion to whether an N-type adverb can occur between [V + NP] and a cognate object which is analyzed as an event argument.

licensing of both kinds of manner adverbs, let us then examine the distribution of each type separately in SVCs.

### 2.2.1 I-type Adverb before First Verb

I begin this discussion by first considering the common position in which the Itype adverb can occur amongst SVC. Consider the following sentences:

(8)	a	Òzó gi <u>é</u> !gié kó!kó Àdésúwà mó!sé Ozo quickly raise Adesuwa be-beautiful 'Ozo quickly raised Adesuwa to be beautiful.'
	b.	Òzó gié!gié dú!nmwún èmà khié!nné Ozo quickly pound yam sell+PL 'Ozo quickly pounded the yams and sold them.'
	c.	Òzó <u>gié!gié</u> gb <u>ó!ó</u> ívìn bòló <u>ó</u> kà Ozo quickly plant coconut peel corn 'Ozo quickly planted the coconut and [he] peeled the corn.'

As illustrated in (8), the I-type adverb can occur before the first verb in SVCs, resultative (8a) and consequential (8b), and covert coordination (8c). The question that arises here is what do these sentences mean, in other words what event or events are the I-type adverbs predicates of ?

In (8a), the interpretation of the sentence is that both actions denoted by the two verbs were quick, i.e., the "raising" and "becoming beautiful" were quick. (8a) cannot have a reading in which only one of the verbs is modified apart from the other. Thus, it cannot be that only the "raising" is quick and not the "becoming beautiful". This integrity of interpretation implies that even though the I-type adverb structurally occurs to the left of the first verb, it must be the case that both verbs express the same event ( a single event) which the I-type adverb is a predicate of. Given my theory of I-type adverb licensing as predicates of events, generated as left-adjuncts to the head of EP, I predict that it should be able to co-occur with the iterative morpheme in this kind of SVC. This prediction is borne out in (9) ( I have switched examples simply because of pragmatics).

Ôzó gi<u>é</u>!gi<u>é</u> ghá suá <u>ògó</u> dé Ozo quickly Iter push bottle fall 'Ozo quickly pushed the bottle down repeatedly.'

(9)

In (9), both the action of pushing and falling are construed as a single event by the adverb and the iterative morpheme. Thus, the interpretation of (9) is that Ozo pushed the bottle down each time to bring about a cumulation of such events. So, it is instructive to observe that 'push-fall' in (9) denotes a single (resultative) event type as evidenced from the meanings of the iterative morpheme and the I-type adverb.

The interpretation of (8b) is similar to that in (8a).<sup>9</sup> The interpretation of (8b) is that both the "pounding" and "selling" were quick. This sentence cannot have the interpretation in which, for example, only the first verb is modified Ozo quickly pounded the yam and then sold it (slowly). Based on the analysis of similar facts in the resultative SVC (8a), it follows that there must also be a sense in which the two verbs of the consequential SVC (8b) express the same event or parts of a macro-event which the I-type adverb in this position before the first verb is predicated of. Once again, the prediction that both the iterative morpheme and the I-type adverb can co-occur is borne out. This is illustrated in (10);

(10) Òzó gié!gié ghá dú!nmwún èmà khié!n Ozo quickly ITER pound yam sell 'Ozo quickly pounded the yams and sold them repeatedly.'

(10) has the meaning that both the pounding and selling were quickly done over and over again (iterated). Given the assumption that iteration quantifies over events, there is a

<sup>&</sup>lt;sup>9</sup> I would like to draw attention to the fact that in both (8a) and (8b), the presence of the adverb seems to trigger tone spreading past the first verb unto the second. This fact indicates a similarity in the underlying properties of the functional head that immediately dominates the first verb. I will expand on this issue as we go along and particularly in sections 2.7.1 and 2.7.2, but see the immediate text for some discussion.

correspondence with the interpretation of the I-type adverb in this position which is that both verbs of the consequential SVC express a single event or parts of a macro-event.<sup>10</sup>

Now, let us compare the foregoing discussion with the covert coordination sentence (8c). There is a sharp contrast between SVCs and covert coordinations based on the fact that the functional head E in (8c) has a different scope interpretation with the I-type adverb. The only interpretation of (8c) is that the planting event was quick without any obvious implication for the event that is denoted by the second verb.<sup>11</sup> This interpretation is consistent with the fact that there is a conjunction of events and suggests a difference in terms of the nature of what the E position is quantifying over in SVCs and covert coordinations. Thus, the essential difference in terms of I-type adverb licensing and the iterative morpheme before the first verb seems to be that the E operator quantifies over both verbs and the events that they denote in both the resultative and consequential 'object sharing' SVCs.<sup>12</sup> However, the E operator in the covert coordination only quantifies over the first verb and the event that it denotes. I will put forward the semantic representation of these differences between the E operators in the different constructions until I have discussed similar facts regarding adverbs in the position before the second verb (section 2.2.2)

In the meantime, I should point out that the facts concerning events and I-type adverbs are, also, predictably matched by the quantification behavior of the iterative morpheme. The relevant example is given in (11);

<sup>&</sup>lt;sup>10</sup> Thus, it appears that this functional head E has similar properties in both resultative and consequential SVCs. One likely speculation for this similarity may be the condition that there should be a single object NP that simultaneously measures out the event denoted by each verb. The significance of this proposal comes from the contrast between these SVCs and the covert coordination sentence (8c). I will not comment on this proposal any further.

<sup>&</sup>lt;sup>11</sup> Observe a phonological corroboration of this interpretation based on the observation that in (8c) the presence of the adverb does not trigger tone spreading past the first verb, unlike in the resultative (8a) and consequential (8b). Again, more on this in section 2.7.1 and 2.7.2 p. 84 ff.

 $<sup>^{12}</sup>$  The difference in the nature of the quantification behavior of this E position between resultative and consequential will be made clear shortly.

(11) Òzó <u>gié!gié</u> ghá gb<u>ó!ó</u> ívìn, bòló <u>ó</u>kà
 Ozo quickly ITER plant coconut pee! corn
 'Ozo quickly planted the coconut repeatedly and [he] peeled the corn.'

The interpretation of (11) is that the event of planting coconut was carried on again and again as he peeled the corn (once). It is important to note that there is a necessary pause before the second verb and this confirms that we are truly dealing with a case of covert coordination (distinct events coordination). Thus, (11) is just like the I-type adverb facts in that the iterative morpheme quantifies over the first verb only.

What we have seen so far from the behavior of I-type adverbs, which is consistently confirmed by the interpretation of the iterative morpheme, is that when they occur to the left of the first verb, they provide evidence for functional structure and the nature of event composition. Thus, in the resultative SVC (8a) the fact that the I-type adverb in the pre-first verb position implicates the action of both verbs suggests that there is a single event of which it is predicated. Similarly, in the consequential SVC (8b) the actions of the two verbs may be formally linked into a single complex event by the functional head E that occurs before the first verb. In the covert coordination (8c), there is a clear-cut sense of two distinct events brought out by these elements.

#### 2.2.2 I-type Adverb before Second Verb

My task in this section is to use the same techniques as the previous section by putting the I-type adverb in the position before the second verb. There are three things I hope to show in this section. First, the distribution and licensing of I-type adverbs will provide evidence for functional structure between the verbs. Second, if we can justify that there is functional structure between the verbs then this should be evidence against the account of 'object sharing' in Baker (1989). This is because the E position would structurally split the two verbs thereby under-cutting Baker's account of true 'internal argument sharing', which is based on a doubly headed VP. Third and finally, I expect that determining whether there is an E position before the second verb should provide evidence for the nature of event composition based on the view that adverbs are predicates of events (cf. Parsons, 1990). This should present a useful insight into the distinction between single versus double events SVCs (Givón 1991).

Consider the following sentences, illustrating the I-type adverb in the position before the second verb for each of the three constructions:

- (12) a \*Òzó sùá ògó gić!gić dé Ozo push bottle quickly fall
  b. \*Òzó kòkó Àdésúwà gić!gić mó!sć Ozo raise Adesuwa quickly be-beautiful
- (13) a. Òzó dùnmwún èmà gié!gié khién
   Ozo pound yam quickly sell
   'Ozo pounded the yam and quickly sold it.'
  - b. Òzó lé èvbàré <u>gié!gié</u> ré
     Ozo cook food quickly eat
     'Ozo cooked the food and quickly ate it.'
- (14) a. Òzó gbòó ívìn gié!gié bó!ló ókà
   Ozo plant coconut quickly peel corn
   'Ozo planted the coconut and he quickly peeled the corn.'
  - b. Òzó gbé <u>èkhù gié!gié</u> lá!á òwá
     Ozo hit door quickly enter house
     'Ozo hit the door and he quickly entered the house.'

As the data above shows, the I-type adverb cannot occur between the verbs in the resultative SVC (12), but it can occur before the second verb in the consequential SVC (13) and in the covert coordination (14).

According to my theory of I-type adverb licensing, (12) constitutes evidence that there is no E position between the verbs in the resultative SVC. We can quickly confirm this fact from the sentences in (15) with the iterative morpheme before the second verb:

(15)	a	*Òzó Ozo	sùá push b	<u>ògó</u> ghá ottle Iter.	dé fall	
	b.	*Òzó Ozo	kòkó raise	Àdésúwà Adesuwa	ghá Iter.	mó!sé be-beautiful

On the basis of the ungrammaticality of the sentences in (12) and (15) with I-type adverb and iterative morpheme which both require the presence of the functional head E in order to be properly licensed, I conclude that resultative SVCs lack an EP projection before the second verb. Given my treatment of adverbs as predicates of events, then the failure of the I-type adverb to occur in the position before the second verb suggests that this second verb does not by itself denote an event that is distinct from the first verb. It is therefore a desirable feature of my theory that the distribution of the I-type adverbs is consistent with the nature of event composition in the resultative SVC.

Thus, I propose that there is only one E position in the resultative SVC and this is because there is a single quantification over a single event. Therefore, it makes perfect sense that the E position where both the I-type adverb and the iterative morpheme are generated dominates both the first and the second verb. The semantic representation of a resultative SVC like (16a) would be as stated in (16b).

- (16) a. Òzó sùá <u>ògó</u> dé
   Ozo push bottle fall
   'Ozo pushed bottle down.'
  - b. <u>The resultative SVC</u>
    (∃e) [ Push-Fall(e) & Agent(e, Ozo) & Theme(e, <u>ògó</u>)].

(16b) can be read in the following way, there is one event (e) and this event is a pushingplus-falling and there is a single Agent of the event which is Ozo and there is also a single Theme which is the bottle, and either the adverb or the iterative morpheme is predicated of this single event. The observations about the resultative SVC are consistent with a doublyheaded VP structure of Baker (1989,1991) as in (17).



(17) is compatible with the semantic facts from I-type adverb and iterative morpheme. However, based on criticisms against this structure such as the fact that it is ternarybranching (cf. Larson 1991) in light of Kayne (1984), I propose a modification as in (18).<sup>13</sup>





In (18), the theta domain of the two verbs is the maximal projection VP that contains both of them. This results from a binding-chain between the top verb position and an empty verb position that is a sister to the V-bar projection that contains the second verb. Therefore, the object NP is in the same maximal projection, VP, that contains the two verbs and so it can be assigned their internal theta roles (cf. Baker 1989). The relations between the top verb and the empty verb in (18) mirrors the Government Transparency Corollary of Baker (1988) but without movement. Thus, (18) implies a non-distinctness between the first and second verb in the resultative SVC, as formalized in (19).

(19) X is distinct from Y only if no part of Y is a member of a chain containing  $X^{14}$ 

<sup>&</sup>lt;sup>13</sup> The analysis of subject NP will be presented in section 2.6.

<sup>&</sup>lt;sup>14</sup> The notion of part corresponds to the VP containing Y.

When (19) is applied to the structure in (18), we observe that the first verb is not distinct from the second verb because the bound empty verb is a sister of the V-bar containing the second verb, under the same VP.<sup>15</sup> I will come back to the reason why I have adopted this binding-chain structure, rather than a Larsonian verb-raising account, when I discuss verb raising in Chapter six. Of primary relevance at this point is that the structure in (18) expresses the fact that the I-type adverb only occurs before the sequence of both the first and second verbs, which together share the same internal argument. Furthermore, the absence of an EP projection between the verbs is consistent with the claim that neither of the verbs in the resultative SVC expresses a distinct event. Thus, for example, an I-type adverb cannot be predicated of either of the verbs individually.

Turning now to the consequential SVC and the covert coordination sentences, we observe based on the data in (13) and (14) that they do contain an EP projection between the verbs. In the consequential SVC in (13), the I-type adverb occurs to the left of the second verb and the sentence has an interpretation in which only the event denoted by the second verb was quick. Thus, for example, in (13a) Ozo may have pounded the yam slowly but the selling event was necessarily quick. This proposal that there is an EP projection between the verbs in the consequential SVC is supported, once again, by the fact that an iterative morpheme can occur before the second verb (20);

<sup>&</sup>lt;sup>15</sup> Another version of this same relation is the head-head relation (Chomsky 1995:177) which is assumed to be a local relation as shown between H and  $X_2$  in (i);



According to Chomsky (1995:177), the structure in (i) can only have arisen by the raising [in my case the binding-chain] of H to adjoin to X. Therefore, H heads a chain CH = (H, ....t), and only this chain, not H in isolation, enters into head c relations. For a head  $\alpha$ , take max ( $\alpha$ ) to be the least full-category maximal projection dominating  $\alpha$ . Thus, in (i) max (H) = max (x) = [XP<sub>1</sub>, XP<sub>2</sub>], the two-segment category XP which is the [VP] projection in (18).

(20) Ôzó dùnmwún èmà ghá khi<u>é</u>!n
 Ozo pound yam Iter sell
 'Ozo produced pounded yam and sold it repeatedly.'

What (20) means is the following: there was a pounding event (which brought about a large amount of pounded yam) and thereafter, Ozo sold the pounded yam in bits; although there was one pounding of yams, there were different events of selling the pounded yam. Thus, just like the I-type adverb before the second verb, the iterative morpheme shows that the intermediate E operator quantifies over the event position of the second verb only. Given my theory which treats each instance of adverb placement as being predicated of an event, it follows therefore that there is quantification over two events in the consequential SVCs and this is compatible with a structure in which there are two EPs. I will come back to the relationship between the two EPs after introducing comparable facts in covert coordinations (CCs).

In the CCs in (14), we observe that the I-type adverb occurs to the left of the second verb. The interpretation of (14a), for example, is that Ozo planted coconut (slowly) and he quickly peeled the corn. The fact that the I-type adverb can occur before the second verb is, once again in the light of my account of adverb licensing, interpreted as evidence of a functional projection dominating the second verb. This, in turn, implies that there is a separate event that is associated with the second verb. We can confirm these observations by putting the iterative morpheme before the second verb (21);

(21) Òzó gbòó ívìn ghá bó!ló ókà
 Ozo plant coconut Iter. peel corn
 'Ozo planted the coconut and he peeled the corn repeatedly.'

The interpretation of (21) is consistent with that of adverb modification in (14); only the event that is denoted by the second verb is quantified over, i.e., Ozo planted the coconut (one time) and he peeled corn over and over again. The conclusion is self-evident, that there are two distinct events as well in CCs.

Now that I have given the overall description of I-type adverb licensing and distribution in consequential SVCs and CCs, I want to go into the issue of the formal representation of the two E positions that have been identified and the relationship between them. The descriptive generalization seems to be the following: in the consequential SVC, the first E position quantifies over two events obligatorily even though there is a real sense that the second event exists and can be separately quantified over. In other words, it seems that the first E quantifies over a macro event that consists of two sub-events denoted by each of the two verbs. However, in the CC there is a quantification over two completely separate events since putting the I-type adverb before the first verb does not imply quantification over the second verb. Therefore, the most obvious question for an adequate formal representation of these facts is what is the relationship between the two E positions?

The answer to this question should provide a clear perspective on the nature of event composition in consequential SVCs and CCs. Happily, there is empirical evidence that can be used to determine the relationship between the two E positions. This is based on the possibility of having the same I-type adverb occur in the two Es in the same clause, and correspondingly whether there can be two separate instances of the iterative morpheme. First, the I-type adverb. The relevant examples are given in (22) and (23):

- (22) a. \*Òzó gié!gié dún!mwún èmà gié!gié khié!n
   Ozo quickly pound yam quickly sell
   'Ozo quickly pounded the yams and quickly sold it.'
  - b. \*Ozó gi<u>é</u>!gi<u>é</u> lé èvbàré gi<u>é</u>!gi<u>é</u> r<u>é</u>
     Ozo quickly cook food quickly eat
     'Ozo quickly pounded the food and quickly ate it.'
- (23) a. Ozó gié!gié gbó!ó ívìn gié!gié bó!ló ókà
   Ozo quickly plant coconut quickly peel corn
   'Ozo quickly planted the coconut and he quickly peeled the corn.'
  - b. Òzó gié!gié gbé èkhù gié!gié lá!á òwá
     Ozo quickly hit door quickly enter house
     'Ozo quickly hit the door and he quickly entered the house.'

This is a striking contrast between (22) and (23). It is ungrammatical for the same I-type adverb to occur before both verbs in the consequential SVC (22), while similar distribution is just perfect in the CC (23). The ungrammaticality of the sentences in (22) is unexpected under the analysis we have been assuming whereby there are two E positions and as such two events that would seem to have independent status. Clearly, (22) implies that there is complex interaction between the two E positions in the consequential SVC. However, there is no restriction in terms of co-occurrence of the same I-type adverb in CCs (23). The contrast between (22) and (23) implies, therefore, that there are truly two distinct events that are conjoined in CCs. Consequently, (23a) will have a semantic representation like (24a) and this may correspond structurally to EP conjunction on the basis of the cvidence thus far, as in (24b),<sup>16</sup>

(24a) <u>covert coordinations</u> (∃e1)[ Planting(e1) & Agt(e1, Ozo) & Th(e1, coconut)] & (∃e2)[ Peeling(e2) & Agt(e2', Ozo) & Th(e2, corn].

(24b)



This leaves us at this time with the problem in (22) where we observe that two Itype adverbs of the same kind cannot occur within the same clause. I will very briefly discuss two possible explanations for this problem and show why they fail.

<sup>&</sup>lt;sup>16</sup> It is also possible that the CC may be a conjunction of two VPs, but since the I-type adverb is licensed by the functional projection EP I will stick to the conjunction of EPs, however, in the subsequent sections I will present evidence that the conjunction actually involve higher constituents, VoiceP and T.

One possible explanation for the facts in (22) is to state a condition such that the lower E position is null just when the top E position is overtly filled. The problem that I find with this approach is that it makes the prediction that there cannot be two distinct I-type adverbs that can occur in the same clause. However, this prediction turns out to be false as the data in (25) shows:

- (25) a. Òzó gé!lé dún!mwún èmà gié!gié khié!n Ozo truly pound yam quickly sell 'Ozo truly pounded the yams and quickly sold it.'
  - Òzó gèlè lè èvbàré giègiè rè Ozo truly cook food quickly eat
     'Ozo truly pounds the food and quickly eats it.'

(25) shows that it is not true that two I-type adverbs cannot occur within the same clause. We can very easily confirm that g d e i is an I-type adverb based on the tonally represented tense contrast between (25a) (past tense with high tones) and (25b) (habitual tense with low tones). Thus, if we adopt the approach that the lower E position is null under any condition, we lose an account of the fact that two I-type adverbs can occur within the same clause in the consequential SVC. In fact, the data below in (26) showing the distribution of the I-type adverb g d e i 'truly' further illustrate the fact that there must be something else that is responsible for the ungrammaticality of these sentences that is not derived from the condition that there cannot be two I-type adverbs within the same clause:

(26)	a.	*Òzó gé!lé Ozo truly	dún!mwú: pound	n èmà yam	gé!lé truly	khi <u>é</u> !n sell
	b.	*Òzó gèlé Ozo truly	lè èvbàré cook food	gèlé truly	r <u>è</u> eat	

I conclude that there must be something about the relationship between the two E positions that is not captured by stating a condition that the lower E position is inert when the upper one is filled. Consequently, I reject such an approach.

Another possible explanation for the ungrammaticality of (22) and (26) comes from adopting some of the ideas about adverb licensing in Cinque (1997). Based on this approach, the claim would be that the adverbs *gèlé* and *gìégié* are licensed in hierarchically ordered functional projections. This sort of explanation would rely on the adverb order in (25) as the basic reflection of functional hierarchy which is, accordingly, violated when the same adverb is repeated twice within the same clause as in (22) and (26).

Attractive as this approach may seem, it also makes incorrect predictions. For example, it predicts on the basis of (25) that it would be ungrammatical to have a consequential SVC where the order between the two adverbs is reversed, and this is contrary to fact. This is illustrated in (27):

- (27) a. Ôzó gé!lé dé íké!ké gié!gié f í Ozo truly buy bicycle quickly ride 'Ozo truly bought a bike and quickly rode it.'
  - Òzó gié!gié dé íké!ké gé!lé f í
     Ozo quickly buy bicycle truly ride
     'Ozo quickly bought a bike and truly rode it.'

In these sentences in (27), two events are present as indicated by the fact that there are two filled adverb positions and the adverbs are freely reversed without any implications for the grammaticality of the sentences. Thus, neither a strict adverb hierarchy account nor stipulating co-occurrence restrictions can fully resolve the puzzle whereby the same adverb cannot occur in the two E positions in the same clause.

As an alternative, I take the strongest position on the analysis of the lower E position and propose to resolve this problem from the point of view that it has to do with the internal structure of the macro event expressed by the two verbs in the consequential SVC. Therefore, I propose an analysis of event binding (cf. Travis 1994) that is based on the fact that when the first E is overtly realized, it necessarily quantifies over the first event, and the second event that is immediately dominated by the lower E, but never vice-versa (the binding chain cannot be bottom-up). A useful insight into my proposal comes from the

analysis of a related quantification problem in Formal Semantics where two quantifiers compete for the same target (x) as stated in (28).<sup>17</sup>

$$\forall x (F x \longrightarrow \exists x G(x))$$

According to (28) there are two quantifications present: universal and existential and both quantifiers target the same variable which is G(x). The solution that is offered for this problem is that the lower quantifier counts and binds the G(x). I adopt a modified version of this assumption and propose that what is happening in (22) and (26) is that there are two quantifications over one 'big' (macro) event represented as E that itself consists of two subevents (event variables), e1 and e2 represented by (x) as in (29).

$$\exists_{\mathbf{x}} (\mathbf{F}(\mathbf{x}) \longrightarrow \exists_{\mathbf{x}} \mathbf{G}(\mathbf{x}))$$

(29) implies that there are two event variables contained within the quantification scope of a macro-operator  $\exists(x)$ . The existential quantifier which is in the first structural E position obligatorily binds the lower structural E operator. Thus, when the top E is overtly filled it quantifies over the macro event that is expressed by both verbs in which case the lower E is bound (rendered redundant) and so it does not require any separate quantification. However, when the first E position is null or filled by a different I-type adverb, then the second E operator counts as the quantifier for the second event. This confirms the observation based on an I-type adverb before the first verb that there is a macro-event that is expressed by the two verbs in the consequential SVC. Thus, given the fact that there is the sense of a single (macro) event it follows that the same variable cannot be quantified-over simultaneously by two separate operators. Therefore, I-type adverbs of the same kind

<sup>&</sup>lt;sup>17</sup> I am grateful to Brendan Gillon for discussing this problem with me and showing me a point where a semantic solution seems to resolve a syntactic problem.

cannot occur in the two E positions (22) (and also (26)). Consequently, the semantic representation of the consequential SVC is as in (30).

(30) The consequential SVC  $\exists_E (\exists e_1 \exists e_2[Buying(e_1) \& Agt(e_1,Ozo) \& Th(e_1,èbé)] \\ \& [Selling(e_2') \& Agt(e_2,Ozo) \& Th(e_2,èbé)] \\ \& [E "consists of" (e_1, e_2)]]]$ 

(30) shows the binding relation between the two E positions as a pair of sub-events (e1,e2) of a single set (E) and that the two events basically share the same syntactic properties.<sup>18</sup>

In terms of the syntactic structure of the consequential SVC, what we take away from this discussion of I-type adverb distribution and licensing is that it cannot be a simple conjunction on a par with CCs. Furthermore, we also know that there is an asymmetric c-command relation between the two E positions. Therefore, we are left with the choice of a complementation structure (31) which is comparable to the position that Collins (1997) takes (with no discussion of functional projections).



An alternative structure to (31) would be an adjunction structure where the lower EP position is adjoined to VP1 and they are both dominated by the first E as illustrated in (32).

<sup>&</sup>lt;sup>18</sup> I will come back to address the specific issues relating to the null object of the second verb in section 2.5 where I discuss object sharing, and my account of Agent argument follows closely in section 2.6.



The obvious question at this point is which of these two structures is the correct one for the consequential SVC? The answer to this question will have to wait until sections 2.5 and 2.6 by which time we have fully come to terms with some of the other aspects of the consequential SVC along with the other constructions, e.g., making a choice about structure would depend greatly an the account of the facts of 'object sharing' and even in some ways on the characterization of the Agent (subject). Before moving on to discuss the distribution of N-type adverbs, I will quickly present a piece of empirical evidence in support of the foregoing discussion about the two E positions and the tentative analysis that has been proposed.

This evidence is aimed at reinforcing the analysis of the relationship between the E positions by correlating the facts from the distribution and licensing I-type adverbs with that of the iterative morpheme. This will be shown to be consistent with the distinction between consequential and CCs. Consider the following:

(33)\*Ôzó gi<u>é</u>!gi<u>é</u> ghá dún!mwún èmà ghá khién a. Ozo quickly Iter pound vam Iter sell 'Ozo quickly pounded the yam repeatedly and sold it repeatedly.' b. \*Ozó gi<u>é</u>!gi<u>é</u> ghá lé èvbàré ghá ré Ozo quickly Iter cook food Iter. eat 'Ozo quickly cooked the food repeatedly and ate it repeatedly.' (34) Ozó gi<u>é</u>!gi<u>é</u> ghá gb<u>ó</u>!<u>ó</u> ívin ghá bó!ló ókà **a**. Ozo quickly Iter. plant coconut Iter, peel com 'Ozo quickly planted the coconut repeatedly and [also] peeled the corn repeatedly.' b. Òzó gi<u>é</u>!gi<u>é</u> ghá gbé <u>è</u>khù ghá lá!á òwá
 Ozo quickly Iter hit door Iter. enter house
 'Ozo quickly hit the door repeatedly and [also] entered the house repeatedly.'

The contrast illustrated in (33-34) clearly replicates the facts discussed above concerning the occurrence of the same adverb in two positions within a single clause. In the consequential SVC (33), observe that it is ungrammatical for the iterative morpheme to occur in the two E positions and I take this as further evidence of the fact that both verbs express a single complex event, in the same clause.<sup>19</sup> However in CCs as in (34), such double appearance of the same element is allowed and this implies that CCs must be made up of two separate events, and by implication, in two clauses.

A perfect illustration of the paradigm of comparison with I-type adverbs would be to provide examples where the sentences in (33) are grammatical when the items in the two E positions are varied, but unfortunately there are no other iterative type morphemes in the language, i.e., the only heads of the category E are ghd and a.<sup>20</sup> In the absence of such evidence, it is still informative to observe the contrast between the consequential SVCs (33) and the CCs (34) based on the function and properties of the two E positions. According to my theory of I-type adverb licensing, the ungrammaticality of (33) comes from the fact that the top E position binds the lower E, and so once the top E is overtly filled it quantifies over the event which is denoted by both the first verb and the second verb. This makes the lower E 'redundant' in such cases; hence the ungrammaticality. However, in the CC where two distinct events are conjoined in a symmetrical structure each event can be separately quantified over by the iterative morpheme in each of the two E positions respectively. Thus, for example (34a) has a meaning like the following: Ozo quickly planted coconuts

<sup>&</sup>lt;sup>19</sup> Note that the adverb is always needed to show clearly that the  $gh\dot{a}$  is iterative and not future tense  $gh\dot{a}$  since both occur contiguously to each other.

<sup>&</sup>lt;sup>20</sup> There is, of course, the iterative [IV] affix (cf. Stewart 1997, Aikhionbare 1988, 1989 Agheyisi 1990 etc.) but since it is an affix it is irrelevant to the point being made here and its discussion would take me far afield. I will not go into any detailed discussion.

over and over again and (being the hard worker that he is) he also engaged in repeated actions of corn-peeling.

Based on the foregoing, we find that there is strong empirical evidence in support of the distinction between consequential SVCs and CCs and the analysis thereof.

# 2.2.3 N-type Adverb after First Verb

In section 2.2, we observed the fact that the I-type adverb can reveal a structural split by showing that there is a functional projection between the two verbs (along with their complements) in consequential SVCs and CCs, but not resultative SVCs. In this section, I will focus on determining the internal structure of VPs, specifically to find out VP boundaries, based on my assumption that the N-type adverb right-adjoins to VP and thus marks the right edge of the VP.<sup>21</sup> Furthermore, based on Parsons (1990) view of adverbs as predicates of events, the distribution of N-type adverbs should also correspond to the nature of event structure that have been identified from the distribution of I-type adverb.

Let us begin, then, by looking at the distribution of N-type adverbs in the position after the first verb plus its object in the three different constructions. Consider the following sentences:

(35)	а	*Òzó kòkó Àdésúwà <u>ègìégìé</u> mòsé Ozo raise Adesuwa quickly be-beautiful
	<b>b</b> .	*Òzó sùá <u>ògó èg</u> ì <u>ég</u> ìé dé Ozo push bottle quickly fall
(36)	a.	Òzó dùnmwún èmà <u>ègiégié</u> khi <u>ènné</u> Ozo pound yam quickly sell+PL 'Ozo pounded the yams quickly and sold them.'
	b.	Òzó lé èvbàré <u>ègiégié</u> ré Ozo cook food quickly eat 'Ozo cooked the food quickly and ate it.'

<sup>&</sup>lt;sup>21</sup> It is possible to state the general licensing of N-type adverbs as right-adjuncts to phrasal category, which will contrast them with I-type adverbs that left adjoin to functional heads.

- (37) a. Òzó gbé <u>è</u>khù <u>èg</u>ì<u>ég</u>ì<u>é</u> làá òwá
   Ozo hit door quickly enter house
   'Ozo broke the door quickly and he entered the house.'
  - b. Òzó gbòó ívìn ègìégìé bòló ókà
     Ozo plant coconut quickly peel corn
     'Ozo planted the coconut quickly and he peeled the corn.'

As the data above indicates, there is a contrast between resultative SVCs (35), and consequential SVCs (36) and CCs (37). In the resultative SVC, we observe that the N-type adverb cannot occur between the verbs. In particular, (35a), for example, shows that the first verb  $k\partial k\dot{o}$  (raise) and the argument  $\dot{A}d\dot{e}s\dot{u}w\dot{a}$  do not constitute a VP apart from the second verb. If they did, then it should have been possible for the N-type adverb to occur as a right adjunct, contrary to fact. Thus, on the basis of the ungrammaticality of the sentences in (35), I propose that there is no VP constituent made up of the first verb and the shared NP argument in the resultative SVC. Furthermore, the failure of the adverb to pick out a VP (35) suggests that the first verb does not by itself denote an event. This conclusion is previously implied in the structure in (18) and repeated here as (38) for the sentence in (35b).





As the structure in (38) shows, the chain formed by the first verb takes as its complement a V' that contains the second verb and so it does not constitute a VP along with the theme (NP) argument to the exclusion of the second verb.

However, in (36-37) we observe that it is possible for an N-type adverb to occur between the verbs in consequential SVCs and CCs respectively. In the consequential SVC (36), the N-type adverb occurs after the first verb plus the object NP before the second verb. Since an N-type adverb may only adjoin to the right edge of a VP, I conclude that the first verb along with its NP complement do constitute a VP in this case. Therefore, the first verb in the consequential SVC denotes a distinct event of which the N-type adverb can be a predicate. This analysis is consistent with the meaning of the sentences. For example, (36a) means that Ozo pounded the yams very quickly, and we know for a fact that he sold them without any implication whether the selling event was quick.

On the basis of the distribution of N-type adverb after the first VP, we now have the first piece of evidence with which to decide on the correct structure of the consequential SVC between a complementation structure (31) and an adjunction structure (32). If we were to chose a complementation structure as in (31), then we will be unable to provide a simple account of the fact that the N-type adverb adjoins to the right of first VP before the second verb. (31) makes the wrong prediction that the N-type adverb cannot occur between the VPs. A complementation structure fails to provide an account of the distribution of Ntype adverb between the VPs.

This problem with a complementation structure (31) does not arise in the case of the adjunct structure (32). In fact, the adjunct structure (32) predicts that the first verb plus its object constitutes VP1 to which the second EP is adjoined. Thus, we derive the correct word order in (36) based on an analysis where the N-type adverb adjoins to the right of VP1 before the second verb. This is illustrated in (39).

## The consequential SVC



Turning now to CCs in (37), here too we observe that the N-type adverb can occur after the first verb plus its object. In terms of meaning (37a), for example, has the reading that Ozo broke the door rather quickly and then walked into the house (perhaps slowly after seeing through the open door that what he was looking for was not in the house). I take the data in (37) illustrating the fact that the N-type adverb occurs after the first verb plus its object as evidence of a VP boundary. This too is consistent with the proposal that the first verb denotes a separate event. All of these facts fit well with the structure in (24b) repeated here as (40) in which there is a conjunction of two EPs.<sup>22</sup>

## (40) <u>covert coordination</u>



 $<sup>^{22}</sup>$  As I indicated before, CCs could also involve the conjunction of VPs in some languages and such fact would not be at variance with the distribution of the N-type adverb.

There is a prediction from this analysis of N-type adverbs in consequential SVCs and CCs, which is that it is possible to have both the N-type and I-type adverb co-occur between the verbs. This prediction can only be derived from an analysis based on an adjunction rather than a complementation structure for the consequential SVC. As it turns out, this prediction is borne out. Consider the following sentences:

- (41) a. Òzó dùnmwún èmà ègìé gìé gé!lé khién
   Ozo pound yam quickly truly sell
   'Ozo pounded the yams quickly and truly sold them.'
  - b. Ôzó lé èvbàré ègìé gìé gé!lé ré
     Ozo cook food quickly truly eat
     'Ozo cooked the food quickly and truly ate it.'
- (42) a. Òzó gbé <u>èkhù <u>ègié gié</u> gé!lé lá!á òwá
   Ozo hit door quickly truly enter house
   'Ozo broke the door quickly and [he] truly entered the house.'
  </u>
  - b. Ôzó gb<u>òó</u> ívìn <u>ègiégié</u> gé!lé bó!ló ókà
     Ozo plant coconut quickly truly peel corn
     'Ozo planted the coconut quickly and [he] truly peeled the corn.'

While the distribution of both adverbs in (42) is not particularly unexpected under most account, however, the same facts in (41) provide strong evidence against previous structural accounts of SVCs like the double-headed VP analysis in Baker (1989, 1991) since there is a complete split between the verbs. Depending on what is assumed to be the head, (41) could also be a problem for the asymmetric bivalent projection analysis in Déchaine (1993) since such a theory does not allow for intermediate functional projection that could disrupt the headedness in a situation in which the second verb is the head. Although Collins (1997) does not discuss functional structure, the facts in (41) are potentially problematic for a complementation analysis along the lines that I have discussed above. However, under my analysis of consequential SVCs, the distribution of manner adverbs in (41) receives a straightforward account. Thus, I will continue my analysis based on the idea that the correct structure of the consequential SVC involves the adjunction of a functional projection EP to VP1 as in (39).

## 2.2.4 N-type Adverb after Second Verb

I now turn my attention to the placement of the N-type adverb after the second verb (plus its object). The basic task is to decide whether the second verb plus its object (where this exists) constitutes a VP apart from the first verb. This will, no doubt, present a useful introduction to the analysis of object sharing in both resultative and consequential SVCs. It is important to state here that when the N-type adverb occurs after the second verb it is ambiguous between modifying the entire sentence, i.e., both verbs, or just the second verb, VP2. I propose that when both VPs are modified the N-type adverb right-adjoins to the EP (or TP), and when only a VP is modified it right-adjoins to VP. This distinction is of primary relevance to the structure of the resultative SVC and also distinguishes it from consequential SVCs and CCs. Consider the following:

(43)	а.	Òzó kòkó Àdésúwà mòsé <u>ègìégìé</u> Ozo raise Adesuwa be-beautiful quickly 'Ozo raised Adesuwa to be beautiful quickly.'
	Ъ.	Òzó dùnmwún èmà khi <u>én ègiégié</u> Ozo pound yam sell quickly 'Ozo pounded the yam and sold it quickly.'
	c.	Òzó gb <u>òó</u> ívìn bòló ókà <u>ègiégié</u> Ozo plant coconut peel corn quickly 'Ozo planted the coconut and peeled corn quickly.'

As (43) illustrates, the N-type adverb can occur after the second verb plus its object (where present) in all three constructions. However, what exactly the sentences mean differs significantly amongst them. In the resultative SVC (43a) where the N-type adverb occurs after the second verb, it is important to note that the interpretation of the sentence is not simply that 'the becoming beautiful alone was quickly'; rather the adverb necessarily modifies both of the events denoted by the first and the second verbs, i.e., 'the raising and the result of being beautiful were quick'. Therefore, I conclude that in the resultative SVC neither the first verb nor the second by themselves constitute a VP; rather they do so by both combining together with the single NP argument. This corresponds to the proposal that there is a single event in the resultative SVC which we have now confirmed is characterized by the combination of both verbs, which the N-type adverb is a predicate of . This interpretation and the distribution of the N-type adverb in the position after the second verb is consistent with the structure of the resultative SVC in (18). According to this structure, there is a V-bar that dominates the second verb to which the N-type adverb cannot right-adjoin (being licensed only as VP adjunct). On the other hand, there is a VP projection that dominates both verbs and it is at this level that the N-type adverb rightadjoins. Now, it seems possible that the N-type adverb in this sentence-final position may actually right-adjoin to the EP rather than the VP. However, there is no way of deciding this difference between EP and VP adjunction since there is a single event. As an illustration of this point, consider the structure in (44) where EP and VP are hierarchically adjacent, therefore the difference between them in terms of interpretation is semantically vacuous.

The resultative SVC & N-Advs.



In the consequential SVC (43b), the N-type adverb can occur after the second verb plus its null object where it is ambiguously interpreted. One interpretation of the adverb in this position is that only the second event denoted by the verb 'sell' was quick. This implies that the second verb plus its null object (that is coreferent with the overt NP, yam) constitutes a VP and the N-type adverb adjoins to its right.<sup>23</sup> A second interpretation of the N-type adverb in (43b) is that the joint events of pounding and selling were quick. This would be a parallel of the case in which the I-type adverb is sensitive to a macro event (E). I propose to account for this interpretation of the N-type adverb being predicated of the big event by adjoining the N-type adverb to the right of VP1 (or EP1) where it can modify the variables associated with two events, e1,e2. These various adjunction possibilities are illustrated in (45) (note that the resulting strings are all identical).

## (45) The consequential SVC & N-Advs.



 $<sup>^{23}</sup>$  Again, we could argue that the adverb adjoins to the EP2 but there would be no way to tell the difference between VP adjunction and EP adjunction in this case because there is a single event in this part.
CCs are like consequential SVCs in that when the N-type adverb occurs after the second verb it is also ambiguous. One interpretation is that only the event denoted by the second verb was quick. This implies that the second verb plus its object constitute a VP. The other interpretation of (43c) is that both the events of 'coconut-planting' and 'corn-peeling' were quick. This implies that the adverb may actually adjoin at the level of conjunction, i.e., to the maximal projection of EP that contains both EPs. This is consistent with the structure of CC (46).<sup>24</sup>



As a general conclusion to the discussion of manner adverbs, I would like to bring out some of the consequences for the analysis of SVCs and CCs. First, the fact that there is evidence from I-type adverbs for a functional projection between the VPs in the consequential SVC seriously questions all existing accounts of this phenomenon and presents some interesting insights into the phenomenon of object sharing and its analysis (see section 2.5). However, the distinction amongst 'object sharing' SVCs between resultative and consequential constructions based on I-type and N-type adverbs is an empirical discovery that needs to be further justified--especially since both types of object sharing SVCs have sometimes been assumed to have the same structure (Baker 1989, Collins 1997, etc.). A second consequence that deserves further attention is the fact that the

 $<sup>^{24}</sup>$  One important fact which this structure brings up is the issue of why an I-type adverb does not left - adjoin to the higher EP and then have the same interpretation in which both verbs are modified. The fact that this interpretation is not possible provides justification for the analysis of I-type adverbs as adjoined to E-head (and not EP).

analysis of adverbs provides a novel structural account of events in SVCs providing illustration for the distinction between single and double events (see Chapter three).

#### 2.3 Distribution of adjuncts: (locative) Prepositional Phrase

This section is intended to confirm the analysis of N-type adverbs as VP adjuncts that identify VP boundaries. Thus, the prediction is that similar adjuncts would exhibit the same pattern of facts. Therefore, I will examine the distribution and licensing of locative prepositional phrases (PPs) which are generally regarded as adjuncts. The basic assumption is that locative PPs are right-adjuncts to VP and are interpreted as VP-modifiers like the N-type adverbs just discussed.

Locative PPs are relevant not only in telling us about syntactic structure but also in confirming the claims about event composition in SVCs and CCs. The idea here is that SVCs in which the verbs must combine to express a single event will not vary in terms of location, i.e., a single event resultative SVC is made up of a 'co-effect' relationship in which there is no variation in place (or location of the events) between the verbs. In structures where the verbs may denote distinct events, they may vary in terms of the location of the two events, i.e., in both consequential SVCs and CCs the action denoted by the first verb may take place in a location different from that of the second event since there is only an implicational relationship between the events, rather than a co-effectual one. The  $\dot{E}do$  locational preposition  $vb\dot{e}$  is illustrated in (47):

- (47) a. Òzó lé èvbàré Ozo cook food 'Ozo cooked the food.'
  - Òzó (\*vbè ùkònì) lé èvbàré vbè ùkònì
    Ozo Loc. Kitchen cook food Loc. Kitchen
    'Ozo cooked the food in the kitchen.'

(47a) illustrates a simple transitive sentence and shows that a locative PP is not part of the argument structure of the verb. Thus, in (47b) the PP is merely an adjunct of place which,

like most adjuncts, can be omitted (compare (47a)); I assume that it adjoins to the right of VP (48).<sup>25</sup>



### 2.3.1 Locative PP after First verb

On the basis of the data and analysis of the locative PP in (47) let us now examine its distribution in SVCs and CCs. First, we consider the position after the first verb plus object:

(48)	<b>a</b> .	*Òzó kòkó Àdésúwà [vbè Èdó] mòsé Ozo raise Adesuwa in Benin be-beautiful
	b.	Òzó dùnmwún èmà [vbè ùkònì] khi <u>é</u> n Ozo pound yam in kitchen sell 'Ozo pounded the yam in kitchen and sold it.'
	c.	Òzó gb <u>òó</u> ívìn [vbè Èdó   bòló <u>ó</u> kà Ozo planted coconut Loc. Benin peel corn 'Ozo planted coconut in Benin and [he] peeled corn.'

The data above illustrates the contrast between single event resultative SVC (48a) and double event consequential SVC (48b), as well as CC (48c). In the resultative SVC (48a), the locative PP cannot occur after the first verb plus object. This implies that the first verb

(i)

<sup>&</sup>lt;sup>25</sup> We can confirm that  $vb\dot{e} \, \dot{u}k\partial n\dot{i}$  forms a PP constituent since it can undergo focus cleft and leave behind the locative resumptive particle  $n\dot{a}$  (i).

vbè ùkònì òré Òzó ná lé èvbàré

Loc. Kitchen Foc Ozo RP cook food

<sup>&#</sup>x27;It is in the kitchen that Ozo cooked the food'

and the object do not constitute a VP, and also that there isn't a distinct event associated with the first verb that can take place in an exclusive location. This is consistent with our previous conclusion that the first verb does not denote a distinct event.

In the consequential SVC (48b), we observe that a locative PP occurs after the first verb plus object indicating that they both constitute a VP. The interpretation of the sentence is that Ozo pounded the yam in the kitchen and sold it (at some unspecified location). This implies that the locative PP adjoins to the right of VP1 which is a separate constituent from the second verb.<sup>26</sup> It can also be inferred that the VP1 denotes a distinct event which can take place in a specified location that excludes that of the second event. This is consistent with the structure of the consequential SVC (49).



In (49), we observe that the fact about locative PP placement between the verbs in the consequential SVC gets a straightforward account based on a structure in which EP2 adjoins to the maximal projection VP1; rather than a complementation structure.

Similarly, in the CC (48c) the locative PP can occur after the first verb plus object. This further confirms the proposal that there is a VP boundary between the two verbs and also implies that the VP denotes a separate event which can have its own location. The

<sup>&</sup>lt;sup>26</sup> This PP placement fact, once again, calls into question an analysis of the consequential SVC such as Baker (1989, 1991) based on doubly-headed VP since the presence of the PP splits the two verbs apart and renders a doubly-headed VP proposal less plausible. Conceivably, Baker's analysis would need to say either that PP is an immediate daughter of VP [VP V NP V' PP], or that V-bar could extrapose; both options requiring extra steps in their derivation compared with the separate VP structures such as the one I have proposed.

interpretation of (48c) is that Ozo planted coconut in Benin and peeled corn (at some other unspecified location). This fact of locative PP insertion is consistent with the structure that I have proposed for CCs (cf. 24b) which involves the conjunction of EPs.

The conclusion based on the distribution of locative PP after VP1 is that the facts are perfectly parallel to those with N-type adverbs, so both analyses can be generalized to account for adjuncts in general in SVCs and CCs. Specifically, we note that the first verb plus object do not form a VP constituent in the resultative SVC but they do so in both consequential SVC and CCs.

#### 2.3.2 Locative PP after Second Verb

When the locative PP attaches after the second verb, we observe further evidence confirming the analysis of N-type adverbs and the distinction between VP adjunction and EP-conjunction in two-event constructions like consequential SVCs and CCs. Consider the following:

(50)	a.	Òzó kòkó Àdésúwà mòsé vbè Èdó Ozo raise Adesuwa be-beautiful in Benin 'Ozo raised Adesuwa to be beautiful in Benin. '
	b. OR	Òzó dùnmwún èmà khi <u>é</u> n vbè Èdó Ozo pound yam sell in Benin 'Ozo pounded the yams and sold them in Benin.' 'Ozo pounded the yams [elsewhere] and sold them in Benin.'
	c.	Òzó gb <u>òó</u> ívìn bòló <u>ó</u> kà vbè É!kó Ozo planted coconut peel corn Loc. Lagos 'Ozo planted coconut and [he] peeled corn in Lagos.'

Like the N-type adverb, a locative PP can occur after the second verb of the resultative SVC (50a). The meaning of the sentence with the PP in this position is consistent with the "co-effect" analysis of the verbs; it means that both the 'raising' expressed by the first verb and the result of 'being beautiful' that is expressed the second verb take place in the same location. It is not possible to have an interpretation of (50a) in which the locative PP

modifies only the second verb such as Ozo raised Adesuwa elsewhere and then she became beautiful in Benin. Thus, the second verb does not constitute a separate VP from the first verb and the object. In addition, it suggests that the second verb does not denote an event that is distinct from the first verb. These facts suggest that the locative PP, like the Nadverbs, adjoins to the VP or the EP in a single event (resultative) SVC with no distinguishable meaning difference (51). This structure and the interpretation of the locative PP after the second verb also further confirms the internal structure of the VP since no phrasal modifier can attach to the intermediate V' containing the second verb.



In the consequential SVC (50b), the locative PP can also occur after the second verb, but it has an ambiguous interpretation which is parallel to that seen with N-type adverbs; depending on the structural position in which it is adjoined. One interpretation of (50b) is that the events denoted by each of the two verbs share the same location, i.e., Ozo pounded the yam [in Benin] and sold it in Benin. This interpretation is consistent with a structural

analysis in which the locative PP adjoins to the right of EP which dominates both the first and second verbs and, therefore, modifies the events denoted by the two verbs.

A second interpretation of (50b) is one in which only the event denoted by the second verb is understood as taking place in the expressed location. This implies that the second verb (plus presumably a null object) constitute a VP, and the locative PP attaches to this VP in order to derive the interpretation given. This further implies that the second verb denotes an event which is distinct from the first verb. Both of these interpretations that I have just discussed are consistent with the structure of the consequential SVC (52), that reflects the distinction between VP-adjoined and EP-adjoined PPs. As in the resultative SVC, the locative PP may either adjoin to the relevant VP or to the EP in the reading where only a single event part is referred to. This is represented by putting the PP adjoined to EP2 in parentheses.



In a similar manner, in the CC (50c), which also consists of two events, the locative PP occurring after the second verb can either modify the second verb alone or both the first and second verbs. I take the interpretation in which the PP modifies only the second verb to be evidence that there is a VP boundary there, while the interpretation in which both verbs are modified implies that the PP can adjoin to EP. These two

interpretations are consistent with the CC structure in (53), and again, the vacuous interpretation of the PP with one part of the big event is represented in parenthesis.



### 2.4 Consequences of Adverb and PP Placements

As a conclusion to the general topic of the structure of SVCs and CCs, I would like to point out that the results from both manner adverb placement and PP insertion make a prediction for SVCs about verb commutability (Ekundayo and Akinnaso 1983). We predict that the verbs in the resultative SVCs are not commutable: the first and second verbs are strictly ordered with respect to one another. However, we predict that VP1 and VP2 are commutable in consequential SVCs and CCs based on the fact that their structures contain distinct projections of EP/VP (and given the right context of interpretation). These predictions are borne out by the following data where I switch the order of verbs in most of the examples that we have seen above in the different constructions:

(54)	a.	*Òzó mòsé kòkó Àdésúwà Ozo be-beautiful raise Adesuwa	'resultative SVC'
	b.	*Òzó dé <u>ògó</u> sùá Ozo fail bottle push	19
(55)	a.	Òzó khi <u>é</u> n ìyán dùnmwún Ozo sell yam pound 'Ozo sold the yams and pounded them.'	'consequential SVC'

- b. Ôzó kpèć èmà d<u>é</u>
  Ozo beat drum buy
  'Ozo played the drum (and then) bought it.'
- (56) a. Ozó bòló <u>ó</u>kà gb<u>òó</u> ívìn
  Ozo peel com plant coconut
  Ozo peeled corn and [he also] planted coconut.'
  - Òzó làá òwá gbé <u>è</u>khù
    Ozo enter house hit door
    'Ozo entered the house and [he] broke the door. '

As can be observed from the resultative SVC (54), the sentences are ungrammatical when the verb order is reversed. This follows from my analysis in which both verbs are daughters of the same VP with the second verb as sister to the tail of the chain that involves the first verb. This is a highly constrained structure which does not allow the insertion of word level or phrase level categories between them and as such they cannot freely permute. In essence, therefore, the verbs in the resultative SVC are an iconically ordered pair of a set that expresses a single event. I will come back to the aspectual properties of the resultative SVC in section 2.8. However, in consequential SVCs (55) where the verbs have been shown to project VPs, they are freely commutable given the right contextual interpretation (although, temporal succession is affected). For example, in the consequential SVC (55b) it can be imagined that Ozo went to a store where drums are sold and he had to test the drum he was interested in buying by first playing it, and afterward, he bought it. Observe that all of the trappings of object sharing are still intact in (55b) since there is a missing object of the second transitive verb.

Verb commutability is a trivial consequence of the CC that is predicted from the nature and structure of the construction, even with difficult sentences to interpret the effect of a pause before the second verb makes verb commutability possible. Thus, for example, (56b) has the reading in which Ozo enters the house and thereafter breaks some door (inside the house).

#### 2.5 Object Sharing

This section focuses on the analysis of objects in SVCs (cf. Stewart, 1963, Sebba 1987, Baker 1989, 1991, Déchaine 1986, 1993, Collins 1997, Carstens 1988, Lefebvre 1991, Li 1991, Larson 1991, Campbell 1996, etc.). The facts of adverb placement, PP insertion and the iterative morpheme provide two kinds of general evidence for distinguishing amongst object sharing SVCs. First, they suggest that there is an iconically ordered pair of verbs of a single VP which is the resultative and that this is different from the consequential SVC in which there are two separate VPs that are commutable. Second, that there is a single EP in the resultative implying a single event but the consequential SVC contains two asymmetric E positions which imply some kind of complex double events. CCs, on the other hand, involve the conjunction of two symmetric EPs.

These distinctions between the different constructions with respect to EPs will now be shown to have deeper structural implications that will bring new evidence to bear on the analysis of true internal object sharing. Basically, I will argue that there are two kinds of object sharing SVCs namely; resultative and consequential. In this regard, I will propose an analysis in which object sharing is mediated by an empty category, *pro*, in the consequential SVC on the one hand, but that there is true internal object sharing, as in Baker (1989), in the resultative SVC with a single syntactic object for both verbs and no empty category, *pro*, contrary to Collins (1997). Furthermore, I propose that CCs never have true internal object sharing because the empty category, *pro*, cannot be licensed in a CC structure. Consider the following sentences:

- (57) a. Òzó kòkó Àdésúwà mòsé Ozo raise Adesuwa be-beautiful 'Ozo raised Adesuwa to be beautiful.'
  - Òzó dùnmwún èmà khi<u>é</u>n
    Ozo pound yam sell
    'Ozo pounded the yams and sold it.'

Ozó gbòó ívìn bòló ókà
 Ozo planted coconut peel corn
 'Ozo planted coconut and [he] peeled corn.'

In the resultative SVC (57a) there is a single object  $\lambda destimation destimation destination of the transitive first verb and the unaccusative second verb. In the consequential SVC (57b) there is also a single overt object <math>ema$  for both transitive verbs and this seems to be a violation of the argument structure of the second verb. The CC (57c) is very different in this respect, since each verb has its own object and so there can be no true object sharing (I will return to the structural account of this fact in section 2.5.2 below ). Therefore, in the remaining part of this section I will only focus on the distinction between resultative and consequential SVCs. Consequently, two related questions will be examined: (a) What tests show the difference in the realization of object sharing between the two SVCs? (b) How does my proposed analysis of these two SVCs relate to the issue of object sharing?

#### 2.5.1 Evidence for Empty Category : Adverbial Particle 'tôbórè'

Thus far, I have been assuming that there is a null object in the projection of the second verb that is coreferential with the overt object NP in the consequential SVC, and that there is only one object NP in the resultative SVC. Therefore, the specific goal of this section is to provide evidence that supports the claim that there is an empty category involved in object sharing in the consequential SVC but not in the resultative SVC. This is based on the behavior of an adverbial particle in  $\dot{E}d\delta$ ,  $t\partial b \underline{\delta} r \dot{e}$  (himself/herself/itself) which has some of the properties of an emphatic anaphor. Although this particle does not occur in argument positions since it typically occurs as NP adjunct, I assume a general analysis of anaphors based on the simplest assumption about Condition A of the binding theory (Chomsky, 1981) that requires it to be locally bound within the governing category. A governing category may be defined as the whole TP or VP-shell (cf. Larson's 1988)

complete functional complex (CFC)). Now consider the behavior of  $t \partial b \underline{o} r \dot{e}$  in simple sentences with object NP coreference:

- (58) a.  $\dot{O}z\dot{o}j$  lé  $(z\underline{e}_k t\dot{o}\underline{b}\underline{o}r\dot{e}_k)$ Ozo cook rice itself/himself 'Ozo cooked the rice by itself (alone) OR 'Ozo cooked the rice by himself'
  - b.  $\underline{\partial g o}_k$  dé  $t_k$  t $\partial b o t e_k$ bottle fall itself 'The bottle fell by itself'
  - Àdésúwàk mòsé tk tòbórèk
    Adesuwa be-beautiful herself
    'Adesuwa is beautiful, herself alone'
  - d. Òzó gbé ékítà<sub>k</sub> nè úyì tá w<u>èé</u> ìs<u>òkèn</u> hòó *pro*<sub>k</sub> tòb<u>ó</u>rè<sub>k</sub> Ozo hit dog that Uyi say that Isoken want pro itself Ozo beat the dog that Uyi said that Isoken wants (itself)

Based on (58a), we observe that the adverbial particle  $t\partial b \underline{o} r \dot{r}$  can occur after the object NP, and in this position it ambiguously takes either the object or the subject as its antecedent.<sup>27</sup> Focusing on the object reading for now, I assume that (58a) has an analysis in which the adverbial particle adjoins to the right of the object NP within the VP that contains both of them. This analysis predicts that an N-type adverb or locative PP cannot occur between the object and the particle on an object sharing reading and such a prediction is borne out (59).

(59) \* $\dot{O}z \dot{o}$  lé  $iz \dot{e}_k \dot{e} g \dot{e} g \dot{e} g \dot{e} / v \dot{e} \dot{o} w \dot{a}$  to  $b \dot{o} r \dot{e}_k$ Ozo cook rice (quickly/at home) itself

The ungrammaticality of the coreference between the object and the particle with either an N-type adverb or locative PP in-between confirms the fact that the particle adjoins to the right of an NP. Based on (58b,c), we observe another interesting property concerning the licensing of the  $t \partial b \underline{o} r \dot{r}$  particle. In particular, we learn that it can also be coreferent with an

<sup>&</sup>lt;sup>27</sup> I will discuss the behavior of this anaphor with subject NP in section 2.6.3

empty category. In other words, the trace of the internal object of unaccusative verbs can serve as the antecedent for the adverbial particle.<sup>28</sup> This is illustrated in (60) for (58c).



Turning now to (58d), we are able to confirm the analysis of the adverbial particle based on the grammaticality of the coreference between the object of the matrix clause and that of the most embedded clause. The only account of this coreference fact is that there is a null *pro* in the embedded clause that is co-indexed with the object of the matrix clause as a result of relativization and  $t\partial b \underline{o} r e$  adjoins to this null object NP.<sup>29</sup> Therefore,  $t\partial b \underline{o} r e$  can be used to reveal the position of an otherwise null NP.

On the basis of this analysis of the licensing of the adverbial particle  $t \partial b \underline{o} r \dot{e}$ , let us examine object sharing SVCs to find out if there is any difference between resultative and consequential. Consider the following :

- (61) \*Ózó kòkó Ádésúwàk mòsé --a. tòbórèk Ozo raise Adesuwa be-beautiful herself \*Ôzó sùá <u>ògó k</u> dé --- tòb<u>ó</u>rèk b. Ozo push bottle fall itself (62)  $\dot{O}z\phi d\underline{e} iyan_k dunmwun pro toborek$ a.
- Ozo buy yam pound pro itself 'Ozo bought the yam and pounded it (itself).'

<sup>&</sup>lt;sup>28</sup> See Baker and Stewart (1997a) for arguments which suggest that stative verbs in Èdó are unaccusatives. <sup>29</sup>The verb 'want' in (58d) can have a null object which I assume, following Baker and Stewart (1997b), is pro rather than NP trace (when the overt object has been wh-moved for focus or topicalization), and the  $t ob \underline{c} r e$  anaphor occurs to the right of the empty category bearing a coreference relation with it.

b. Òtàsówié dé éwùk yó prok tòbórèk Otasowie buy dress wear pro itself 'Otasowie bought the dress and wore it (itself).'

Based on the analysis of the unaccusative verbs in (58b,c) and the structure in (60), we predict that in the resultative SVC the adverbial particle should be able to occur after the second verb and this is contrary to fact. In (61a,b) we observe that it is ungrammatical for the adverbial particle to occur after the second verb in the resultative SVC. The contrast between (58b,c) and (61a,b) is direct evidence against Collins (1997) who claims that the resultative SVC contains an empty category that is generated in the Spec of the second VP. The evidence is simple. It has been established that *tobóre* can adjoin to the right of *pro* in a simple sentence with an unaccusative verb (58b,c), but it fails to do so in the resultative SVC (61a,b). The same facts apply to the sentences in (63):

- (63) a. Òzók dé tòb<u>ó</u>rèk
  Ozo fall himself
  'Ozo fell by himself.'
  - b. \*Òtà sùá Òzók dé tòbórèk Ota push Ozo fall himself

(63a) replicates the fact that the adverbial particle can occur in the trace position of the internal object of an unaccusative and be coreferent with it, and this is in sharp contrast with the resultative SVC where the second verb is unaccusative (63b). I have taken the ungrammaticality of sentences like (63b) as evidence that there is no empty category in the resultative SVC. Consequently, I conclude that there is no evidence for an empty category in the resultative SVC and object sharing is only compatible with a single NP object. Therefore, I reject the analysis of the resultative SVC in Collins (1997). <sup>30</sup>

Turning now to the consequential SVC (62), we observe that the distribution of the adverbial particle suggests the presence of an empty category, *pro*, which serves as the object of the transitive second verb and is co-indexed with the overt object in VP1. This is represented in (64) for (62a).

<sup>&</sup>lt;sup>30</sup> More arguments against Collins' (1997) analysis of the resultative SVC are presented in section 6.2.



This structure in (64) provides the basis for a contrast with the resultative SVC (61) and (63) where we observe that although the tob<u><u>ó</u></u>re particle can occur after an unaccusative verb in isolation (simple clause), it fails to occur in the resultative SVC where the second verb is unaccusative. The conclusion is that there is no empty category involved in object sharing in the resultative SVC. Contrastively, in (62) based on the fact that  $tob\underline{o}re$  particle can occur after the transitive second verb, I conclude that this is clear evidence that object sharing is mediated by an empty category in the consequential SVC (see Baker and Stewart (1997b) for some theoretical elaboration of this analysis of the consequential SVC). I turn now to the issue of how the empty category, *pro*, is licensed.

# 2.5.2 Null pro licensing: unaccusativity versus transitivity

In this section, I will present further evidence in support of the proposal that resultative SVCs do not contain a *pro* empty category. This is based on an examination of the issues of the licensing of *pro* and the unaccusative-transitive contrast in the position of the second verb between resultative and consequential SVCs. Recall the generalization from Chapter one that the second verb of a resultative SVC is typically unaccusative while transitive verbs are favored in corresponding position in consequential SVCs. I now

69

provide a structural account for this contrast based on Rizzi (1986:524) and propose that it derives from Case-licensing of the null pro (65).<sup>31</sup>

(65) pro licensing condition

pro is Case-marked by X<sup>o</sup>.

My basic claim is simply that (65) is possible with the null object of a transitive verb but not with an unaccusative verb since unaccusatives, by what has become known as Burzio's generalization, cannot assign accusative Case (cf. Burzio 1986). Therefore, (65) is satisfied only by a transitive verb but not an unaccusative verb, i.e. pro-drop is possible when the second verb is a transitive verb but not an unaccusative.

The empirical evidence in support of the proposal that unaccusative verbs cannot license pro-drop while transitive verbs can, comes from looking at other contexts of prodrop in the language such as the conditional construction and donkey anaphora (cf. Collins 1997, Baker and Stewart 1997b). In these sentences, I imitate both the resultative notion as well as the consequential notion:

- kòkò ékítà; èsé, \*proj/íràn ghá mó!sé \*proj/tj (66) a. Ozó ghá Ozo COND raise dogs well, \*(s/he) they will be beautiful 'If Ozo raises dogs very well, they will become beautiful.'
  - $\dot{O}z \dot{O}_k$  ghá kòkò ìyán; ní!bún,  $\dot{O}_k$  ghá dún!mwún proj b. Ozo COND gather yams many, he will pound 'If Ozo gathers enough yams, he will pound them.'

In these sentences there are two clauses that are linked by some kind of event quantification which is the conditional (COND) in the matrix clause (cf. Heim 1982, Baker and Stewart 1997b). In particular, the sentences illustrate the significant contrast between unaccusatives

Null pro licensing condition (Baker and Stewart, 1997b) (i) pro is licensed in Edó if and only if; (i) It is governed by a verb, and (ii) It is locally bound by an operator

(formal licensing) (identification of content)

<sup>&</sup>lt;sup>31</sup> This analysis builds on Baker and Stewart (1997b) who propose the following conditions on null pro licensing:

and transitives in terms of the licensing of pro. In (66a), a resultative notion is implied by the composition of the two verbs and the second verb is an unaccusative verb, however, we observe that the null pro object cannot be co-indexed with the indefinite object of the matrix clause  $\acute{ekita}$  (dogs) because pro NP has to move away from the V-governor for Case. However, when we put an overt pronoun iran (they) in place of pro in the subject position the sentence is acceptable with a coreference reading between the indefinite object of the matrix clause and the overt pronoun in subject position that must move there for Case reasons. Therefore, I conclude that an unaccusative verb cannot license pro arguments in Èdó. So, there can be no empty category, pro, in the structure of the resultative SVC where the second verb is always unaccusative contrary to the analysis in Collins (1997). Object sharing in the resultative SVC involves a single structural NP argument which is governed by the two verbs. This is formalized as a condition on 'true' internal object sharing in SVC (67).

#### (67) <u>True Internal object sharing in SVCs</u> A single structural NP is the object of two verbs, the second of which must be unaccusative

Turning our attention to the sentence in (66b), we see clearly that the object of the transitive second verb can be *pro*. This implies that an empty category analysis is compatible only with the consequential SVC where the second verb is always transitive. This is possible, based on (65), because the transitive verb can assign accusative Case to its object. I assume the analysis of the identification of *pro* in Baker and Stewart (1997b) that is based on an indexing relation between two operators: a top one and the bottom one, as represented in the structure in (68).

EP1 E' Spec VPI Ek VPĺ EP2 NP Spec E' dé iyánı ÉŁ VP2 ŇΡ dùnmwún proj

(68)

This analysis of the identification of *pro* draws on the distinction between resultative and consequential SVCs with respect to the E position before the second verb. Thus, the absence of the lower E in the resultative SVC implies that there is no local operator that can identify the content of *pro* and so *pro* cannot be licensed. However, there is evidence that there are two E positions in the consequential SVC and the first E asymmetrically quantifies over the lower E position. It is assumed that this lower E is referentially dependent on the first E in a sort of quantifier-indexing relation (cf. Heim, 1982). Therefore, the lower E is a local operator that binds *pro* and this is only possible in the consequential SVC, but not the resultative. Now, we can extend the foregoing discussion to the issue of what forces [NP V1 NP V2 *pro*] to be analyzed as a consequential SVC rather than a CC. I propose that this difference should be linked to the fact that the right context for the licensing of null *pro* in Èdó is not met in a CC structure where there are two symmetric EPs ( see Baker and Stewart 1997b for elaborate discussion of this and other related issues).

As a conclusion to this discussion of object sharing, I would like to highlight the fact that based on the condition on *pro* licensing (65), we now have a structural account for the observation that the second verb of the resultative SVC is typically unaccusative while that of the consequential SVC is always transitive. Therefore, we have a well-rounded argument for the distinction between resultative and consequential SVCs and for the fact that there is an empty category *pro* in the latter but not in the former.

#### 2.6 Subject NP

A cursory look at all the analyses of SVCs reveals that the subject position has been given relatively little attention. Very often, its analysis is determined by the theoretical framework assumed, which in any case merely seeks to replicate a very traditional idea that is based on the descriptive fact that the verbs in series share a single structural subject (cf. Bamgbose 1973, 1974, Awobuluyi 1973, Schachter 1974, Stahlke 1974, Williamson 1965, Bendix 1973 etc.). This idea has been largely incorporated into subsequent analysis, thus, for example, Baker (1989, 1991) and Sebba (1987) basically accept the traditional view by base-generating a single subject in the Specifier of IP, while Collins (1997) and Larson (1991) adopt an analysis of subjects based on the VP internal subject hypothesis (Kuroda 1988, Sportiche 1988 etc.) where the subject is generated in the Specifier of an empty VP into which the first verb raises for external theta role assignment (cf. Larson 1988).

Against this background, this section intends to investigate more carefully the notion of subject of a clause and how this relates to Agent of an event. This analysis will be done in three parts: the preliminary discussion centers on the syntactic evaluation of subjects and the semantic interpretation of Agents by looking at what the SVC sentences mean from the perspective of the 'doer'. Then, the next two parts provide empirical and theoretical grounding to the intuitions about subject and Agent, and concludes on this basis that there are differences between SVCs and CCs.

#### 2.6.1 Interpretation and Analysis of 'Subject'/'Agent'

This primary goal of this section is to provide a detailed description of SVC sentences from the point of view of how the events or actions denoted by the verbs are interpreted by native speakers, i.e., the psychological state of the doer of the actions described by the chain of verbs and how the subsequent events denoted by the verbs are

perceived. I assume the standard interpretation of 'subject of' as a structural notion represented by the NP that is assigned nominative Case and occurs on the surface in Spec of IP or TP (in a Nominative-Accusative Case system like English). On the other hand, the Agent of an event is roughly the entity that brings about a state of affairs e.g., in a sentence like 'The malaria killed Bill' [The malaria] is interpreted as the entity that brings about the change of state which is Bill being dead. I will operate with this general idea of Agent of events but with one proviso, that the Agent is a willful causer (cf. Gruber, 1965). This will get me the general fact about SVCs the subject is typically animate, but I will come back to this point below.<sup>32</sup>

Against this background, let us consider the interpretation of the subject of a sentence in SVCs and CCs, beginning with the resultative:

(69)	a.	Òzó kòkó Àdésúwà mòsé Ozo raise Adesuwa be-beautiful 'Ozo raised Adesuwa to be beautiful.'
	b.	Èsósà gbé émá!t <u>ó</u> n p <u>èrhé</u> Esosa hit metal be-flat 'Esosa hit the metal flat.'
	c.	<u>ògó</u> dé gù <u>oghó</u> bottle fall break 'The bottle fell down and broke.'

The interpretation of (69a) is that Ozo performed one action which is 'raise Adesuwa to be beautiful'. This implies that both the 'raising' and the 'becoming beautiful' are the expressed goal of Ozo, not that he raised Adesuwa and then by chance she became beautiful'. Consequently, we conclude that there is one subject which also bears the role of Agent in (69a). This generalization also holds true for the sentence in (69b), however (69c) with two unaccusative verbs has a different interpretation which is that 'the bottle fell (due to unknown external impact) and broke'. I assume that Agent is introduced by a CAUSE

<sup>&</sup>lt;sup>32</sup> The exceptions that I know of are sequences of two unaccuatives such as ' $\underline{\partial}\underline{g}\underline{o}$  dé gu $\underline{\partial}\underline{g}\underline{h}\underline{o}$  (bottle fall break) which I also provide an analysis for below.

operator and that an unaccusative lacks a CAUSE component in its lexical decomposition, therefore in the case of double unaccusative verbs in (69c) I predict that the subject is not the Agent of the event denoted by the two verbs. This prediction is borne out because in (69c), it is clear that the subject NP is the theme (the entity that undergoes transition (Gruber 1965) and not an Agent (the causer) given the definition that Agent is a willful animate entity that brings about a state or transition. Thus, in (69c) we observe that the subject of the resultative need not be the Agent of the event denoted by the verbs in this sub-class of resultative SVCs made up of two unaccusative verbs.

Structurally, I assume that CAUSE is the same thing as Voice in Kratzer (1996) and so I propose a structure for the resultative SVC in which the external argument is introduced by a VoiceP and raises into Spec TP for nominative Case (cf. Chomsky 1993) since Èdó is like English with a Nominative-accusative system.<sup>33</sup> This would provide an elegant account for (69a,b) as represented in (70).



<sup>&</sup>lt;sup>33</sup> Comparable accounts of this same fact is that CAUSE is a small V (Chomsky, 1995) or CAUSE is an upper empty V (Larson 1988) etc.

(70) captures the fact that there is a single subject for the clause and is also consistent with the fact that the subject of the sentence bears an Agent theta role. The subject gets the agent from the head of VoiceP which has the '+Agent' feature (cf. Kratzer 1996). Observe that VoiceP is generated below EP. This is intended to express the idea that an EP which formally represents an event must contain the Agent of such event, somewhat like the VP internal subject hypothesis (cf. Kuroda 1988 and others). Aside from the fact that this proposal is intuitive, there is actually evidence based on the distribution of iterative ghá generated in E and the subject-oriented (interpretation of the) adverbial particle  $t \partial b \underline{o} r \dot{e}$  (see section 2.6.3 below) which indicate that EP must dominate VoiceP.<sup>34</sup>

However, the analysis is somewhat different for (69c) where there are two unaccusative verbs and the subject bears a theme theta role. This example brings out one of the assumptions built into the structure of the resultative SVC. Recall that the two verbs are generated as daughters of the same VP and based on Burzio's generalization, unaccusative verbs do not assign Accusative case. Thus, in (69c) the single object NP receives the internal theme theta roles of both unaccusative verbs but must move up to get Case and so it moves upward to Specifier of TP for nominative Case. I assume in this case of double unaccusatives, that the head of VoiceP is inert or '-voice', and so it does not assign an Agent role.<sup>35</sup> In this way, the NP <u>ògó</u> in (69c) ends up being the subject of the clause which bears a theme role. This is represented in (71).

(i) Òzók giégié (\*tòbórèk) ghá tòbórèk tílé èbé
 Ozo quickly Iter. himself read book
 Ozo quickly read the book himself'

<sup>&</sup>lt;sup>34</sup> In a simple sentence like (i) *ghá* occurs in E which is below Tense and *tòb<u>ó</u>rè* adjoins to the trace of an internal subject in Spec of VoiceP which is below EP.

Observe that the particle cannot occur before the E head which is occupied by gha. This implies, therefore, that the subject NP is not generated in Spec of EP but below it. I propose that this lower projection is VoiceP. Thus, there are two functional projections below Tense, EP and VoiceP.

<sup>&</sup>lt;sup>35</sup> I adopt the assumptions that underlie the term 'Holder' in Kratzer (1996), which she uses for verbs like 'own' and so I propose that -voice is a concrete illustration of the notion 'Holder'.



Now, let us examine the semantic interpretation of subjects in consequential SVC such as the examples given below:

- (72) a. Òzó d<u>é</u> LGB tié Ozo buy LGB read 'Ozo bought LGB and read it.'
  - Òzó lé èvbàré ré
    Ozo cook food eat
    'Ozo cooked the food and ate it.'
  - Òzó mú èmà kpèé
    Ozo carry drum play
    'Ozo took the drum and played it.'

The semantic interpretation of these sentences show a consistent pattern. In (72a), the sentence can only have the meaning in which Ozo intentionally set about performing one complex task which consists of two stages: buying the book (*Lectures on Government and Binding*) and reading it'. An impossible interpretation of (72a) is one in which 'Ozo went to the store where there are several books, picked up a copy of LGB with the intention of giving it to someone as a gift, then later changed his mind and read it himself'. This

interpretation is completely unacceptable for (72a). Similarly, in both (72b,c) the only possible interpretation is one in which Ozo knew what he wanted to do from the beginning and he did it: in (72b) he cooked the food with the intention of eating (and he did eat), while in (72c) he carried the drum with one intention in mind which is to play it, and play it he did. What the interpretations of these consequential SVC sentences imply is that there is only one dose of Agency for the actions denoted by the first and second verbs.

Structurally, these interpretations are consistent with the fact that there are two transitive verbs, and I assume that transitive verbs have '+Voice' feature, it follows that what licenses the complex event in the consequential SVC must also come from the way in which the two verbs combine under a single '+Voice' head.<sup>36</sup> Like in the general case of the resultative SVC, I propose that subject of the consequential SVC is generated in the Specifier of VoiceP where it is assigned an Agent role (agreeing with Kratzer's (1996) sketchy proposal based on SVC data from Ewe). This is illustrated in (73).



<sup>&</sup>lt;sup>36</sup> This is similar to the conclusion based on I-type adverb licensing that the two verbs are within the scope of the higher E head.

According to this structure, both the events denoted by the first verb and the second verb are dominated by the single VoiceP where the external argument is assigned and so by implication the Agent of both events is the same as the subject of the clause, Ozo.

Finally, let us now examine the interpretation of the subject NP in CCs with the relevant examples given in (74):

- (74) a. Òzó lé íz<u>è</u> rrí <u>ó</u>rè Ozo cook rice eat it 'Ozo cooked rice and ate it.'
  - b. Ôzó gb<u>òó</u> ívìn bòló <u>ó</u>kà
    Ozo plant coconut peel corn
    'Ozo planted coconut and peeled corn.'
  - c. Òzó vú èrhán khi<u>é</u>n <u>órè</u>n Ozo uproot tree sell it 'Ozo uprooted a tree and sold it.'

The semantic interpretation of these sentences illustrates one more difference between SVCs and CCs. All the sentences in (74) are clearly understood as a conjunction of two events (with a pause before the second verb). Thus, in (74a) Ozo cooked rice (maybe to sell it for money), and then afterward ate instead. I suppose that this is possible because CCs express a sequence of events that do not have to be related semantically, in contrast with SVCs. This interpretation difference is even clearer in (74b) where there are two unrelated objects; this has the meaning that Ozo planted coconut, and he also peeled corn, a conjunction of two events with some kind of intuitive subject associated with each event. (74c) is like (74a) where Ozo performed the first event with a different intention in mind and ends up performing the second event which happens to share the same Agent with the first and a (coreferent) overt object NP as well.

These facts suggest that there are two subject positions in CCs. I propose an Across-the-board (ATB) analysis in which the overt subject occupies the Specifier of TP and leaves traces in the Specifier of each VoiceP (75).



# 2.6.2. Distribution of Subject Pronoun

The subject position is very different from the object position in SVC languages and one of such difference is illustrated by the following contrast based on NP extraction:

- (76) a. Ôzó lé èvbàré Ozo cook food 'Ozo cooked food.'
  - èvbàré òré Òzó lé pro (\*órè) food Foc. Ozo cook
     'It food that Ozo cooked.'
  - C. Ôzó <u>ò</u>ré <u>ó</u> lé èvbàré
    Ozo Foc he cook food
    'It's Ozo that cooked food.'

According to the data above, when an NP object undergoes wh-extraction (cleft), it leaves an empty category (*pro*) behind (76b). Furthermore, (76b) shows that it is ungrammatical to have an overt object pronoun occur in the position of the "moved" object. However, when the subject NP undergoes similar movement it obligatorily leaves a subject (resumptive) pronoun behind (cf. Koopman and Sportiche 1982, 1986, Agheyisi 1990 etc.) and so subjects can never be null. This contrast suggests that subjects will always be visibly marked in Èdó, even in constructions like SVCs and CCs, while objects can be dropped as in the consequential SVC. Therefore, I assume that the subject resumptive pronoun  $\underline{6}$  will be able to occur in a subject position where such a position exists. I will use this as a test to develop my analysis based on the interpretation of subjects. I will argue that SVCs are different from CCs in that the latter involves conjunction in general and can actually be a conjunction of VoicePs, in contrast with SVCs in which there is a single VoiceP.<sup>37</sup> This analysis is based on the fact that the subject pronoun occurs above the future tense morpheme ghá which is generated in Tense (77a) except in inversion context (77b):

- (77) a. <u>Ò</u> ghá y<u>ó</u> éwù h/se will wear dress 'S/he will wear a dress.'
  - b. ghá <u>Ò</u> y<u>ó</u> <u>é</u>wù will h/se wear dress 'Who wore a dress.'

Let us now use the subject pronoun test to confirm that there is a single subject/Agent in the resultative SVC. Compare (69a) with (78):

- (78) a. Òzók kòkó Àdésúwà (\*Ók) mòsé
  Ozo raise Adesuwa he be-beautiful
  'Ozo raised Adesuwa and he became beautiful.'
  - b. \*Òzó kòkó Àdésúwà Ó mòsé
    Ozo raise Adesuwa he/she be-beautiful
    'Ozo raised Adesuwa and s/he became beautiful.'

As (78) illustrates, a subject pronoun cannot occur in the Specifier of the second verb, whether it is coreferent with the subject (78a) or not (78b).<sup>38</sup> This confirms the analysis of sentences like (69a) as resultative SVC which has a single TP, and not a covert TP

<sup>&</sup>lt;sup>37</sup> The underlying assumption is that the subject pronoun starts out from the Specifier of VoiceP but must move up to Specifier of TP prior to spell-out.

<sup>&</sup>lt;sup>38</sup> The only interpretation possible here is with a heavy pause before the subject pronoun and  $\underline{\phi}$  is disjoint from  $\partial z \phi$ , hence the sentence is then a CC with the interpretation Ozo raised Adesuwa, and he Ozo is beautiful, describing two separate eventualities, in contrast to a resultative SVC.

coordination structure.<sup>39</sup> It also confirms the structure I have proposed where the Specifier of the second V-bar is occupied by the theme object and there is no VP-internal subject position. Thus, there is only one subject position in the resultative SVC which requires a "minimal structure" that is not a conjunction of full clauses.

This conclusion can be generalized also to the consequential SVC by applying the subject pronoun test as shown in (79):

- (79) a. \* $\dot{O}zo_k$  mú èmà (\* $\underline{O}k$ ) kpèé Ozo carry drum he beat
  - b.  $\dot{O}z \acute{o}_k \acute{Q}r \acute{Q}_k m \acute{u} \acute{e}m \acute{a} (*\acute{o}) kp \acute{e}\acute{O}z \acute{O}z \acute{O}$  Foc. he carry drum (he) beat 'It is Ozo who carried the drum and played it.'

In (79a), we observe that a subject pronoun cannot occur before the second verb in the consequential SVC. This implies that there is no subject position that is associated independently with the second VP. In fact, (79a) cannot have the covert coordination reading. The lack of covert coordination reading is consistent with the fact that the second verb of the consequential SVC is transitive as the verb does not have an object (incomplete sentence). (79b) illustrates a trivial consequence of this test which is that when the subject undergoes wh-movement, it leaves only one resumptive pronoun in the position before the two verbs and none before the second verb. This implies that there is no Tense position before the second verb to host another subject. The conclusion is that there is a single subject in the consequential SVC which is Case-licensed in Specifier of TP.

CCs are, however, very different from SVC in terms of the distribution of subject positions. Consider the following sentences:

(80) a. Òzó lé ízè Ó rrí órè Ozo cook rice he eat it 'Ozo cooked rice and he ate it.'

<sup>&</sup>lt;sup>39</sup> A related observation holds in Igbo that resultative V-V compounds do not make good multi-event constructions (cf. Manfredi 1991). This fact is discussed further in Chapter five.

Òzó gb<u>òó</u> ívìn <u>Ó</u> bòló <u>ó</u>kà
 Ozo plant coconut, he peel corn
 'Ozo planted coconut and [he] peeled corn.'

As (80) shows, it is quite possible to have a subject pronoun in the position before the second verb; this clearly sets CCs apart from SVCs in terms of the structure, CCs can involve the coordination of larger phrases such as TP/VoiceP while SVCs have a single TP/VoiceP.<sup>40</sup>

## 2.6.3 Subject-oriented Interpretation of 'tobore'

In this section, I provide empirical evidence in support of the analysis of subjects in SVCs and CCs based on the distribution and analysis of the adverbial particle  $t\partial b \underline{o} r \dot{e}$ , an element previously discussed with respect to objects in 2.5.1 above. Recall that the particle is licensed as a right-adjunct to an NP which may be overt or null. Thus, it can be used to detect the presence of null NPs. Let us begin, then, by finding out whether there is a null subject NP before the second verb.<sup>41</sup> Consider the following:

(81)	a.	*Òzók kòkó Àdésúwà tòb <u>ó</u> rèk mòsé Ozo raise Adesuwa himself be-beautiful
	b.	*Èsósàk gbé émá!tón tòbórèk pèrhé Esosa hit metal himself be-flat
(82)	а.	*Òzók d <u>é</u> LGBj tòb <u>ó</u> rèk tié proj Ozo buy LGB himself read 'Ozo bought LGB and he himself read it/he read it himself.'
	b.	*Òzók lé èvbàréj tòb <u>ó</u> rèk ré proj Ozo cook food himself eat 'Ozo cooked the food and he himself ate it.'

<sup>&</sup>lt;sup>40</sup> (80) says nothing about the presence of a subject position in the VP1. I assume that it is there by inference, especially so since the t $\partial b \dot{o} r \dot{e}$  particle can occur there (see next section for discussion of t $\partial b \dot{o} r \dot{e}$  with subjects).

<sup>&</sup>lt;sup>41</sup> Note that it is quite possible for the particle to be also construed with the object of V1 in the examples in (81)-(83). For example, (83b) can also have this interpretation 'Òzó planted the coconut itself and peeled the corn'. This interpretation is a trivial consequence of the analysis that I have proposed whereby the particle can always adjoin to the right of an NP. However, I am not interested in this sort of interpretation.

- (83) a. Òzók lé íz<u>è</u> tòb<u>ó</u>rèk rrí <u>ó</u>rè
  Ozo cook rice himself eat it
  'Ozo cooked rice and he himself ate it.'
  - Òzók gbòó ívìn tòbórèk bòló ókà
    Ozo plant coconut himself peel corn
    'Ozo planted coconut and he himself peeled corn.'

The sentences in (81) indicate that it is ungrammatical for the adverbial particle  $i\partial b \delta r \hat{e}$  that is co-indexed with the subject NP to occur before the second verb of the resultative SVC. This implies that there is no null NP in this projection and consequently that there is no subject position between the verbs to which the particle can adjoin. This is consistent with the structure of the resultative SVC as in (70). More significantly, in the consequential SVC (82), we also observe that the particle  $i\partial b \underline{\delta} r \hat{e}$  when co-indexed with the subject NP, cannot occur before the second verb. This is evidence that there is no null subject NP and as such there is no subject position in the projection of VP2. This is consistent with the structure of the consequential SVC (73). Finally, in CCs (83) we observe that it is possible for the  $i\partial b \underline{\delta} r \hat{e}$  particle that is co-indexed with the subject NP to occur before the second verb. This is evidence of the presence of a subject NP trace to which the particle right-adjoins. This fact supports the structure of CCs, as in (75). Thus, I conclude that there is a subject position that dominates the projection of VP2 and this is occupied by the trace of an ATB movement; this is what the  $i\partial b \underline{\delta} r \hat{e}$  particle right-adjoins to.

Let us now turn to the position before the first verb which is contiguous to the overt structural subject. My analysis predicts that the particle should occur immediately after the subject before the I-type adverb and Iterative morpheme in which case it adjoins to the subject NP, or before the first verb and below E (which hosts I-type adverb and Iterative morpheme) in which case it is adjoins to the trace of the subject in Specifier of VoiceP. This predictions are borne out, as illustrated in the following sentences:

(84) a. Ozók töbórek giélgié ghá suá ogó dé
 Ozo himself quickly Iter. push bottle fall
 'Ozo, by himself, quickly pushed the bottle down repeatedly.'

- b. Özók gi<u>é</u>!gi<u>é</u> ghá tòb<u>ó</u>rèk suá <u>ògó</u> dé
  Ozo quickly Iter. himself push bottle fall
  'Ozo, quickly, by himself, pushed the bottle down repeatedly.'
- (85) a. Ozók töbórèk gié!gié ghá dé LGB tié Ozo himself quickly Iter. buy LGB read 'Ozo, by himself, quickly bought LGB and read it repeatedly.'
  - Ózók gi<u>é</u>!gi<u>é</u> ghá tòb<u>ó</u>rèk d<u>é</u> LGB tié
    Ozo quickly Iter. himself buy LGB read
    'Ozo, quickly, by himself, bought LGB and read it repeatedly.'
- (86) a. Ozók töbórèk gié!gié ghá gbó!ó ívìn bòló ókà Ozo himself quickly Iter. plant coconut peel corn 'Ozo, by himself, quickly planted coconut repeatedly and peeled corn.'
  - b. Özók gi<u>é</u>!gi<u>é</u> ghá tòb<u>ó</u>rèk gb<u>ó</u>!<u>ó</u> ívìn bòló <u>ó</u>kà
    Ozo quickly Iter. himself plant coconut peel corn
    'Ozo, quickly, by himself, planted coconut repeatedly and peeled corn.'

These sentences are somewhat complicated because of the items in E which has to be this way because the adverb is required in order to have the iterative reading of gha. Once we get past this, observe that the behavior of the  $t\partial b \underline{o} r \dot{e}$  particle in the position before the first verb provides further evidence for the structures that I have proposed. In the resultative SVC examples in (84), the particle can either occur before E (84a), in which case it adjoins to the right of the subject NP in Specifier of TP, or it can occur after E. I assume that when it occurs below E it adjoins to the subject trace in Specifier of VoiceP. These possibilities are borne out by the different word order and consistent with the structure of the resultative SVC where TP dominates EP which in turn dominates VoiceP.

This same conclusion can be generalized for the consequential SVC. This is supported by the word order and grammaticality contrast in (85) where we observe, like in the resultative SVC the TP>EP>VoiceP order as represented in the structure in (73). Finally, in the CC sentences (86), we observe the same distribution of tob<u>o</u>re in the position before the first verb. This similarity is based on the fact that none of the constructions really differ with respect to the TP>EP>VoiceP order and I have no evidence to decide if (86) does not involve VoiceP coordination rather than TP. I will leave this issue open, but whatever option is correct it will still reflect the fact that TP>EP>VoiceP seems to be the order of functional projections before the first verb which is compatible with the distribution of  $t\partial b \underline{o} r \dot{e}$ .

As a conclusion I restate the fact that whereas resultative SVCs differ from consequential SVCs with respect to the internal structure of the clause, they are however similar in terms of higher functional projections and the position and interpretation of the subject of the clause. In this way, SVCs contrast remarkably with CCs which could involve coordination at any of the higher-level functional projection, in particular it must contain two subject positions.

## 2.7 Phonology-Syntax Interface.

In the previous section, I argued that SVCs along with the feature of object sharing also share the common property of a single E head that quantify over the event(s) that the verbs denote and that the verbs also share a single Voice. These properties are in sharp distinction to what we observe in CCs in which there are two distinct E heads as well as two Voice heads. In this section, I will present empirical evidence which confirm these analyses and distinctions. This is based on tonal changes on the verbs in two environments (a) when Tense is lexical, i.e., filled by an overt morpheme or I-type adverb.

(b) when the object NP undergoes wh-movement as in focus cleft in a single object SVC.<sup>42</sup>

Under these two special circumstances, special high-downstep-high tones occur on the verbs (cf. Haik 1990, Haik, Koopman, and Sportiche 1985, Clements 1984, Tuller 1985 etc.). I will refer to this special tone effect as "relative tones" which can be defined as the tonal trace of lexicalized Tense head (where tense tones are otherwise generated) or as the tonal trace of wh-movement. I will argue for a particular analysis of these tone changes

86

<sup>&</sup>lt;sup>42</sup> I assume that focus cleft and questions involve movement to the Specifier of CP (A'-movement) (cf. Manfredi, 1993).

based on the proposal that they are linked to the nature of the c-commanding E head (which is also tense-related).

# 2.7.1. E-head and Relative Tones

This section examines what happens in SVCs and CCs when E head is filled by the I-type adverb. These tone effects were represented in the preceding sections but never discussed. The relevant data is given in (87)-(89):

(87)	а.	Òzó kòkó àdésúwà mòsé Ozo raise Adesuwa be-beautiful 'Ozo raised Adesuwa to be beautiful.'
	b.	Òzó gi <u>é</u> !gi <u>é</u> kó!kó àdésúwà mó!sé Ozo quickly raise Adesuwa be-beautiful 'Ozo quickly raised Adesuwa to be beautiful.'
(88)	а.	Òzó kòkó ìyán dùnmwún Ozo gather yam pound 'Ozo gathered the yams and pounded them.'
	b.	Òzó gi <u>é</u> !gi <u>é</u> kó!kó ìyán dún!mwún Ozo quickly gather yam pound 'Ozo quickly gathered the yams and pounded them.'
(89)	а.	Òzó rhàán ùw <u>ò</u> nmw <u>è</u> n dùnmwún ìyán Ozo warm soup pound yam 'Ozo warmed the soup and pounded the yams.'
	b.	Òzó gié!gié rhá!án ùwònmwèn dùnmwún ìyán Ozo quickly warm soup pound yam 'Ozo quickly warmed the soup and pounded the yams.'

In the data above, an I-type adverb which is licensed in E occurs before the first verb in a resultative SVC (87), a consequential SVC (88) and a CC (89). A quick summary of the event-modification facts is as follows. In the resultative SVC, the I-type adverb modifies both the first and the second verbs. Similarly, in the consequential SVC both the first and the second verbs are modified, whereas in the CC when the adverb occurs before the first verb only this first verb is modified.

What is relevant for present purposes is the interaction between these scope facts associated with the adverb in the upper E position before the first verb and the consequent tone spreading that it triggers onto the verbs. I propose that there is correlation between tone spreading triggered by the I-type adverb on the verb(s) and what it is predicated of.

In the resultative SVC (87b), the presence of the I-type adverb in the head of EP triggers tone spreading of an additional high tone to both the first and second verbs that it is a predicate of. This contrasts with a similar sentence in (87a) where there is no I-type adverb and the tone sequences on the verbs are the standard low-high pattern that marks past tense. Thus, we observe that the domain of tone spreading associated with the presence of the I-type adverb matches the interpretation of the adverb, and I propose that this is so because the tone changes arise from the single E head in the structure of the resultative SVC.

Similarly, in the consequential SVC (88b) when the I-type adverb occurs before the first verb and is licensed in the head of EP1, it triggers the extra high tone on both the first and the second verb which it modifies. Therefore, I conclude that both the first and the second verb are surely within the scope of the head of EP1 and thus tone spreading onto the verbs is triggered by the c-commanding head of EP1. This, too, is consistent with the structure of the consequential SVC.

The contrast which is observed when the I-type adverb occurs before the first verb in the CC (89b) confirms the significance of this discussion of the phonology-syntax interface. The striking fact is that in (89b), the presence of the INFL-type adverb before the first verb only triggers high tone on the first verb, spreading onto the second verb is excluded in CCs. This fits well with the structural claim that the second verb is not ccommanded by the head of EP1, being a conjunction of phrases.

The conclusion is that based on the interface between phonology and syntax we have an account of special tone effects which correlate with the position and interpretation of I-type adverbs, and this derives the distinctions between SVCs and CCs.

## 2.7.2 Wh-extraction and Relatives Tones

Focus and questions have been discussed in Èdó (Omoruyi 1988, 1989) but no account has been proposed for the special tone effect observed on verbs in such contexts. Therefore, this section shows that the same analysis of the special relative tones triggered by the I-type adverb arise when the object undergoes wh-extraction.<sup>43</sup> I propose that this is because the object moves through the Specifier of EP on its way to Spec, CP and so the relative tone surfaces on the verbs in the c-command domain of the E head as a signal of wh-movement. First, let us consider the resultative SVC as illustrated in the following sentences:

(90)	a.	Özó kòkó àdésúwà mòsé Ozo raise Adesuwa be-beautiful 'Ozo raised Adesuwa to be beautiful.'
	,	

- àdésúwàj <u>ò</u>ré Ozó kó!kó tj mó!sé
  Adesuwa Foc. Ozo raise be-beautiful
  'It's Adesuwa that Ozo raised to be beautiful.'
- Dè òmwànj nè Ôzó kó!kó tj mó!sé?
  Q person that Ozo raise be-beautiful 'Who did Ozo raise to be beautiful?'

(90a) is a typical illustration of the resultative SVC in which there are two verbs that share a single object NP. Notice also from this sentence that the past tense tones on the verbs normally consists of a low-high pattern. In (90b), the shared object  $\dot{a}d\dot{e}s\dot{u}w\dot{a}$  undergoes wh-movement to the Specifier of CP for syntactic focus and it leaves behind a phonetically null trace (possibly *pro*). However, there are noticeable tonal differences between the verbs in (90a) and those in (90b). In (90b) where the direct object has moved, the tone on the first segment of each verb changes from low to high along with a floating tone which causes tonal downdrifts (downstep) within the words. Exactly these same observations can

<sup>&</sup>lt;sup>43</sup> This is not limited to just objects, the same point can be made with subject extraction but for present purposes I will limit the discussion to objects.

be made concerning the sentence in (90c) where the direct object undergoes wh-movement to the Specifier of CP for questioning. Notice that once again the tones on both verbs change in comparison with (90a) where the direct object is unmoved.

Before going into the discussion of my proposed analysis, let me also introduce similar facts in the consequential SVC. This is illustrated by the following sentences:

(91)	a.	Òzó kòkó ìyánj dùnmwún proj Ozo gather yam pound 'Ozo gathered the yams and pounded them.'
	b.	ìyánj <u>ò</u> ré Òzó kó!kó tj dú!nmwún proj yam Foc Ozo gather pound 'It is yams that Ozo gathered and pounded.'
	c.	Dè èmwìnj nè Òzó kó!kó tj dú!nmwún proj? Q thing that Ozo gather pound 'What did Ozo gather and pound ?'

(91a) is a typical example of the consequential SVC where there is one surface direct object and a null *pro* which serves as the object of the second transitive verb. Observe also the sequences of low-high tones on the verbs which is characteristic of past tense on disyllabic verbs in the language. In (91b), the surface direct object undergoes wh-movement for syntactic focus and it leaves behind a null *pro* (or a phonetically null trace). Interestingly, just as in the resultative SVC, there are tone changes on the first tone-bearing segment of both verbs along with an accompanying floating tone which causes downstep within the verbs. Similar observations also hold for (91c), where the shared surface direct object undergoes wh-movement for questioning.

On the basis of the data in (90) and (91), it appears that there must be a common trigger for the tone changes on the verbs that we have observed in both resultative and consequential SVCs respectively. Furthermore, it also seems reasonable to infer that these tone changes are linked to the extraction (A'-movement). These observations find empirical support from consideration of similar extraction facts in the CC. Compare the following sentences:
- (92) a. Òzó hìín èrhán kpàán ívìn
   Ozo climb tree pluck coconut
   'Ozo climbed the tree and plucked a coconut.'
  - b. èrhánj <u>ò</u>ré Ózó hí!ín tj kpàán ívìn tree Foc. Ozo climb pluck coconut
     'It's a tree that Ozo climbed and plucked coconut.'
  - c. ívìnj <u>ò</u>ré Òzó hí!ín èrhán kpá!án tj
     coconut Foc. Ozo climb tree pluck
     'It's a coconut that Ozo climbed the tree and plucked.'
  - d. Dè èmwìnj nè Òzó hí!ín tj kpàán ívìn Q thing that Ozo climb pluck coconut 'What did Ozo climb and pluck the coconut?'
  - e. Dè èmwinj nè Òzó hí!ín èrhán kpá!án tj Q thing that Ozo climb tree pluck 'What did Ozo climb the tree and pluck?'

(92a) is a typical illustration of the conjunctive SVC where each verb has its own distinct object. In (92b), the object of the first verb undergoes wh-movement and there are tone changes only on the first verb. This contrasts with SVCs where tone shifts always surface on both verbs. On the other hand (92c) where the object of the second verb undergoes similar movement, the tone changes do occur on both the first and second verbs.<sup>44</sup> Similar observations can be made in the question sentences involving the object of the first verb in (92d) and the object of the second verb in (92e).

If these tone changes that I have described in the data in (90-92) are indicative of underlying syntactic structures, then exactly what do they imply and how should we interpret them? I will now offer an analysis that illustrates one perspective from which we can view the issue of tones as signals of wh-movement. The data on object extraction in CC is the preferred place that I chose to begin the illustration of my analysis because it reveals the interaction of tones with syntactic structure as each verb is dominated by a

<sup>&</sup>lt;sup>44</sup> It has often been noted that the covert coordination does not obey the Coordinate Structure Constraint (CSC) of Ross (1967) (cf. Baker 1989, Collins 1997 etc.); this is true for these examples as well. In fact, the point must be made, at least based on Èdó, that the CSC effect is variable because it cannot be observed consistently in [different] forms of covert coordinations. Therefore, I do not put too much weight on the CSC in the analysis of covert coordinations.

unique EP. Therefore, we should be able to capture the facts relating to isolated cases of tone changes. In this regard, the key contrast is between the extraction of the object of the first verb in (92b,d) and the object of the second verb in (92c,e). What appears to be going on is that the signal of wh-movement shows up as a floating high tone on the verb or verbs if the moved object goes through the tense-related Specifier of EP. This licensing relation between the object and Specifier of EP is illustrated in the simplified structure without the TP and VoiceP in (93).



Abstracting away from all other issues, for example, those associated with the licensing of the object NP in the resultative SVC, I assume that the structure in (93) is the underlying structure for object wh-extraction. Therefore, I propose an account of the relative tone from wh-movement of the object based on a movement analysis. Accordingly, the theme direct object moves through the Specifier of EP to get to the landing site in Spec, CP. The signal of this movement is a high tone (relative tone) that spreads downward to the verb(s) which it c-commands/quantifies over.45

- Òzó sùá úyì ùsuámwgn
  - Ozo push Uyi nom-push-nom
- 'Ozo gave Uyi a push'

b.

- ??/\*úyì òré Òzó sú!á ùsuámwèn
  - Uyi Foc. Ozo push nom-push-nom
  - 'It is Uyi that Ozo gave a push'

One possible explanation for the contrast in (i) is that both the nominal argument 'dsuámwen' and the moved direct object 'úyi' are in competition in the syntax for the single Spec, EP position. Thus, we find

 $<sup>^{45}</sup>$  Evidence that the direct object extraction involves movement through the Spec of EP comes from the fact that it is not felicitous to have a sentence in which the direct object has been A'-moved and a nominal cognate object of the verb that is involved in predicate cleft construction (see Chapter three) is also present; (i) a.

Based on this proposal, I propose that in (92b,d) the signal of wh-movement which is the relative tone only surfaces on the first verb but not the second verb since the extraction of the object of the first verb moves through only EP1 and E1 c-commands only the first verb. The presence of the relative tone on both verbs in (92c,e) can be derived from a sort of successive cyclic movement or Across the Board extraction whereby two EPs are crossed. The moved object of the second verb is extracted by ATB through both Spec EP1 and Spec EP2 and leaves the wh-traces on both of the verbs that are uniquely ccommanded or quantified over by the respective E. However, considering that CCs have a symmetrical structure this poses a bit of a problem for this proposal because we would not expect what goes on in one part of the conjunct to affect the other. I will leave this issue open for further research.

However, this analysis extends in a straightforward manner to account for the presence of relative tones on both verbs in resultative and consequential SVCs. The resultative SVC contains a single projection of EP which dominates the two verbs and the shared object. When the theme direct object is extracted out of the VP it goes through the Specifier of EP to get to its landing site in the Specifier of CP and this triggers relative tones on both the first and second verbs that are bound by the single E head.<sup>46</sup>

Consequently, the idea of a single EP for both verbs of the resultative is consistent with the fact that the extraction of the shared (single) object triggers the trace of whmovement in the form of the relative tone on both verbs and not one or the other. This is illustrated in the simplified structure without TP and VoiceP in (94).

direct evidence in support of the analysis of cognate object in Chapter three as well as for the licensing of relative tone being discussed here.

<sup>&</sup>lt;sup>46</sup> In fact this counts as further supporting evidence for the analysis of the syntactic relations between both verbs of the resultative SVC as governor (theta assigner) of the object.



Concerning the licensing of relative tones with object extraction in the consequential SVC (91), we observe that there is a consistency between the interpretation of I-type adverbs before the first verb, the fact that relative tones show up on both verbs, and the syntactic structure proposed. The simplified structure of the consequential SVC without VoiceP is given in (95).

(95)



According to the structure in (95), when the direct object of the first verb undergoes whmovement, relative tones show up on both the first and the second verbs. What is not immediately obvious in this structure is how the extraction of the object of the first verb would come to trigger the relative tone pattern on the second verb since object sharing is mediated by an empty category, *pro* that serves as the object of the second verb. The explanation that I propose is based on already established fact that the first E node quantifies over the events denoted by both verbs. Consequently, I propose that the same conditions which govern relative tone licensing and object extraction in resultative SVCs (and a part of CCs) also hold here. When the overt object is extracted out of VP1 it moves through the Specifier of EP1 enroute to Spec, CP. Now, since EP1 c-commands and quantifies-over the event arguments of both verbs, it follows that the trace of wh-movement will show up on both verbs by the same principle that distributes tense marking to both verbs in an SVC (see Chapter six below). Based on this analysis, the appearance of relative tone on the second verb is not due to the movement of *pro* or on a chain-link between the object NP and *pro*, but rather is linked to the scope facts associated with the projection of EP1 namely, both the first and the second verbs are in the c-command domain of a single E head which licenses the trace of wh-movement through its Specifier to show up on both verbs.

As a conclusion, we note that there is a correlation between object extraction, the licensing of I-type adverbs, relative tones, and the syntactic structures of SVCs and CCs. A single functional head unites the two verbs in SVCs while each verb is within the scope of a distinct functional head in CCs. This difference is clearly illustrated by the interaction of syntactic structures with phonological features such as relative tones, which identify syntactic boundaries in terms of tense-related functional heads, like E.<sup>47</sup>

<sup>&</sup>lt;sup>47</sup> In section 7.2.3, I will also show that Tense head exhibits this property as well.

# 2.8 Resultative SVCs, Aspect, and Temporal Adverbs

In the previous sections, I examined the details of the structure of SVCs and CCs from the bottom-up by presenting syntactic tests which foster our understanding of the similarities and differences amongst them. In this section, I introduce a different kind of argument for the proposed distinction and this is to show what the resultative SVC has that excludes consequential SVCs and CCs.

### 2.8.1 Differences in Aspectual Properties

The standard view about resultative constructions in languages like English is that they denote single events, that are internally complex, being made up of parts. Like AP resultatives, resultative SVCs presumably must fit into an event ontology. In particular, resultatives are generally classified as denoting accomplishments, which are complexes that are made up of a process and a transition (Pustejovsky 1991, Tenny 1987, Levin and Rappaport 1995 etc.). Therefore, the resultative SVC is predicted to obey certain constraints. I will examine three such well-established aspectual properties of resultative SVCs

## (96) aspectual properties of the resultative SVC

- a. The first subevent in the resultative construction must be either a process or an activity.
- b. The event denoted in the resultative construction can only be delimited once.
- c. A resultative construction can only be modified by a 'in a hour' type of temporal adverb, not 'for an hour'.

On the other hand, I will argue that consequential SVCs and CCs are made up of two events and so there are no deep interactions or constraints. Thus, they lack any set of characteristics that is based on event ontology.

### 2.8.2 Process-Activity Restriction on First Verb

Quite generally, it is assumed that the main verb of the resultative secondary predicate denotes a process or activity (cf. Pustejovsky (1991), Levin and Rappaport (1995), Baker (1997b) etc.). This predicts, then, that a resultative AP secondary predicate cannot occur with stative verb as the first verb. This prediction is borne out by the data in (97).

- (97) a. \*Mary owns chickens fat
  - b. Mary feeds chickens fat

Stative verbs such as *own* in (97a), express single eventualities that are not evaluated relative to any other event. Consequently, the internal argument *chickens* cannot measure out the event (Tenny, 1987) because states are single eventualities that do not involve change. However, a process verb such as *feed* can be substituted for the stative verb *own* as in (97b) and the sentence is grammatical with a resultative reading: *Mary feeds her chickens until they become fat*. In other words, (97b) expresses an accomplishment that involves a function from a process to a transition (<P, T>) (cf. Pustejovsky (1991)). The initial subevent (denoted by the verb) consists of the process of *feeding the chickens*, and this is delimited by the resultative predicate *fat*, which constitutes the second subevent.

The generalization from the foregoing is that the main verb of the resultative construction should involve some notion of change.<sup>48</sup> When this conclusion is applied to SVCs and CCs it predicts that the first verb cannot be a stative verb in the resultative. This prediction is borne out, as the following contrasts illustrate:

(98) a. \*Özó hòśmwśn àdésúwà wú Ozo love Adesuwa die

<sup>&</sup>lt;sup>48</sup> There are various proposals concerning the formalization of this observation about the main verb (cf. Pustejovsky (1991), Levin and Rappaport (1995), Baker (1997) etc.) I will not embark on a review of these proposals since they do not bear directly on the point I am making about the restriction on the verb.

b.	*Òzó gh <u>ò</u> gh <u>ó</u>	ùbi <u>é</u> mw <u>è</u> n- <u>ò</u> m <u>ó</u>	khuòmwín
	Ozo be-happy	nom-birth-nom-child	be-sick

- (99) a. Òzó guàló <u>ó</u>kà d<u>é</u>
   Ozo find corn buy
   'Ozo seek for corn and bought it.'
- (100) a. Òzó gbé èkhù làá òwá
   Ozo hit door enter house
   'Ozo broke the door and went into the house.'
  - b. Ôzó mòsé làá òwá
     Ozo be-beautiful enter house
     'Ozo was beautiful as he entered into the house.'

In the resultative SVC (98), we observe that a stative verb cannot occur as the first verb. This ungrammaticality arises from the fact that stative verbs such as  $h\partial \underline{e}mwen$  'love' in (98a), express single eventualities that are not evaluated relative to any other event. Consequently, a stative in the first position of the resultative SVC does not provide the required process subevent and as such the sentence lacks the causative force that is needed in resultative constructions. Furthermore, based on Tenny (1987) I propose that in resultative SVCs in which the first verb is a stative verb, for example (98b), the internal argument  $ubi\underline{e}mw\underline{e}n-\underline{o}m\underline{o}$  'child-birth' cannot measure out the event because the state does not involve change. Thus, I conclude that resultative SVCs are constrained by event ontology which requires the first verb to be a process, and since stative verbs denote states they are, therefore, excluded. Thus, (98) is out for essentially the same reason as (97).<sup>49</sup>

However, no predictions are made on the basis of aspectual properties for the sequential SVC. Thus, observe that it is grammatical for either a stative or process verb to occur as the first verb. This is true for the consequential SVC (99a) where the first verb is process/activity verb.<sup>50</sup> Similarly, in CCs a process/activity verb can occur as the first verb (100a), while a stative verb occurs as the first verb in (100b).

<sup>&</sup>lt;sup>49</sup> See Pi and Stewart (1998) for a detailed discussion of Macro-events in resultative and consequential SVCs.

 $<sup>^{50}</sup>$  The examples of consequential SVC that meet these restrictions are very rare to come up with. I suppose that this has got to do with the fact that the consequential SVC has a restriction of its own that both verbs be transitive.

This distinction between resultative SVCs, consequential SVC and CCs that is based on aspectual constraints on the first verb makes one further prediction namely, that the two verbs which express states cannot co-occur with each other in the resultative SVC although they could in CCs. This prediction, if proven to be correct, would constitute evidence in support of single-event resultative and two-event CCs distinction and the effect of a pause before the second verb in CCs. Consider the following:

- (101) a. \*Òzó hoèmwén àdésúwà khuòmwín
   Ozo love Adesuwa be-sick
   'Ozo loved Adesuwa to death (his love killed her).'
  - b. \*Òzó rènrén úyì tùnién
     Ozo know Uyi be-short
     'Ozo knows Uyi to be short.'
- (102) a. Ozó hoèmwén àdésúwà khuòmwín
   Ozo love Adesuwa be-sick
   'Ozo loved Adesuwa and [till] he became sick.'
  - b. Ózó ghòghó égié khuòmwín
     Ozo be-happy title be-sick
     'Ozo became sick after rejoicing over his title.'

Resultative SVCs like (101) which involve the sequence of two verbs that express states are ungrammatical because they violate the aspectual constraint which requires the first verb to be a process-activity verb. Contrastively, the CCs (102) do not need to obey the same restrictions, thus it is perfectly grammatical for two verbs expressing states to occur with a phonological pause between them. This difference underscores the nature of event composition in resultatives SVC, consequential SVCs, and CCs: there are sub-events which combine into a single event in the resultative, consequential SVCs are composed of separate events which are formally connected as a complex event, and CCs are made up of two distinct events.

## 2.8.3 Event Delimiter

This section examines the nature of possible event type that the second verb can denote in the different constructions. It has been observed that there is an aspectual restriction on the resultative construction which prevents including a second resultative phrase. This is illustrated in (103);

(103) \*Mary pounded the metal flat [into pieces]

(103) shows that it is not possible to add a resultative secondary predicate to an achievement which in this case is composed of a process verb and an AP result predicate. This is based on the idea that an event can only be delimited once (Tenny 1987). When this aspectual condition on event delimitedness is applied to SVCs, I predict that it is impossible to have double resultative SVCs, whereas iteration of events is possible in consequential SVCs and CCs. I will illustrate this prediction by first examining the resultative SVC. Consider the following examples:

- (104) a. Òzó sùá òmó dé
   Ozo push child fall
   'Ozo pushed the child down.'
  - b. \*Òzó sùá òmó dé wú
     Ozo push child fall die
     'Ozo pushed the child down to its death.'
  - <u>òmó</u> dé wú
     child fall die
     'The child fell down to its death.'
- (105) a. Òzó gbé àkhé gu<u>òghó</u> Ozo hit pot break 'Ozo broke the pot.'
  - b. \*Ozó gbé àkhé guòghó khànmwán Ozo hit pot break be-short 'Ozo broke the pot into small pieces.'
  - c. àkhé gu<u>òghó</u> khànmwán pot break be-short 'The pot broke into small pieces.'

Pustejovsky (1991) proposes an account of event composition in resultative secondary predicates as composed of a main verb that is specified as a process or activity verb and the resultative secondary predicate which receives a state interpretation that functions as a delimiting expression of the event. Similarly, in the resultative SVC, for example (104a) and (105a), I propose that the unaccusative verbs which occur as the second verb receives an inchoative (change-of-state) interpretation, and function as a delimiting expression of the single event. However, in both (104b) and (105b) we note that it is ungrammatical for there to be a second unaccusative verb which also characterizes a resulting state. This is an Èdó reflex of the basic fact that resultatives by definition can only have one delimiter (cf. Tenny (1987). The sentences in (104c) and (105c) are given as useful controls to show that these sequences are otherwise acceptable but they are ungrammatical in the context of double resultatives.

In contrast, iteration of events is quite possible in consequential SVCs and CCs. This is because they are composed quite differently from the resultative. Thus, whereas the verbs in the consequential SVC and CCs express sequences of events, those in the resultative combine in a unique process-state relation to express a single event. Consequently, the second verb of the consequential SVC or CC do not have to define a state interpretation and is not an event delimiter. Therefore, the single delimiter constraint does not apply to consequential SVCs and CCs, and multi-events are possible. Consider the following sentences:

- (106) a. Ôzó dé ìyán lé
   Ozo buy yam cook
   'Ozo bought yams and cooked them.'
  - b. Ôzó d<u>é</u> ìyán lé ré
     Ozo buy yam cook eat
     'Ozo bought yams, cooked and ate them.'
  - c. Ôzó miện ìyán dế lé rế
     Ozo find yam buy cook eat
     'Ozo sought yams, bought them, cooked and [then] ate them.'

- (107) a. Òzó gbé <u>è</u>khù làá òwá
   Ozo hit door enter house
   'Ozo hit the door and he entered the house.'
  - b. Ôzó gbé <u>èkhù làá òwá rhié úkpòn</u>
     Ozo hit door enter house take dress
     'Ozo broke into the house and took a dress.'
  - c. Ôzó gbé <u>è</u>khù làá òwá rhié úkpòn y<u>ó ó</u>rè
     Ozo hit door enter house take dress wear it
     'Ozo broke into the house and took a dress and wore it.'

As the foregoing data show, it is possible to stack the verbs in the consequential SVC (106) as well as CCs (107). What makes these stacking relations possible is the fact that they, unlike the resultative SVC, do not impose the aspectual condition of process-result (state) on the two verbs and so allow the sequencing of multi-event.

## 2.8.4 Temporal adverbs

One final prediction that arises from the aspectual properties of the resultative is based on the *telic* vs. *atelic* between verbs as brought out by the type of temporal adverb modifier that they allow (Tenny 1987). Consider the following English sentences:

- (108) a. Peter pushed the cart for an hour/ \*in a hour.
  - b. Peter made a cart in an hour/ \*for an hour.

As the contrast in (108) shows, a verb expressing a telic action like 'make' can only be modified by the temporal adverb 'in an hour' but not 'for an hour' (108b). On the contrary, a verb expressing an atelic action can only be modified by the temporal adverb 'for an hour' but not 'in an hour' (108a).

This distinction can also be tested in the SVC: since a resultative SVC expresses an accomplishment, it is telic and so is predicted to be modified only by the temporal adverb 'in an hour'. The consequential SVC and CC on the other hand, expresses a sequence of events which can be either telic or atelic, so they may be modified by either temporal adverbs 'in an hour/'for an hour.' What is striking about this idea is the fact that most of

the verbs in the resultative SVC particularly allow only the temporal adverb 'for an hour' in isolation of the resultative SVC context, rather than 'in an hour'. This is illustrated in (109):

- (109) a. Òzó kòkó àdésúwà (\*vbè ùkpó ìsén) là ùkpó ìsén Ozo raised Adesuwa (\*in year five) for year five Ozo raised Adesuwa for a period of five years.'
  - b. Àdésúwà mòsé (\*vbè ùkpó ìsén) là ùkpó ìsén Adesuwa be-beautiful (\*in year five) for year five 'Adesuwa stayed beautiful for a period of five years.'

Now, consider the behavior of the resultative SVC compared with the consequential SVC

and CC in terms of temporal adverb modification:

- (110) a. Òzó kòkó àdésúwà mòsé vbè ùkpó ìsén (\*là ùkpó ìsén)
   Ozo raised Adesuwa be-beautiful in year five (\*for year five)
   'Ozo raised Adesuwa to be beautiful in five years.'
  - b. Òzó sùá Èsósà dé vbè òwàrókpá (\*là ífuánró ìsén )
     Ozo push Esosa fall in one minute (for minute five)
     Ozo pushed Esosa down in one minute.'
- (111) a Òzó lé èvbàré khién là úzólá èvá /vbè ífuánró ìsén
   Ozo cook food sell for week two/ in minutes five
   'Ozo cooked the food and sold it for two weeks/in five minutes.'
  - Òzó hìín èrhán kpàán ívìn là úzólá èvá /vbè ífuánró ìsén
     Ozo climb tree pluck coconut for week two/ in minutes five
     Ozo climbed the tree and plucked coconuts for two weeks/in five minutes.'

According to the data above, only the temporal adverb 'in an hour' which is compatible with a telic action can occur with the resultative SVC (110). This is true regardless of whether the second verb is a stative (110a) or an eventive-unaccusative (110b). On the basis of the contrast with (109), I propose that the telic interpretation of these resultative SVC sentences does not arise from the verb class of the second verb, but rather it comes from the overall interpretation of the event expressed by the resultative SVC, i.e., both the process-result sub-events combine into a single event and it is this event that the telic temporal adverb modifies. Notice that this result is compatible with there being one E node, structurally.

The consequential SVC is not constrained by the aspectual properties listed in (96), and as such both telic and atelic temporal adverbs are compatible with either the consequential SVC (111a) or the CC (111b). This proposal that there are no aspectual constraints beyond those imposed by the lexical properties of the component verbs is consistent with my analysis of the consequential SVC and CCs in which the verbs head separate VP projections and also express distinct events with different E-nodes.

#### 2.9 Conclusion

The basic argument that was made is the distinction between resultative and consequential SVCs, in contrast to CCs, on the basis of cumulative empirical evidence. It was shown that there is a single object in the resultative SVC--true internal argument sharing, while in the consequential SVC internal argument sharing involves an empty category that serves as the object of the second verb. However, SVCs differ from CCs in that the subject is introduced by a single Voice head, whereas in CCs there are two subject positions linked by an ATB movement to derive one overt subject.

The consequence of this chapter is that we now have sufficient syntactic tests that can help us to identify true SVCs from other surface verb sequencing constructions (see chapter seven); for example SVCs are those constructions in which a single E head quantifies over the verbs, and the verbs combine under a single Voice that licenses the subject (and Agent) that sets about a plan of one macro event which may be resultative or consequential.

104

#### Chapter three

## Predicate Cleft and Event Quantification in SVCs\*

#### 3.1 Introduction

This Chapter examines more evidence for the distinction between single-event resultative SVCs and two-event consequential SVCs, and CCs, based an observed asymmetry between the two with respect to predicate clefts.<sup>1</sup> The phenomenon of Predicate cleft is a way of focusing a verb that involves moving a category (XP or X<sup>O</sup> depending on the analysis) that is associated with it. (cf. Piou (1982), Koopman (1984), Hutchinson (1989), Lumsden & Lefebvre (1990), Lefebvre and Larson (1991) Ameka (1992), Dekydtspotter (1992), Manfredi (1993), DeGraff (1993), Lefebvre (1994) etc.).<sup>2</sup> Predicate clefts have been attested in several African languages or language families, including Kwa and Kru. Predicate clefts have been said to express several meanings such as contrastive, emphatic and factive--although I do not know of an analysis that attempts to provide a unified account for all these meanings, or even if they are all from the same underlying predicate cleft structure (cf. Collins 1994, Lefebvre 1994).<sup>3</sup> The sentences in (1)- (4) illustrate the predicate cleft construction in Èdó: <sup>4</sup>

<sup>\*</sup> I acknowledge Claire Lefebvre for reading a draft of this chapter and providing extensive comments.

<sup>&</sup>lt;sup>1</sup> The predicate cleft contrast in SVCs was introduced by Laniran and Manfredi (1988) when they observed an extraction asymmetry with predicate clefts from instrumental SVCs (cf. also Awoyale 1987, Manfredi 1991, 1993, Déchaine 1993 etc.) However, in much of the Yoruba data used in discussing this phenomenon in SVCs it is observed that only the second verb fails to undergo predicate cleft in most cases. This is one area where this thesis will provide significant empirical contribution since the predicate cleft asymmetry is uniformly observed by both verbs in relevant contexts in the Èdó data.

<sup>&</sup>lt;sup>2</sup> Predicate cleft is not an option open to other categories like prepositions, adjectives or adverbs (cf. Lord, (1973), Déchaine (1986), Sebba (1987), Manfredi and Laniran (1988)).

<sup>&</sup>lt;sup>3</sup> I will only give the contrastive meanings in the translations. However, in the discussion of splitting verbs in Chapter five I will make reference to the factive interpretations of predicate cleft. I will leave the issue open for future research whether there are similarities in terms of underlying structure between these two meanings of predicate cleft since this will require a systematic investigation that will take me too far off course.

<sup>&</sup>lt;sup>4</sup> The English glosses in these sentences and those below in the text are the best approximations that I can give, even then the English translations are largely ungrammatical in the light of the fact that predicate cleft is not possible in the English language.

- a. Òzó kp<u>òló</u>
   Ozo be-big
   'Ozo is big.'
  - b. ùkpólómwèn òré<sup>5</sup> Òzó \*(kpóló) nom-be-big-nom Cop. Ozo be-big
     'It is fat that Ozo is fat, (not say having an obesity sickness).'
- (2) a. Òzó dé Ozo fell
  - b. ùdémwèn òré Òzó \*(dé) nom-fall-nom Cop. Ozo fall
     'It is falling that Ozo did, (not say rolling).'
- (3) a. Òzó khi<u>é</u>n èbé Ozo sell book 'Ozo sold the book.'
  - b. ùkhiénmwèn òré Òzó \*(khién) èbé
     nom-sell-nom cop. Ozo sell book
     'It is selling that Ozo did to the book, (not say give as gift).'
- (4) a Òzó só Ozo shouted
  - b. ùsómwèn òré Òzó \*(só) nom-shout-nom cop. Ozo shout
     'It is shouting that Ozo did, (not say wail).'

As the data above show, predicate cleft is the movement to sentence initial position of some item that is morphologically cognate to the verb, and it applies to verbs from all basic classes in Èdó: stative, unaccusative, transitive and unergative. (This is contrary to the observations in Haitian cf. Lefebvre 1990, Lefebvre and Larson 1991).<sup>6</sup>

As a way to illustrate the nature of the morphologically cognate item that is moved, let us consider predicate clefts from SVCs and CCs as shown in (5)-(7):

3s Cop house

<sup>&</sup>lt;sup>5</sup> It has been suggested that the copula 'dré' may actually have internal structure (R-M Déchaine, p.c.), but there are several arguments that this cannot be true. For example, this approach implies a decomposition as in (i)

<sup>&#</sup>x27;Ò ré òwá'

<sup>(</sup>i) is intended to read as 's/he is at home' but this is in fact ungrammatical. Other arguments against the proposal that the copula is not monomorphemic comes from tone facts as well as the different kinds of copula that occur in the language (see Baker and Stewart 1997a and Baker 1997b).

<sup>&</sup>lt;sup>6</sup> In fact Claire Lefebvre (p.c.) says that there are splits between Haitian speakers as to the restrictions on predicate clefts. In this regard, see DeGraff (1995).

resultative SVC

- (5) a Òzó sùá Àdésúwà dé Ozo push Adesuwa fall 'Ozo pushed Adesuwa down.'
  - b. \*ùsúámw<u>èn ò</u>ré Òzó sùá Àdésúwà dé nom-push-nom Foc. Ozo push Adesuwa fall
  - c. \*ùdémw<u>èn ò</u>ré Òzó sùá Àdésúwà dé nom-fall-nom Foc. Ozo push Adesuwa fall

consequential SVC

- (6) a. Òzó lé èvbàré ré
   Ozo cook food eat
   'Ozo cooked the food and ate it.'
  - b. ùlémwèn òré Òzó lé èvbàré ré nom-cook-nom Foc. Ozo cook food eat 'It is cooking that Ozo cooked the food and ate, (not shred it).'
  - c. ùrémw<u>èn</u> <u>ò</u>ré Òzó lé èvbàré ré nom-eat-nom Foc. Ozo cook food eat 'It is eating that Ozo cooked the food and did, (not sell it).'

covert coordination

(7)	a.	Ozó gbé <u>é</u> khù lá òwá Ozo hit door enter house 'Ozo hit the door and he entered the house.'		
	b.	ùgbémw <u>èn</u> òré Òzó gbé <u>é</u> khù lá òwá nom-hit-nom cop. Ozo hit door enter house 'It is hitting that Ozo did to the door to enter the house.'		

 c. ùlámwèn òré Òzó gbé ékhù lá òwá nom-enter-nom cop. Ozo hit door enter house
 'It is entering that Ozo hit the door and did into the house.'

Whereas predicate clefts are allowed from simple clauses (1)-(4), (5) shows that it is ungrammatical to cleft either of the verbs from the resultative SVC. On the contrary, based on (6) and (7) we observe that predicate clefts of either of the verbs in consequential SVC and CC respectively are grammatical.

The generalization which emerges from this contrast is that predicate cleft is constrained in a single-event (resultative) SVC but possible in two-event consequential SVC and CCs. Based on the proposal that the morphologically cognate item that is moved in predicate cleft is the nominal argument of the event (Bamgbose 1972, Manfredi 1993, Lefebvre 1994), I will argue for an analysis of the asymmetry in (5)-(7) that relates to the basic difference in event quantification. I propose that the nominal argument of an event is generated as a complement within the VP. On its way to the Specifier of CP/FP (Focus Phrase) it must move at LF through the Specifier of EP, after the verb has also raised into the functional head E, in order for it to be licensed under Spec-head and thereafter moves at S-structure to Specifier of CP/FP to check [+Focus] feature. Crucially, I assume that adjunction to Specifier is not allowed and so there are no multiple Specifiers of EP. Given this, since resultative SVCs have a single EP projection, the event argument of one of the two verbs will not be licensed and so predicate cleft from a single-event resultative SVC is ungrammatical. However, in the two-event consequential SVCs and CCs whose structures contain two EP projections, either of the two verbs can be clefted, fully licensed under Spec-head in the separate EPs.

In order to fill in the details of this analysis of the predicate cleft asymmetry, I find it useful to first present the necessary background about the predicate cleft construction in general and its analysis, and then work my way back to the problem illustrated in (5)-(7).

## 3.2 Èdó Predicate Cleft Construction.

The data on predicate cleft from simple clauses in (1)-(4) highlights three major syntactic properties of the construction. First, predicate cleft involves category conversion because it is the nominalized form of the verb that is clefted. Thus, as can be observed in (1)-(4), each verb undergoes nominalization via the affixation of the  $\hat{u}$ -mwén circumfix. Second, predicate cleft is morphologically related to focus clefts in general through the presence of a copula-type focus morpheme. This is illustrated by the morpheme  $\underline{\partial}r\dot{e}$  that occupies the position immediately after the clefted predicate in (1)-(4). Third, a copy of the verb must be left behind in or near the position from which the derived nominal has moved. Working with the ultimate goal which is to use predicate clefts to study SVCs and CCs, I will only deal with the following specific issues:

(a) nominalization and category conversion: predicate cleft as a species of XP-movement

(b) the thematic and syntactic status of the deverbal nominal which appears in verb focus

(c) the licensing of predicate clefts in SVCs and CCs

The leading idea of my analysis is based on Larson and Lefebvre (1991), who elaborate on Chomsky's (1977) proposal that cleft and focus constructions have a universally similar quantificational analysis. This approach derives a quantificational analysis of predicate clefts whereby a predicate phrase undergoes clefting with an accompanying quantification over events. As a result, predicate cleft will reveal another way to account for the difference between one and two event SVCs that is consistent with the difference in E(vent) P(rojections) that I have already argued for in Chapter two.

## **3.2.1.** Evidence for Category Conversion

An interesting issue in the syntax of predicate cleft is the apparent category conversion from V to N. This is relevant to the question of whether predicate cleft is a kind of  $X^{O}$  (verb) movement as claimed by Koopman (1984), or NP movement (cf. Hutchinson 1989, Manfredi 1993 etc.). While a language like Yoruba provides clear morphological evidence for this category change, there is controversy concerning the morphological analysis of the *Se* morpheme and predicate cleft in Haitian (cf. Lumsden and Lefebvre (1990), Larson and Lefebvre (1991), Manfredi (1993)). This is another area in which Èdó can contribute to the descriptive and analytic literature on predicate cleft construction because there is clear syntactic and morphological evidence in support of the proposal that the clefted predicate is a nominal copy of the base verb.<sup>7</sup>

<sup>&</sup>lt;sup>7</sup> In Yoruba, where the nominalization of a verb such as 'ra' (buy) is 'rira' (buying), the nominalizing affix is a prefix, so it is not very clear in SVCs in which two verbs are nominalized whether the prefix is on both of them or only the first. However, in Èdó the affix is a circumfix which surrounds the verb, in what I will call complete nominalization.

The primary argument that supports the proposal that the clefted predicate is a nominal copy of the verb is based on two bits of morphological evidence. The first is based on a language-internal morpho-phonological constraint on nominals. This constraint requires all nominals to begin with a vowel (open syllable), in contrast to all the other lexical categories (cf. Agheyisi 1990, Amayo 1976, Omoruyi 1986, 1987, Elugbe 1976, etc.) This points to why the first part of the nominalizing affix is a vowel prefix. The second evidence is based on the regular derivational relationship between the verb and its nominal counterpart. As described in Agheyisi (1990), Amayo (1976) and Omoruyi (1989), one productive pattern of verb nominalization in Èdó is based on the morphological process of affixation (cf. Baker and Stewart 1997a). Quite commonly, a circumfixal morpheme is attached to the verb in order to derive the nominalization that is the same as that frequently found in predicate clefts. (8) provides a typical illustration with selected verbs.

(8)	<u>verb</u>		nominalizing affix		derived nominal
a.	kp <u>o</u> lo	"be-big"	ù-mw <u>é</u> n	=	ùkp <u>ó</u> l <u>ó</u> mw <u>è</u> n
b.	de	"fall"	ù-mw <u>é</u> n	=	ùdémw <u>è</u> n
c.	khi <u>e</u> n	"sell"	ù-mw <u>é</u> n	=	ùkhi <u>é</u> nmw <u>è</u> n
d.	so	"shout"	ù-mw <u>é</u> n	=	ùsómw <u>è</u> n

These derivations in (8) show that stative, unaccusative, transitive, and unergative verbs respectively can each have a corresponding nominalization form given in the last column. These forms are otherwise used as event denoting nominals: rather than some other type. When these forms are compared with the nominal copy in the predicate cleft examples in (5)-(7), we observe a perfect correspondence between the items in the final column in (8) and the items that occur in the sentence initial positions; they both have the  $\hat{u}$ -mwén form of nominalization. Therefore, I take this similarity as evidence that predicate cleft involves

the movement of a nominal copy of the verb which is derived by a productive and regular morphological process.

This conclusion can be further complemented by the restriction on the category that can occur in initial position in basic underived sentences. I illustrate this based on something that was alluded to in Chapter two and whose general properties have been discussed already; the contrast between I- versus N-type adverbs.

(9)	a.	*gì <u>égìé</u> , Òzó b <u>ég</u> hé!é èmèrí quickly Ozo see Mary	clefting of I-type adverb
	b.	<u>ègìégìé</u> , Òzó béghé!é èmèrí quickly Ozo see Mary 'Quickly, Ozo sighted Mary'	clefting of N-type adverb

As the contrast in (9) shows, an I-type adverb cannot occur in an adjoined sentence-initial position (9a); only the N-type adverb can be adjoined to TP (9b). By abstracting away from the difference in the properties of the licensing of these two adverbs clause internally and simply focusing on how they are licensed in sentence initial position, we are able to explain this contrast in (9) in a way that is relevant to the issue of predicate clefts.

I suggest that the ungrammaticality of (9a) is related to a structure preservation constraint which requires a head ( $X^{O}$ ) to adjoin to another head and allow only an XP to adjoin to an XP. Pre-theoretically, we conclude that in (9b) the N-type adverb is an XP, but the I-type adverb in (9a) is an X<sup>O</sup>. Therefore, since heads can only adjoin to heads and XPs to XPs, it follows that (9a) will be ungrammatical because there is no relevant head to which the I-type adverb can adjoin. On the contrary, the XP adverb in (9b) adjoins to an XP category (TP) and hence the grammaticality contrast in (9). The conclusion from this simplified discussion of the contrast in (9) is a structure preservation constraint that allows only XPs to occur in sentence initial position as syntactic adjuncts and also that only XPs can occur in Specifier positions. Returning now to the issue of predicate cleft and nominalization we observe that only a deverbal nominal derived by the affixation of the  $\dot{u}$ - *mwén* morpheme to yield an XP category can undergo predicate cleft to Specifier of CP, and this implies that the moved copy cannot be the same as the base-verb which is an  $X^{O}$  category.

## 3.3 The Status of the Derived Nominal

This section examines the semantic and syntactic properties of the nominal copy of the verb that is involved in predicate cleft. In other words, why does predicate cleft move a deverbal (derived) nominal and then leave a copy of the verb behind?

## **3.3.1 Semantic Interpretation**

Let us begin by first examining the semantic interpretation of the deverbal nominal. Consider the examples in (1)-(4) two of which are repeated as (10) and (11):

(10)	a.	Òzó dé Ozo fell	(=2)
	b.	ùdémw <u>è</u> n <u>ò</u> ré Òzó *(dé) nom-fall-nom Cop. Ozo fall 'It is falling that Ozo did, (not say rolling).'	
(11)	а.	Òzó khi <u>é</u> n èbé Ozo sell book 'Ozo sold the book.'	(=3)
	b.	ùkhiénmwèn òré Òzó *(khién) èbé nom-sell-nom cop. Ozo sell book 'It is selling that Ozo did to the book, (not sa	ay give as gift)

On the basis of meaning, there is one relevant aspect of the predicate cleft construction that I want to consider. Observe in the sentences above that the deverbal nominal refers to the event denoted by the verb. This can be illustrated by a comparison of both transitive and unaccusative verbs. For example in (10), the nominal refers to the 'falling' event, and in (11) it is the 'selling' event that is nominalized.<sup>8</sup> The point I am making here is that the

.'

<sup>&</sup>lt;sup>8</sup> In fact, some older Edó speakers confirmed to me that it is possible to have sentences in which the object appears with the nominalized verb as shown in (i).

<sup>(</sup>i) a. úkhiénébémwèn <u>ò</u>ré Òzó \*(khién) èbé

nominalization of the verbs refers to the events, i.e., a 'selling' event in (10b) or a 'bookselling' event (footnote). Consequently, based on Laniran and Manfredi (1988), Manfredi (1993) and as established in the Yoruba descriptive literature (e.g. Bamgbose 1972), I propose that predicate cleft involves movement of a nominal argument which denotes the event of the verb (cf. Lefebvre 1994).

# 3.3.2 Cognate Object as Event Argument

This section shows that the derived nominal involved in predicate cleft has exactly the same meaning as cognate objects. On the basis of evidence from a morphological blocking relationship between the irregular (vowel-initial type) and the regular ( $\hat{u}$ -mwén) forms, I propose that predicate clefts are derived from cognate objects.

One piece of language-internal evidence that buttresses the event argument interpretation of the derived nominal that is involved in predicate cleft comes from cognate objects. Like the deverbal nominal in the predicate cleft construction, a cognate object is an object that is semantically and morphologically derived from a verb. Here are some relevant examples of cognate objects in Èdó:

- (12) a. Ózó hi<u>ó</u> (àhi<u>ó)</u> Ozo urinate (urine) 'Ozo urinated.'
  - b. Òzó tu<u>é</u> (òtu<u>é)</u> Ozo greet (greeting) 'Ozo greeted.'
  - Ozó kpá (èkpá)
     Ozo vomit (vomit)
     'Ozo vomited.'
  - d. Özó khián (dkhián) Ozo walk (walk) 'Ozo walked.'

'It is book-selling that Ozo did, (not say give as gift)' c. ùlévbárémw<u>èn</u> <u>ò</u>ré Òzó lé èvbàré nom-cook-food-nom cop. Ozo cook food It is food-cooking that Ozo did, not throw the food away' e. <u>Òmó</u> wén (èwén) child suck (breast) 'The child suckled.'

In the examples above, the direct objects are exact copies of the verb, except that they, like all nominals, begin with a vowel prefix. Strikingly, they make no obvious contribution to the interpretation of the meaning of the sentence and yet they are present at S-Structure as these sentences are grammatical without the cognate objects. The question then is, what function do these cognate objects fulfill in the syntax? I propose that cognate objects are like predicate clefts in Èdó, the difference being in the form of their nominalizations.

One argument in favor of this proposal comes from Lefebvre (1994) who points out that a bare cognate object may occur in some cases instead of the nominalized copy in a construction such as the predicate cleft which otherwise would involve the nominal copy of the verb. Thus, cognate object is another morphological realization of the nominalization of a verb. This is illustrated in the following predicate cleft examples involving cognate objects:

- (13) a. àhi<u>ó</u> òré Òzó hi<u>ó</u> urine Foc Ozo urinate
   'It is urine that Ozo urinated, not (say) blood.'
  - b. ôtué <u>ò</u>ré Òzó tué greeting Foc Ozo greet
     'It is greeting that Ozo greeted, not (say) a sneer.'
  - c. èkpá òré Òzó kpá vomit Foc Ozo vomit
    'It is vomit (food) that Ozo vomited, not (say) blood which would require another kind of word for vomit 'bí'.)
  - d. òkhián <u>ò</u>ré Òzó khián
     walk Foc Ozo walk
     'It is walking that Ozo walked, not (say) get a ride.'
  - èwén òré Òmó wén
     breast Foc child suck
     'It is breast that the child suckled, not (say) feed on the bottle.'

The semantic interpretation of the sentences in (13) is one in which the cognate object is a nominal realization of the event denoted by the verb, just as in predicate clefts. Furthermore, observe that these cognate objects have similar contrastive focus. This semantic similarity reflects the two morphological choices in predicate clefts: regular nominal  $\dot{u}$ -mwén form and irregular nominal vowel prefix form as in cognate object. Therefore, I conclude that the forms involved in predicate clefts should be grouped along with the forms that characterize cognate objects. This predicts that it is ungrammatical to have the  $\dot{u}$ -mwén form of nominalization in the predicate cleft of any of these verbs in (13) because it will be ruled out by morphological blocking. This prediction is borne out, as illustrated by the ungrammaticality of the sentences in (14):

- (14) a. \*àhiómwèn òré Òzó hió nom-urinate Foc Ozo urinate
  - b. \*ùtuémwèn <u>ò</u>ré Òzó tué nom-greet-nom Foc Ozo greet
  - c. \*ùkpámw<u>èn</u> <u>ò</u>ré Òzó kpá nom-vomit-nom Foc Ozo vomit
  - d. \*ùkhiánmw<u>èn ò</u>ré Òzó khián nom-walk-nom Foc Ozo walk
  - e. \*ùw<u>é</u>nmw<u>è</u>n <u>ò</u>ré <u>Ò</u>m<u>ó</u> w<u>é</u>n nom-suck-nom Foc child suck

In addition, the conclusion that predicate clefts are related to cognate objects is buttressed by the fact that cognate objects can also be used as event nominalizations in  $\dot{E}d\delta$ on a par with the discussion of the forms in (8). This is illustrated in (15).

		verb cognate derived nominal
(15)	a.	hi <u>ó</u> 'pee' = àhi <u>ó</u> 'peeing'
	b.	kpá 'vomit' = èkpá 'vomiting'
	c.	tu <u>é</u> 'greet' = òtu <u>é</u> 'greeting'
	d.	khián 'walk' = òkhián 'walking'

In the rest of this Chapter I will assume that the term nominal copy that is used in most descriptions also refer to cognate objects (cf. Manfredi 1993 for Yoruba, Lefebvre 1994 for Haitian and Fon, Koopman 1984 for Kru languages, etc.).

# 3.4 The Syntax of Predicate Clefts

In this section, I will present four arguments based on the syntactic interpretation of predicate clefts and cognate objects which will then form the basis of the syntactic analysis of predicate clefts in general.

First, observe that the nominal copy of the verb can occur with a thematic direct object of a transitive verb.<sup>9</sup> This is illustrated by the following sentences:

- (16) a. Òzó gbé <u>èkhù ùgbémwèn</u>
   Ozo hit door nom-hit-nom
   'Ozo hit the door a hitting'
  - Òzó rrí èvbàré ùrémwèn
     Ozo eat food nom-eat-nom
     'Ozo ate the food a eating'
  - Òzó tu<u>é</u> úyì òtu<u>é</u>
     Ozo greet Uyi greeting
     'Ozo greeted Uyi a greeting'

(16) illustrates the fact that both the regular cognate objects (16a,b) and the irregular cognate object (16c) can co-occur with the direct object of a verb. Note that even when they occur in situ along with the object of the verb, these nominals still have the event nominal interpretation, for example (16a) means that Ozo hit the door (a) hitting, describing the hitting event. I like to point out that these sentences in (16) indicate two very important observations concerning the syntax of the nominal copy of the verb. First, they point to the fact that the nominal copy must occur after the direct object of the verb in linear word order (*Pace* Larson's 1988 analysis of double objects). We can confirm this from the

<sup>&</sup>lt;sup>9</sup> I would like to point out the fact that I do not give more examples of irregular form of cognate objects since they are mostly derived from intransitive verbs.

ungrammaticality of the corresponding sentences to (16) where the order is reversed between the direct object and the nominal copy:

- (17) a. \*Òzó gbé ùgbémwèn èkhù Ozo hit nom-hit-nom door
  - b. \*Òzó rrí ùrémw<u>è</u>n èvbàré Ozo eat nom-eat-nom food
  - c. \*Òzó tié ùtiémw<u>è</u>n èbé Ozo read nom-read-nom book
  - d. \*Òzó fĩ ùfímw<u>è</u>n ím<u>ó</u>tò Ozo drive nom-drive-nom car

The data in (17) confirms the ordering relation between the thematic direct object and the nominal event argument. Since the direct object must receive a thematic role from the verb, it follows that adjacency between the two is required. The second point arising from the data in (16)-(17) is that the nominal copy is apparently an event argument which does not get any theta role from the verb. Thus, whereas two nominalizations cannot co-occur with one another (see example (20) below), one of them can occur with a transitive verb, as long as it appears linearly outside of the thematic-argument [V-NP] complex. I suggest that predicate clefts are derived by moving cognate (derived) objects which are base-generated within the VP as in sentences like (16).

A clue to the structural analysis of the cognate object comes from their interaction with N-type adverbs, which I have argued above marks the right edge of a VP (section 2.2).<sup>10</sup> Consider the following:

- (18) a. Òzó gbé èkhù ùgbémwèn ègìégìé Ozo hit door nom-hit-nom quickly 'Ozo hit the door a hitting quickly.'
  - b. \*Òzó gbé <u>èkhù ègìégìé</u> ùgbémw<u>èn</u> Ozo hit door quickly nom-hit-nom

<sup>&</sup>lt;sup>10</sup> Observe that predicate cleft is quite possible in the context of an N-type adverb. For example, (18a) can also be, 'ùgbémwèn dré Òzó gbé èkhù ègiégié'

- c. Òzó rrí èvbàré ùrémw<u>èn</u> <u>ègiégié</u> Ozo eat food nom-eat-nom quickly 'Ozo ate the food a eating quickly.'
- d. \*Ôzó rrí èvbàré ègìégìé ùrémwèn
   Ozo eat food quickly nom-eat-nom
- Ozó tu<u>é</u> úyì ôtu<u>é</u> ègìégìé
   Ozo greet Uyi greeting quickly
   Ozo greeted Uyi a greeting quickly.
- f. \*Òzó tu<u>é</u> úyì <u>èg</u>ìégìé òtué
   Ozo greet Uyi quickly greeting

As we observe in (18), only those sentences in which the N-type adverb occur after the direct object and event nominal are correct. This implies that both NPs are within the same VP. However, it is ungrammatical for the N-type adverb to occur between the direct object and the event nominal. Therefore, I conclude that the event nominal is generated in the complement position within the VP, while the direct object occupies the Specifier position.<sup>11</sup> This is illustrated in (19).



The third evidence for the analysis of predicate clefts as cognate objects comes from the fact that two nominalizations cannot co-occur together: the regular derived nominal form of the verb and the irregular cognate derived object are apparently in competition for the same position. Consider the following;

<sup>&</sup>lt;sup>11</sup> See footnote (16) for evidence that the cognate object acts as an event delimiter, like argument PPs. However, I do not assume that cognate objects are the same thing as argument PPs except that they are both generated in the inner complement position within the VP, but licensed differently.

- (20) a. \*Òzó việ èvê ùviệmwện Ozo cry cry nom-cry-nom
  - b. \*Òzó khián òkhián ùkhiánmwèn Ozo walk walk nom-walk-nom
  - c. \*Òzó hi<u>ó</u> àhi<u>ó</u> ùhi<u>ó</u>nmw<u>è</u>n Ozo urinate urine nom-urinate-nom
  - d. \*Òzó tu<u>é</u> òtu<u>é</u> ùtu<u>é</u>mw<u>è</u>n Ozo greet greeting nom-greet-nom
  - e. \*Òzó kpá èkpá ùkpámw<u>è</u>n Ozo vomit vomit nom-vomit-nom

The fact that the two forms of nominalizations cannot co-occur as shown in (20) can be taken as surface evidence that they are generated and licensed in the same structural position.<sup>12</sup>

The fourth and final evidence that can be inferred from the discussion above is the fact that the predicate cleft construction and the cognate object are mutually exclusive; a fact which has been well demonstrated in Lefebvre (1994) with data from Fongbe. Consider the following examples from Èdó:

- (21) a. \*ùhi<u>ó</u>mw<u>è</u>n <u>ò</u>ré Òzó hi<u>ó</u> àhi<u>ó</u> nom-urinate-nom Foc Ozo urinate urine 'It is urinating that Ozo urinated.'
  - b. \*ùtuémwèn dré Òzó tué dtué nom-greet-nom Foc Ozo greet greeting 'It is greeting that Ozo greeted.'
  - c. \*ùpkámw<u>èn òr</u>é Òzó kpá èkpá nom-vomit-nom Foc Ozo vomit vomit 'It is vomiting that Ozo vomited.'
  - d. \*ùkhiánmwèn <u>ò</u>ré Òzó khián òkhián nom-walk-nom Foc Ozo walk walk 'It is walking that Ozo walked.'

 $<sup>^{12}</sup>$  It is possible that these sentences are out anyway because of morphological blocking but I assume that they are ungrammatical because of the condition on Spec-head matching whereby one of the nominalizations fail to be licensed at LF in a manner to be described shortly.

In all these sentences above, it is ungrammatical for both the nominal copy of the verb and the cognate object to occur together. I propose that these examples are bad for the same reason as those in (20) based on the assumption that predicate clefts are derived by fronting the cognate object of the verb and licensed uniformly.

One the basis of the observed properties of predicate clefts, I propose a syntactic account whereby I associate the licensing, function, and properties of the nominal expression of the event denoted by the verb with the EP projection. This has an intuitive appeal since EP stands for Event Phrase and so it seems reasonable that the event argument be licensed there. I propose (22) as the canonical structure illustrating the relationship between a verb and its cognate object at LF.



In this structure (22), I propose a Spec-head matching condition in EP that is based on the assumption that there are no multiple Specifiers. Consequently, the functional head E attracts the verb that it quantifies over at LF thereby creating a complex head and the nominal copy of the predicate raises also at LF to the Specifier to be checked against the complex event operator E (cf. 'E-binder' in Grimshaw (1990)). This creates the required Spec-head matching for the licensing of the cognate object in predicate cleft construction. This analysis relies on a distinction between thematic role licensing of obligatory arguments

and the licensing of optional arguments like these cognate object in predicate clefts, the former is assigned directly by the verbs to its thematic NP under sisterhood relation, or in a configurational approach--in the Specifier of VP. On the other hand, the licensing of optional non-thematic NPs takes place at LF under a Spec-head relationship.<sup>13</sup>

The empirical evidence in support of this distinction with respect to thematic arguments and non-thematic arguments comes from the fact that sentences in which the event nominal occurs along side with the direct object are different from real cases of double object constructions. This contrast is illustrated in (23) and (24):

- (23) a. Òzó hàé úyì íghó Ozo pay Uyi money 'Ozo paid Uyi money.'
  - b. úyì <u>ò</u>ré Òzó hàé <u>é</u>rè íghó Uyi Foc. Ozo pay clitic money 'It is Uyi that Ozo paid money.'
  - c. íghó <u>ò</u>ré Òzó hàé úyì money Foc Ozo pay Uyi pro
     'It's money that Ozo paid to Uyi.'
- (24) a. Òzó tu<u>é</u> úyì òtu<u>é</u>
   Ozo greet Uyi greeting
   'Ozo greeted Uyi a greeting.'
  - b. úyì <u>ò</u>ré Òzó tuć (\*érè) òtuć
     Uyi Foc. Ozo greet clitic greeting
     'It is Uyi that Ozo greeted a greeting.'
  - c. ôtué òré Òzó tué úyì greeting Foc Ozo greet Uyi
     'It is a greeting that Ozo greeted Uyi.'

(23) illustrates a double object construction (see chapter four) while (24) shows the predicate cleft construction. In both cases a wh-movement test is being applied to the object NPs. In the double object construction (23b), the extraction of the first object leaves a resumptive pronoun in its base position. This contrasts sharply with the extraction of the

<sup>&</sup>lt;sup>13</sup> The proposal made in Massam (1990) that (some) cognate objects are thematic arguments certainly does not apply to these sort of cognate objects and so that analysis does not carry-over in this case as well (see the immediate text below for the empirical evidence).

first object in the predicate cleft construction (24b) where it is, in fact, ungrammatical for the resumptive pronoun to occur. This means that the first object cannot be pronominalized and so it is typically different from normal double object constructions.<sup>14</sup> I interpret this contrast as evidence that these two constructions are not the same although the phrase structure may be the same (see Chapter four for the analysis of double object construction).

Furthermore, observe from (23c) that the clefting of the direct object leaves a null *pro* behind as its trace,<sup>15</sup> and there is no compelling reason not to assume that the clefting of the cognate object in (24b) doesn't do the same thing. However, based on the discussion of the data in (16)-(18) we know that the cognate object is not a direct object of the verb, and therefore must be something else. In addition, given the fact that there is c-command between double objects in Èdó (see Chapter four) whereby the indirect object c-commands the direct object, the cognate object in the predicate cleft construction (which is neither an indirect object nor a direct object) cannot be a thematic object of the verb since it cannot be pronominalized by the object pronoun (\*Òzó tuế úyì érè ('Ozo greeted Uyi it)).<sup>16</sup> Thus, by implication the cognate object cannot be the direct object and so (24a) is not a double object construction since we cannot establish a c-command between the two NPs. This conclusion is further confirmed by the fact there is no special morphology such as an applicative in Èdó to license two symmetrical arguments and so the extra argument in the predicate cleft construction in (24) must be licensed differently.

Consequently, I conclude that an optional argument such as the cognate object in the predicate cleft construction is a non-thematic argument of the verb that is licensed differently from thematic arguments in the manner that I have proposed in (22).

<sup>&</sup>lt;sup>14</sup> In (24a) it is possible for the cognate noun to host a relative clause just like the corresponding lgbo examples (V. Manfredi p.c.) e.g. 'Ozó greeted Uyi a greeting which was surprising'.
<sup>15</sup> cf. Baker and Stewart (1997b).

<sup>&</sup>lt;sup>16</sup> Here, it is important to note that unlike Yoruba predicate cleft, an adverb or adjective cannot occur within the nominalization, although an adjective can be adjoined by a Kase head óghé (see Chapter four) to the NP. This difference between Edó and Yoruba draws attention to the observation I made earlier in footnote (6) regarding the fact that Yoruba nominalizer occurs as a reduplicative prefix whereas the one in Edó is a circumfix which is a complete nominalization and I assume that is what makes it impossible for adjectives or adverbs to occur.

### 3.4.1 On Event Quantification

Larson and Lefebvre (1991) propose a semantic analysis of predicate clefts as involving a quantification which, rather than quantifying over familiar individuals like persons and chairs, quantifies over events. This analysis of event quantification with predicate clefts has a direct mapping into the one that I propose. According to my analysis, predicate cleft involves the movement of a nominal argument which is the event argument of the verb. Therefore, the quantificational value of the cleft is derived by moving the clefted event through the relevant EP projection.

Furthermore, according to Larson and Lefebvre (1991:252), there is an intuitive parallel between clefts and focus which is this: they both have truth-conditions that are formed by dividing the sentence information into a *focus*--a contrasted subject of assertion--and a *presupposition*--a property that is asserted of the subject. This similarity is exemplified by the following sentences;

(25)	а.	It was John that saw Mary	(L&I	(29=را	'cleft'
	b.	JOHN saw Mary	( "	30)	'focus'

(25a) and (25b) have the same truth-conditions: they add to the simple assertion 'John saw Mary', the claim that John--and not some other person-- saw Mary, as well as the assumption that Mary was in fact seen. This can be represented as in (26);

(26) John = FOCUS x saw Mary, for some x = PRESUPPOSITION

The representation in (26) strips away the surface difference in the morphology of focus and clefts. This implies that there is no semantic difference between focus and clefts, and in fact Larson and Lefebvre (1991) and Dekydtspotter (1992) based on Chomsky (1977) also make the corresponding claim that there is similar syntactic analysis for both constructions (at LF). Taking the general structure for clefts as in (27a), and based on Larson and Lefebvre (1991) and Dekydtspotter (1992), I propose the interpretation for predicate clefts in (27b), with the mapping to focus and presupposition shown in (27c):

## The semantic representation of predicate cleft

(27) a. It be  $XP_i$  [-  $EP t_i$  -----]

- b.  $[\exists e_i \text{ cooking } (e), [...e..]]$
- c. cooking (e) = RESTRICTION = FOCUS --e-- (for some e) = SCOPE = PRESUPPOSITION

According to (27), the clefted phrase which is an NP (a cognate object) represents the quantifier restriction and corresponds to the focus.<sup>17</sup> The phrase containing the "trace" of the nominal event argument corresponds to the presupposition. I assume, as with focus, that the existential quantifier comes as part of the general scheme for interpreting the structure.

We can illustrate (27) with the sentence in (28a) and the corresponding structure in (28b):

(28) a. [ùlémwèn]<sub>NP</sub> òré Òzó [<sub>EP</sub> lé èvbàré]] nom-cook-nom Foc. Ozo cook food 'It is cooking that Ozo did to the food'

 $<sup>^{17}</sup>$  I will not discuss in detail the exact category label of the landing site of the moved verb-focus, since it has no bearing on SVCs. It could be Spec CP or FP (focus phrase), what is important is that the head of the projection has a [+Focus] feature that needs to be checked by the cognate object at S-structure.



According to (28b), the NP is generated in the complement position within the VP with a [+Foc] feature and moves through EP (at LF) on its way to Specifier of FP overtly, prior to spell-out. Thus, by the LF raising of the verb which is attracted to the E head the relevant Spec-head matching exists at the EP level for the licensing of the event argument.<sup>18</sup>

## 3.5 Predicate Cleft and SVCs

Perhaps, the best way to illustrate the connection between the syntactic structure (28b) and the semantic representation in (27) is by using it to provide an account of the asymmetry between single-event resultative SVCs and two event consequential SVCs as well as CCs, that was introduced in (5)-(7). I will now examine each construction:

125

<sup>&</sup>lt;sup>18</sup> It has been observed that predicate cleft exhibits wh-dependencies (cf. Koopman (1984), Manfredi (1993), Dekydtspotter (1992) etc.) and such as Islands effects can also be observed in Edó. However, I propose that the ungrammaticality of such sentences should be attributed to the same Spec-head matching condition in my analysis, i.e., how many EPs does the sentence contain and by implication how many of them are crossed?

## 3.5.1 Resultative SVCs

This section will show that predicate cleft is ungrammatical in resultative SVCs which express complex events because the nominal argument of one sub-event is not properly licensed since there is a single Specifier of EP and so after the two verbs raise to E at LF, there is no Spec-head matching. However, I will also show that it is possible to have a nominalized compound of the event argument that is generated as complement and moves through EP and satisfy the Spec-head matching condition.

For the sake of exposition, let me first, show that it is perfectly grammatical to have the predicate cleft with the relevant verbs in simple clauses:

(29)	а.	Òzó sùá Àdésúwà Ozo push Adesuwa 'Ozo pushed Adesuwa.'
	b.	ùsúámw <u>è</u> n <u>ò</u> ré Òzó sùá Àdésúwà nom-push-nom Foc. Ozo push Adesuwa 'It is pushing that Ozo did to Adesuwa, not (say) kicking.'
(30)	a.	Àdésúwà dé Adesuwa fall 'Adesuwa fell.'
	b.	ùdémw <u>èn ò</u> ré Àdésúwà dé nom-fall-nom Foc Adesuwa fall

'It is falling that Adesuwa did, not (say) rolling (on the ground).'

As we observe in both (29) and (30), the transitive verb  $s\dot{u}\dot{a}$  'push' and the unaccusative verb  $d\dot{e}$  'fall' can undergo predicate cleft in simple clauses. Now, consider the predicate clefts from resultative SVCs as shown in (31) and (32):

(31)	a	Òzó sùá Àdésúwà dé Ozo push Adesuwa fall 'Ozo pushed Adesuwa down.'
	b.	*ùsúámw <u>è</u> n <u>ò</u> ré Òzó sú!á Àdésúwà dé nom-push-nom Foc. Ozo push Adesuwa fall
	c.	*ùdémw <u>è</u> n <u>ò</u> ré Òzó sú!á Àdésúwà dé nom-fall-nom Foc. Ozo push Adesuwa fall
- (32) a Òzó kòkó Àdésúwà mòsé Ozo raise Adesuwa be-beautiful 'Ozo raised Adesuwa to be beautiful.'
  b. \*ùkókómwèn <u>ò</u>ré Òzó kó!kó Àdésúwà mó!sé nom-raise-nom Foc. Ozo raise Adesuwa be-beautiful
  - c. \*ùmósémw<u>è</u>n <u>ò</u>ré Òzó kó!kó Àdésúwà mó!sé nom-be-beautiful-nom Foc. Ozo raise Adesuwa be-beautiful

According to the data in (31-32), it is ungrammatical to cleft either of the verbs from a resultative SVC. This result is not surprising and is in fact consistent with my analysis. In the resultative SVC, both verbs are attracted at LF by the functional head E that existentially quantify over the two of them E. Based on the condition that there must be Spec-head matching for the NP to be licensed, it follows, therefore, that there will be no match if there is only the nominalization of one of the verbs in Spec EP and so the predicate cleft is ungrammatical. In this regard, Èdó is unlike Yoruba (see below) because the nominalization of both verbs is ruled out by a morphological filter that bans V-V compounds in the language.

I will illustrate the analysis of (32) by reproducing the semantic representation of the resultative SVC from section 2.2.2 as in (33) which corresponds to the structure in (34).<sup>19</sup>

(33) <u>The resultative SVC</u>

( $\exists$ e) [ Push-Fall(e) & Agent(e, Ozo) & Theme(e, bottle)]

<sup>&</sup>lt;sup>19</sup> I should point out that the ternary-branching structure in the E node has not arisen via successive headmovement, rather both verbs raise together as a unit to E as co-heads of the same VP.



Based on (30-31) we know that each of the verbs in (32) sùá "push" and  $d\acute{e}$  "fall" have an event argument that can appear in a predicate cleft construction. Now, in (32) these verbs combine to express a single "fused" event under a single EP, where the E head existentially quantifies over the fused event. Therefore, the clefting of the event nominal of the first subevent (first verb) as in (32b) implies that there is no Spec-head matching since both verbs have raised but the nominalization of only one of them can be in the Specifier and so the sentence is ungrammatical. Parallel reasoning accounts for the ungrammaticality of the predicate cleft of the second verb from the resultative SVC (32c) as well.

The immediate consequence of this analysis is that we predict that cognate objects in SVCs will also be grammatical (in the non-predicate cleft context). This prediction is borne out, as illustrated below:

(35) a \*Òzó sùá Àdésúwà ùsúámwèn dé Ozo push Adesuwa nom-push-nom fall

	b	*Òzó sùá Àdésúwà dé ùdémw <u>è</u> n Ozo push Adesuwa fall nom-fall-nom
(36)	a.	*Òzó kòkó Àdésúwà ùkókómw <u>è</u> n mòsé Ozo raise Adesuwa nom-raise-nom be-beautiful
	b.	Òzó kòkó Àdésúwà mòsé ùmósémw <u>è</u> n Ozo raise Adesuwa be-beautiful nom-be-beautiful-nom 'Ozo raised Adesuwa to be beautiful.'

Just as in predicate cleft, (35) and (36) show that it is ungrammatical for either of the verbs in the resultative SVC to have a cognate object. This is consistent with the analysis of resultative SVC in which the two verbs are co-heads, and so they raise to the head of E but there is no Spec-head matching between the NP and the two verbs, so the sentences with cognate objects are also ungrammatical.<sup>20</sup>

However, it is possible to have predicate clefts from resultative SVCs if a nominal copy of both verbs can be clefted together.<sup>21</sup> This is possible in Yoruba where it has been observed that predicate cleft is allowed from 'certain' SVCs when both verbs are

- (i) Òzó khián (\*òkhián) là áwà ókpá
   Ozo walk nom-walk for hour one
   'Ozo walked for an hour'
- (ii) Özó gbé émá!tón (\*ùgbémwèn) là áwà ókpá
   Ozo hit metal nom-hit-nom for hour one
   'Ozo hit the metal for an hour'

<sup>21</sup> This shows that there are indeed restrictions on the predicate cleft construction contrary to the claim I made earlier that predicate cleft is possible, in principle, with verbs from all aspectual classes. For example, Larson and Lefebvre (1991) propose a restriction based on Stage- vs. Individual-level contrast, i.e., only predicates denoting temporary characteristics can participate in contrastive predicate clefts. (cf. also DeGraff (1993). However, Dekydtspotter (1992) argues against the stage vs. individual level contrast based on evidence from Yoruba, e.g. color and stative predicates can be clefted contrastively. Similarly, as I have shown earlier stative verbs are cleftable in Èdó. The alternative proposed by Dekydtspotter (1992) that only 'individual level predicates with experiencer subjects' cannot undergo predicate clefts is based on controversial data involving predicate clefts of verbs such as mo "know", pé "think" and fifé "want" which even he reports as ungrammatical for some speakers. The clefting of such predicates in Èdó produce sharply ungrammatical sentences. From the general perspective, it appears to be true that predicate cleft is possible for any verb which can be nominalized. Therefore, I propose that the failure of a verb to undergo predicate cleft must be due to the fact that the verb cannot be nominalized based on the nature of complex single event. This is true for Èdó resultative SVCs and splitting verbs in Yoruba as well as Èdó.

<sup>&</sup>lt;sup>20</sup> This data illustrates one property of these cognate objects; that they act as event delimiters so that for example when a cognate object is added to an activity verb it makes it telic/delimited:

These sentences show that when the cognate object is added to a (potential) activity verb it makes it delimited/telic so that it is no longer compatible with a for-time phrase (see Chapter four for more discussion)

nominalized as a single unit (cf. Baker (1989), Gruber and Collins (1996), Manfredi (1993)).<sup>22</sup> This is illustrated by the following sentences:

(37)	а.	Okuta gba ogiri fo (Gruber & Collins (1996)= 48) Stone hit wall break 'The stone broke the wall.'
	b.	*gbi-gba ni Okuta gba ogiri f <u>o</u> nom-hit NI Stone hit wall break
	c.	*fifò ni Okuta gba ogiri f <u>o</u> nom-break NI Stone hit wall break
	d.	gbi-gba-fo ni Okuta gba ogiri f <u>o</u> nom-hit-break NI Stone hit wall break

Note that this example has a resultative meaning and we observe that the predicate cleft of either of the verbs is ungrammatical; rather they must be clefted together. According to my analysis, (37a) is possible because the nominalized compound of both verbs is generated as a complement and then moves through EP at LF where the two verbs have raised after being attracted by the E head. Consequently, the nominalized event argument is properly licensed because there is Spec-head matching with the two verbs. This is illustrated in the structure (38).

'It was hitting and breaking that the stone did to the wall'

 $<sup>^{22}</sup>$  I do not know for a fact whether it is the same range of verb sequences that have been classified under the resultative SVC that are exactly the ones which allow this sort of compound nominalization in Yoruba. I will leave this open for future research. However, see the discussion of splitting verbs in Yoruba in section 5.3, so it might just be that sentences of the kind in (37) are splitting verbs.



The Yoruba data in (37d) and the structure in (38) are both desirable consequences of my analysis because the fact that both verbs must be clefted together implies that they license a single nominal argument of the complex single event in Spec of EP. However, this V-V compound structure is not attested in Èdó as the ungrammaticality of sentences similar to those in Yoruba show:

- (39) a. Okútá gbé úkpù gu<u>o</u>gh<u>ó</u> stone hit cup break The stone broke the cup.'
  - b. \*ùgbémw<u>èn ò</u>ré Òkútá gbé úkpù gu<u>ó</u>!gh<u>ó</u> nom-hit-nom Foc stone hit cup break
  - c. \*ùguóghómwèn òré Ôkútá gbé úkpù guó!ghó nom-break-nom Foc stone hit cup break
  - d. \*ùgbégu<u>óghó</u>mw<u>èn ò</u>ré Òkútá gbé úkpù gu<u>ó</u>!gh<u>ó</u> nom-hit-break-nom Foc stone hit cup break

According to (39), it is ungrammaticality to cleft any of the verbs separately (39b,c) or both verbs together (39d) in the resultative SVC. In fact, there are no such morphological compounds in Èdó of the form V-V as in Yoruba, i.e., V-V compounds are not licensed at a morphological level in Èdó: \*[V1-V2].

I propose that this basic difference between Èdó from Yoruba carries-over to the predicate cleft construction (and also to the possibility of verb movement as I will show in Chapter six). This morphological difference between Èdó and Yoruba with respect to V-V compounds has interesting consequences for Ìgbo where almost exceptionlessly Èdó resultative SVCs translate as V-V compounds (Chapter five for discussion). This discussion provides a window of opportunity for me to comment directly on an observation by R-M Déchaine (p.c.) that the fact that Yoruba allows V-V clefting with resultative SVCs incorrectly predicts that Yoruba should also permit resultative V-V compounds like Ìgbo. Under my analysis, this prediction can only be true for Yoruba if it can be shown that there is a Tense head that attracts, i.e., a head above the VP that triggers V-V raising, and this is contrary to fact (see Chapter six for discussion ).

# **3.5.2.** Consequential SVCs and CCs

In consequential SVCs where there are two EP projections and two event quantifiers, we expect that predicate cleft of either of the verbs will be possible since each event argument would be properly licensed in the relevant Specifier. Adopting the same strategy as the one used for resultative SVCs, let us first examine the process of predicate cleft with two transitive verbs in a simple clause:

- (40) a. Ôzó lé èvbàré Ozo cook food 'Ozo cooked the food.'
  - b. ùlémw<u>èn</u> <u>ò</u>ré Òzó lé èvbàré nom-cook-nom Foc Ozo cook food 'It is cooking that Ozo did to the food, not shredding.'

(41) a. Òzó rrí èvbàré Ozo eat food 'Ozo ate the food.'

> b. ùrémw<u>èn</u> <u>ò</u>ré Òzó rrí èvbàré nom-eat-nom Foc Ozo eat food 'It is eating that Ozo did to the food, not selling.'

From the data in (40) and (41), we observe that the transitive verbs lé 'cook' and ré 'eat'

can each undergo predicate clefts. This implies that they both can license event arguments.

Now, let us examine predicate clefts with transitive verbs from both consequential SVC

and CC:

consequential SVCs

(42)	a.	Òzó lé èvbàré ré Ozo cook food eat 'Ozo cooked the food and ate it.'
	b.	ùlémw <u>èn</u> <u>ò</u> ré Òzó lé èvbàré ré nom-cook-nom Foc. Ozo cook food eat 'It is cooking that Ozo cooked the food and ate, (not shred it).'
	с.	ùrémw <u>èn</u> <u>ò</u> ré Òzó lé èvbàré ré nom-eat-nom Foc. Ozo cook food eat 'It is eating that Ozo cooked the food (and ate it), (not sell it).'
(43)	а.	Òzó d <u>é</u> èmà kpèé Ozo buy drum beat 'Ozo bought the drum and played it.'
	b.	ùdémwèn òré Òzó dé èmà kpé!é nom-buy-nom Foc. Ozo buy drum beat 'Its is buying that Ozo bought the drum and played it, (it was not a gift).'
	c.	ùkpéémw <u>è</u> n <u>ò</u> ré Òzó d <u>é</u> èmà kpé!é nom-beat-nom Foc. Ozo buy drum beat 'Its is playing that Ozo bought the drum (and played it) (not give it away).'
CCs		
(44)	a.	Òzó gb <u>òó</u> ívìn bòló <u>ó</u> kà Ozo plant coconut peel corn 'Ozo planted coconut and peeled corn.'

b. ùgb<u>óó</u>mw<u>è</u>n <u>ò</u>ré Òzó gb<u>ó</u>!<u>ó</u> ívìn bòló <u>ó</u>kà nom-plant-nom cop. Ozo plant coconut peel corn 'It is planting that Ozo planted coconut and peel corn.'

- c. ùbólómwèn òré Òzó gbóló ívìn bólló ókà nom-peel-nom cop. Ozo plant coconut peel corn
   'It is peeling that Ozo planted coconut and peeled the corn, (not slice it).'
- (45) a. Òzó lé èvbàré rrí <u>ó</u>rè
   Ozo cook food eat it
   'Ozo cooked the food and ate it.'
  - b. ùlémwèn òré Òzó lé èvbàré rrí órè nom-cook-nom Foc. Ozo cook food eat it
     'It is cooking that Ozo cooked the food and he ate it, (not shred it).'
  - c. ùrémw<u>èn</u> <u>ò</u>ré Òzó lé èvbàré rrí <u>ó</u>rè nom-eat-nom Foc. Ozo cook food eat it 'It is eating that Ozo cooked the food (and ate it), (not sell it).'

Unlike in the resultative SVC, in these examples there are two separate EP projections which, according to my analysis, implies that the nominal argument of the event of each verb can be potentially licensed in its own Spec EP, and, therefore, predicate cleft is acceptable. However, there appears to be a paradox with the higher EP in consequential SVC since in this case it can have restricted function to only the event argument of the first verb. This behavior is, in fact, consistent with the semantic representation of the consequential SVC in (46b) and the corresponding syntactic structure in (46c):

- (46) a. Òzó dé èbé khién
   Ozo buy book sell
   'Ozo bought a book and sold it'
  - b. <u>The consequential SVC</u>
    ∃<sub>E</sub> ( ∃e<sub>2</sub> ∃e<sub>2</sub>[Buying(e<sub>1</sub>) & Agt(e<sub>1</sub>,Ozo) & Th(e<sub>1</sub>,èbé)] & [Selling(e<sub>2</sub>) & Agt(e<sub>2</sub>,Ozo) & Th(e<sub>2</sub>,èbé)] & [ E 'consists of (e<sub>1</sub>, e<sub>2</sub>)]]]



135

The simplest illustration of the semantic and syntactic representations comes from the prediction that cognate objects are possible in consequential SVC, unlike the resultative SVC, since there are two events that are connected. The relevant data is given in (47) and (48):

(47)	а.	Òzó lé èvbàré (ùlémw <u>è</u> n) ré Ozo cook food (nom-cook-nom) eat 'Ozo cooked the food (a cooking) and ate it.'
	b.	Òzó lé èvbàré ré (ùrémw <u>è</u> n) Ozo cook food eat (nom-cook-nom) 'Ozo cooked the food and ate it (a eating).'
	c.	Òzó lé èvbàré (ùlémwèn) ré (ùrémwèn) Ozo cook food (nom-cook-nom) eat (nom-cook-nom) 'Ozo cooked the food (a cooking) and ate it (a eating).'
(48)	a.	Òzó dé èmà (ùdémwèn) kpèé Ozo buy drum (nom-buy-nom) beat

'Ozo bought the drum and played it.'

- b. Òzó dé èmà kpèć (ùkpćémwèn)
   Ozo buy drum beat nom-beat-nom
   'Ozo bought the drum and played it.'
- c. Ôzó dé èmà (ùdémwèn) kpèé ùkpéémwèn
   Ozo buy drum nom-buy-nom beat nom-beat-nom
   'Ozo bought the drum and played it.'

The grammaticality of these sentences compared with the ungrammatical resultative SVC (35)-(36) provides confirming evidence for the proposal that there are two events in the consequential SVC whose nominalizations can be licensed separately in the relevant EP. I assume that the higher E head attracts only the first verb at LF in this case because there is a separate E head before the second verb and it counts as the closest operator that would license the event argument of the first verb. Therefore, predicate cleft is licensed by the closest E head that does the attracting. This is represented in the simplified structure in (46c) where the LF raising of the verbs and their event arguments are illustrated by the arrows.<sup>23</sup>

However, in Yoruba where V-V compounding is tolerated, it has been observed that the two verbs can also raise to the head of the first EP in predicate cleft from consequential SVCs.<sup>24</sup> According to my analysis, this can be derived from the scope of the higher E node in the consequential SVC because it existentially quantifies over the events denoted by both verbs and so it attracts both verbs at LF as illustrated in (48) and (49):

(48) a. <u>Bó</u>lá sè <u>e</u>ran tà Bola cook meat sell 'Bola cooked the meat and sold it.'

 $<sup>^{23}</sup>$  This structure is also simplified in terms of getting the first verb before the object. This is derived by the same binding chain between the top verb and empty verb that is illustrated in the resultative SVC structure.

 $<sup>^{24}</sup>$  There are a lot of things unclear about the Yoruba data. For example, a striking fact that is very often never translated into the commonly cited data is that it is possible for a clitic Ó to occur before the second verb in the predicate cleft of a sentence like (47);

sí-sè-tà ni Bólá sè eran O tá

nom-cook-sell that Bola cook clitic sell

<sup>&#</sup>x27;It was cook and sell that Bola cooked that meat and he sold (it)'

also see footnote (25) for more observations concerning Yoruba predicate cleft.

- b. Sí-sè-tà ni <u>Bó</u>lá sè <u>e</u>ran tà (Baker (1989=74a)) nom-cook-sell that Bola cool meat sell 'It was cook and sell the meat that Bola did.'
- (49) a. Ade ra iwe fún Bisi Ade buy book give Bisi 'Ade bought a book and gave it to Bisi.'
  - c. Rí-rà-fún ni Ade ra iwe fún Bisi (Baker (1989=74ab)) nom-buy-give that Ade buy book give Bisi 'It was buy the book and give to Bisi that Ade did.'

As the data above shows, it is possible to cleft both verbs in the consequential SVC just like the resultative SVC where both verbs can raise to the same E in order to license a single event argument which shows up in predicate clefts.<sup>25</sup> Therefore, I propose that the predicate cleft of both verbs is possible in the consequential SVC by assuming the same analysis as in the resultative SVC where the higher E existentially quantifies over the macro event denoted by the two verbs. Under this assumption, what makes the clefting of both verbs possible in the consequential SVC is the fact the existential quantifier represented in the first E can bind the two events denoted by the verbs, a property that is shared by true SVCs.

However, R-M Déchaine (p.c.) has observed that this possibility of clefting the two verbs in the consequential SVC in Yoruba poses a problem for my analysis. This is in light of the fact that the structure of consequential SVCs is the same in Yoruba and Èdó which then predicts that both languages should permit the first and second verbs to be clefted. This prediction is not borne out: while Yoruba permits both verbs of the consequential SVC to be clefted at the same time, Èdó does not. It is not immediately obvious why there is this difference between the two languages, however it is quite possible that it could be that the same mechanism that allows V-V clefting also licenses VP predicate cleft in Yoruba (VP

 (i) fifun ni Ade ra iwe fun Bisi nom-give that Ade buy book give Bisi
 'Its is buying that Ade bought the book for Bisi, not say, that he stole the book'

 $<sup>^{25}</sup>$  I have also been informed (M. Olusegun (p.c.)) that it is possible to have the predicate clefts of these verbs separately like in Èdó contrary to popular belief that the second verb of the SVC almost always never undergo predicate clefts (cf. Manfredi and Laniran 1988, Déchaine 1993, etc.);

predicate cleft involves clefting of verb + object + other modifiers (adverb, adjective)). Yet, it stands to reason that whichever way the actual mechanism that is responsible for this difference is stated, it should be such that we can derive the fact that whereas Yoruba permits units of V, V-V and VP to be clefted, Èdó only allows a single V to be clefted (V-V and VP are both ruled out). I will leave these issues open for future research.

Turning now to CCs (44) and (45), we notice, like in the consequential SVCs, that it is grammatical to have the predicate cleft of either of the verbs. According to my analysis, this is possible because each event is independently quantified over by the head of distinct EPs. This is consistent with the semantic representation of CCs given in (50a) and the corresponding syntactic structure (50b) for a sentence like (45).

(50) a. <u>covert coordination</u>

( $\exists$ ei)[ Planting(e1) & Agt(e1, Ozo) & Th(e1, coconut)] & ( $\exists$ e2)[ Peeling(e2) & Agt(e2, Ozo) & Th(e2, corn].



Based on the structure in (50b), it is possible to predicate cleft either of the two verbs because there are two parallel projections of EP where the nominal realization of the event of each verb can be properly licensed. Unlike the consequential SVC, it is not possible to cleft both verbs in CCs in Yoruba. Thus, there is a clear distinction between SVCs and CCs that can be derived from the difference in the nature of event quantification associated with each E head.

## 3.6 Conclusion

In conclusion, the results from predicate cleft in SVCs are consistent with those from adverb and PP placements: there are two major kinds of SVC that can be distinguished on the basis of the difference in the projection of EP. A resultative SVC has one EP projection, while the consequential SVC has two projections of EP. This functional projection, according to the predicate cleft test, is compatible with the nature of event composition and quantification: resultatives express a single-event and hence one projection of EP while consequential SVCs express two (connected) event with two asymmetrical EP projections. CCs are like consequential SVCs because they contain two EP projections and there are two events, however the relationship between the functional projection in CCs is a symmetrical one and so while it is possible to cleft both verbs together or individually in consequential SVC, CCs allow only individual clefting. In this regard, resultative SVCs are typically different, they either uniformly do not the allow clefting or where possible the two verbs must be clefted together. All of these differences correlate with the scope properties of the head of the functional projection EP, and they define the event quantifications realized by predicate cleft .

## **Chapter Four**

## **Double Objects and Object sharing SVCs**

# 4.1 Introduction

In the last two chapters, I discussed two general syntactic tests (adverb placement and predicate clefting) that were used to motivate the distinction between resultative and consequential SVCs, as well as CCs. However, there are several important similarities between resultative and consequential SVCs that are worth emphasizing. First, the facts from I-type adverb and predicate clefts show that in both resultative and consequential SVCs a single E operator can have scope over both (sub)-events defined by the two verbs. This is true in spite of the fact that there is a sense in which the events denoted by the verbs in the consequential SVC are less sub-atomic than those in the resultative SVC. Second, both resultative and consequential SVCs appear to be similar in that they have a single Agent (that is introduced by a single voice head), there being no evidence for SVC-internal null subject. Finally, the verbs in both resultative and consequential SVCs are involved in some kind of 'object sharing' although, this sharing has been shown to be mediated by an empty category *pro* in consequential but not in resultative.

Against this background, this chapter presents evidence based on the possibilities of double objects in Èdó that will further confirm the distinction between resultative and consequential SVCs. In particular, I will illustrate how double object constructions provide a strong argument against a unified analysis of both resultative and consequential SVC, such as that of Baker (1989, 1991). In addition, I will consider some of the general constraints on object sharing in SVCs as well as examine the double object constructions in a wider context. For example, I will argue for an analysis in which double object constructions (DOCs) are a kind of resultative construction having similar event structure as resultative SVCs. Furthermore, I will propose an analysis that distinguishes between underlying and derived objects.

## 4.2 SVCs and Double Objects Constructions (DOCs)

In this section, I provide evidence that further validates my analysis of SVCs, in particular the distinction between resultative and consequential SVCs. This is based on the examination of novel data from double objects in SVCs. I will argue for an analysis of double objects where the Indirect object (goal) is in Spec of AspP (cf. Travis 1991, Baker 1997 etc.) while the Direct object (theme) occupies the Specifier of VP.<sup>1</sup> Let us begin then with an introduction to double object constructions (DOCs) in Èdó in order to present the analysis and its interaction with SVCs.

# 4.2.1 DOCs and the Associative Construction

Simple triadic verbs in several Kwa languages are few and somewhat complicated, but those given in (1) from Èdó will suffice:  $^{2}$ 

(1)	а.	Ózó	hàé	úyì	íghó	
		Ozo	pay	Uyi	money	
		'Ozo	paid (	Jyi so	me money	<i>.</i> '

b. úyì màá ìs<u>òkèn</u> èbé
 Uyi show Isoken book
 'Uyi showed Isoken the book.'

These are like English DOCs which seem semantically to imply a null preposition 'to' between the objects and so for example (1a) has a reading in which Ozo paid money to Uyi, where Uyi is the goal (recipient) and *ighó* is the theme. However, there are other DOCs which have an adversative interpretation in which there seems to be a null 'from' preposition between the double objects. For example, the Èdó equivalent of the verb "give"

<sup>&</sup>lt;sup>1</sup> In the framework developed in Travis (1991) and adopted by (1997), there is a distinction between Inner and Outer aspect. Therefore, Aspect is mostly a separate projection from E. However, in my analysis the top E (first E) can be compared to an Outer Aspect, while Inner Aspect is distinctly the projection that hosts the Indirect object (goal) of a double object construction.

 $<sup>^2</sup>$  In Igbo, the double object construction is very often the environment where one will find the applicative suffix on the verb, while the double object construction has been observed to be entirely absent n Yoruba (cf. Manfredi 1991, V.Carstens p.c.). See also Saah and Eze (1997) for discussion of asymmetries between Akan and Igbo DOCs.

is not a simple verb since it consists of a verb "rhie" and some kind of dative particle "na", each of which has an object after it. This is illustrated in (2):

- (2) a. Òzó rhié íghó nè úyì Ozo take money to Uyi 'Ozo gave the money to Uyi'
  - b. Òzó rhié úyì íghó
     Ozo take Uyi money
     '#Ozo gave Uyi money'
     '√Ozo took Uyi's money'

An analysis of DOCs such as Larson (1988) would predict that (2b) is derived from (2b) on a par with the analysis of the ditransitive verb 'give' in English. However, this is not quite the case with these Edó example in (2). In (2a), we observe that the theme object *ighó* is also the direct object of the verb rhié, while the goal object is the oblique object of the dative particle  $n\dot{e}$ . However, in (2b) where we attempt to derive a double object construction from (2a) by moving the indirect object to the immediate position after the verb, and omitting the preposition, we observe that the resulting double object sentence has a completely different meaning from the one expected in a language like English on a par with those in (1), i.e., Ozo gave Uyi's money. Rather, (2b) now has a source/ adversative reading; Ozo took Uyi's money or Ozo took money from Uyi. I propose that what has happened here is that the verb 'give' is expressed as a complex verb in Edó made up of the verb rhié 'take' and the particle 'ne' and the prepositional particle cannot be incorporated into the verb as argued for in Larson (1988). Thus, there is no simple verb 'give' in Èdó, rather it is a complex verb whose parts cannot be omitted and so it cannot license double objects like its English counterpart. The interesting thing that this discussion raises is the fact that (2b) is actually related in meaning to what is known as the Associative construction (somewhat like the construct-state construction) (Agheyisi 1990).

The Associative Construction is commonly found in several Kwa and Bantu languages (cf. Schaefer 1997, Hyman 1996 etc.). It involves the putting together of two

nouns in an ordered semantic relation of possessed-possessor (alienable, inalienable), to create a single phonological and syntactic domain. This is illustrated by the derivations in (3) in which HT stands for a floating high tone that is characteristic of this construction.<sup>3</sup>

(3)	possessed NP	<u>Asso</u>	ociative ton	<u>ie</u>	Possessor_NP	=	Associativ	e Construction
a.	èbé 'book'	+	´ (HT)	+	úyì	=	èbú!yì	'Uyi's book'
b.	òw <u>è</u> 'leg'	+	(HT)	+	Òtà	=	ðwó!tà	'Ota's leg'
c.	òsé 'beauty'	+	- (HT)	+	ékítà 'dog'	=	òsé!kítà	'dog's beauty'
d.	íhién 'finger'	+	(HT)	+	àdésúwà	=	íhián!dés	úwà 'A's finger

In simple descriptive terms, what happens in the Associative construction as illustrated in (3) is that two lexical NPs are joined together by a construction high tone. Thus, the two NPs become one phonological unit as well as one syntactic constituent. The possessed NP is always on the left while the possessor occurs on the right. For example, in (3b) the possessed NP  $\partial w \dot{g}$  'leg' is associated semantically with a possessor NP  $\partial t \dot{a}$  via the associative high tone.

The evidence for the phonological unit comes from the delinking of the tone of the final syllable in the possessed NP, coupled with the elision of the tone bearing segment. The initial vowel of the possessor NP replaces the final vowel of the possessed NP and thereafter the associative high tone spreads to the left displacing the lexical tone on the final syllable of the possessed NP. This delinking of an autosegment creates a floating tone which is realized as a downdrift within the derived word represented orthographically by an exclamation mark (by convention).

This phonological analysis of the associative construction as a constituent is supported by two pieces of evidence from syntax. First, observe that the derived NP

<sup>&</sup>lt;sup>3</sup> Agheyisi (1986, 1990) suggests that the source of this high tone is the genitive marker '<u>ó</u>ghé' which is like the English 'of' insertion in complex NPs, although the Èdó '<u>ó</u>ghé' may sometimes have a partitive reading.

constituent can be determined as a whole and when it is separated the second noun is introduced by a preposition head. Consider the following:

- (4) a. né!né òwó!tà bí<u>gó!ó</u> The Ota's leg be-bent 'The (particular) Ota's legs are bent'
  - b. né!né dwè óghé né!né Òtà the leg genitive the Ota 'The (particular) leg of (a specific) Ota

In (4a), we observe that the associative construction can be a DP since the NP contained therein can be determined by the definite article  $n\acute{e}!n\acute{e}$  'the' (cf. Omoruyi 1987). This contrasts minimally with (4b), where we find that it is possible for each noun to have a determiner when they are linked by an overt genitive marker  $\acute{o}gh\acute{e}$  'of'. Consequently, on the basis of the data in (4), I propose a structure for the associative construction in which the possessed NP plus construction tone and possessor NP form an NP, and this NP can be a complement of a Det head. This is illustrated in (5) for a sentence like (4a).

(5) DP D NP né!né N KP/PP Òwè K/P NP/DP '/óghé | N Òtà

The second syntactic evidence for treating the associative construction as a single DP is based on the fact that it behaves as a syntactic constituent in terms of movement. Consider the following;

(6) a. Ozó tá wèć owó!tà bigòó Ozo say that Ota's leg be-bent 'Ozo said that Ota's leg is bent'

- b. Dè èmwin nè Özó tá w<u>èé ó</u> bi<u>gòó</u> Q thing that Ozo say that it be-bent 'What did Ozo say that is bent'
- c. dwó!tà dré Ozó tá w<u>eé ó</u> bigdó Ota's leg Foc. Ozo say that it be-bent 'It is Ota's leg that Ozo said that is bent'
- d. \*òwè! <u>ò</u>ré Òzó tá w<u>èé</u> Òtà big<u>òó</u> leg Foc. Ozo say that Ota be-bent
- e. Otà <u>ò</u>ré Ozó tá w<u>è</u>é òw<u>è</u> <u>é</u>rè bi<u>gòó</u> Ota Foc. Ozo say that leg his be-bent Its is Ota that Ozo said that his leg got bent'.

What these simple movement illustrations suggest are as follows: In (6b), we observe that it is possible to question the entire DP involved in the associative construction by whmovement. The presence of a subject resumptive pronoun  $\underline{o}$  also provides evidence that we are dealing with a DP projection (see section 4.3.2 below for more on resumptive pronouns). Similarly, the NP-movement by focus cleft in (6c) shows that the associative construction as a whole is a constituent. These facts are consistent with the standard assumption that only constituents can be wh-moved. In (6d) we note that it is ungrammatical to move the head of the constituent which by itself is not an NP, but in contrast observe that the possessor NP/DP can move (6e) and a resumptive also shows up in the position from which it has moved. Therefore, I conclude that the associative construction is a DP with a structure like the one in (5).

Now that we have discussed the associative construction, we are in a position to reexamine the meaning of the double object sentence in (2b). Consider the following examples where (2b) is repeated as (7b).

(7)	a.	Òzó rhié èbú!yì Ozo take book of Uyi 'Ozo took Uyi's book'	'Associative Construction'
	b.	Òzó rhié úyì èbé Ozo take Uyi book 'Ozo took the book of Uyi'	'DOC'

(8)	a	Òzó gu <u>ò</u> gh <u>ó</u> òwú!zò Ozo break leg antelope 'Ozo broke the antelope's leg'	'Associative Construction'

Òzó guòghó úzò òwè 'DOC'
 Ozo break antelope leg
 'Ozo broke the leg of the antelope'

In these sentences (7) and (8), we notice that there is a systematic alternation between the Associative Construction ((a) sentences) and surface double object constructions ((b) sentences). Based on what we know about the Associative Construction, it is easy to see that there are tonal contrasts between the (a) and (b) sentences apart from the apparent word order variations. Significantly, in the double object construction version there is no floating high tone which is the associating tone link between the nouns in the (a) sentences. I take this difference in tone to be a signal of a difference in syntactic structure as well. Thus, I propose that the (b) sentences. Furthermore, (2b) is simply a double object construction which derives from a comparable underlying structure to the one in (5), and this illustrates the sentence in (7a)

I propose an analysis of double objects derived from the associative construction based on Travis (1991) whereby the derived object (indirect (possessor) NP) occupies the Specifier of a functional projection, AspP. Furthermore, I also assume an analysis parallel to that of double objects derived from dative and locative alternations in Baker (1997) whereby the source (goal) object undergoes NP-movement to AspP in the syntax after Pincorporation (of the null preposition head) has taken place. Therefore, double objects are derived from associative constructions by moving the possessor NP out of a DP to the Specifier of AspP. According to Travis (1991), AspP (i.e., Inner Aspect) occurs between Larson's (1988) VP-shells, as such the structure of double objects that is derived from the associative construction is given in (9). (I omit VoiceP for ease of exposition.)

146



According to the structure in (9), the AspP projection splits Larson's (1988) VP-shells but I assume that the verb is base-generated in the top verb position from where it binds the lower empty verb.<sup>4</sup> Furthermore, I propose that the null preposition implies something like "from" which is the corresponding version of the dative "to" in English (cf. Larson 1988). Subsequently, the floating associative high tone deletes by an operation similar to P-incorporation, or alternatively it could be just never inserted. The end result is a structure where the (possessed) object NP is left behind in the DP of the Specifier of the lower VP and thus creates a DOC on the surface.<sup>5</sup>

147

<sup>&</sup>lt;sup>4</sup> As pointed out by R-M Déchaine (p.c.), the fact that the assumption that V is base-generated in the top V position does not follow from the more general claim that there is never V-movement in Èdó. In fact, I do admit that V-to-T movement is distinct from so-called short verb movement, i.e., V-to-V raising. However, I propose that both movements are different and thus triggered by very different principles. Therefore, V-to-V movement could also be a case of head movement, but according to my theory it is not triggered by attract and so it is not relevant to the issue of the serial verb parameter. Consequently, I will continue to assume that V is base-generated in the top V position and especially so since there is no evidence in support of V-to-V movement in Èdó or V-movement within the VP.

<sup>&</sup>lt;sup>5</sup> Direct evidence against a possessor raising analysis for DOCs derived from associative construction comes from sentences of the kind in (i)

 <sup>(</sup>i) Òzó rhié àdésúwà èbé úyì
 Ozo take Adesuwa book Uyi
 'Ozo took from Adesuwa Uyi's book'

We can verify this analysis immediately by using the movement test to show the difference between associative construction and the DOC.

- (10)Ózó gu<u>òghó</u> úzò òw<u>è</u> a. Ozo break leg antelope 'Ozo broke the leg of the antelope.'
  - Dè èmwin nè Òzó guòghó órè òwè b. Q thing that Ozo break it leg 'What did Ozo broke its the leg.'
  - òw<u>è</u> òré Ózó guòghó úzò c. leg Foc. Ozo break antelope 'Its a leg that Ozo broke of the antelope.'

According to the data above, it is possible to move either of the NPs from the sentence in (10a) by wh-movement (10b) or similar movement in focus cleft (10c) in contrast to (6). Therefore, I conclude that the NPs in (10a) must be separate constituents in the derived structure, of the form [NP V NP1 NP2].<sup>6</sup>

#### 4.2.2 **DOCs in Resultative and Consequential SVCs**

This section discusses the asymmetry between resultative and consequential SVCs with respect to double objects derived from the associative construction. The contrast is illustrated in the following sentences:

(11)	a.	úyì sùá Òzó <u>è</u> wé Uyi push Ozo goat 'Uyi pushed Ozo's goat. '	(DOC)
	b.	*úyì sùá Òzó <u>è</u> wé dé Uyi push Ozo child fall 'Uyi pushed Ozo's child down.'	(resultative)

It is evident that Adesuwa is not the possessor of the theme NP 'Uyi's book' and so the DOC could not have arisen from possessor raising, and this is compatible with the analysis I have proposed. In fact, a possessor raising account predicts the sentence in (ii) (ii)

- \*Òzó rhié èbé úyì àdésúwà
- Ozo take book Uyi Adesuwa

However (ii) is ungrammatical and this is evidence, therefore, against a possessor raising analysis of the DOCs.

<sup>&</sup>lt;sup>6</sup> I defer discussion of the clitics that show up in the extraction site of the indirect object till section 4.3.2.

	c.	Òzó gbé Àdésúwà émá!t <u>ó</u> n Ozo hit Adesuwa metal 'Ozo hit Adesuwa's metal.'	(DOC)
	d.	*Òzó gbé Àdésúwà émá!tón pèrhé Ozo hit Adesuwa metal be-flat Ozo hit Adesuwa's metal flat.'	(resultative)
(12)	a.	úyì vb <u>ó ò</u> kh <u>ó</u> !kh <u>ó</u> ìgàn Uyi pluck chicken feather 'Ozo plucked the chicken's feather.'	(DOC)
	b.	úyì vb <u>ó òkhó</u> !khó ìgàn khiện Uyi pluck chicken feather sell 'Ozo plucked the chicken's feather and sold	(consequential)
	c.	úyì kòkó Àdésúwà ùpk <u>ò</u> n Uyi gather Adesuwa cloth 'Uyi gathered Adesuwa's cloth.'	(DOC)
	d.	úyì kòkó Àdésúwà ùpk <u>ò</u> n vi <u>ó</u> Uyi gather Adesuwa cloth take 'Uyi gathered Adesuwa's cloth and took the	(consequential) em.'

This contrast that is illustrated by (11) and (12) is a remarkable one for the analysis of SVCs, in particular the distinction between resultative and consequential SVCs. What we observe is that it is ungrammatical to have double objects derived from the associative construction in a resultative SVC, while this is perfectly grammatical in the consequential SVC. There are two general issues arising from this contrast that I see as relevant to the goal of distinguishing between resultative and consequential SVCs: (a) How are double objects licensed in SVCs and what are the constraints ? (b) What are the consequences for object sharing.

I begin with the first issue of how double objects are licensed in SVCs and the constraints on it. In order to present a clear discussion, I will first examine double objects in the resultative SVC. According to the data, the problem cannot be that it is because the first verb does not allow double objects, since it clearly does in simple clauses (11a,c). One possibility is to derive the ungrammaticality of double objects in resultative SVCs from the restriction on the second verb as an unaccusative since there are constraints on unaccusative verbs with respect to dative shift (cf. Baker 1992). However, based on the syntactic tests

from chapter two, we know that the object position is simultaneously licensed by both verbs--the object is the single internal NP argument of both verbs. Therefore, I suggest that whatever is responsible for the ungrammaticality must be something that is relevant to both verbs.

Against this background, I propose that the ungrammaticality of double objects is based on the general aspectual restriction that there can be only one delimiter in a clause. The idea is that the same aspectual restriction which rules out a second unaccusative verb from occurring in the resultative SVC also excludes other delimiter phrases-- including double objects, where one of the objects has often been analyzed as an event delimiter (Tenny 1987, Hoekstra 1992 etc.). Consider the following English examples:

- (13) a. John split a coconut [open]
  - b. John split Mary a coconut.
  - c. \*John split Mary a coconut open.

(13a) is an example of an AP-resultative in English, while (13b) is an example of a double object construction with the same verb as in (13a) and both sentences are grammatical. However, (13c) shows that it is ungrammatical to have double objects inside the resultative sentence. This paradigm is just like the resultative SVC in (11) in Èdó.

Therefore, it follows that the ungrammaticality of double objects should be derived from a general fact about resultative constructions. In section 2.8, I argued that the resultative SVC is a single event that can only be delimited once. (14) also show that the second object of the DOC is an event delimiter:

- (14) a. Ozó gbé émá!tón là áwà ókpá Ozo hit metal for hour one 'Ozo hit the metal for an hour.'
  - b. \*Ôzó gbé àdésúwà émá!t<u>ó</u>n là áwà <u>ó</u>kpá Ozo hit Adesuwa metal for hour one 'Ozo hit Adesuwa's metal for an hour.'

# c. \*Òzó gbé émá!tón pèrhé là áwà ókpá) Ozo hit metal be-flat for hour one) 'Ozo hit the metal flat for an hour.'

In (14a) the atelic activity verb can be modified by a temporal adverb 'for an hour', however as (14b) shows when there are double objects with the same verb the 'for an hour' temporal adverb is ungrammatical on a par with (14c). In (14c), a resultative SVC is incompatible with the same 'for an hour' temporal adverb. Therefore, I conclude that there is a single complement position in the resultative SVC which may either be filled by the unaccusative second verb, or in the case of the DOC the indirect object that starts out as a PP complement which is in a canonical delimiter position, as in (9). Thus the ungrammaticality of the DOC in resultatives come from a competition for the single delimiter (complement) position between the unaccusative second verb and the PP which contains the indirect object. This is illustrated in (15).



The other side of the question that I am discussing is the issue of possible constraints on the licensing of double objects in the consequential SVC. Recall the discussion of the fact in section 2.8 that there are no aspectual restrictions on the

consequential SVC (but see Pi and Stewart 1998). Thus, unlike the resultative SVC, stacking of verbs of the right kind is allowed in the consequential SVC. Similarly, the possibility of double objects in the consequential SVC (12b,d) implies that it has a structure that is less tight than the resultative. Since the second verb projection is not generated in the complement position, it is not incompatible with a DOC. According to my analysis of the DOC, the AspP projection occurs between the upper verb and the base-generated verb position that it binds. At this point, we are faced with two theoretical issues: are there two separate AspPs for each VP, and which of these objects is involved in object sharing? I will address each of these questions in turn.

The positioning of manner adverb in a consequential SVC with double objects can help to tell us whether there are two projections of AspP. This is based on the fact that the placement of an N-type adverb should correspond to the presence of a VP boundary since it is licensed as a right-adjunct to the VP.<sup>7</sup> Furthermore, the position of the I-type adverb should provide evidence for a functional projection. These predictions are illustrated by the following sentences:

(16)	a.	*úyì vb <u>ó òkhó!khó ègìégìé</u> ìgàn khién Uyi pluck chicken quickly feather sell
	b.	úyì vb <u>ó òkhó</u> !khó ìgàn <u>ègìégìé</u> khi <u>é</u> n Uyi pluck chicken feather quickly sell 'Ozo plucked the chicken's feather quickly and sold it.'
	c.	*úyì vb <u>ó òkhó</u> !kh <u>ó gié</u> !gié ìgàn khién Uyi pluck chicken quickly feather seil
	d.	úyì vbó òkhó!khó ìgàn gié!gié khién Uyi pluck chicken feather quickly sell 'Ozo plucked the chicken's feather and quickly sold it.'

From the data in (16) we observe that neither the N-type nor the I-type adverb can occur between the two objects (16a,c). This implies that both objects are within the projection of

<sup>&</sup>lt;sup>7</sup> I ignore, at the moment, the earlier observation that N-adverb can also attach as right adjunct to EP since it has no immediate consequences for the analysis of DOCs.

the first verb, i.e., the first object does not constitute a VP along with the verb in a leftbranching structure. The correct order of N-type adverb placement is after the second object (16b), and this implies that both objects constitute a VP along with the first verb. Furthermore, there is the expected fact based on the structure of the consequential SVC that the second EP adjoins to this VP containing the double objects and this is shown in (16d) where the I-type adverb occurs before the second verb, after the double objects. All of these are consistent with the proposal that there is an AspP projection in the extended projection of VP1 according to the analysis that I have assumed for double objects. However, adverb placement does not tell us about the internal structure of the second verb phrase.

In order to find out more about the internal structure of the second verb phrase, i.e., whether it contains another projection of AspP, I now turn my attention to the exact interpretation of the consequential SVC with double objects with respect to object sharing. Consider the following:

- (17) úyì vb<u>ó</u> <u>òkhó!khó</u> ìgàn a. Uyi pluck chicken feather 'Ozo plucked feathers from the chicken.' Ь. \*úyì vb<u>ó òkhó!khó</u>; ìgàn khi<u>é</u>n proj Uyi pluck chicken feather sell 'Ozo plucked the chicken's feather and sold it (the chicken).' c. úyì vb<u>ó ò</u>kh<u>ó</u>!kh<u>ó</u> ìgàn<sub>i</sub> khién proj Uyi pluck chicken feather sell 'Ozo plucked the chicken's feather and sold it (the feather).' (18) úyì kòkó Àdésúwà ùpkòn a. Uyi gather Adesuwa cloth 'Uyi gathered Adesuwa's cloth.' b. \*úyì kòkó Ádésúwà; ùpkòn mú proj Uyi gather Adesuwa cloth carry 'Uyi gathered Adesuwa's cloth and carried her.' úyì kòkó Àdésúwà ùpk<u>ò</u>nj c. mú proj
  - Uyi gather Adesuwa cloth carry 'Uyi gathered Adesuwa's cloth and carried them (away).'

In the simple double object construction (17a) and (18a), the possessed NP is the theme while the possessor NP is typically a source. Now, when the DOC occurs in the first part of a consequential SVC, the source phrase cannot be understood as the shared object. This is evident from the ungrammaticality of the sentences (17b) and (18b) in which a coreference reading is intended between the null *pro* object of the second verb and the indirect object of the first verb. The ungrammaticality of this coreference implies that the indirect (goal or source) object is not the shared object.<sup>8</sup> However, in (17c) and (18c) where the coreference is between the null *pro* and the theme direct object the sentences are perfectly grammatical. (Observe that in (18) I switched the second verb to  $m\dot{u}$  (carry), this is to provide a fair chance for  $Ad\acute{esuwa}$  to be interpreted as the shared object since  $vi\acute{o}$  normally would only take plural count nouns, whereas Uyi mu Adesuwa is grammatical.)

Consequently, I conclude that since the facts of object sharing are the same in both single and double object consequential SVCs namely, the (theme) direct object is always shared, therefore, there is no evidence for another functional projection in the second VP. This conclusion is illustrated clearly by the following contrast:

(19)	a.	Ozó mi <u>é</u> n ùkp <u>ò</u> nk kòkó prok Ozo see cloth gather 'Ozo saw the clothes and gathered them.'
	b.	*Òzó mi <u>é</u> n ùkp <u>ò</u> nk kòkó àdésúwà prok Ozo see cloth gather Adesuwa

In (19a), the theme direct object is shared between the two verbs and the null *pro* is generated in the Specifier of VP2. However, (19b) is ungrammatical with the same coreference as (19a) because in this case the indirect object is present. This implies that there is no landing site for the source object in VP2 and so there is no (Inner) AspP

<sup>&</sup>lt;sup>8</sup> Collins (1997) makes a similar remark that the object of a preposition cannot be the shared object. See also Baker and Stewart (1997b) who demonstrate that no object apart from the (theme) direct object can be involved in object sharing.

projection in the projection of the second verb. The representation for (17c) is given in the simplified structure in (20) in which TP-VoiceP are omitted.



In these double objects, as with single object, consequential SVC I assume the same process of null *pro* licensing argued for in Baker and Stewart (1997b) that was alluded to in footnote (31) in section 2.5.2 of chapter two and repeated below for convenience.

(21)	Null pro_licensing condition_					
• •	pro is licensed in Edó if and only if;					
	(i) it is governed by a verb, and	(formal licensing)				
	(ii) it is locally bound by an operator	(identification of content)				

In the structure in (20), the null *pro* object of the second verb is properly licensed because it is governed by the verb and also bound by the higher E operator. Accordingly, I claim that coreference is not possible between source and *pro* because *pro* is not properly licensed (see section 4.3.2). Thus, the facts of double objects reinforce the significant nature of the two E heads in the consequential SVC. Furthermore, we observe that the presence of the functional projection AspP in the first VP whose Specifier is occupied by the goal (indirect) object does not interrupt object sharing. One interesting side remark about the structure of consequential SVC and double objects is that it provides evidence against a complementation-like analysis with c-command that can be used to establish the control (coreference) relation between the shared objects under an analysis like Collins (1997). This is based on the interaction of coreference between the double objects in the projection of VP1 with a pronoun or an anaphor contained in the VP2. Consider the following:

- (22) a. Özó fián <u>è</u>wé <u>è</u>rhùnrhùnmwùn Ozo cut goat tail 'Ozo cut the tail from the goat'
  - Ózók fián <u>è</u>wéj <u>è</u>rhùnrhùnmwùni rhié proi nè <u>é</u>rèj/\*k ègb<u>é</u>rèk/\*j
     Ozo cut goat tail give it to him himself

Very simply, what the sentence in (22b) illustrates is that despite the fact that an anaphor embedded within a PP inside the VP2 can be bound by the subject (implying that the whole TP is one governing category) yet the same anaphor cannot be bound by the indirect object (goal) NP  $\underline{\dot{e}}$ wé (goat). This implies that the indirect object of VP1 does not c-command the prepositional object contained within VP2. Then, by the transitivity of c-command we infer that the (theme) direct object of VP1 also does not c-command the *pro* direct object of VP2. (see Baker and Stewart (1997b) for extensive discussion and illustration). Therefore, I conclude that the structure of the consequential SVC can only be like in (20) where VP2 is not contained in VP1.

We can verify the c-command relation between the two objects of VP1 based on some of the tests from Barss and Lasnik (1986) which show that the indirect object ccommands the direct object. Consider the following:

# anaphor binding

(23) a. I màá úyì ègbérè vbè ùghègbè I show Uyi himself in mirror 'I showed Uyi himself in the mirror.'

b.	*Ì	màá	ègbérè	úyì	vbè	ùghègbè
	I sl	how	himself	Ŭyi	in	mirror

## each-the-other construction

(24)	a.	Í màá	Íràn	ùghúghán	íbátà	nókpá		
		I show	them	each	shoe	other		
		'I show	'I showed them each other's shoes.'					

b. \*Ì màá íbátà nókpá Íràn ùghúghán I show shoe other them each

# negative polarity and pronoun coreference

- (25) a. <u>Ômó rrôkpaj</u> má rrí èmà érèj Child neg. Pol not eat yam his 'No childj ate hisj yam.'
  - b. \*íyé <u>òmó</u> r<u>ò</u>kpàj má b<u>òó</u> <u>é</u>rèj mother child neg.Pol not comfort him

The (a) sentences show that there is c-command between the double objects, for example in (23a) the indirect object must linearly precede and c-command the direct object. Now, in the (b) sentences, where the order is either reversed (23b) and (24b) or more structure is introduced (25b) which violate the c-command condition, these sentences are ungrammatical. Thus, I conclude that there is c-command relation between the objects in the double object construction consistent with my analysis and structure, as in (20).

This analysis of the structure of the consequential SVC that is based on double objects and object sharing also provide a window of opportunity to re-examine certain claims of Baker (1989). In his study of the SVC, Baker (1989) observes that a phrase headed by a triadic verb such as 'give' can appear embedded within a projection of some other verb. When this happens, he argues, the theme argument of the three-place verb may appear as the object of the higher verb, but the goal object must always appear in the first (lowest) projection of the dative verb. Baker took this as evidence for a hierarchical relationship between the goal and the theme, consistent with the thematic hierarchy (cf. Larson 1988). The following Èdó sentences from Baker (1997) illustrates these claims:

(26)	a.	$\dot{O}z\delta [V_x \text{ rhie igho } [V_x \text{ hae uyi}]]$	(Baker = (61))
		Ozo take money pay Uyi	
		'Ozo took money and paid it to Uyi'	

 b. \*Ôzó [V<sub>x</sub> guàló úyì [V<sub>x</sub> hàé íghó]]
 Ozo find Uyi pay money 'Ozo found Uyi and paid him money'

According to Baker's analysis, in (26a) the object of the dative verb is assigned the internal theme roles of both the first verb *rhié* 'take' and the second verb  $ha\dot{e}$  'pay' and the theme argument is projected higher than the goal argument. This is consistent with the thematic hierarchy, and therefore grammatical. However, (26b) is ungrammatical because the goal argument of the second verb *hae* 'pay' c-commands the theme argument of the same verb, contrary to the thematic hierarchy. While much of Baker's observations and analysis are very insightful, it, most definitely can use some improvements. I will focus specifically on the facts from double objects.

First of all, the possibility of double objects show Baker's structure for the consequential SVC to be incorrect. As an illustration, his structure for (26a) is as in (27).

(27) IP Spec I' Òzó I VP v NP V' rhiế íghó V NP hàế úyì

(27) is a doubly-headed VP structure in which the two verbs are co-heads of the same VP and the theme object c-commands the goal argument. However, this structure cannot accommodate the fact that double objects are possible before the second verb in the consequential SVC. For one thing, the structure is in most respects like that of the resultative SVC. Second, even if a second NP position could be accommodated before the VP2, Baker would have to say that it is interpreted as an argument of the second verb. This is not correct. Finally, Baker necessarily predicts that all complements of the first verb should c-command any inner complements of the second verb; but (22b) shows this is false.

Apparently, Baker's structure is consistent only with the resultative SVC where there is in fact only a single object that is shared by the two verbs, not with the consequential SVC. Thus, his analysis has the same weakness as all others which fail to make the distinction amongst object sharing SVCs between resultative and consequential. The conclusion concerning double objects and object sharing is summarized in (28).

(28)\*[NP VI NP1 NP2 V2] if it is a resultative SVC a.

> [NP V1 NP1<sub>k</sub> NP2<sub>i</sub>] [  $pro_{i}/*_{k}$  V2] if it is consequential SVC b.

According to my analysis (28a) is ruled out because both NPI and V2 are delimiters which compete for the same unique inner complement position (see similar discussion of inherent complements in Igbo in chapter five). However, in the consequential SVC only the theme direct object can be shared but not the indirect object. Thus, resultative SVCs are similar to consequential SVC based on the fact that object sharing involves the theme direct object, although the underlying causes are somewhat different in the two cases.<sup>9</sup>

<sup>&</sup>lt;sup>9</sup> There is an interesting comparison with AP secondary predicates in English as shown in (i) and (ii)

John; gave Maryk the meat [hungry];/\*k (i)

<sup>(</sup>ii) John gave Mary; the cup<sub>k</sub> [empty]<sub>k</sub>/\*; (i) and (ii) show that in the double object construction the depictive AP secondary predicate can be predicated of the subject but not the indirect object (i). In addition, the AP resultative secondary predicate can be predicated of the theme direct object but not the indirect object (ii). The similarity in direct object sharing that excludes the indirect object could be taken as evidence for analyzing depictives as adjuncts to VP or as evidence for a possible similarity between consequential SVCs and depictive secondary predicates which I will not pursue in this thesis. I will leave the issue open for further research but see chapter six for a unified analysis of both resultative AP and VP constructions.

## 4.2.3 More on DOCs and SVCs

In section 4.2.1 above, I argued that DOCs are derived from associative constructions by moving the object of a null preposition that means "from" to the DP in AspP between VP-shells. In this section, I will illustrate one other source of double objects in which the null preposition has a benefactive meaning. I will argue that when this DOC occurs in SVCs, it obeys the same restriction described in (28): resultatives do not allow double delimiters like double objects while double objects are perfectly grammatical in consequential SVCs. (29) illustrates the benefactive goal construction:

- (29) a. Òzó guòghó émá!tón nè úyì Ozo break metal for Uyi 'Ozo broke the metal for Uyi.'
  - Òzó gu<u>òghó</u> úyì émá!tón
     Ozo break Uyi metal
     'Ozo broke Uyi's metal. '
  - Ozó kòkó èbé nè úyì
     Ozo gather book for Uyi
     'Ozo gathered the books for Uyi.'
  - d. Òzó kòkó úyì èbé
     Ozo gather Uyi book
     'Ozo gathered Uyi's books.'

(29a,c) are examples of what I call the benefactive goal construction. Descriptively, it contains a transitive verb with its object and a PP, and inside the PP is the benefactive goal argument. I assume an underlying representation for these sentences as the one proposed for similar sentences in English in Larson (1988), without short verb raising. The simplified structure is given in (30) for a sentence like (29c) (for ease of exposition, I have not included AspP between the VP-shells since it is unfilled in this case).



It is clear from the data in (29b,d) that double objects can be derived from these underlying sentences. Here, as with double objects derived from the associative construction I assume an analysis based on Travis (1991) that the derived object moves into the Specifier of AspP that occurs between the VP-shells. Furthermore, based on Baker (1997) I assume that there is P-incorporation into the verb prior to NP-movement by the benefactive goal argument. Therefore, DOCs derived from benefactive goal construction like (29d) have the representation in (31) just as the double objects derived from source/possessor. (VoiceP is omitted)

(31) TP Spéc Òzó Г Т ÌР AspP kòkój Spec Asp' úyìk VP Asp ŃP V' èbé ΡP ej ŇP Ρ ø <sup>t</sup>k

161

As it turns out, there is a sharp contrast between the resultative and consequential SVCs when these double objects (29b,d) occur. Indeed, this same contrast is also evident when the underived sentences (29a,c) occur. These are illustrated in the following sentences:

(32)	а.	*Òzó gu <u>òghó</u> èmá!t <u>ó</u> n nè úyì p <u>è</u> rh <u>é</u> Ozo break metal for Uyi be-flat
	b.	*Òzó gu <u>òghó</u> úyì èmá!t <u>ó</u> n p <u>è</u> rh <u>é</u> Ozo break Uyi metal be-flat
	c.	Òzó kòkó èbé nè úyì khi <u>é</u> n Ozo gather book for Uyi sell 'Ozo gathered the books for Uyi and he (Ozo) sold it.'
	d.	Òzó kòkó úyì èbé khién

Ozo gather úyi book sell 'Ozo gathered Uyi's books and sold them.'

(32a,b) illustrates the resultative SVC while (32c,d) are examples of the consequential SVC, and we observe that there is a grammaticality contrast both when there is a DOC and when there is underived NP PP order. For example, in (32a) the (benefactive) goal phrase occurs between the two verbs of the resultative SVC and the sentence is ungrammatical. I propose that the ungrammaticality of (32a) stems from the aspectual constraints on the resultative SVC in general which prevents double delimiters from occurring. This is based on my assumption that the (benefactive) goal phrase has already made the event denoted by the transitive verb into an accomplishment that requires no further endpoint. According to my analysis of the resultative SVC, the second verb is also a delimiter, therefore, based on the general condition that there should be only a single delimiter the sentence is ungrammatical. The two delimiters are in competition with one another, and structurally the ungrammaticality is consistent with the fact that both PP and V' need to be inner complements of the verb. We can confirm this fact based on the observation that (32a) becomes grammatical when one of the two delimiters is dropped. (29b) illustrate this fact by the dropping of the unaccusative verb and the DOC is grammatical, while (33) shows the same thing when the PP is dropped.
Òzó gu<u>òghó</u> émá!t<u>ó</u>n Ozo break metal (33) pèrhé be-flat 'Ozo smashed the metal flat'

The ungrammaticality of (32a) and the explanation just offered carries the implication that double objects would be bad in the resultative SVC, again based on the observation that the first object in the DOC is a delimiter. This implication is borne out in (32b), where having double objects derived from the benefactive goal construction is ungrammatical. The conclusion from this sentence is as in the one before, there is structurally a single inner complement position in the resultative SVC.

Recall the fact that there are no similar aspectual limitations on the consequential SVC. In this case, the second VP is not an event delimiter, so the issue of double delimitation does not arise. Accordingly, in (32c), the (benefactive) goal phrase that is the object of a PP occurs between the two verbs of the consequential SVC and the expected object sharing reading holds of the (theme) direct object of the transitive verbs (but not of the object of PP). This is consistent with the observation above that only the (theme) direct object can be shared since it is the NP that can be a null *pro* (cf. Baker and Stewart 1997b). Consequently, based on the analysis of object sharing that is mediated by an empty category, *pro*, it follows that sharing in (32c) is with the direct object rather than the object of PP.

In (32d), we observe that double objects derived from the (benefactive) goal construction in (32c) can occur between the verbs in the consequential SVC. The striking fact here is that the (theme) direct object is involved in object sharing and not the derived object. One might have thought that after P-incorporation and NP-movement by the goal phrase into the Specifier of AspP, there would be an ambiguity in terms of which object is involved in object sharing since the derived object linearly occurs in the direct object position of the verb and this is contrary to fact. Thus, double objects provide evidence both for the distinction between resultative and consequential SVC and the fact that object

sharing applies only to the underlying rather than derived object. The structural representation of (32d) is given in (34). ( I omit TP and VoiceP and the binding relation between the top and lower verb in VP, for simplicity).



While we are able to draw the conclusion that the facts from double objects clearly show the distinction between resultative and consequential SVCs, there are still two issues that arise concerning the analysis of DOCs in the consequential SVC. First, observe from (34) that the derived (goal-source) object is structurally higher than the theme object. Second, whether there is any evidence to support the structural distinction between underlying (direct object in Spec, of VP) and derived object (Indirect object in AspP)?

Concerning the first question, the analysis of double objects that I have assumed proposes that the derived object comes to be structurally higher than the theme at s-structure via NP-movement following Larson (1988). Since the thematic hierarchy (Larson 1988) and other constraints on theta role assignment such as the UTAH (Baker 1988) hold at dstructure then the surface order in which the goal-source phrase appears higher is properly

164

accounted for in this way. I turn now to the discussion of the more interesting second question.

### 4.3 On the syntax of DOCs

In this section, I provide empirical evidence to support the observation that the derived object cannot be involved in object sharing, in contrast to the underlying object. I have been assuming that this difference between the objects stems from the fact that in the double object construction the underlying object is in the Specifier of VP, while the derived object is in the Specifier of AspP. Accordingly, based on the analysis in Baker and Stewart (1997b) and as observed in Collins (1997) only the direct object can be involved in object sharing and not, for example, the object of a preposition in the consequential SVC. In this section, I will give additional supporting evidence for this assumption.

#### **4.3.1** The Asymmetry between Theme and Goal Objects.

I define the problem with having double objects in the consequential SVC by drawing attention to an implicit assumption in Baker (1989, 1997), and this is the presupposition that there are clear diagnostics for distinguishing themes and goals across languages. However, this is an area of potential problems because there are several unclear facts surrounding the structural characterization of which of the two arguments of the three-place verb is its direct object. This problem derives from the observation that a number of three-argument verbs allow alternations as to which argument shows up as the unmarked, immediately post-verbal object. This sort of alternation is amply documented in the literature on English dative (35) and locative (36) constructions:

- (35) a. Peter gave the money to the children.
  - b. Peter gave the children the money.
- (36) a. Sue loaded the hay onto the truck.
  - b. Sue loaded the truck with hay.

165

One specific problem with these corresponding (a) and (b) sentences has been how to decide on which of them should be taken as revealing the underlying (basic) structure (cf. Larson (1988), Baker (1988, 1997), Collins and Thráinsson (1993) etc.). For example, in order to account for the dative alternations in (35), Baker (1997) following Larson (1988) propose a movement solution in which sentences like (35b) are derived from underlying structures like (35a) by a combination of P-incorporation and NP movement in the syntax. However, in contrast, Baker claims that the locative alternation in (36) results from two different conceptions of the event in question: one in which the hay is seen as primarily affected, and one in which the truck is seen as primarily affected (Rappaport and Levin 1985). In both cases, the affected argument (i.e., the theme) is consistently generated as the direct object.

There are at least two very obvious connections between data of the kind in (35) and (36) and the issue of double objects in the consequential SVC. First, it is also possible to derive DOCs from both dative--and something in the semantic domain of locative constructions; which I will call source--constructions, as shown in (37) and (38) respectively:

- (37) a. úyì hàć íghó nè ìs<u>òkèn</u>
   Uyi pay money to Isoken
   'Uyi paid the money to Isoken.'
  - b. úyì hàé ìs<u>òkè</u>n íghó
     Uyi pay Isoken money
     'Uyi paid Isoken the money.'
  - úyì hàé ìs<u>ò</u>kỳn íghó dó -rhié
     Uyi pay Isoken money steal
     'Uyi paid Isoken the money and stole it.'
- (38) a. <a href="https://doi.org/10.1111/journal.com">doi:10.1111/journal.com</a>
   (38) a. <a href="https://doi.org/d
  - b. òzó vb<u>ó òkhókhò</u> ìgàn
     Ozo pluck chicken feather
     'Ozo plucked the chicken's feather.'

c. òzó vb<u>ó òkhókhò</u> ìgàn khi<u>é</u>n
 Ozo pluck chicken feather sell
 'Ozo plucked the chicken's feather and sold it (the feather).'

(37a,b) shows that we can derive double objects from a typical dative construction. In (37c) we observe that the double objects derived from the dative alternation can occur in the consequential SVC and the expected pattern of object sharing with the underlying (theme) object still holds, i.e., the theme of 'pay' is *ighó* (money) and is also the theme of dó-rhié (steal) hence, object sharing. Similarly, (38a,b) shows that we can derive double objects from the source construction. This too can then be put in the consequential SVC (38c), where the theme (underlying) object is still the shared object.

The second, and by far the most significant connection to be made between double objects and consequential SVCs is the fact that in both of these constructions (37) and (38), as in all other double objects previously discussed, there is a constraint that is based on whether an argument is the *syntactic direct object* that is assigned the *theme role* in order for it to be the shared object. I propose that it is the combination of both properties of syntactic and semantic licensing that allows object sharing in the consequential SVC as well as NP-movement in both the dative and source alternation.

### 4.3.2. Double Objects and the Empty Category Principle (ECP)

In this section, I will provide Èdó-internal evidence which shows the distinction between underlying and derived objects. I will argue for an analysis of this distinction based on the difference between moving out of the Specifier of a functional projection and moving out of the Specifier of a lexical projection.

Languages differ in the way that they treat null positions or that left behind after wh-movement (cf. Chomsky 1981, Rizzi 1986 etc.). In English and many other languages, empty positions of moved NPs are said to be occupied by syntactic traces, while null *pro* is often assumed in languages with very rich systems of agreement (cf. Georgopoulus 1985, 1991 etc.). In Èdó (and some other Kwa languages), the position left behind after a noun has been extracted by wh-movement may be either null or filled by a resumptive pronoun. This is illustrated in the following examples by focus cleft and question formation as instances of wh-movement. First, the subject NP:

(39)	a.	òzó gbé <u>ò</u> kh <u>ó</u> kh <u>ò</u> Ozo kill chicken 'Ozo killed a chicken.'	
	b.	òzó <u>ò</u> ré <u>ó</u> gbé <u>òkhó</u> !kh <u>ò</u> Ozo Foc. subj. cl. kill chicken 'Ozo killed a chicken.'	'subject cleft'
	c.	ghá <u>ò</u> ré <u>ó</u> gbé <u>òkhó</u> !kh <u>ò</u> who Foc. subj.cl. kill chicken 'Who killed a chicken?'	'subject question'
	d.	Dè òmwàn nè ó gbé òkhó!khò Q person that subj.cl. kill chicken 'Who killed a chicken?'	1 11 1

(39) shows some of the different ways in which a subject NP can be extracted in  $\underline{\dot{E}}d\dot{o}$ . The basic sentence is given in (39a) where we observe the normal SVO pattern. (39b) illustrates the extraction of the subject by focus cleft and we observe that a resumptive pronoun or subject clitic  $\underline{o}$  's/he/it' occurs in the position out of which the subject has moved. (39c,d) shows two separate ways of asking a question in the language.<sup>10</sup> However, what is relevant here is that in both cases of questioning we observe the same form of syntactic variable which is the subject clitic (resumptive pronoun). Consequently, I conclude that an overt pronoun trace always spells-out subject extraction. Now, let us consider the extraction of the direct object NP. The relevant data is given in (40) from the basic sentence in (39a):

(40) a. <u> $\dot{Q}$ kh<u>Q</u>kh<u>Q</u> <u> $\dot{Q}$ ré</u> <u> $\dot{Q}$ zó</u> <u>gbé</u> ---chicken Foc. Ozo kill 'It is a chicken that Ozo killed'</u>

<sup>&</sup>lt;sup>10</sup>The ghá form is often used when the subject is a human being, while the  $d\hat{e}$  - form can be used in general for all subjects.

Dè èmwìn nè Òzó gbé ---- Q thing that Ozo kill
 'What did Ozo kill'

(40) shows that when the complement of a verb is extracted, it leaves a phonetically null trace or *pro* behind. This is evident in (40a) with the clefting of the complement (theme object) and similarly in (40b) where the object is questioned. It appears that Èdó is like English in leaving null traces in the object position. This would tend to point toward an ECP account along the lines of Koopman (1982). However, there is room to suspect an ECP account because if Èdó is really like English, then we should expect that a null trace would also be left in the empty position of a moved oblique contrary to fact. Consider the following:

- (41) a. òzó rhié ìgàn yè ìjòkórò
   Ozo take feather on small chair
   'Ozo put the feather on the small chair.'
  - ìgàn òré òzó rhié -- yè ìjòkórò feather Foc. Ozo take on small chair It's a feather that Ozo put on the small chair.'
  - c. ìjòkórò <u>ò</u>ré òzó rhié ìgàn <u>yì</u> small chair FM Ozo take feather on 'It is on the small chair that Ozo put the feather.'

(41b) confirms that indeed the complement (theme) of the verb has a consistent behavior in the context of clefts, i.e., it leaves behind a phonetically null trace or *pro*. However, in (41c) when the object of the preposition is moved the trace is a clitic which is realized as a vowel change on the preposition from [e to i] with the same low tone maintained. We can justify this observation from a comparison between the forms of the preposition in (41a) and (41c). Since the only difference between them is whether the object of the preposition is in place or not, we can therefore attribute the presence of the vowel change on the preposition to the moved oblique NP. I conclude then that the extraction of the object of a

preposition leaves behind some kind of overt trace (clitic) which makes it different from the direct object, under the assumption that the preposition is a lexical governor..

Therefore, we find that there are clear differences between the different syntactic positions in terms of extraction: subject position is always filled by a subject resumptive pronoun, while the object position is always null, and the extraction from the object of a preposition leaves an 'overt trace'. Abstracting away from all other issues, I propose a specific account of the difference between objects and subjects in terms of the restrictions on the extraction site : whether it is from a functional or a lexical projection. Thus, following Koopman and Sportiche (1982, 1986) I propose that the extraction out of the Specifier of a functional projection which is not a lexically governed position will always result in overt traces (such as the resumptive pronoun) while extraction out of the Specifier of the VP, which is a lexically governed position can permit a null variable. This is summarized in (42)

### (42) <u>Conditions on argument extraction and traces.</u>

- a. Extraction from a functional projection (as with subject) leaves behind an overt resumptive pronoun (the subject clitic).
- b. Extraction of the object (theme) leaves behind a phonetically null trace (or pro)
- c. Objects of prepositions trigger phonological changes on the lexical governor

Against the background of the descriptive generalization in (42) we are now in a position to verify two aspects of the analysis of DOCs assumed thus far. First, we can examine the issue of structural differences between underlying versus derived objects with respect to DOCs. Second, it has been noted across languages that dative and locative (source) constructions may sometimes undergo an alternation that yields a surface DOC (cf. Baker 1997), and there are wide-ranging views on whether the unshifted (non-double object) one is the basic or vice versa. I believe that this is one area where the generalization in (42) can be very useful in constructing a theory of dative/locative alternations. The

prediction is that the theme-direct object will behave differently from the goal-oblique object with respect to the nature of traces that they leave behind in wh-extraction. I will now address these issues in turn.

### 4.3.3 Underlying vs. Derived Objects

In this section, I provide evidence for the analysis in which the derived object is in Specifier of AspP and the underlying object is in Specifier of VP. This is based on the differences in the traces left behind by the double objects when they are wh-extracted, e.g., by focus cleft. Consider the following:

(43) a.	Òzó vb <u>ó òkhó</u> !khó ìgàn (khién) Ozo pluck chicken feather sell 'Ozo plucked the chicken's feather (and sold it (the feather)).'
b.	<u>àkhó!khó àré Òzó vbó é</u> rè ìgàn (khi <u>é</u> n) chicken Foc. Ozo pluck obj.cl. feather sell 'It is a chicken that Ozo plucked its feather (and he sold it (the feather)).'
c.	ìgàn <u>d</u> ré Òzó vb <u>ó dkhó!khó</u> (khi <u>é</u> n) feather Foc. Ozo pluck chicken sell 'It is feather that Ozo plucked from the chicken (and sold it (the feather)).'
The basic	sentence is (43a), and in (43b) we observe that when the first objectwhich l
have analy	zed as the derived objectis extracted, an object resumptive pronoun surfaces as

the spell-out of the trace.<sup>11</sup> In (43c) the underlying object is extracted for focus cleft and there is a gap in the variable position, like with the extraction of the object of monotransitive verbs. This asymmetry between the derived object and the underlying object follows from (42) and consistent with my analysis: the former is in the Specifier of a functional projection and so not lexically governed, while the latter is in the Specifier of a VP. This difference is also seen in the case of double objects from dative alternation (44) and associative construction (45) when put in consequential SVC structure:

(44) a. Òzó hàć ìsòkèn íghó (dó -- rhić)
 Ozo pay Isoken money steal
 'Ozo paid Isoken some money (and stole it (again))'

<sup>&</sup>lt;sup>11</sup> I should point out that the interpretation of this sentence is incompatible with a reading in which sharing is with the 'chicken'.

- b. ìs<u>òkèn òré</u> Òzó hàé <u>é</u>rè íghó (dó -- rhié)
   Isoken Foc. Ozo pay obj.cl. money steal
   'It is Isoken that Ozo paid some money (and stole it (again))'
- c. íghó <u>ò</u>ré Ózó hàé ìs<u>ò</u>k<u>è</u>n --- (dó -- rhié) money Foc. Ozo pay Isoken steal 'It is money that Ozo paid to Isoken (and stole it (again))'

The behavior of the objects follows very nicely from the analysis that I have proposed such that when the underlying object is extracted a null *pro* is left behind which is co-indexed with the null *pro* of the second verb if there is one as in (44), and (45) below:

- (45) a. Òzó rhié úyì èbé (khién) 'Ozo took Uyi's book (and sold it)'
   Ozo take Uyi book sell
  - b. úyì <u>ò</u>ré Òzó rhié <u>é</u>rè èbé (khi<u>é</u>n) pro j
    Uyi Foc. Ozo take obj.cl. book sell
    'It's Uyi that Ozo took his book (and sold it)'
  - c. èbé òré Òzó rhié úyì --- (khién)
     book Foc. Ozo take Uyi sell
     'It is book that Ozo took from Uyi (and sold it)'

On the basis of these sentences in (43)-(45) we conclude that (46) is indeed the structure of the consequential SVCs with double objects. (I omit projections above EP for simplicity.)

(46)



According to (46) the extraction from a lexically governed position as with the direct object leaves behind a null *pro*, and so *pro* is properly licensed (cf. Baker and Stewart 1997b). However, when a derived object is extracted from a functional projection, as with subjects (cf. Koopman and Sportiche 1982, 1986, etc.), a resumptive pronoun is left behind, since such position is not lexically governed.

This conclusion is particularly instructive given the fact that in English, it has been suggested that it is rather bad to extract the first object of a double object construction as illustrated by the following sentences(cf. Stowell 1981, Baker 1988, among many others).

(47) a. Which man do you think I should ?give/\*buy t the shirt

b. Which shirt do you think I should give t/buy t to/for Mary

According to my analysis the difference between Edó and English lies in the possibility of having a resumptive pronoun to occur in the position of the derived object which is in the Spec of AspP. This option is available in Edó but not in English. Thus, while it is possible to extract the first (derived) object in Edó the same is not so good in English. This is a rather striking fact in the light of the fact that extracting the NP out of an NP-PP structure in English is grammatical (47b). In fact, Baker (1997) argues that the same restriction in NP-NP applies to the distinction between locative and dative constructions. For example, it is perfectly grammatical to extract the direct object from either version of a locative alternation as in (48) but the goal object of a double object construction exhibits unique syntactic behavior. This is also seen in the inability of the goal object to undergo heavy NP shift:

(48) a. Which boxes do you think I should load t onto the truck?

b. Which truck do you think I should load t with hay?

(49) \*I gave t candy every child that came to the door

In Édó, there is no distinction amongst the various kinds of double objects which is consistent with their having the same structure in a consequential SVC like (46). I

conclude, therefore, that the Èdó data illustrates the asymmetry between the objects in the double object construction based on the clear diagnostic showing the difference in extraction between Specifiers of functional projections and lexical projections, i.e., Spec of AspP versus Spec of VP.

# 4.4 Conclusion

This chapter confirms the proposal that there are clear contrasts between resultative and consequential SVCs, and this goes against a unified analysis as in Baker (1989). This is based on the facts of double objects which show that there is competition for the single inner complement delimiter position in the resultative SVC in contrast with the consequential SVC where there are no such structural limitations. Furthermore, I argued that only the theme (direct object) can be shared in both constructions, although the cause is different in each case. Finally, I provided Èdó-internal evidence which sheds some light on the distinction between underlying and derived objects, consistent with the analysis I have proposed (with consequences for goals and themes in dative-locative alternations, DOCs, and heavy NP shift in English), that is, underlying objects occupy Specifier of VP, while derived objects occur in the Specifier of AspP.

#### Chapter five

#### **Cross-linguistic Extensions of Analysis**

#### 5.1 Introduction

This section is intended to illustrate the cross-linguistic relevance of my analysis of SVCs. The empirical materials to be presented will consistently confirm the distinction between two kinds of object sharing SVCs--resultative and consequential, and the more general CC construction that involves only subject sharing.

Overall, my analysis make some specific predictions about SVC types that are not only observable in other languages but solve an apparent puzzle in Kwa languages: the controversy surrounding Igbo as an SVC language on a par with its neighbors (geographically and genetically) Èdó and Yoruba. In this regard, I will present illustrative data from Yoruba, Igbo, and Chinese which confirm the distinctions I have proposed. Primarily, I will focus on the distinction between resultative and consequential SVCs, but I will also present some suggestions about so-called multi-event SVCs as a cross-linguistic parallel of CCs.<sup>1</sup> Furthermore, I will show that there is a close parallel of these distinctions based on the event(s) that the verbs express: the two verbs in the resultative SVC express a single event with a single structural E node, while those in the consequential SVC express two events realized via a binding relation between two structurally asymmetric E nodes. Finally, the verbs in CCs express two separate (unbounded) events and the two E nodes are structurally symmetrical.

### 5.2 On Structurally Ambiguous 'SVCs'

As a preliminary step to extensions proper, let me illustrate the distinction between resultative SVC and CC by showing that my analysis can clarify unclear cases of

<sup>&</sup>lt;sup>1</sup> In chapter seven, I will argue that some cases of what others have classified as two-event SVCs actually involve Control, i.e., clausal complementation.

ambiguous SVCs. This is based on the observation that some surface SVC sentences are structurally ambiguous (Baker 1989, Stewart 1996, Collins 1997, etc.). Some examples are given with data from Èdó, Yoruba, and Ewe, in (1):

(1)	а.	<ul> <li>Òzó gbé èkhù làá òwá</li> <li>Ozo hit door enter house</li> <li>a. 'Ozo hit the door into the house' (onl)</li> <li>b. 'Ozo hit the door and entered the house</li> </ul>	y door goes into the house).' se' (Ozo goes into the house).'
	b.	Olú lu màálù kú Olu beat cow die a. 'Olu beat the cow dead' (the cow die b. 'Olu beat the cow and died' (Olu died	Yoruba (Baker (1989)) d).' d).'
	c.	F <u>é</u> mi tì Akin subú Femi push Akin fall a. Femi pushed Akin down (Akin fell) b. Femi pushed Akin and fell (Femi fell)	Yoruba (M. Olusegun (p.c.) see Lord (1974))
	d.	Ekpe fo kopo yi xo-me <sup>2</sup> rock hit cup go room-in a. 'A rock hit a cup into the room' (cup g b. 'A rock hit a cup and then went into the	Ewe (Collins 1997) goes into the room) he room' (rock goes into room)

These sentences in (1) have two meanings; object sharing (the (a) sentences) and subject sharing (the (b) sentences). On the basis of meaning, it seems that the sense in which the theme (object) is shared expresses a result, while the sense in which the subject is shared is like a CC. Based on the Èdó sentence (1a), I will now show that the tests that I have proposed provide clear confirmation for this interpretation of the ambiguity. For the sake of keeping the discussion simple, I will mainly focus on the position between the verbs since this is where the crucial differences arise.

# A. I-type adverb

(intended reading: door into house)
 (2) a. \*Òzó gbé èkhù gié!gié lá!á òwá (resultative)
 Ozo hit door quickly enter house

 $<sup>^2</sup>$  The underlined vowels are not intended to translate laxness (as in Èdó) but are merely rough transcriptions of the Ewe sound that is spelled with a backwards c.

(intended reading: Ozo into house) b. Òzó (giś!giś) gbć èkhù giś!giś lá!á òwá (CC) Ozo (quickly) hit door quickly enter house 'Ozo (quickly) hit the door and he quickly entered the house'

The I-type adverb cannot occur between the verbs in the resultative SVC (2a), and this is consistent with the proposal that there is no projection of EP present there in the structure. Thus, we confirm that there is a single E position which quantifies over the single event. However, (2b) shows that the I-type adverb can occur between the verbs and also (optionally) occur with the same I-type adverb before the first verb. This is consistent with the idea that the subject sharing interpretation is in fact a CC, with the verbs denoting two separate events.

# B. N-type adverb

(3)	а.	*Òzó gbé <u>è</u> khù <u>èg</u> ì <u>ég</u> ì <u>é</u> làá òwá Ozo hit door quickly enter house	(resultative)
	b.	Òzó gbé <u>è</u> khù làá òwá <u>ègiégié</u> Ozo hit door enter house quickly 'Ozo hit the door into the house quickly'	(resultative)
	c.	Òzó gbé <u>èkhù ègìégìé</u> làá òwá Ozo hit door quickly enter house 'Ozo hit the door quickly and he entered the house'	(CC)
	d.	Òzó gbé <u>èkhù</u> làá òwá <u>ègiégié</u> Ozo hit door enter house quickly 'Ozo hit the door and he entered the house quickly'	(CC)

On the object-sharing reading, an N-type adverb cannot occur after the first verb plus object because they do not constitute a VP (3a); however the adverb can occur after the second verb as a predicate of the single event denoted by both verbs, where both verbs are coheads of the same VP (3b). On the subject-sharing reading, it acts like a CC and the N-type adverb can occur after each verb plus its object in a manner that is consistent with each verb denoting a separate event expressed by separate VPs. These conclusions are, predictably, confirmed by the distribution of locative PPs, as shown in (4).

### C. locative PPs

(4)	a.	*Òzó gbé <u>è</u> khù vbè Èdó làá òwá Ozo hit door in Benin enter house	(resultative)
	b.	Òzó gbé èkhù làá òwá vbè Èdó Ozo hit door enter house in Benin 'Ozo hit the door into the house in Benin City'	(resultative)
	c.	Òzó gbé <u>è</u> khù vbè Èdó làá òwá Ozo hit door in Benin enter house 'Ozo hit the door in Benin and he entered the ho	(CC) use (location not implied) <sup>3</sup>
	d.	Òzó gbé <u>è</u> khù làá òwá vbè Èdó Ozo hit door enter house in Benin	(CC)
		'Ozo hit the door (someplace) and he entered the	ne house in Benin'

On the object-sharing reading the locative PP cannot occur between the verbs (4a), but only outside the VP (4b) where it implies the location for the single event expressed by the two verbs. However, on the subject-sharing reading the locative PP can either occur between the verbs (4c) where it expresses the location of the first event, or after the second verb (4d) where it could either attach to VP2 and express the location of the second event or adjoin higher to EP and express the location of the two events.

# D. subject NPs and tobore anaphor

(5) a. \*Òzók gbé èkhù tòbórèk làá òwá (resultative) Ozo hit door himself enter house
b. Òzók gbé èkhù tòbórèk làá òwá (CC) Ozo hit door himself enter house 'Ozo hit door and he [himself] entered the house'

Again, there is a clear contrast between the object-sharing (resultative) reading and the subject-sharing (CC) reading. The failure of the particle to occur before the second verb and be coreferent with the overt subject implies that there is a single subject in the resultative that is also the Agent of the event. However, observe that the particle can occur

<sup>&</sup>lt;sup>3</sup> However, the normal implication in these sentences is that they take place at the same gross location.

in this position and be coreferent with the overt subject in the CC reading. This is consistent with the proposed ATB analysis whereby there is the trace of a subject in the VoiceP projection dominating the second verb to which the particle right-adjoins.

#### E. predicate clefts

(6)	а.	*ùgbémw <u>èn ò</u> ré Òzó gbé <u>è</u> khù lá!á òwá nom-hit-nom Foc. Ozo hit door enter house	(resultative)
	b.	*ùláámw <u>èn ò</u> ré Òzó gbé <u>è</u> khù lá!á òwá nom-enter-nom Foc. Ozo hit door enter house	(resultative)
	c.	ùgbémw <u>è</u> n <u>ò</u> ré Òzó gbé <u>è</u> khù làá òwá nom-hit-nom Foc. Ozo hit door enter house 'It is hitting that Ozo hit the door and then entered the house'	(CC)
	d.	ùláámw <u>è</u> n <u>ò</u> ré Òzó gbé <u>è</u> khù lá!á òwá nom-hit-nom Foc. Ozo hit door enter house 'It is entering that Ozo hit the door, and did, into the house'	(CC)

The contrasts in (6) confirms the proposal that what prevents predicate clefts in the resultative SVC is not based on the individual verbs but rather on the nature of the event. Therefore, with the same verbs in (6) predicate cleft of either one under an object-sharing (resultative) reading is ungrammatical (6a,b) whereas under the subject-sharing CC reading such cleftings are grammatical. This contrast brings out the difference between single event SVCs and two event SVCs and the correlating distinction between single EP and two EPs where Spec-head matching takes place in licensing predicate clefts.

As a conclusion, we observe that the tests that I have proposed clearly and consistently distinguish object-sharing (resultative) SVCs from subject-sharing CCs. I turn now to the big picture regarding what obtains in other languages.

# 5.3 On "Splitting Verbs" as Resultative SVCs

In the discussion of resultative SVCs thus far, we have seen examples of transparent verbs where the first verb expresses a process or activity and the second verb expresses a state, and I have simply been assuming their idiosyncratic properties. In this section, I will illustrate some of the idiosyncrasies in the resultative SVC based on what is classified as 'splitting verbs' in Yoruba (cf. Awobuluyi 1969) and Èdó. Descriptively, the term 'splitting verbs' denotes one lexical item that splits into two parts and the object appears in the middle. Thus, on the surface they have the linear order [NP V NP V] which resembles object sharing SVCs in general. I will argue that, in fact, splitting verbs can only have a resultative SVC structure and not consequential SVC type. The relevant examples are given below:

(7)	а.	O bá kekee mi je s/he spoil bicycle me spoil 'S/he spoiled my bicycle.'	Yoruba
	b.	Òzó mín émi!ówó ré Ozo ? meat eat 'Ozo swallowed the meat.'	Èdó
	c.	Òzó bí <u>è</u> khù gbé Ozo push door hit 'Ozo shut the door.'	Èdó

Quite generally, these sentences in (7) are classified as SVCs that are made up of "splitting verbs" in Yoruba (cf. Oyelaran 1982, George 1975, 1976), and Èdó (Agheyisi 1986).<sup>4</sup> In the Èdó examples, neither 1-type nor N-type adverbs, locative PPs, or double objects can occur between these splitting verbs. In addition, neither of the verbal elements can undergo predicate clefts. These facts are illustrated in (8):

(8)	a.	*Òzó bí <u>è</u> khù gi <u>é</u> !gi <u>é</u> gbé Ozo push door quickly hit 'Ozo pushed the shut quickly'	I-type adverbs
	b.	*Òzó bí <u>è</u> khù <u>èg</u> ìégìé gbé Ozo push door quickly hit	N-type adverbs
	c.	*Òzó bí <u>è</u> khù vbè òwá gbé Ozo push door at house hit 'Ozo pushed the door shut at home'	locative PP

<sup>&</sup>lt;sup>4</sup> Agheyisi (1986) proposes that these kinds of sentences show the residues of grammaticization in Èdó, this is comparable to my analysis which treats splitting verbs as idiomatic SVCs.

d.	*Òzó bí úyì <u>è</u> khù gbé Ozo push Uyi door hit 'Ozo pushed Uyi's door shut'	double objects
e.	*ùbímwèn òré Òzó bí èkhù gbé nom-push-nom Foc. Ozo push door hit	pred. cleft of V1
e.	*ùgbémw <u>èn ò</u> ré Òzó bí <u>è</u> khù gbé nom-hit-nom Foc. Ozo push door hit	pred. cleft of V2

On the basis of the ungrammaticality of these sentences, I conclude that splitting verb constructions are a type of resultative SVC, and have the same syntactic structure. The relevance of this analysis of splitting verbs as resultative SVCs comes out very clearly when they are compared with those resultative SVCs with lexically independent verbs. This time, I will illustrate with examples from Yoruba . (9) is one such example of Yoruba resultative SVC.

(9)	Okuta gba ogiri fo	(Gruber & Collins 1996=48)
	stone hit wall break	
	'The stone smashed the wall'	

The comparison between resultative splitting verbs and resultative SVCs can be illustrated by the similarity of their behavior in predicate clefts. This is shown in (10) for (7a) and (11) for (9).

(10)	a.	*bi-bá	ni	0	bá	kekee	mi	je
		nom-spoil	Foc	s/he	spoil	bicycle	me	spoil

- b. \*ji-je ni O bá kekee mi je nom-spoil Foc s/he spoil bicycle me spoil
- c. bi-bá-je ni O bá kekee mi je nom-spoil Foc s/he spoil bicycle me spoil 'It was spoiling that S/he spoiled my bicycle'
- (11) a. ?/\*Gbi-gba ni okuta gba ogiri fo nom-hit Foc stone hit wall break
  - b. \*fi-fo ni okuta gba ogiri fo nom-break Foc stone hit wall break
  - c. Gbi-gba-fo ni okuta gba ogiri fo nom-hit-break Foc stone hit wall break 'It was hitting and breaking that stone did to the wall'

The data above is intended to show that predicate cleft is not possible with any of the idiosyncratic SVCs, except when they are clefted together as a unit just like other productive resultative SVCs. Therefore, such similarity from the behavior in predicate clefts can be taken to imply that in both resultative splitting verbs and SVCs the verbs combine to express a single event. Consequently, I conclude that splitting verbs (though idiosyncratic) will have the same structural representation as resultative SVC. This is illustrated abstractly in (12), for the sentence in (7a). (VoiceP is omitted)



The consequence of this analysis of idiosyncratic splitting verbs in Yoruba and Edó is that it naturally extends to resultative V-V compounds in Igbo.

#### 5.4 Igbo

Based on certain facts of verbal inflection, it has often been claimed that Igbo (which shares the same boundary with Èdó) lacks SVCs altogether (cf. Lord 1975), however Déchaine (1992, 1993), Ihionu (1992) Manfredi (1991) etc. use the same property of verbal inflection to argue that Igbo does indeed have two kinds of SVCs: single event (instrumental) and multi-event SVCs.

In this section, I will show that Igbo so-called SVCs receive an enlightening interpretation based on my proposed distinction between resultative SVCs, consequential SVCs, and CCs. Specifically, I will argue that Edó resultative SVCs correspond predictably to Igbo resultative V-V compounds, that Igbo systematically lacks consequential SVCs, and that the so-called multi-event SVCs in Igbo may either be cases of CCs or involve clausal complementation.<sup>5</sup>

### 5.3.1 Resultative V-V Compounds.

An interesting consequence of my analysis of SVCs comes from the observation that resultative V-V compounds in Igbo are exceptionlessly made up of transitive plus unaccusative verbs (Ihionu 1992:174).<sup>6</sup> This implies that the same unaccusative second verb restriction holds in both resultative V-V compounds and resultative SVCs. This similarity is lost under a unified analysis that does not distinguish between resultative and consequential SVCs (Baker 1989, Collins 1997, Manfredi 1991, etc.). Therefore, I propose that the class of resultative SVCs, resultative splitting verbs, and resultative V-V compounds are all reflections of the same structure in different languages (see section 5.5.1 below for extensions of this generalization to Chinese resultative V-V compounds). Some of the Igbo action-result (resultative) V-V compound sentences are given in (13):

Obi kwá-da-ra Ézè (13) a Obi push-fall-rV Eze 'Obi pushed Eze down.' (Ihionu ,1992=18)

<sup>&</sup>lt;sup>5</sup> See Chapter seven for evidence showing a clausal complementation analysis of instrumental constructions in Edó which is consistent with the analysis of Igbo so-called multi-event constructions in section 5.4.3. <sup>6</sup> The only known exception is the set of idiosyncratic V-V compounds of the same class as ri-gbú, lit. 'eatkill' (Lord 1975, Émenanjo 1978, Nwachukwu 1987). However based on the analysis of splitting verbs in Edó, in particular those in which the second part is gbé (hit) as in (7c) I propose that the Igbo verb 'gbú' which is semantically cognate to Edó gbé 'kill' is ambiguous between a purely transitive meaning and a transitive-unaccusative meaning. This is based on the ambiguity of the verb gbé kill' in Èdó between the idiosyncratic use in (7c) and a transitive use (i) (i)

Òzó gbé ékítà

Ozo hit dog

<sup>&#</sup>x27;Ozo killed the dog'

Therefore, I equate all Igbo resultative V-V compounds with resultative SVCs in Ed6 (transparent and idiosyncratic).

- b. Ó rí-ju-ru afo (Déchaine, 1993 =125) 3s eat-be.full-rV stomach 'S/he ate [her/his] stomach full.'
- c. Ó kú-gbu-ru Ezè
   3s beat-kill-rV
   'S/he beat Ézè mercilessly/ to death.'
- d. Àdhá tì-gbu-ru Ézè (Manfredi, 1991=31a) Adha beat-kill-rV Eze 'Àdhá beat Ézè to death.'
- e. Ádhá gbà-ju-ru mótò pour-fill-rV car (with fuel) 'Àdhá fueled the car (with petrol).'

It is striking to note that unless these verbs in (13) are compounded, for example, the construction (13e) could only refer to consecutive events (14a) or become outright ungrammatical (14b).

- (14) a. Àdhá tì-ri Ézè íhe (wè-é) gbu-o yá (Manfredi=33a) beat-Asp thing take-Asp kill-Asp 3sg-Gen
   'Àdhá beat Ézè and (then) killed him (as a separate action)'
  - b. \*Àdhá gbà-ra fúèl ju-o mótò pour-Asp petrol fill-Asp car

I consider these alternatives in (14) as evidence of the fact that the class of consequential SVC is systematically absent in Igbo. Thus, the option seems to be between V-V compounds which express a resultative single event (13e) or two separate verbs involved in some form of covert conjunction (CC) (14a).

I propose that Igbo resultative V-V compounds originate underlyingly from the same structure as that of the resultative SVC. Essentially the same proposal is in Déchaine (1993) based on Manfredi (1991) who argue that Igbo V-V compounds in general have the same D-structures as their Yoruba (or Èdó) serial counterparts, but with the first verb serving as the head of the complex verb projection. Thus, abstracting away from my analysis the structure for (13a) is (15). (VoiceP is omitted)

.

18



The question that arises from this analysis is how can we account for the surface V-V compounds? Before I provide an answer to this question, I will introduce a competing analysis of the resultative V-V compounds in Igbo which will serve as useful background.

Déchaine (1993), based on Manfredi (1991) proposes a bivalent verb projection analysis in which V-V compounds are derived from covert serial constructions by Vmovement, as shown in (16).



According to the structure in (16), the first verb heads the bivalent projection and the second verb moves out of its projection and incorporates into the first verb. While this analysis seems plausible, it does have certain wrinkles which I will argue are weaknesses. This is based on how it handles the facts of inherent complements.

#### **5.4.2 Inherent Verb Complement**

The facts from Inherent Verb Complement (IVC) as discussed in Manfredi (1991), Ihionu (1992), Nwachukwu (1987) etc. provide empirical evidence in support of my proposed account of the resultative V-V compound. The basic fact to be explained is this: the interaction of V-V compounding with 'inherent complements' in determining the surface order of arguments (cf. Lord 1974, 1975, Déchaine 1993). This is illustrated in the following sentences:

(17) a. Ó tì-rì nwóké áhù \*(okpó) (Déchaine, =139) 3s hit-rV man that blow 'S/he hit that man.'
b. Ó gbà-rà ényì yá \*(egbè) "
3s shoot-rV friend 3s.Gen gun 'S/he shot at his/her friend.'

What (17) shows is that, in isolation, verbs such as 'hit' and 'shoot' take a fixed and obligatory DP complement since it is ungrammatical to omit the nouns okpó 'blow' in (17a) or egbé 'gun' in (17b). However, these inherent complements are not possible with V-V resultative compounds (Lord 1975:33). This is illustrated in (18)

- (18) a. Ó tì-gbu-ru Ézè (\*okpó)
   (Déchaine, = 140)
   3s hit-kill-rV blow
   'S/he struck Eze dead.'
  - b. Ó gbà-gbu-ru Ézè (\*égbè)
     3s shoot-kill-rV gun
     'S/he shot Eze dead.'

In order to account for the ungrammaticality of the sentences in (18) Manfredi and Déchaine are forced to say something special about inherent complements by proposing a complicated aspectual explanation (cf. Déchaine 1993: 243). I will now show that these facts can be handled insightfully under my analysis and structure.

11

Déchaine (1993) observes that the ungrammaticality of (18) should not be attributed to a failure of Case assignment, given the well-formedness of (17). Therefore, she proposes (following Manfredi (1991)) that the problem is aspectual in the sense that after the second verb has moved and incorporated into the first verb as shown in (16), the resulting complex verb can only have one affected argument. Thus, in (18) only the affectum of the second verb (gbú 'kill') surfaces while the affectum of the first verb (tí 'hit', gbá 'shoot') is suppressed because it is recoverable as a lexical constant.

Under my analysis, the ungrammaticality of the IVC in (18) follow from the same general constraint which rules out double complements in resultative constructions. Essentially, this is a reflex of the proposal that there can be only one delimiter for the single event expressed by the two verbs derived from the aspectual properties of the resultative SVC (section 2.8). Therefore, Manfredi/Déchaine's idea that the ungrammaticality of (18) has something to do with aspect receives a consistent analysis from my structure and this fact is cross-linguistically verifiable with resultative constructions.<sup>7</sup> Thus, the reason why inherent complements cannot occur in the structure in the resultative V-V compound is because there is only one object position and as such if there is a second object, it would compete for the inner complement delimiter position which is already filled by the second verb of the resultative V-V compound in underlying structure. Consequently, I propose that IVCs never surface in certain kinds of V-V compounds namely, the resultative kind.

Returning now to the question of how to account for the surface V-V compounds in Igbo that parallels the resultative SVCs in Èdó, I propose an account that is similar in some respects to that in Manfredi (1991) and Déchaine (1993) in that the locus of variation between languages is ultimately traceable to differences in the nature of V-I movement. In anticipation of the discussion in chapter six concerning the serial verb parameter, I propose that verb raising to INFL is not possible in SVC languages like Yoruba or Èdó but obligatory in Igbo. In Igbo, both verbs are in essence co-heads of the same VP realized by V-incorporation of the second verb into the first (cf. Manfredi 1991, Déchaine 1993), and the incorporated form obligatorily raise to check strong tense features in INFL As I will

<sup>&</sup>lt;sup>7</sup> Peter Ihionu (p.c.) observes that IVCs are delimiters and so cannot occur in a resultative construction like the V-V compound which allows only one delimiter.

show in Chapter six this intermediate step involving verb incorporation allows us to provide a clear-cut basis for the distinction between Èdó and Igbo: no verb raising implies an SVC structure, while obligatory verb raising to INFL implies a V-V compound structure.

This analysis of Igbo resultative V-V compounds that is based on obligatory V-I movement can be illustrated by the following grammaticality contrast.

- (19) a. \*Obi kwá-da Ézè Obi push-fall Eze
  - b. Obi kwá-da-ra Ézè
     Obi push-fall-RLS Eze
     'Obi pushed Eze down'

In (19a) we observe that it is ungrammatical to have a resultative V-V compound in which the verbs fail to raise to INFL (or some projection of tense). We know this based on the fact that the verbs in (19a) do not bear the -rV tense inflection. This is in sharp contrast with (19b) in which the V-V compound inflects for perfective aspect that is realized by the -rV suffix. Consequently, I take this grammaticality contrast as indicative of the fact that V-V compounding is dependent on verb movement in Igbo. Therefore, the s-structure representation for (19b) is as shown in (20). (VoiceP is omitted)





The differences between my structure and that of Déchaine given above are as follows: in my analysis V-incorporation takes place between items that are daughters of the same projection and so I assume that the head-movement constraint (HMC) of Travis (1984, 1990) would be less stringent in allowing the [V1+V2] incorporation as opposed to a structure where incorporation applies across two separate projections like in Déchaine's. Also, the weight of empirical evidence from Èdó and Yoruba support only a co-headed structure rather than a bivalent VP projection. Finally, under my analysis, it does not matter whether the perfective suffix is in a tense -related functional projection like EP or if it is INFL (cf. Déchaine 1992, 1993, Manfredi 1991, Welmers 1973, Winston 1973, Emenanjo 1975, Nwachukwu 1976, Ezikeojiaku 1979 etc.). All that is crucial is that the head of the functional projection has strong (tense) features and this triggers overt V-I movement. Of additional relevance here is the proposal that it is not possible to raise only the first verb in Igbo resultatives because it forms a co-headed VP structure with the second verb. The combination of these two properties is sufficient to license verb raising by both verbs in Igbo, creating V-V compounds (see Chapter six).

As a conclusion, I have shown that one consistent analysis of action-result V-V compounds in Igbo can be derived by a comparison with resultative SVCs in Èdó and Yoruba. This analysis is also shown to be consistent with that proposed for splitting verbs which are analyzed as idiosyncratic SVCs in Èdó-Yoruba. The primary parameter distinguishing both cases is that both verbs must raise to some projection of INFL in Igbo, thus creating the surface V-V compound and it is predicted that overt verb raising by both verbs to INFL is not allowed in Èdó SVCs (see Chapter six).

# 5.4.3 Verb Inflection and Multi-event Constructions

The purpose of this section is to show that so-called multi-event SVC involve some kind of clause-chaining (cf. Hale 1991) whereby verbs denoting consecutive (sequential) events are syntactically linked together. Therefore, I will argue that so-called multi-events

in Igbo may involve clausal complementation, IP-adjunction or CCs but not true serialization. This is based on a particular analysis of the open vowel suffix (OVS). I also propose that it is these possibilities of structure that underlie the term 'consecutivization' that is normally associated with the sentences (Welmers 1973, Lord 1975 etc.). However, my goal is not to provide an exact syntactic analysis of these constructions but simply to point out the empirical facts surrounding the proposal that these so-called multi-event SVCs form a complex of different constructions.

The relevant core of the data on Igbo multi-event constructions is given in (21) from Déchaine (1993:238-240):

- (21) a. Ó jí ´mmà bhá-a jí
   3s hold knife peel-A yam.Gen
   [S/he peeled yam[s] with [a] knife.'
  - b. Ó wè-re úkwu gà-á áhyá 3s take-rV leg go-A market.Gen 'S/he went to [the] market on foot.'
  - c. Ó jì-ri ohuhu rí-e ihé 3s use-rV hurry eat-A thing.Gen 'S/he hurriedly ate [something].'
  - Ó kwù-ru ókwu khwa-a akhwá
     3s speak-rV word cry-Á tears.Gen
     'S/he spoke and cried.'
  - e. Ó wè-re ite byá 3s take-rV pot come-Á 'S/he came with [a] pot.'
  - f. Ógu go-ro okúkò gbú-o si-e ri-e buy-rV chicken kill-Á cook-Á eat-Á
    'Ogu bought [a] chicken, killed [it], cooked [it] and ate [it].'

There are three empirical observations concerning these examples that point to the fact that there are underlyingly different structures.

The first observation concerns a very obvious fact that is right at the core of the analysis of Igbo non-resultative "SVCs". This is the issue of the analysis of verbal inflection, in particular how to interpret the harmonizing high tone vowel suffix, traditionally called the Open Vowel Suffix (OVS) and glossed as -Á in (21) following the convention in Déchaine (1993). The OVS on the second verb has been interpreted as a quasi-conjunctive 'consecutive' marker (Hyman 1971, Welmers 1973, Lord 1975 etc.), while Déchaine (1993) proposes that the second verb obligatorily bears OVS because it marks the event as completed (some kind of Aspect category).

However, what is taken for granted in (21) is the fact that the second verb always bears the OVS inflection while the first verb obligatorily bears the -rV inflection. Reversals of these morphemes with OVS on the first verb and -rV on the second verb are unattested in the literature and are in fact ungrammatical (22).

Furthermore, observe that in the multiple event stacking example in (21f) all the subsequent verbs have the OVS inflection and only the first verb bears the -rV inflection. Thus, even if this can be taken as a consequential SVC with *pro* -drop, one would have to explain the function of the verbal inflections. I propose that the ordering relation between the -rV and OVS inflections and the fact that in some cases the verbs bear different verbal inflection are indications of structural relations between two separate clauses. The inflectional morphemes determine the ordering of functional projections in these multi-event constructions, and so I conclude that (21f) is not a consequential SVC but like the other multi-event constructions is either CC or clausal complementation.

One piece of evidence that can help in deciding the structure of these kinds of sentences comes from the contrast between (21) and real covert coordinations with respect to the distribution of the inflectional affixes: -rV and OVS. According to Déchaine (1993: 242), in verb-doubling examples like (23a) both verbs bear the -rV inflection and such sentences are generally analyzed as clausal coordination. A clausal coordination analysis for sentences such as (23a) is buttressed by the fact that it is ungrammatical to extract the object

of the second conjunct (23b), because of the coordinate structure constraint (CSC) of Ross (1967).

- (23) a. Àdhá shì-ri jí, shì-ri édè
   boil-rV yam boil-rV cocoyam
   'Àdhá cooked both yams and cocoyams.'
  - b. \*édè Àdhá shì-ri jí, shì-ri (yá)
     cocoyam-Rel. boil-rV yam boil-rV 3s
     'The cocoyam which Àdhá cooked both yams and (it).'

In spite of the previous observation that CSC effects appear to be variable in sentences that seem to involve coordination, yet based on sentences like (23a) we observe that clausal coordination in Igbo will be realized by verbs each marked separately by its own -rV inflection. Since this is not the case with the examples in (21), it seems reasonable to propose that they involve syntactic subordination (possibly IP adjunction), or they are CCs involving the coordination of something smaller than IP (see Collins 1997 for similar proposal that covert coordination involves I-bar conjunction).

The second observation about the sentences in (21) is that there is a functional projection dominating the second verb that is the source of the OVS (see also (14a)). It would seem, therefore, that the overt realization of event sequencing (clause-chaining) is the presence of strong tense and aspect features in the head of a functional projection. Thus, multi-events are realized by overt functional projections in Igbo which is morphologically spelled-out by the OVS. In fact, the OVS seems to have approximately the same tense-aspect properties like the Èdó INFL headed by ya (see chapter seven).<sup>8</sup>

192

<sup>&</sup>lt;sup>8</sup> In brief, the points of comparison between Èdó INFL headed by yá and the Igbo OVS are the following. First, observe that the second verb in (21) has some special aspect-tense interpretation that has been argued to be derived from the OVS inflection (cf. Déchaine 1993), i.e., the OVS aspectually marks completed events. Second, there is a rather intricate connection between the OVS and the presence of genitive tones in Igbo. Both of these properties are similar to the connection between the presence of yá and infinitive tones in Èdó. In both Èdó and Igbo, there are special tone effects which is the same high-downstep-high that is realized on the object of the second verb in Igbo (the genitive tone, Manfredi 1991, Déchaine 1992, 1993) but on the second verb itself in Èdó (see chapter seven). The likely conclusion based on the Èdó-Igbo comparison is that some of the sentences in (21) may involve clausal adjunction in Igbo, although on language and theory internal grounds I will argue for clausal subordination in Èdó (Chapter seven).

This brings us to the third observation concerning the sentences in (21). This is based on an examination of other contexts in which the OVS can occur, data from Déchaine (1993: 240):

(24)	a.	Ámaakó a-nu-o-na mmìi	(Déchaine=130)
		3s e-drink-A-perf wine	
		'Amaakó has drunk [some] wine.'	

- M nu-o mmìi
   1s drink-Á wine
   '[If] I drink wine.'
- c. Ányì cho-ro sí unu ga-a ahyá 1p want-rV say 2p go-Á market.Gen 'We want that you (p) go to market.'
- Ányì ga-ra áhya zu-o jí
   1p go-rV market buy-Á yam.Gen
   'We went to market and bought yam[s].'
- e. Ó jì-(ri) ´mmà bhá-a jí 3s hold-rV knife peel-Á yam.Gen 'S/he peeled yam[s] with [a] knife.'

There are two types of data in (24), those in which there is only one verb (24a,b) and those involving the sequence of two verbs (24c-e). Déchaine (1993) proposes a single analysis for the OVS in all these sentences whereby they are governed by some higher functional head:

[In the perfective, V is governed by the perfective Aspect morpheme; in an *if* -clause, V is governed by a conditional operator (presumably in Comp); in subjunctive clauses, V is governed by dependent Tense, and by hypothesis, in serial constructions V2 is governed by non-adjacent tense] (Déchaine 1993: 240)

When this generalization is interpreted in the context of the data in (21), I propose that the second verb is governed by a tense related functional projection. The OVS is generated in the head of this functional projection and the surface order of Verb+inflection is realized by obligatory verb raising to this functional head. This is illustrated in (25a) with an adjunction structure for (21c-f) and (25b) with a clausal complementation structure for (21a,b).<sup>9</sup>

<sup>&</sup>lt;sup>9</sup> The choice of syntactic structures follow from the discussion in footnote 8. For example, I am assuming a common analysis for instrumental constructions (25b) as distinct from CCs and SVCs (but see the discussion in Chapter seven which reveals a more complicated analysis of instrumental constructions).



In conclusion, therefore, we find that the distinction between resultative SVC, consequential SVC, and CCs presents valid insights into the analysis of Igbo verb-verb constructions which can either have the resultative structure or a clausal complementation or adjunction structure but not the consequential SVC structure.

### 5.5 (Mandarin) Chinese

(Mandarin) Chinese has often been classified as an SVC language (cf. Craig and Hale 1988) and it is also a language that has many V-V compounds (cf. Li 1990). This mixture of V-V compounds with SVCs in a non-Kwa language presents a window of opportunity to test some of the empirical predictions of my analysis based on a structural distinction between resultative and consequential SVCs. My basic claim is the same as the one proposed in Igbo: resultative SVCs, and not consequential SVCs, translate as

194

resultative V-V compounds in Chinese. This implies that Chinese sentences in which there are two transitive verbs that share a single surface object should not occur as V-V compounds, but sentences with two verbs which express a resultative meaning should always occur as V-V compounds. This distinction is based primarily on the fact that resultatives are close-knit constructions and they surface as compounds in Igbo, whereas consequential SVCs are more loosely connected and so the verbs can occur independently. This is borne out by the templatic data in (26) and (27):

- (26) a. wo qie rou mai I cut meat sell 'I cut the meat and sold it.'
  - b. \*wo qie-mai rou I cut-sell meat
- (27) a. \*wo da Zhangsan si I hit Zhangsan die
  - b. wo da-si Zhangsan I hit-die Zhangsan 'I struck Zhangsan dead.'

Observe the fact that it is ungrammatical to express sequential events in a V-V compound (26b) while (27) shows that Chinese is like Igbo because it is ungrammatical to express action-result V-V compounds as consecutive (separate) events. This intuitive distinction that is based on the possibility of separating two events or being realized as one single event is consistent with the difference between resultative and consequential SVCs.

### 5.5.1 Resultative V-V compounds

I will now present data that shows that the generalization from section 5.3 and 5.4.1 can be replicated over a wide range of examples. Consider, first, the case of resultative V-V compounds illustrated in (28) :

(28)	a.	Zhangsan tui-dao-le Lisi	(data from Teresa Wu p.c.)
		Z push-fall-Asp Lisi	
		'Zhangsan pushed Lisi down.'	

- b. Libai tang-ping-le chenshan L iron-flat-Asp shirt 'Libai ironed the shirt flat.'
- c. Zhangsan da-si-le Lisi Z strike-die-Asp Lisi 'Zhangsan struck Lisi dead.'

There are two cross-linguistically comparable pieces of evidence for proposing a single analysis for both Igbo and Chinese resultative V-V compounds on a par with resultative SVCs in Èdó. First, it has been observed that no aspect markers or measure words may intervene between the parts of Chinese resultative V-V compounds (Thompson 1973:362). Thus, Chinese resultative V-V compounds are like Èdó resultative SVCs which do not allow adverbs, PPs or double objects to occur between the verbs. Second, whereas Chinese is like Èdó in having very little inflectional morphology, in the resultative V-V compound an aspect (perfective) marker occurs obligatorily as a suffix to the whole compound. In this way, Chinese is like Igbo whereby both verbs must raise to a higher functional category with strong tense/aspect features. In the case of Chinese, though, the head of the functional projection bears aspectual features spelled-out by -le. By abstracting away from my analysis of the Èdó resultative SVC, the representation for (28b) is given in (29). (VoiceP is omitted).<sup>10</sup>

196

<sup>&</sup>lt;sup>10</sup> I have omitted EP projection based on the fact that EP is a kind of 'outer' Aspect in the framework of Travis (1994) and so the projection of AspP is sufficient to license the same thing as EP (in this case). One consequence of this analysis is that we are able to derive resultative V-V compounds which have idiosyncratic meanings and some verb + resultative particles like those listed in (i) from Thompson (1973:376).

<sup>(</sup>i) jì-zuo 'mail out', kai-zuo 'drive out', mai-dao 'succeed in buying'

Although I do not consider particles in my analysis but I assume that they have the same structure and analysis as resultative constructions in general (cf. Dikken 1995).



# 5.5.2 Multi-event Constructions

In this section, I will present empirical evidence which show that the consequential SVC is systematically absent in Chinese, rather there are only resultatives and CCs which are realized as CP adjunct (Wu, forthcoming). Chinese exhibits what appears on the surface to look like consequential SVCs. Relevant examples are given in (30):

(30)	a.	wo zhu ji che I cook chicken eat 'I cooked the food and ate it'
	b.	Li mai sha song Zhangsan Li buy book give Z 'Li bought a book and gave it to Zhangsan'
	c.	Libai mai sha kan L buy book read 'Libai bought a book to read'
	d.	Zhangsan na qian fu Lisi Z take money pay Lisi 'Zhangsan took the money and paid it to Li'

It is interesting to note that the events expressed by these sentences can never be realized as V-V compounds. This is illustrated for (30c) in (31).

(31)	*Libai mai-kan-le shu	(Teresa Wu, p.c.)
	L buy read-Asp book	
	'*Libai buy- read a book'	

This contrast between (30c) and (31) underscores the difference between putative SVCs and resultative V-V compounds. However, these putative SVCs behave differently from their Èdó equivalents (see Wu, forthcoming) in two major respects namely, negation placement (32) and tense/aspect inflection (33).

(32)	a.	*Òzó d <u>é</u> ìyán má lé Ozo buy yam neg. cook	'Èdó'
	b.	*Òzó d <u>é</u> ìyán í lé Ozo buy yam neg. cook	18
	c.	Libai mai-le LGB bu kan L buy-Asp LGB not read 'Libai bought LGB not to read it'	Chinese
	d.	Libai bu mai LGB kan L not buy LGB read 'Libai does not buy LGB to read it'	
(33)	a.	*Òzó d <u>é</u> !(ré) ìyán lé!(ré) Ozo buy+rV yam cook+rV	Èdó
	b.	Libai mai-(le) LGB kan-(le) L buy-Asp LGB read-Asp 'Libai bought LGB and he did read it'	Chinese

In the Èdó consequential SVC sentences in (32a,b), we observe that past and non-past negative markers can never precede the second verb, however when compared with similar looking constructions in Chinese we see that negation can occur before the second verb. Furthermore, there is also a contrast between Èdó and Chinese with respect to verb inflection. The perfective -rV suffix is ungrammatical on any of the verbs in the consequential SVC in Èdó (33a) (see discussion in chapter six), but the aspect marker -le can occur on both verbs in similar looking sentences in Chinese (33b).

Consequently, I suggest (following Wu, forthcoming) that Chinese multi-event constructions do not have exactly the same syntactic structure as Èdó consequential SVCs.

198
I will, however, not speculate on the syntactic structure of these Chinese sentences in any detail.

# 5.6 Conclusion

As a general conclusion, I have shown in some detail that there are systematic variations between SVCs that can be replicated across languages. In particular, resultative SVCs are realized as resultative V-V compounds in languages where verb raising to a functional head (above the VP) and bearing a strong V-feature is obligatory (Igbo, Chinese). Furthermore, I have argued that the class of consequential SVC is systematically absent in these languages where the V must raise to check a V-feature of Infl, thus such languages have only resultative V-V compounds and possibly some form of clausal adjunction/conjunctions. Now, this point would be completely lost under a unified analysis of SVCs and so one of the strengths of this thesis is the predictions it makes based on the distinction between resultative and consequential SVCs.

# Chapter Six

# The Serial Verb Construction Parameter

### 6.1 Introduction

This chapter centers around the fundamental question of what is the serial verb parameter? In other words, why does the grammar of Èdó allow verb serialization whereas English does not? Or, to take a finer-grain version of the question, why do Èdó resultative SVCs show up consistently as V-V compounds in Igbo and Chinese? The basic claim that I will argue for is that the serial verb parameter can be derived from differences in the nature and "strength" of Tense-bearing functional heads across languages (cf. Muysken 1988, Déchaine 1993, Collins 1997, also Baker 1989, traceable to Roberts 1985).

As a preliminary step, I would like to summarize what I consider to be some of the major and distinctive syntactic properties that characterize the structures of the three constructions as discussed in the earlier Chapters. These are listed in (1)-(3).

- (1) **Resultative SVCs** 
  - a. Both verbs are syntactic co-heads and they assign their internal theta roles to a single object NP within a VP that is contained within a single EP projection.
  - b. The second verb is typically unaccusative, and it combines with the first verb to express a single event.
  - c. It is incompatible with double object construction, since the DOC is also a resultative.

# (2) Consequential SVCs

- a. The verbs head separate VPs which are dominated by separate projections of EPs whose heads are bound by the higher EP; as such the verbs express two connected events
- b. There is a single Agent for the two connected events event that is introduced by Voice (cf. Kratzer 1996)
- c. The verbs in the consequential SVC must be transitive; have distinct objects which are coreferent, that of the second verb being realized as *pro*.

d. Double object construction is possible but object sharing is with the underlying (theme) object rather than the derived (goal) object

#### (3) **Covert coordinations**

- a. The verbs head separate VPs which are dominated by separate (symmetric) projections of EPs and they express separate events.
- b. Each event 'potentially' has its own Agent, thus there are separate VoicePs associated with each VP.
- c. Each verb can have a 'separate' object in principle depending on the argument structures of verbs being conjoined.<sup>1</sup>

# 6.2 Serial Verb Constructions and Secondary Predicates

Larson (1991) proposes that serial constructions should be analyzed as a form of secondary predication. Consequently, he suggests that the SVC parameter reduces to a matter of what secondary predicate categories are allowed. Basically, this implies that the "serialization parameter" separating Èdó and English should involve some respect in which verbs and nominals differ with respect to predication (Larson, 1991:206). Based on standard generative feature matrices for lexical categories, serializing languages will have secondary predicates that are either [-N] or [+V], whereas non-serializing languages will have secondary predicates that are either [+N] or [-V].

While Larson's proposal is attractive and introduces part of the research agenda being pursued in this chapter, it however makes empirically incorrect predictions. For example, it predicts that NP secondary predicate constructions analogous to 'John arrived a perfect wreck' are entirely absent in a SVC language and this is contrary to fact as illustrated in the following examples:<sup>2</sup>

- (i) <u>ó</u>bólòkàn òré Òzó rrí èvbàré raw Foc. Ozo eat food 'It is raw that Ozo ate his food'
- (ii) óhóghà òré Òzó fi ímótò empty Foc. Ozo drive car
   'It is empty that Ozo drove the Car'

<sup>&</sup>lt;sup>1</sup> According to my analysis, the object the second verb cannot be pro in a CC because pro is licensed in the domain of a single E head (see also Baker and Stewart 1997b).

<sup>&</sup>lt;sup>2</sup>Independent evidence that the phrases in brackets are NPs include the fact that they are vowel-initial, and they can be clefted (i) and (ii).

- (4) a. Òzó rrí èvbàré [<u>ó</u>bólòkàn]<sub>NP</sub>
  Ozo ate food raw/empty (without meat etc.)
  'Ozo ate the food raw/bare.'
  - b. Òzó fĩ ímótờ [óhóghà]<sub>NP</sub> Ozo drive car empty 'Ozo drove the car empty (of passengers).'

In (4a,b), the bracketed NPs are predicated of the object of the verb and these sentences translate as standard examples of depictive SPs in English, i.e., in (4a) John ate the food and/while the food was raw/empty (object depictive), and in (4b) Ozo drove the car and/while the car was empty of passengers (object depictive). The possibility of sentences like (4) suggest that Larson's proposal as it is stated cannot be correct since SVC languages do in fact use [+N] lexical categories for secondary predication along with VP and AP.<sup>3</sup>

However, the syntactic properties of the different constructions (1)- (3) presents some interesting features which suggest that a partial cross-linguistic connection can be made between SVCs and SPs. I will discuss these similarities based on data from Èdó SVCs and English SPs.

The primary similarity comes from the analysis of event structures in both SVCs and SPs. As will be made clear shortly, some SPs are similar to SVCs in terms of the composition of the events that they express. The initial assumption here is that SPs resemble SVCs in terms of being made up of two general classes: the resultative versus depictive distinction for SPs (5-6) can be compared with the resultative versus consequential SVCs and CCs (7):

- (5) a. John pounded the metal  $[flat]_{AP}$ 
  - b. John beat the metal [into a sword]<sub>PP</sub>
- (6) a. Peter gave the meat to Mary  $[raw]_{AP}$ 
  - a. John got to the party [a perfect wreck]<sub>NP</sub>

<sup>&</sup>lt;sup>3</sup> See also Baker (1997a), (1997b) and Baker and Stewart (1997a) for general discussion of the inadequacy of using generative feature matrices for capturing the distinction between verbs and adjectives.

- (7) a. Òzó sùá àdésúwà dé
  Ozo push Adesuwa fall
  'Ozo pushed Adesuwa down.'
  - b. Ôzó d<u>é</u> émà kpèé
    Ozo buy drum play
    'Ozo bought a drum and played it.'
  - Òzó dé émà khién égógó
    Ozo buy drum sell bell
    'Ozo bought a drum and sold the bell.'

By comparing the resultative SP in (5) with the resultative SVC in (7a), we notice the fact that APs, PPs, and VPs all express what the theme comes to be like as a result of the event that is denoted by the main (first) verb. For example, because of the action of the event that is denoted by the verb *pound*, the metal becomes flat in (5a), while the metal is shaped into a sword in (5b). Similarly, in (7a) the object  $\dot{a}d\dot{e}s\dot{u}w\dot{a}$  undergoes a transition into the state of having fallen, which expresses the result of the action of the first verb.

In the examples of depictive SPs in (6), both the NP and the AP characterize what the object or subject is like at the time of the event. In general, a depictive SP can either be object-oriented (characterizing only the object) as in (6a) where the AP tells us about the state the object was in when it was given to Mary, or it can be subject-oriented (characterizing only the subject) as in (6b) in which the NP tells us about the state in which the subject arrived at the party. On an intuitive level, this contrast amongst depictives seems to reflect the difference between consequential SVCs and CCs. In (7b), the two transitive verbs share a single surface object, while in (7c) each verb has its own object and they only share the same surface subject. The facts from double objects provide further evidence for an intuitive similarity between consequential SVCs and object-oriented depictives. In both constructions, when double objects occur only the theme (underlying) object is shared but not the goal (derived) object as shown in (8).

- (8) a. John gave Mary the meat raw/hungry = AP predicated of theme only, \*goal Agent goal theme
  - b.  $\dot{O}z\dot{o} vb\dot{o} \dot{o}kh\dot{o}!kh\dot{o} igan khién = VP predicated of theme only, *source Agent source theme$

203

While there seems to be an intuitive parallel between consequential SVCs and objectoriented depictive SPs, the same cannot be said for conjunctive SVCs and subject-oriented SPs since they differ in terms of interpretation (and presumably in the syntax as well).<sup>4</sup>

However, the strongest resemblance of all is between resultative SPs and resultative SVCs which I consider to be relevant to the whole issue of the serial verb parameter. This resemblance comes from the nature of sub-atomic events in both AP and VP resultatives. In both cases, the VP as a whole is an accomplishment, i.e., a single event composed of a process part and a result part in which the process sub-event is realized by a verb in both constructions and the result sub-eventuality is expressed by a second predicate which may be AP, PP, or VP. The fact that an accomplishment is constrained by the (universal) ontology of events provides the basis for comparing the structures of the two types of resultatives. Consequently, I propose that this resemblance between resultative SPs and SVCs underscores the structural relations between the two constructions from which we can narrow in on the serial verb parameter. My proposal is supported by four different bits of structural evidence.<sup>5</sup>

First, it has been observed that the adjectival predicate in a resultative SP bears a close thematic relation to the verb (cf. Rapoport 1990, Bolinger 1971, Dowty 1979, Simpson 1983, Rothstein 1983 etc.). This is compatible with the structure of the resultative construction presented in the simplified structure in (9).

<sup>&</sup>lt;sup>4</sup> I will put this issue aside for further studies.

<sup>&</sup>lt;sup>5</sup> In order to make the comparison more systematic, I will continue my discussion based on AP resultatives in English compared with VP resultatives in Èdó. I exclude PP resultatives because the status of prepositions in Èdó needs more study before it can be used in a controlled comparison and this is outside the immediate scope of this work. Furthermore, this choice of restricting the discussion to the AP category is based on two other factors (a) AP is the only category obligatorily interpreted as a predicate (Williams 1981), (b) Èdó has the option where either AP or VP can occur as the second predicate of the resultative thereby producing theory-internal evidence for the distinction between AP and VP resultative constructions.



As represented in (9) the main verb that is in a bound chain between a top verb position and an empty (lower) position takes as its complement the resultative predicate [AP or PP] and the two together, as complex predicator, assign a thematic role to the object NP. This analysis of the AP resultative SP is supported by evidence from English based on the middle construction discussed in Hale and Keyser (1987) and illustrated in (10).

- (10) a. This kind of metal hammers smooth fast.
  - b. This counter wipes dry quickly.

Rapoport (1990) takes these examples as the basis for the argument that the resultative predicate [V + AP] "affects" (and obviously theta marks) its object NP since this is a necessary condition on the object NP becoming the subject in a middle construction. According to Hale and Keyser (1987), a well-formed middle implies that the underlying object NP is theta-marked by the verb and in the case of (10) by both the verb and the AP resultative predicate. We can confirm the fact that the AP is making a crucial contribution based on the ungrammaticality of corresponding sentences without AP.

- (11) a \*This (kind of) nail hammers fast.
  - b. \*This counter wipes quickly.

I conclude, therefore, that the verb and the AP resultative form a complex predicate with a single object just as the verbs in the resultative SVC.

The second bit of evidence for a close relationship between AP resultative SP and resultative SVC is based on the fact that both constructions appear to share striking similarities in the predication relation between the direct object argument and the second predicate [AP or VP]. For example, in the resultative SVC the shared argument is the direct object that is assigned the internal theme theta roles of both the first and second verbs. A similar restriction--the Direct Object Restriction (DOR), also obtains in the resultative SP whereby the AP is predicated of the direct object that is assigned a theme role (cf. Williams 1980, Rothstein 1983, Simpson 1983, Levin and Rappaport-Hovav 1995, Collins 1997, Bowers 1993, Baker 1997a/b, etc.).<sup>6</sup>

The third evidence for proposing a structural correspondence between resultative SVCs and AP resultative SPs comes from the observation that double object constructions (or double complements more generally) are ruled out from both constructions. Hoekstra (1992) discusses this point for the AP resultative SP based on the data in (12) and (13) which is parallel to the Èdó paradigm in (14):

- (12) a. The teacher taught the boy a lesson. (Hoekstra=57)
  - b. The teacher taught the boy crazy.
  - c. \*The teacher taught the boy crazy a lesson.
- (13) a. John split Mary a coconut.
  - b. John split a coconut open.
  - c. \*John split Mary a coconut open.
- (14) a. Ozó sùá èsósà ògó
  Ozo push Esosa bottle
  'Ozo pushed Esosa's bottle.'
  - Òzó sùá <u>ògó</u> gu<u>òghó</u>
    Ozo push bottle break
    'Ozo pushed the bottle broken.'
  - c. \*Ôzó sùá èsósà ògó guòghó Ozo push Esosa bottle break

<sup>&</sup>lt;sup>6</sup> The details of theta role assignment and whether the AP has any theta role differ in these analyses and I will not comment on any of these approaches in detail.

According to my analysis of the resultative SVC, double object constructions are incompatible with resultatives because of a general constraint on delimitedness. In English AP resultative SPs and Èdó resultative SVCs, there is only one structural position for a delimiter and this is filled by AP and VP respectively. Based on the assumption that the first object in the double object construction is also a delimiter, therefore both categories compete for the same structural position in the pre-movement structure and hence the ungrammatical sentences in (12c), (13c) and (14c).

The fourth and final bit of structural evidence bearing on the relationship between resultative SVCs and AP resultative SPs comes from the nature of category-restrictions on the second predicate in the two constructions. For example, while the categories AP and PP have been observed to make good resultative SPs in many languages,<sup>7</sup> however, it is ungrammatical for a VP to occur as a resultative SP in a non-serializing language (cf. Larson 1991, Baker 1997b etc.). On the contrary, it is very common to find the category VP serving as a resultative predicate in serializing languages such as Èdó. Furthermore, a parallel restriction which is internal to the category VP can be found in the resultative SVC namely, only an unaccusative verb (which includes stative verbs as well as traditional unaccusatives) can occur as the second predicate. I propose that these are not superficial differences but rather they reflect a deeper connection between AP resultatives and VP resultatives. Thus, once we understand the nature of the difference in the domains in which the category-restrictions hold we should find that resultative SVCs resemble resultative SPs with respect to a structural relationship between unaccusative second verb and adjective secondary predicate.

As a conclusion to this section, I propose that whatever else can be said of the analysis of these constructions, it should be the case that a (partially) unified account

<sup>&</sup>lt;sup>7</sup> It is a general fact that NPs do not make good resultative SPs in English, apart from the type of examples in Carrier and Randall (1992) e.g. They painted the barn a hideous shade of green' which is controversial as a resultative predicate (Mark Baker p.c.).

should be given that would explain at least these four correspondences that I have just discussed.

# 6.3 Towards the SVC Parameter: AP vs. VP Resultatives

One key fact in the analysis of AP and VP resultatives as the basis for deriving the serial verb parameter is that along with VP resultatives, there are also AP resultative SPs in Èdó. This is important because it provides language-internal evidence concerning the distinction between AP and VP resultatives which can then be generalized into non-SVC languages like English which have only AP but never VP resultatives, thereby presenting us with some very useful insights into the serial verb parameter. Consider the following:

- (15) a. Òzó kòkó àdésúwà mòsè
  Ozó raise Adesuwa beautiful (A)
  'Ozo raised Adesuwa to be beautiful.'
  - Òzó gbé émá!tón wénrén
    Ozo beat metal tiny (A)
    'Ozo beat the metal thin'
  - c. Òzó giá írúnmwùn gi<u>èghè</u> Ozo cut grass short (A) 'Ozo cut the grass short.'
- (16) a. Ozó kòkó àdésúwà mòsé
  Ozó raise Adesuwa beautiful (V)
  'Ozo raised Adesuwa to be beautiful.'
  - b. Ôzó gbé émá!t<u>ó</u>n w<u>è</u>nr<u>é</u>n Ozo beat metal tiny (V) 'Ozo beat the metal thin.'
  - c. Òzó giá írúnmwùn gièghé Ozo cut grass short (V) 'Ozo cut the grass short.'

The sentences in (15) and (16) are similar since they are all examples of resultative constructions. In (15), the second predicate is an adjective, like the AP resultative SP in English, while (16) illustrates the resultative SVC in Èdó in which the second predicate is a an unaccusative verb. The only obvious surface difference between the adjectival resultative

SP (15) and the verbal resultative (16) is the tone on the final word. The final words in (16) are verbs and they all have the same pattern of tonal inflection which consist of a low-high sequence, while the final words in (15) are morphologically related adjectives whose tones vary but are mainly level tones with either all low or all high patterns without any tone contours (e.g. floating tone or downstepped tone etc.). We can establish the relevance of tones on these categories based on the contrast between (15a) and (16a) in which there are disyllabic words. Observe that while the tones on the verbal predicate in (16a) matches those on the first verb, those on the adjectival predicate do not match the tone sequence on the verb in (15a).

Adjectives contrast significantly with verbs in Èdó in terms of their packaging in the lexicon. Adjectives like nouns have stable (invariant) tone patterns which are set in the lexicon, i.e., the tones on nouns and adjectives come as part of the knowledge of language. For example, Agheyisi (1986), Omoruyi (1986) lists the following items as adjectives:

(17) (a) w<u>ò</u>r<u>ò</u> 'long' (b) gi<u>è</u>gh<u>è</u> 'tiny' (c) mòsè 'beautiful' (d) w<u>é</u>nr<u>é</u>n 'slim'

I propose that when one knows the Édó language such knowledge includes the representation of the tone patterns and the meanings that are associated with these words in (17); in other words, adjectives are not derived by any special syntactic (or morphological) process. In contrast, all verbs have grammatical tones which convey grammatical information such as Tense. I will come back to the analysis of how these Tense tones come to be inflected on verbs in section 6.5.

It is important to note that the general structural properties that were discussed in relation to AP resultatives in English and VP resultative in Èdó in section 6.2 also hold for these Èdó AP resultatives. Consider the following:

(18) a. \*Özó kökó àdésúwà mòsè wénrén Ozó raise Adesuwa beautiful slim

- Òzó kòkó àdésúwà èmó Ozó raise Adesuwa children 'Ozo raised Adesuwa's children.'
- Ozó kòkó èm<u>ó</u> mòsè
  Ozó raise children beautiful (A)
  'Ozo raised the children to be beautiful.'
- d. \*Òzó kòkó àdésúwà èm<u>ó</u> mòsè Ozó raise Adesuwa children beautiful (A)

What the data in (18) show is the fact that Èdó AP resultative SP has the same behavior like its English counterpart. For example, (18a) illustrates the aspectual generalization that it is ungrammatical to have two delimiters in the resultative construction. Thus, the sequence of two APs each being predicated of the same single object is ungrammatical. This same fact is illustrated by the contrast between (18b) and (18d) where we observe that although double objects can occur with the verb  $k\partial ko'$  'raise' as in (18b), the same sentence with double objects is ungrammatical (18d) when there is an AP resultative. Again, the ungrammaticality of (18d) stems from the assumption that the source object in the double object construction is a delimiter which competes for the same structural position as the AP. Furthermore, notice that (18c) is grammatical as there is a single object of which the single AP is predicated of, i.e., a single delimiter. Finally, in the grammatical sentences the AP characterizes the theme and not the Agent. Thus, I conclude that the AP resultative SP in Èdó has the same structure as the one proposed for English in (9), repeated here as (19) for the sentence in (18c) (only the verbal projection is relevant at this point).

(19) VP V'  $k \partial k \partial k$  NP V'  $e_k$  AP  $e_k$  AP  $e_k$  AP  $e_k$  AP AP  $e_k$  AP Given this similarity between Èdó and English in terms of AP resultative SP, we are now directly confronted with the issue of the contrast between AP and VP resultative since they essentially have the same underlying structure. For the purpose of analytical clarity, the relevant part of the structure of the resultative SVC is repeated in (20).



As a way to further justify the proposal that AP and VP resultatives have almost the same underlying structures in relevant respects, I adopt a strategy based on illustration of the weaknesses of two other competing analysis done within the same general framework.

Collins (1997) proposes that one piece of evidence that supports his analysis of resultative SVCs in Ewe as involving control of a null *pro* comes from the observation that English AP resultative SPs should probably be analyzed as involving control as well. Based on Bowers (1993), it is proposed that a sentence such as (21a) could be analyzed with the structural representation in (21b).

#### (21) a. John watered the tulips all flat



According to Collins (1997), *all* in (21a) must be attached to an empty category which is represented as PRO in (21b), assuming the theory of floated quantifiers in Sportiche (1988).

The primary objection to this unified analysis for both AP and VP resultatives is based on the empirical observation that goes against positing an empty category in the VP resultative (SVC) and this has been discussed at length in section 2.2.2 of chapter two. In addition, there are two separate basis for a theoretical objection to this analysis. First, Bobaljik (1997) argues that *all* in a sentence like (21a) which is assumed to be evidence for NP trace in Sportiche (1988) is actually an adverb. Now, if this analysis is correct it means that there is really no evidence for an empty category in the AP resultative SP and so it cannot be compared with VP resultatives on this ground. Second, it is not exactly clear (nor explicitly stated) what theory of control of PRO is being assumed for the analysis of the English resultative SP. Based on one analysis of PRO--the PRO theorem of Chomsky (1981)--the structure in (21b) would require some functional structure to protect PRO from being governed and Collins' (1997) analysis is silent on this issue as it shows no evidence for such functional structure. Therefore, I reject the analysis of AP and VP resultatives given in Collins (1997) that is based on an empty category mediating object sharing.

Another relevant analysis of the resultative construction is Baker (1997b) who also considers the relationship between AP and VP resultatives. The main claim of this analysis, which adopts a lexical decomposition approach, is that resultative constructions arise when a second adjective is adjoined to the adjectival component of the verb in a pre-conflation representation. Thus, the most deeply embedded part of the verb is itself an adjective which combines in a sort of conjunction with another element that must be of the same syntactic/semantic type [adjective]. This analysis can be illustrated for the sentence in (22a) by the pre-conflation structure in (22b), that follow from the assumption that standard transitive verbs are typically decomposed into (at least) three elements as in (22c). (22) a. I polished the metal smooth



c. [x CAUSE [ y BE [ADJECTIVE]]]

In (22c) x is the agent and y is the theme which is followed by a BE operator. The way this works is illustrated by the representation in (22b) in which the adjectival component of the verb moves out of the complex AP to combine with BE and CAUSE to derive the verb realized by *polish*. However, VP resultative (SVCs) do not fit neatly into the structure in (22b) since this particular analysis hinges on a complex predicate relation between two adjectives whereas the second predicate of the resultative SVC is a verb. As an alternative, Baker (1997b) proposes a modification to (22b) and includes a second BE element so that the two verbs in the resultative SVC could be created by conflating into their separate BEs. Thus, a resultative SVC like (23a) would have the structure in (23b).<sup>8</sup>

(23) a. Ozó h<u>òó</u> ùkp<u>ò</u>n hùán
 Ozo wash cloth be-clean
 'Ozo washed the cloth clean.'

<sup>&</sup>lt;sup>8</sup> Observe that the lower part of the V'-structure replicates a bivalent projection analysis of resultatives in Déchaine (1993: 141 ff.)



My primary objection to the structure in (23b) is based on the fact, which Baker rightly observes, that it involves the coordination of two phrasal categories (non-atomic heads). Therefore, moving the first verb which is made up of BE+WASH to conflate with CAUSE violates the Coordinate Structure Constraint (CSC) of Ross (1967). Furthermore, a related problem is that (23b) would require an explicit formulation of why only the first verb and not the second verb would need to raise up to CAUSE and this does not follow in any way from his analysis except by stipulation. Therefore, I reject the analysis of VP resultatives in Baker (1997b).<sup>9</sup> One point of convergence between Baker's analysis of AP and VP resultatives and the one I have proposed in (19) and (20) is the fact that both analyses avoid the awkward feature of Collins (1997) since there is no empty category; structurally there is a single NP object for both predicates.

There are two challenges facing the correct unified analysis of AP and VP resultatives. The first is to show the sorts of constraints on the relation between AP and VP resultatives, and the second is to justify the structural difference between both of them. I expect that the answers to these questions will lead to an explicit formulation of the serial

<sup>&</sup>lt;sup>9</sup> There are two peculiar properties of the resultative construction which Baker's (1997b) analysis provides an elegant account for namely, the causative meaning associated with the main verb (V1) and the predicative power of adjectives (cf. Levin and Rappaport-Hovav, 1995). Such features are not in any way obvious in my analysis of the resultative SVC, and I do not consider them so as to keep the discussion of the serial verb parameter and the structures fairly simple. However, I should point out that these properties can be technically derived from my analysis if I also assume a lexical decomposition approach. Thus, the first verb would have a CAUSE operator which is typically missing from the unaccusatve second verb. The second predicate itself can have the uniform decomposition of [ y INCH<BE[Adj]] such that in AP resultatives there is no conflation, while in VP resultative conflation applies to a simple Adj+BE to yield statives like mosé 'be-beautiful', or it may go higher to INCH and yield unaccuastives like wú 'die'.

verb parameter distinguishing AP resultatives (Èdó and English) from VP resultatives (Èdó only) and, more generally, to distinguish SVC languages from non-SVC languages.

### 6.4 The Serial Verb Construction Parameter

In this section, I will focus on two crucial empirical evidence which clearly illustrate the constraints on AP and VP resultatives that will be generalized as the serial verb parameter, for example, distinguishing Èdó VP resultatives from English AP resultatives. The two areas that I will consider are; (a) the interaction between verb-raising and the bare stem condition, and (b) the issue of tense (tone) inflection.

#### 6.4.1 The Bare Stem Condition

In this section, I provide a systematic account of the distinction between AP and VP resultatives with respect to the distribution of morphological tense inflection (cf. Baker and Stewart 1997a). I will argue that this property of morphological inflection is correlated with overt verb raising in Èdó and that verb raising is an obligatory parameter that relates to what I will call the 'bare stem condition' (BSC) and is relevant to the licensing of SVCs. The relevant data introducing the contrast is given in (24)-(27):

- (24) a. Ozó kökó Adésúwa möse Ozo raise Adesuwa beautiful (A) 'Ozo raised Adesuwa to be beautiful.'
  - b. Àdésúwà dré Ózó kó!kó(-ró) mòsè
    Adesuwa Foc. Ozo raise-RV beautiful (A)
    'It's Adesuwa that Ozo raised to be beautiful.'
- (25) a. Ôzó má ákhé w<u>òrò</u>
  Ozo mould clay-pot long (A)
  'Ozo moulded the clay-pot to be long'
  - ákhé <u>ò</u>ré Òzó má(-ré) w<u>ò</u>r<u>ò</u> clay-pot Foc. Ozo mould-RV long (A)
     'It is the clay-pot that Ozo moulded to be long.'

- (26) a. Òzó kòkó Àdésúwà mòsé Ozo raise Adesuwa be-beautiful (v) 'Ozo raised Adesuwa to be beautiful.'
  - b. Àdésúwà <u>ò</u>ré Òzó kó!kó(\*-ró) mòsé(\*-ré) Adesuwa Foc. Ozo raise-RV be-beautiful (V) 'It's Adesuwa that Ozo raised to be beautiful.'
- (27) a. Òzó má ákóbi<u>é</u> wó
  Ozo mould clay-doll be-hard (V)
  'Ozo moulded the clay-doll to be hard.'
  - b. ákóbi<u>é</u> <u>à</u>ré Òzó má(\*-ré) wó(\*-ró) clay-doll Foc. Ozo mould-RV be-hard-RV (V) 'It's a clay-doll that Ozo moulded to be hard.'

(24)-(25) are examples of the AP resultative SP while (26)-(27) illustrate the resultative SVC. Both constructions are acceptable in simple past tense clauses as seen in the (a) sentences where the verbs bear the normal past tense tones for disyllabic verbs. However, we notice a clear contrast between adjectival resultative SPs and resultative SVCs in the (b) sentences when the verbs are inflected for the past perfective tense realized as an [-RV suffix].<sup>10</sup> The adjectival predicate is possible when the main verb is inflected for past perfective tense that is realized by the -rV suffix (24b) and (25b); notice that there is no affix on the A, and it still keeps its lexical tones. However, when the resultative predicate is a VP, i.e., a stative verb in (26b) and (27b), the first verb cannot be in the past perfective form and the ungrammaticality does not improve even when the second verb is similarly inflected (to capture tense matching that I have alluded to earlier on and will come back to under 6.5). This fact that the verbs in the resultative SVC cannot bear overt morphological inflection is what I call the bare stem condition, stated descriptively as in (28).

#### (28) Bare Stem Condition (BSC)

No verb in the serial verb construction can bear morphological tense inflection<sup>11</sup>

<sup>&</sup>lt;sup>10</sup> Agheyisi (1990) proposes that the direct objects are clefted in these examples because the -rV (past perfective) affix elides before overt NPs as a result of fairly normal phonological rules (cf. Omoruyi 1991) but I will provide a syntactic explanation below.

<sup>&</sup>lt;sup>11</sup> According to this generalization, tone marking is not inflectional.

In order to understand the BSC (28), it is important that we examine the special status of the past perfective in Èdó. As stated in Baker and Stewart (1997a), the past perfective verbal suffix is the only tense/aspect category in Èdó that is realized as an inflectional affix with segmental content. In particular, it is a suffix consisting of the approximant consonant r and a low-tone vowel that harmonizes with the last vowel of the verb stem.<sup>12</sup>

(29)	Èdó tense paradigms (partial) (cf. Baker and Stewart, 1997)					
		One syllable verb (cry)	two syllable verb(cry-PL13)			
simple	past	só	sòló			
present	(habitual)	sò	sòlò			
simple	future	ghá sờ	ghá só!ló			
past per	rfective	só-rò	sòló-rò			

The tense paradigm in (29) is fairly accurate for most classes of verbs (transitives, unergatives, statives, and unaccusatives). What (29) clearly shows is the fact that there is only one inflectional affix with segmental content--the past perfective suffix -rV--and that is incompatible with serial verb construction given the BSC (28). The next question is what is the theoretically satisfying account of the descriptive generalization stated in the BSC.<sup>14</sup> To pursue this, I turn now to the issue of verb raising in Èdó.

<sup>&</sup>lt;sup>12</sup> All other tense/aspect features in Èdó are indicated either by tonal morphemes, such as the simple past and present that have been used thus far in all of my examples, or by independent auxiliary particles or preverbs such as the future ghá or past habitual ghá!á.

<sup>&</sup>lt;sup>13</sup>PL refers to plurality of the object that is indicated as an agreement on the verb (cf. Stewart 1997). Under certain conditions PL may also refer to the iteration of the action denoted by the verb based on the plural interpretation of the cardinality of the object.

<sup>&</sup>lt;sup>14</sup> Baker and Stewart (1997a) attempt to draw a link between the BSC and similar effects in the come/goplus-bare-infinitive construction of American English (i) that has been discussed in Jaeggli and Hyams (1993) and Pollock (1991) among others.

<sup>(</sup>i) a. Come talk to me today

b. He will come talk to me today

c. They will come talk to me every day

d. \*He comes talk(s) to me every day

e. \*He came talk(ed) to me every day

f. \*He has gone talk(ed) to her more than once

As will become obvious from my discussion of verb raising in Èdó, it would seem that these sentences only bear surface resemblance to SVCs but are actually determined by other facts since there is no evidence for V-I movement at S-structure in English (cf. Pollock 1989, Arnold 1998 and references therein, Roberts 1993 etc.) One account of this difference between Èdó and English with respect to overt V-I movement is discussed in this section.

#### 6.4.2 Verb Raising-to-Infl

In this section, I will argue that the BSC is a constraint on SVCs that is a direct consequence of overt V-to-I movement. Once again, the minimal contrast between AP and VP resultatives with respect to BSC is given in (30):

- (30) a. Àdésúwà dré Òzó kó!kó(-ró) mòsè Adesuwa Foc. Ozo raise-RV beautiful (A) 'Its Adesuwa that Ozo raised to be beautiful.'
  - b. Adésúwa dré Ozó kó!kó(\*-ró) mdsé(\*-ré) Adesuwa Foc. Ozo raise-RV be-beautiful (V) 'Its Adesuwa that Ozo raised to be beautiful.'

I propose that the reason why the main verb in the resultative adjectival SP can bear the past perfective tense (30a) is because it is the only category that can bear such inflection in the structure and so raises overtly to T to check (strong) tense features as illustrated in (32). Recall the structure in (19), repeated here as (31), which represents the underlying structure of the AP resultative.



I assume that verb raising in the AP resultative SP derive from this basic structure of the VP in (31) and proceeds in the manner illustrated in (32).



In this structure in (32) I assume that there is no short verb movement in the sense of Larson (1988). As an alternative, I adopt the approach from the previous chapters in which the top verb binds an empty (lower) verb and this is based on the assumption that what is crucial in the Larsonian approach is that the verb must be before the object at S-structure, and I derive this fact without any movement. This is based on the fact that there is evidence for overt V movement in the Èdó language; when the verb moves at S-structure it must be to Tense (see discussion of example (34f) below). Therefore, since there is no evidence for Larsonian V-to-V raising within the VP, I assume in (32) that the (main) verb undergoes overt verb movement to T to support the -rV affix (or for head-head checking of the tense features realized by the -rV suffix).<sup>15</sup>

<sup>&</sup>lt;sup>15</sup> One relevant question at this point is whether there is ever any evidence for raising to E? According to my analysis, the answer would be no because if a verb raises to the outer Aspect, E then it must get to T.

On the other hand, we have observed that verb raising to T is in principle incompatible with the verbal resultative (SVC) (30b). Consider the underlying structure of the VP in the resultative SVC given in (20) and repeated here as (33).



When the fact about the incompatibility of the -rV suffix with resultative SVCs is linked with this underlying structure of the VP and then contrasted with the analysis of the AP resultative in the structure in (32), the conclusion is that there can be no verb movement (Vto-I raising) in SVCs.

I turn now to the empirical evidence for verb movement to Tense in the resultative adjectival SP but not in the resultative SVC. I propose that such evidence comes from the surface distribution of verbs with respect to I-type adverbs (cf. Pollock (1989), Bowers (1993), Koizumi (1993), etc.). In Èdó, the I-type adverb clearly has a fixed position in syntactic structure as a left-adjunct to the functional head E (see section 2.2.1) where it also inflects for tense tones. I will begin this description of the basic facts of verb raising with a simple sentence illustration as shown in (34):

- (34) a. Èsósà khién èbé Esosa sell book 'Esosa sold the book.'
  - b. Èsósà giégié khién èbé Esosa quickly sell book 'Esosa quickly sold the book.'
  - c. \*Èsósà giégié khiénrén èbé Esosa quickly sell-RV book

- d. èbé <u>ò</u>ré Èsósà khi<u>é</u>nr<u>é</u>n gi<u>é</u>gi<u>é</u> book Foc. Esosa sell-RV quickly 'It is a book that Esosa quickly sold.'
- e. \*èbé òré Èsósà giégié khiénrén book Foc. Esosa quickly sell-RV
- f. \*èbé <u>ò</u>ré Èsósà khi<u>é</u>n gi<u>égié</u> book Foc. Esosa sell quickly

(34a) is an example of a typical transitive verb and I have chosen a transitive verb in order to make a specific statement about what has been previously assumed to be a phonological problem in Èdó (cf. Omoruyi 1991, Agheyisi 1990 etc.).<sup>16</sup> Recall the discussion in section 2.2.1 in which I showed that an adverb cannot occur between a verb and its argument and so, for example, (35) is ungrammatical because the position of the I-type adverb is fixed;

(35) \*Èsósà khiến giếgiế èbé Esosa sell quickly book

This generalization about the position of the I-type adverb being fixed can be further illustrated by the contrast between (34b) and (34f). (34b) shows that the I-type adverb can occur in the position before the verb plus object, while (34f) shows that it can never be after the verb even when the direct object is extracted.

The crucial part of the discussion of verb raising begins with the sentence (34c) in which the verb bears the past perfective tense inflection but the sentence is ungrammatical. Since the minimal contrast between (34b) and (34c) is the presence of the past perfective - rV affix, it appears that the -rV inflection cannot occur on a verb when its direct object is present. However, as (34d) shows the -rV suffix can only be inflected on a verb when its direct object is direct object is clefted and the verb comes to be in a position before the I-type adverb. Omoruyi (1991) analyzes the contrast between (34c) and (34d) based on the proposal that the ungrammaticality of (34c) comes from a phonological incompatibility between the -RV suffix on the verb and direct object. I will argue that this explanation is incorrect and that it

<sup>&</sup>lt;sup>16</sup> This account of verb raising is true for all verbs in Èdó, not only transitive ones. More discussion of this generalization is provided based on the analysis of aspectual verbs in chapter seven.

is at the very best merely descriptive. In fact, if we assume that -rV and -lV (cf. Stewart 1997) occupy the same position then such an analysis will make the wrong prediction that the phonologically similar affix -lV suffix and its nasal counterpart  $-n\underline{e}$  would be ungrammatical when the verb has a direct object and this is contrary to fact (36);

(36) Èsósà khiénné èbé
 Esosa sell book
 'Esosa sold the books'

As an alternative, I propose that the contrast between (34c) and (34d) is evidence for overt verb movement whereby the verb comes to be in a position before the I-type adverb that always occur in a fixed position in E. (34c) is ungrammatical because the verb can only bear the -rV suffix if it has moved to Tense past the adverb, but in this case this has not happened, because the I-type adverb precedes the verb. In comparison with (34d) we observe that the sentence is grammatical because the verb has moved upward to Tense and now bears the -rV inflection. Accordingly, I propose that the reason why the object has to be clefted is not because of any phonological incompatibility but rather because Case cannot be assigned to the direct object via the trace of the verb (cf. Koopman 1992 for similar observations in Bambara, a Mande language spoken in Mali).<sup>17</sup> This explanation by itself already provides confirmation for verb raising and its theoretical relevance in the grammar of Èdó.

(34e) and (34f) provide evidence for two more properties that are associated with verb raising in Èdó. The contrast between (34e) and (34c) shows that verb raising to Tense and object cleft are obligatory when the verb bears the -rV inflection. I propose that the verb cannot move overtly to Tense unless the -rV is there to trigger it. Thus, (34e) is ungrammatical because although the object has moved to avoid Case filter violation, the -rV

<sup>&</sup>lt;sup>17</sup> R-M.. Déchaine (p.c.) observes that since the movement of the object to the Focus position seems to be the thing that permits case-assignment, therefore this predicts that A-bar movement correlates with Case retraction contradicting (42), p. 165. However, I do not think that this is a problem because I assume that wh-traces do not need Case (Borer 1983).

inflection on the verb is not possible because the verb has not moved past the adverb. (34f) shows the other side of the obligatory nature of verb movement to Tense and the surfacing of the -rV inflection and this is that even when the object is clefted and the verb raises to a structural position which is not T but presumably E, the sentence is ungrammatical because the verb must obligatorily raise as far Tense in order to support (or check) the -RV.

Consequently, I conclude that verb raising to Tense is obligatory when the verb bears the -RV inflection. On the basis of this conclusion, I now return to the distinction between AP and VP resultatives. I will begin with a description and analysis of verb raising in the AP resultative SP where there is clearly only one candidate for V-raising. Consider the following:

(37)	a.	À désúwà k	<u>ò</u> ré Òzó	kó!kóròj	gi <u>égié</u> tj t	k mòsèmòsè <sup>18</sup>
		Adesuwa	Foc. Ozo	raise-RV	quickly	beautiful (A)
		'Its Adesuwa that Ozo raised quickly to be beauti				

- b. \* $\dot{A}desuwa_k \dot{Q}rec \dot{Q}zo ko!koj giegie tj t_k mosemose$ Adesuwa Foc. Ozo raise quickly beautiful (A)'Its Adesuwa that Ozo raised quickly to be beautiful'
- c. \*Àdésúwà<sub>k</sub> <u>ò</u>ré Òzó gi<u>égié</u> kó!kóròj tj t<sub>k</sub> mòsèmòsè Adesuwa Foc. Ozo quickly raise-RV beautiful (A) 'Its Adesuwa that Ozo quickly raised to be beautiful'

In (37a) we observe that the verb precedes the I-type adverb that is adjoined to the E position rather than the normal order in which the adverb occurs before the verb. This switch in word order goes along with a morphological tense inflection which is the -RV that marks past perfective. I interpret these facts as the empirical evidence that the verb has raised past the fixed position of the adverb to support the -RV suffix in Tense. (37b) confirms this conclusion based on the fact that the verb can only raise to Tense if the -rV suffix is present and not to any other functional projection above E. (37c) completes the argument concerning the obligatory nature of verb raising in Èdó when -rV is in Tense.

<sup>&</sup>lt;sup>18</sup> Note that mosè and mosèmosè are the same thing (adjectives). Both are related via a fairly regular process of reduplication.

This shows that if a verb bears the morphological features of tense such as the -rV suffix, the only way this can happen is by the verb raising to Tense. Thus, the presence of such inflection implies verb raising to Tense and only Tense. I take all of these facts to be visible manifestation of Chomsky's (1995) ATTRACT which can be descriptively stated as a condition that requires the relevant functional head to 'attract the closest thing'. I formalize the relevant definition of attract as in (38): <sup>19</sup>

### (38) ATTRACT

X attracts Y only if Y could check a feature of X, and all Z such that Z could check a feature of X, Y asymmetrically c-commands  $Z^{20}$ 

In the case at hand, X refers to the functional head Tense that bears the -rV suffix and Y is the first or main verb as the case may be in resultative SVC and AP resultative SP respectively while Z may be the second verb in SVC. (38) provides a very elegant account for the facts about verb raising in the AP resultative SP (37). Verb raising is possible in (37a) based on (38) because the verb is the only candidate that qualify as Y to check the -RV feature on X (Tense). There is no Z to consider in this case because the adjective could not check the tense feature and so the c-command condition on the target of ATTRACT is satisfied vacuously. Therefore, I attribute the presence of the -rV suffix on the verb to the

<sup>&</sup>lt;sup>19</sup> R-M.. Déchaine (p.c.) has observed that this definition of ATTRACT may be at odds with general considerations of economy (e.g. only move if you must). However, this is not quite correct because although it may seem conceptually more economical to attract one verb in the resultative SVC (for example), yet the syntax sees the two verbs as a single head. This explains ATTRACT is defined as in (38). I thankfully acknowledge the contribution of Mark Baker in helping me to work out a suitable definition of ATTRACT.

<sup>&</sup>lt;sup>20</sup> I assume the notion of c-command (as first discussed and defined by Aoun and Sportiche 1983) which is stated as follows;

C-command

A c-commands B iff

<sup>(</sup>a) A does not dominate B and B does not dominate A; and

<sup>(</sup>b) the first maximal projection dominating A also dominates B (i.e., for every maximal projection C, if C dominates A then C dominates B).

Chomsky (1986) calls this M-COMMAND, distinguishing it from another notion of c-command in all categories that contain the "commander" must also contain the "commandee." We can ignore the difference for the most part.

fact that the verb has raised up to Tense. Therefore, I propose the structure in (39) for the AP resultative SP sentence with verb movement.



This structure in (39) accounts for the facts of verb movement in Èdó based on a series of head-movements (cf. Travis 1984) by the verb through functional heads to the Tense projection that is doing the attracting.<sup>21</sup> Once again, the adjective does not count as Z since it cannot check the relevant feature of Tense which is the past perfective that is morphologically realized by the -rV suffix, and also there is no c-command because the A is in a maximal projection.

Let us now compare the foregoing facts and analysis with verb raising possibilities in the resultative SVC based on the definition of ATTRACT in (38). For descriptive clarity two separate sets of data are presented, the first set (40) illustrates the possibility of verb

<sup>&</sup>lt;sup>21</sup> I assume that E cannot attract because it has weak V-features.

raising with the first verb, while the second set (41) shows comparable facts with simultaneous V-raising involving both the first and second verbs:

- (40) a. \*Àdésúwàk <u>ò</u>ré Òzó kó!kóròj giégié tj tk mó!sé
  Adesuwa Foc. Ozo raise-RV quickly be-beautiful (V)
  'Its Adesuwa that Ozo raised quickly to be beautiful.'
  - b. \*Àdésúwàk òré Òzó kó!kój giégié tj tk mó!sé Adesuwa Foc. Ozo raise quickly be-beautiful (V) 'Its Adesuwa that Ozo raised quickly to be beautiful.'
  - c. \* $\dot{A}desuwa_k \underline{\partial}rec \dot{O}zoc giegie kolkoroj tj t_k molse$ Adesuwa Foc. Ozo quickly raise-RV be-beautiful (V)'Its Adesuwa that Ozo raised quickly to be beautiful.'
- (41) a. \*Àdésúwàk <u>à</u>ré Òzó kó!kórðj mó!sérél giégié tj tk tl Adesuwa Foc. Ozo raise-RV be-beautiful-RV (V) quickly 'Its Adesuwa that Ozo raised quickly to be beautiful.'
  - b. \*Àdésúwàk <u>ò</u>ré Ôzó kó!kój mó!sél giégié tj t<sub>k</sub> tl Adesuwa Foc. Ozo raise-RV be-beautiful-RV (V) quickly 'Its Adesuwa that Ozo raised quickly to be beautiful.'
  - c. \*Àdésúwàk òré Òzó giégié kó!kóròj mó!sérél tj tk tl Adesuwa Foc. Ozo quickly raise-RV be-beautiful-RV (V) 'Its Adesuwa that Ozo raised quickly to be beautiful.'

(40a) shows that the first verb of the resultative SVC cannot undergo verb raising to Tense, while (40b) recalls the fact that only the -rV tense can attract and thus trigger verb raising. Thus, (40b) shows that it is ungrammatical for the verb to move past the adverb and adjoin to any other element. Furthermore, we also observe from (40c) that it is ungrammatical for the verb to bear the -rV inflection in its base position (or if it has short-moved to an empty V position in a Larsonian shell). This implies that verb movement to Tense is obligatory when there is a strong feature such as the past perfective morphologically spelled-out by the -rV suffix. Consequently, I conclude that the first verb of the resultative SVC never raises to Tense.

Based on the definition of ATTRACT in (38), I propose that inability of the first verb to undergo verb movement to Tense in the resultative is because there are two potential candidates for ATTRACT; the first verb (Y) and the second verb (Z). According to my analysis of the resultative SVC which is illustrated in (42) both verbs c-command (m-command) each other.



The first verb is realized by a binding chain between an empty verb position and an overtly filled top verb and I assume that both positions count as the same, they are non-distinct. Thus, in (42) the first verb (Y) c-commands the second verb (Z) and vice versa since it is the same first branching node that dominates both V(e) and the V' containing the second verb within the same VP projection. Consequently, Tense (X) cannot ATTRACT the first verb because it does not asymmetrically c-command the second verb, and so verb movement by the first verb is ungrammatical.

A similar analysis accounts for the ungrammatical sentences in (41) where both verbs in the resultative SVC simultaneously move to Tense.<sup>22</sup> As (41) shows, such simultaneous verb movement in the resultative SVC is ungrammatical based on the fact that there is mutual c-command between the two verbs and so neither of them can be the object of ATTRACT. Consequently, neither the first nor second verb of the resultative SVC can bear the -rV inflection.

<sup>&</sup>lt;sup>22</sup> Conceivably, one could attempt to move the second verb past the adverb position before the first verb and have it also move past the first verb. However, this does not make sense since the resulting word order goes against the current of our expectation and lacks any empirical motivation whatsoever cf. \*àdésúwà <u>o</u>ré Òzó mó!séré gié!gié kó!kó.

On the basis of the fact that none of the verbs in the resultative SVC can bear the rV inflection (the BSC (28) that linked to the inability of Tense to attract any of the verbs), I conclude that the serial verb parameter deriving resultative SVCs can be found in a language where the functional projection bearing Tense feature fails to attract. This is formalized as in (44):

(44) Verb-raising serial verb parameter

A verb serializing language is one in which Tense (or other Infl type categories) does not need to be checked, i.e., T has no V-feature.<sup>23</sup>

There are two ways that I propose to illustrate the serial verb parameter in (44). First, one can make minimal comparisons between resultative SVCs in Èdó and resultative V-V compounds in Igbo on the one hand, and then with AP resultative SPs in English. Second, one can examine the issue of V-to-I movement in the consequential SVC and CCs in Èdó.

# 6.4.2.1 Re: Igbo Resultative V-V Compounds

Recall my proposal in section 5.4.1 that Igbo resultative V-V compounds originate from the same underlying structure as the resultative SVC prior to V- incorporation. Furthermore, I proposed along similar lines as in Déchaine (1993) and Manfredi (1991) that the locus of variation between a language such as Èdó with resultative SVCs and Igbo with resultative V-V compounds lies in the two-step process of V-incorporation and subsequent obligatory V-V raising to Tense. I will now show that these two-steps are predicted (and in fact required) by the verb-raising serial verb parameter (44) distinguishing Èdó from Igbo. This is illustrated based on the following Igbo sentences:<sup>24</sup>

 $<sup>^{23}</sup>$  So, ATTRACT is not relevant, even at LF (see text below for discussion of the parameter applied to English).

<sup>&</sup>lt;sup>24</sup> I thank Oyenma M. and Peter Ihionu (p.c.) for providing judgments on these sentences.

- (45) a. \*Obi kwá-ra Ézè da Obi push-RV Eze fall
  - b. \*Obi kwá Ézè da-ra Obi push Eze fall-RV
  - c. \*Obi kwá-ra Ézè da-ra Obi push-RV Eze fall-RV
  - d. \*Obi kwá-da Ézè Obi push-fall Eze
  - e. Obi kwá-da-ra Ézè Obi push-fall-RV Eze 'Obi pushed Eze down'
- (46) a. \*Ó rí-ri afo ju 3s eat-RV stomach be.full
  - b. \*Ó rí afo ju-ru 3s eat stomach be.full-RV
  - c. \*Ó rí-ri afo ju-ru 3s eat-RV stomach be.full-RV
  - d. \*Ó rí-ju afo 3s eat-be.full stomach
  - e. Ó rí-ju-ru afo 3s eat-be.full-RV stomach 'S/he ate [her/his] stomach full.'

The verbs in (45) and (46) combine as sequences of action-result which are generally classified as resultative V-V compounds (cf. 5.4.1). The underlying representation that I have proposed for the Igbo resultative V-V compound prior to V-incorporation is (47).

 $(47) \qquad VP \\ v \\ kwá_k \\ NP \\ Ezè \\ v \\ e_k \\ | \\ v \\ ra$ 

Turning now to the data, observe that the (a) sentences in which only the first verb of the compound bears the -rV suffix are ungrammatical. Having the tense affix on the first verb

implies that it has moved to tense, however such movement by the first verb alone is ruled out by my analysis in which only the uniquely closest verb is attracted. However, as shown in (47) there is mutual c-command relation between the two verbs of the compound since the first maximal projection that dominates the empty V position of the first verb also dominates the second verb and vice versa, so both count as the closest. Consequently, Tense fails to attract only the first verb because it does not asymmetrically c-command the second verb which is a potential candidate to be attracted, and so the sentence is ungrammatical. The same explanation can be extended to account for the ungrammaticality of the (b) sentences in which only the second verb is attracted. Again, according to my analysis the second verb also c-commands the first verb which makes it also a potential candidate for ATTRACT along with the first verb. Therefore, based on the asymmetric ccommand condition on ATTRACT the sentences are ungrammatical because the second verb alone cannot be attracted by Tense excluding the first verb. The ungrammaticality of these sentences where each verb by itself cannot check the Tense feature borne by the -rV suffix implies that V-incorporation is obligatory in order to derive Igbo resultative V-V compounds (Manfredi 1991, Déchaine 1993).

The sentences in (c) provide further evidence that V-incorporation is obligatory. Here, we observe that the -rV suffix is borne by each of the verbs. These sentences are predicted to be ungrammatical because it is assumed that there is a single Tense which can attract. Having two separate inflections on each verb would imply that there are two separate Tense projections and this is contrary to fact.

In the (d) sentences, we notice a step that is crucially impossible in Edó where Vincorporation takes place thereby creating a single complex word, i.e., the V-V compound. The consequence of this V-incorporation is that the c-command condition on ATTRACT is now nullified. I assume a head-movement account of V-incorporation whereby the second verb incorporates to the right of the first verb, an order which I further assume is allowed by the head movement constraint (HMC) (Travis 1984) because movement it internal to a maximal projection and furthermore is forced by the constraints on the ontology of events, resultatives are accomplishments. However, I also assume that the trace of head-movement by the second verb for incorporation is not relevant to c-command because traces have no feature and so are invisible to ATTRACT (Chomsky 1995: chapter 4). The resulting structure from V-incorporation is given in (48).



(48) illustrates the motivation and process of obligatory V-incorporation in the Igbo resultative V-V compound, however as we notice from the ungrammaticality of the (d) sentences this structure is still not fully licensed since the verbal compound has not raised to check the relevant feature of Tense. Based on the grammaticality of the (e) sentences in which V-incorporation has taken place and the verbal compound has raised to Tense, I propose that V-incorporation is a necessary but not sufficient step in deriving resultative V-V compounds. Crucially, the verb must obligatorily raise to Tense as is generally the case in Igbo. This is consistent with the ATTRACT condition (38) because in the (d) sentences, although the c-command condition has been circumvented, the sentences are still ungrammatical because the verbal compounds have failed to raise to Tense.

Consequently, I propose that Igbo is a language in which the strong feature of Tense does ATTRACT (the verb-raising serial verb parameter (44)) and this creates surface V-V compounds in Igbo resultatives, whereas the failure of Tense to ATTRACT creates resultative SVCs in Èdó. As a summary, ATTRACT is defined so that movement of all potential heads capable of checking a given feature is forced. This accounts for the fact that resultatives surface as V-V compounds in Igbo (both Vs must raise), but as SVCs in Èdó (neither V raises). Described this way, R-M Déchaine (p.c.) observes that if the Èdó -rV suffix is located in Tense and forces V-to-T, then one might expect resultative compounds to be possible with -rV morphology in Èdó of the form [---V1-V2-rV], a pattern which is unattested. This is to be expected in Èdó given the discussion and analysis in Chapter two (p. 128) that there are no morphological compounds in Èdó of the form V-V as in Yoruba, i.e., V-V compounds are not licensed at a morphological level in Èdó, i.e., \*[V1-V2]. Therefore, by implication Igbo-type resultative V-V compounds are ruled as potential targets of ATTRACT in Èdó due to the morphological filter that prevents the derivation of such structures in the first place.

# 6.4.2.2 English AP resultative SPs

Turning to English, let us now address the issue of why there are no VP resultatives in English, in order words why English does not serialize? Again, I propose that the answer comes from the parameter in (44) which is the prose version of the ATTRACT condition in (38). The relevant data is given in (49):

- (49) a. John hammered the metal [flat]AP
  - b. John cut the watermelon into pieces.
  - c. John cut the watermelon [\*shred/\*flatten]

The sentences in (49) show that APs or PPs, but not VPs are possible as resultative predicates in English. Limiting my discussion to AP-VP contrast, I assume a unified analysis of all adjectival resultative SPs and propose that (49a) has the same analysis as Èdó AP in (39). This is illustrated for the English example (49a) in (50) where verb raising occurs at LF.



233

According to the structure in (50), there is only one potential candidate to be attracted by Tense and this is the main verb. Thus, the main verb raises to Tense at LF to check the relevant feature as has been standardly assumed (Roberts 1985a/b, Chomsky 1995, Pollock 1989, Koizumi 1993 etc.). However, I assume on a parity between Èdó and English resultative SPs that there is no Larsonian "short verb movement" in this construction. Therefore, since there is no potential Z that can occur along with Y under the same (x) Tense in English, it follows that English is a language in which the weak feature in Tense need not attract and hence it fails to serialize, i.e., have a verb (Z) as the secondary predicate under a single tense. Observe that while the structure in (50) is acceptable, the interesting thing is the ungrammaticality of the other example (49c) where the secondary predicate is a verb. If the verb raises at LF as has been assumed for English, then this creates the same problem that Èdó has at s-structure because both verbs would c-command each other and so none of them can be attracted, therefore there are no resultative SVCs in English. <sup>25</sup>

I would like to conclude this discussion of the V-raising serial verb parameter based on a summary that is presented in the form of a typology of languages.

#### (51) <u>A typology of verb-raising serial verb parameter</u>

SVCs occur if a functional	projection has Tense features that do not need to be
checked. ATTRACT is def	ned as follows;
X attracts Y only if	Y could check a feature of X,
and all Z such that Z	C could check a feature of X,
Y asymmetrically c	commands Z
T with strong V-feature	{French, Igbo, Chinese} <sup>26</sup> no SVC
T with weak V-feature	{English} no SVC
T with no V-feature	{Èdó, Yoruba, Ewe} SVC possible

Concerning this typology in (51), it has been suggested by R-M Déchaine (p.c.) that for languages where T has no feature, the verb-raising parameter predicts that T should be able to merge with any type of projection, e.g. T need not be followed by (an extended projection of) V. According to her, some languages do seem to allow this, e.g. the nominal tense of Athabaskan, Salish and Algonquian. However, I do not agree that this is actually the case since according to Baker and Stewart (1997a), nouns and Adjectives need a *pred* head in order to be properly licensed.

 $<sup>^{25}</sup>$  This is consistent on the surface with the serialization parameter in Collins (1997) that a single Tense can license multiple Vs. However, my proposal is different from Collins (1997) because it is more specific about the exact nature of Tense in SVC languages, based on the connection with Pollock-style verb raising. In fact, Collins (1995) which develops some specific proposals about the relationship between Tense and the verbs proposes the opposite of what I am saying because the verbs would have to raise in his account which is contrary to the facts from verb raising in Edó.

<sup>&</sup>lt;sup>26</sup> Observe that Igbo and Chinese are grouped together with French as languages in which T has strong Vfeature. While this accounts for the fact that Igbo and Mandarin Chinese both have resultative V-V compounding, French is clearly different since it does not have V-V compounding and so the question is why? The simple answer that I will give is that French is like Edó in having a morphological filter (plus, possibly other conditions) that prevents the forming of V-V compound structures.
### 6.4.2.3 V-raising in Consequential SVC and CCs

The goal of this section is to show that the V-raising serial verb parameter also extends in an appropriate way to other consequential SVCs and CCs. In particular, I will show that the serial verb parameter can distinguish between a true SVC (consequential) and a covert coordination (CC). I will begin with the consequential SVC. Consider the following sentences which illustrate verb movement of the first verb:

(52)	a.	Ôzó gèlé lé èvbàré khi <u>é</u> n Ozo truly cook food sell 'Ozo truly cooked the food and sold it.'
	b.	*èvbàré <u>ò</u> ré Òzó lé!ré gé!lé khi <u>é</u> n food Foc. Ozo cook-RV truly sell
	c.	*èvbàré <u>ò</u> ré Òzó lé gé!lé khi <u>é</u> n food Foc. Ozo cook truly sell
	d.	*èvbàré <u>ò</u> ré Òzó gé!lé lé!ré khi <u>é</u> n food Foc. Ozo truly cook-RV sell

(52a) shows that the adverb position before the first verb can be filled. Let us now examine the possibility of moving the first verb, whereby it comes to precede the adverb and bear the -rV inflection. (52b) shows that the first verb cannot occur before the adverb and bear the -rV inflection. I take this as evidence that the first verb cannot raise to Tense. (52c) examines the possibility of a Larsonian short verb movement. However, such movement is not empirically motivated as the resulting sentence (52c) with this order of verb before adverb is ungrammatical. (52d) confirms the proposal that the presence of a morphological tense inflection on the verb implies verb movement to Tense; it is ungrammatical for the first verb to bear the -rV inflection in its base position.

The facts just discussed about the first verb in (52) consistently re-enforce the predictions of the V-raising serial verb parameter. In order to illustrate this point let us consider the structure of the consequential SVC for the sentence (52a) as shown in (53).



In this structure in (53), I have written the I-type adverbs under their respective E rather than adjoining them to E simply for the convenience of presentation. Now, based on this structure with respect to the ATTRACT condition in (38) I assume a very strict interpretation of c-command which refers to the lowest segment of the first maximal projection (I have in mind the category/segment distinction in May (1985)). Thus, if we analyze the ATTRACT condition such that X is Tense and Y is the first verb while Z is the second verb, we will observe from the structure in (53) that Y does not asymmetrically ccommand Z since the lowest segment of the first maximal projection dominating the first verb does not dominate the VP2 that Z is contained in. It follows, therefore, that Tense cannot attract the first verb because it does not qualify as the closest element with the desired features. This analysis extends in a straight-forward manner to capture similar facts of verb movement by the second verb as illustrated by the data in (54):

(54) a. \*èvbàré òré Òzó lé khién!rén giégié food Foc. Ozo cook sell-RV quickly

- b. \*èvbàré òré Òzó lé khién giégié food Foc. Ozo cook sell-RV quickly
- b. \*èvbàré <u>ò</u>ré Òzó lé giégié khiénrén food Foc. Ozo cook quickly sell-RV

As the sentences in (54) show, it is ungrammatical for the second verb to move past the adverb that normally occurs before it. It can be assumed that these sentences are simply ungrammatical because although there is an adverb position before the second verb there is no lower Tense position structurally that can trigger verb movement (in contrast to covert coordinations below).<sup>27</sup> In fact, this is precisely the point about the ATTRACT condition on SVCs because it is based on the strength of a single Tense head that dominates and can potentially attract two verbs to check its features; neither of which is the uniquely closer verb in an SVC.

Perhaps the best way to illustrate the V-raising serial verb parameter is to show what happens when both verbs bear the -rV inflection in the consequential SVC. This is given in (55):

(55) Ozó giégié lé èvbàré khién a. Ozo quickly cook food sell 'Ozo quickly cooked the food and sold it' \*èvbàré <u>ò</u>ré Òzó lé!ré b. khiénrén giégié food Foc. Ozo cook-RV sell-RV quickly \*èvbàré òré Ózó lé!ré giégié khiénrén c. food Foc. Ozo cook-RV quickly sell-RV

(55a) illustrates the fact that there is an adverb position immediately after Tense and it ccommands and has scope over the two verbs. Therefore, compare with (55b) in which both verbs have moved past the adverb and now bear -rV inflections indicating that they have both moved to Tense. The fact that this sentence is ungrammatical is consistent with the

237

<sup>&</sup>lt;sup>27</sup> Concerning the second verb, the ATTRACT story rules out the ridiculous sentence in (i) (i) \*èvbàrék <u>d</u>ré  $\dot{O}z\delta$  khién-rén; gél!é lé t<sub>k</sub> t<sub>j</sub>

where the second verb raises past the first verb plus adverb to Tense.

BSC (28) and the serial verb parameter (44). Based on the analysis of the consequential SVC, neither verb c-commands the other and so neither of them can be attracted by Tense and so they cannot inflect for the -rV suffix (55b) or in the reverse order (55c).

On the basis of the foregoing, I will conclude that resultative SVCs and consequential SVCs illustrate two opposite ways that ATTRACT can fail to apply. Whereas the two verbs in the resultative mutually c-command each and so neither could count as the closest thing for ATTRACT, on the other hand neither of the verbs in the consequential SVC c-command the other and so Tense fails to attract any of them.

Let us now turn to the CC where we will observe a striking contrast with object sharing SVCs. Consider the following:

(56)	а.	Ózó gi <u>é!gié</u> gb <u>òó</u> ívìn gi <u>é</u> !gié bòló <u>ó</u> kà Ozo quickly plant coconut quickly peel corn 'Ozo quickly planted coconut and quickly peeled corn.'
	b.	ívìn <u>ò</u> ré Òzó gb <u>óó</u> !ré gi <u>é</u> !gi <u>é</u> bòló <u>ó</u> kà coconut Foc. Ozo plant-RV quickly peel corn 'It's coconut that Ozo planted quickly and peeled corn.'
	c.	<u>ókà ò</u> ré Òzó gb <u>òó</u> ívìn bó!lóró gi <u>é</u> !gi <u>é</u> corn Foc. Ozo plant coconut peel-RV quickly 'It's corn that Ozo planted coconut and peeled quickly.'

(56a) illustrates the already established fact that adverbs of the same kind can occur before each verb. This was taken (amongst other factors) to imply that CCs involve the conjunction of two separate and symmetric EPs. However, evidence from verb raising indicate that the level of adjunction may actually be higher and should in fact have to do with Tense (cf. Collins 1997). According to the data above, either the first verb or the second verb can undergo V-raising. Based on my analysis, the presence of the -rV suffix on a verb implies that the verb must have raised to Tense. Consequently, I propose the structure of CCs as in (57):



According to the structure in (57) CCs involve the conjunction of T's and this allows each of the verbs to be separately attracted by the different Tense heads.<sup>28</sup> The implication of this analysis is that CCs are quite different from object sharing SVCs.

The cumulative evidence provided in this thesis should resolve the going back and forth on whether to call covert coordination constructions a type of SVC due to the lack of systematic definition of SVCs. The contrast between CCs and object sharing SVCs with respect to V-raising is striking and so in the light of the serial verb parameter (44) I conclude that CCs are simply covert T' coordinations and not SVCs in any more substantive sense.

# 6.4.2.4 Consequence of V-raising Analysis

The immediate consequence of the V-raising serial verb parameter is that it accounts for a long standing stipulation in SVCs namely; that there is only a single negation for the chain of verbs (cf. Bamgbose 1973, 1986). There are inconsistent analyses of this fact which attempt to equate the possibilities of interpretations from scope and presupposition

<sup>&</sup>lt;sup>28</sup> For simplicity, I have represented the I-type adverbs as E-head (rather than adjuncts).

with syntactic structures of SVCs (cf. Manfredi 1987, Bamgbose 1986 etc.). I illustrate my point by first introducing a basic analysis of negation in Èdó:

- (58) a. Òzó dùnwmún ìyán Ozo pound yam 'Ozo pounded yam.'
  - Òzó ghá dún!wmún ìyán
     Ozo Fut. pound yam
     'Ozo will pound yam.'
  - Ozó í dún!wmún ìyán
     Ozo neg. pound yam
     Ozo will not pound yam.'
  - d. \*Òzó ghá í dún!wmún ìyán Ozo Fut. neg. pound yam
  - Òzó má dún!mwún ìyán
     Ozo neg. pound yam
     'Ozo did not pound yam.'
  - f. \*Òzó má dùnmwún ìyán Ozo neg. pound yam

(58a) illustrates an ordinary disyllabic transitive verb and the fact that simple past tense is indicated by the low-high sequence on the verb. However, in (58b) tense is morphologically realized by the future tense morpheme  $gh\dot{a}$  and we observe that the tones on the verb change to a high-downstep-high sequence (there is a fuller discussion of this tone behavior in chapter seven). I propose an analysis in which both the simple past tense tones and future tense particle are generated in the Tense head as shown in the simplified structure in (59) (here EP and VoiceP are omitted).

(59) TP Т Spec Т VP ghá dunmwun ìyán

According to this analysis verbs have no lexical tones and only get their tonal value based on the nature of Tense. In the relevant example which is (58b), Tense is morphologically realized by a morpheme bearing a high tone which is then copied onto the verb. Now, observe that exactly the same thing happens in (58c) in which a negation morpheme occur before the verb. I propose that the fact that both future and negation morphemes are found in the same position and trigger the same tone pattern on the verb is not an accident; rather it is evidence that they are both generated in the same position, which is the Tense head. This proposal allows us to make the general descriptive statement that when the Tense functional head is overtly (morphologically) filled, it triggers relative tones on the verb. The ungrammaticality of (58d) provides further empirical evidence for the proposal that both the negative morpheme and the future morpheme occur in the same position, i.e., Tense. The ungrammaticality arises from the fact that there is competition for the same position. The implication of this analysis is that negation is not a separate head in Edó but rather is the overt realization of the Tense head. This conclusion is supported by the fact that there are two negation morphemes in Edó; i which is a non-past tense negation and má which is past tense negation.

The past tense negation is illustrated in (58e) and we can observe again that a filled Tense head triggers high-downstep-high tones on the verb. (58f) is ungrammatical because of the fact that although Tense is filled by the past tense negation the associated tone changes on the verb do not occur. The reason for the ungrammaticality of (58f) is because there is a clash between the tones triggered by past tense negation in Tense and the normal low-high tones associated with simple past tense on the verb. Consequently, I conclude that negation is not a separate head from Tense in Èdó and so the potential fillers of Tense should be expanded in (59) to include these facts about negation. This is illustrated in (60).

241



This analysis of negation in Èdó makes one prediction for the account of verb movement that I will now examine. Based on the analysis of (58d) and (58f) in which we assume that there is a competition for the same Tense head, we equally predict that in a sentence in which Tense is overtly (morphologically) filled with an independent particle it will be ungrammatical for verb movement to occur. This prediction is based on the simple logic that, for example, future tense and non-past tense negation morphemes cannot co-occur (58d). This implies that only one morpheme can occur in the Tense head. Consider the following sentences which illustrate the interaction of verb movement to Tense with the overtly (morphologically) filled Tense head:

- (61) a. Ozó giégié lé èvbàré Ozo quickly cook food 'Ozo quickly cooked food.'
  - b. èvbàré <u>ò</u>ré Òzó lé!ré gi<u>égié</u> food Foc. Ozo cook-RV quickly 'It is food that Ozo cooked quickly.'
  - Ozó má giégié (\*má) lé èvbàré
     Ozo neg. quickly cook food
     'Ozo did not quickly cook food.'
  - d. \*èvbàré <u>ò</u>ré Òzó má lé!ré gi<u>égié</u> food Foc. Ozo neg cook-RV quickly

As the data above shows, we get a very nice confirmation for the prediction that is based on the analysis of Tense, verb movement and negation. (61a,c) both illustrate the ordering relation between the I-type adverb and negation and we confirm the structural analysis in which negation is generated in Tense higher than EP. (61b) is included as a control sentence showing that verb movement to Tense is grammatical if it contains only a single morphological feature, in this case the -rV suffix. However, observe in (61d) that it is ungrammatical to have verb movement to Tense when both past tense negation and the past perfective suffix are present. The basic argument here is that there is a competition between the negation and the -rV suffix, which are two possible instantiations of a single Tense head. They cannot co-occur, so there can be no trigger for verb movement when negation is present.

Based on the foregoing analysis of negation, it is predicted that there can be only one negation morpheme in both resultative and consequential SVCs because there is only a single Tense head. However, two negation morphemes are predicted for the covert coordination structure because there are two separate Tense heads. These predictions are indeed borne out, as shown in the following sentences:

- (62) a. Ozó má kó!kó àdésúwà mó!sé Ozo neg. raise Adesuwa be-beautiful 'Ozo did not raise Adesuwa to be beautiful.'
  - b. \*Òzó kó!kó àdésúwà má mó!sé Ozo raise Adesuwa neg. be-beautiful
- (63) a. Òzó má kó!kó ìyán dún!mwún Ozo neg gather yam pound 'Ozo did not gather the yam and pound it.'
  - b. \*Òzó kó!kó ìyán má dún!mwún Ozo gather yam neg. pound
- (64) a. Òzó má gb<u>ó</u>!<u>ó</u> ívìn bó!ló <u>ó</u>kà
   Ozo neg plant coconut peel corn
   'Ozo did not plant coconut and peel corn.'
  - b. Özó má gb<u>ó!ó</u> ívìn, bòló <u>ó</u>kà
     Ozo neg plant coconut peel corn
     'Ozo did not plant coconut and (he) peeled corn.'
  - c. Ózó gb<u>òó</u> ívìn má bó!ló <u>ó</u>kà Ozo plant coconut neg. peel corn 'Ozo planted coconut and did not peel corn.'

In the resultative SVC (62), we observe that negation can only occur before the two verbs (62a), not before the second verb (62b). This contrast is consistent with my analysis of the resultative SVC in which a single Tense head dominates both verbs, so that in (62a) the past tense negation is in the Tense head and has scope over both verbs. (62b) is ungrammatical because there is no Tense head projection before the second verb. A striking fact which is consistent with my analysis of overtly filled functional heads is the fact that in the grammatical sentence in (62a) relative tones occur on both the first and second verbs. I propose that this is a phonological manifestation of the scope of negation. Thus, in the resultative SVC both verbs inherit the high tones from negation because they are in the scope domain of a single Tense head. In this way, I have now provided an account for the observation that there is only a single negation for both verbs based on the nature of the Tense head that is central to the serial verb parameter

Similarly, we observe in the consequential (63) that the past tense negation can only occur before the two verbs (63a), not between them (63b). Again, this is structurally accounted for based on the proposal that there is only a single Tense head which c-commands both verbs. In (63a), the past tense negation occurs in this Tense position and takes scope over both verbs. Here also, observe the fact that in the grammatical sentence (63a) both verbs tonally inflect for the high tones from the overtly filled Tense head. Again, this is consistent with the parameter deriving serial verbs which proposes that the two verbs are in the domain of, and in this case inherit the features from, a single Tense head.

The CC data is the most interesting because it presents yet another difference between these kinds of sentences and object sharing SVCs. Observe from (64a,b) that the past tense negation can occur before both verbs. In this position, it either takes scope over both of them (64a) or it takes scope over the first verb only (64b). This ambiguity in the scope interpretation of negation is to be expected in cases of real symmetrical conjunction (cf. Déchaine 1993). What is interesting, however, is the empirical observation that the difference in the scope interpretation of negation in this position matches the tones on the

244

verbs. Thus, in (64a) in which negation scopes over both verbs we notice that the two verbs have the same high-downstep-high tone patterns. In contrast, in (64b) in which negation only scopes over the first verb we observe the high-downstep-high tone pattern only on the first verb, but not on the second.

An even more striking contrast from covert coordination sentences comes from (64c) in which we observe that negation can occur before the second verb unlike in resultative and consequential SVCs. This is consistent with my analysis which proposes that covert coordinations involve conjunction at some level of the Tense projection. In particular, each verbal projection can be dominated by a separate Tense head. It follows, therefore, under the analysis in which negation occupies Tense that the negative morpheme can occur before the second verb in covert coordinations, since we already know that there is such a structural position present based on the facts from V-raising.

### 6.5 'Rule R' and Tense Matching in SVCs

The general theme of section 6.4.2 has been that V-raising to Tense is never possible in SVCs and this was formalized as the serial verb parameter in (44), repeated here as (65).

### (65) Verb-raising serial verb parameter

A verb serializing language is one in which Tense (or other Infl type categories) does not need to be overtly checked. ATTRACT is defined as follow; X attracts Y only if Y could check a feature of X, and all Z such that Z could check a feature of X, Y asymmetrically c-commands Z

The question to be addressed in this section, then, is how the verbs in SVCs are licensed otherwise since they do not raise to Tense to check features even at LF. I propose something similar to Rule R in Chomsky (1981) which takes the essential details of ATTRACT and allows tense tone generated in T to float down to the verbs. This is formalized as the tense copying/matching condition in (66).

### (66) Tense matching condition

copy the tone feature on X to Y only if X c-commands Y and there is no other tone-bearing element Z such that Z c-commands Y and X c-commands Z

By the condition in (66) we are able to explain in a straightforward manner the facts of tense tone matching in SVCs (cf. Stewart 1996). Consider the following data from the previous section:

(67)	а.	Òzó kòkó ìyán dùnmwún Ozo gather yam pound 'Ozo gathered the yam and pounded it.'	(past tense tones)
	b.	Òzó kòkò ìyán dùnmwùn Ozo gather yam pound 'Ozo gathers the yam and pounds it.'	(habitual tense tones)
(68)	а.	Òzó gb <u>òó</u> ívìn bòló <u>ó</u> kà Ozo plant coconut peel corn 'Ozo planted coconut and peeled corn.'	(past tense tones)
	b.	Òzó gb <u>òò</u> ívìn bòló <u>ó</u> kà Ozo plant coconut peel corn 'Ozo plants coconut and peeled corn.'	(habitual+past tense tones)

I have taken the consequential SVC (67) as the representative example for object sharing SVCs, but the same analysis generalizes over to the resultative SVC. According to the tone matching condition (66), X refers to Tense, and Y can refer to either verb in the SVC. Therefore, the tense feature that is realized tonally is generated in the Tense head as a floating affix in the case of simple past in (67a) and is then copied onto the verbs. I assume, therefore, that the ban on V-raising is a general fact of SVCs and so by the lack of V-raising-the serial verb parameter, the tone matching condition kicks in as a supplement derived from the failure of ATTRACT.

However, in the covert coordination (68) in which there are two projections of Tense (X) the same process of tone copying apply within the domain of each Tense projection and so we get a combination of a habitual tense projection being combined with another event that is in the past; both actions performed by the same subject in Spec, TP.

One specific consequence of (66) is that we can derive the tonal differences between AP and VP resultatives in a manner that is consistent with the analysis of the serial verb parameter. Recall the previous claim that verbs do not have inherent tones while adjectives have inherent lexically specified tones. Therefore, based on (66) we are able to delineate AP resultatives from VP resultatives on language internal grounds.<sup>29</sup> Now, consider the following sentences:

- (69) a. Òzó kòkó àdésúwà mòsé
   Ozo raise Adesuwa be-beautiful
   'Ozo raised Adesuwa to be beautiful.'
   b. Òzó kòkó àdésúwà mòsè
  - Òzó kòkó àdésúwà mòsè
     Ozo raisc Adesuwa beautiful
     'Ozo raised Adesuwa to be beautiful.'

We observe a clear tonal contrast between the resultative SVC (69a) and the resultative adjectival SP (69b). Whereas the tones on both verbs match in the resultative SVC (69a) there are no matching tones between the verb and the adjective in AP resultative SP (69b). This contrast can be accounted for based on the tone matching condition in (66) such that both verbs of the resultative SVC are legitimate targets for the tone-copying rule. However adjectives are not, because of their inherent tones.

## 6.6 Conclusion

This chapter proposed a serial verb parameter and this is shown to capture the systematic differences between Èdó, Igbo, and English. It is argued that the need to check features of Infl triggers obligatory verb movement in non-SVC languages. In SVC languages, Infl features do not have to be checked by the verb because of a condition on tense copying which can be likened to Rule R of Chomsky (1981). Although the primary basis for the cross-linguistic discussion was limited to AP versus VP resultative contrast, it

<sup>&</sup>lt;sup>29</sup> These same observations can also be verified from the difference in inflectional morphology between verbs and adjectives in English; verbs but not adjectives inflect for tense

was shown that the parameter also distinguishes between verb sequencing constructions, language-internally. Furthermore, it is derived from the parameter an elegant consequence for the analysis of negation which has been traditionally stipulated (cf. Bamgbose 1974, 1986, etc.).

#### **Chapter Seven**

#### **Re-analysis of 'Serial Verb Constructions'**

### 7.1 Introduction

This chapter will show that some constructions that have been called serial verb constructions in the literature exhibit properties that are quite different from those that we have observed with resultative, consequential, or covert coordinations. Thus, *re-analysis* is used ambiguously to imply a synchronic analysis of cases of lexical and structural re-analysis, and on the other hand it further implicates a new way to assess (re-analyze) what used to be classified as SVCs. Consider the following sentences:

- a. Òzó hìá lé èvbàré
   Ozo try cook food
   'Ozo managed to cook the food.'
  - b. Òzó miànmián kié <u>èkhù</u>
     Ozo forget open door
     'Ozo opened the door inadvertently.'
- (2) a. Ôzó yá ábé fián èmió!wó
   Ozo use knife cut meat
   'Ozo cut the meat with the knife.'
  - b. Òzó rhié úghánmwàn ghu <u>òghó</u> ówá Ozo take axe break stall 'Ozo broke the stall with an axe.'

The sentences in (1) contain two verbs in what has been wrongly analyzed as a single clause in which the first verb is restricted to the class of 'modal-aspectual' verbs and the second verb is relatively unrestricted. These sentences have been classified as SVCs because there are two verbs on the surface and there is no overt marker of subordination or coordination (cf. Oyelaran 1982, George 1975, 1976, Agheyisi 1986 etc.). The sentences in (2) exemplify what are generally classified as Instrumental SVCs, in which the first verb is always from 'a closed class' (cf. Lefebvre 1991) and whose object is interpreted as the

instrument that is used to accomplish the event referred to by the second verb (cf. Lefebvre 1991, Baker 1989, 1991, Collins 1997 etc.).

In the first part of this chapter, I will argue for a *restructuring* account for the sentences of the kind in (1), in which the second verb that is in an embedded infinitival clause undergoes re-analysis, resulting in a unification with the matrix verb. This restructuring and Control analysis is proposed on the basis of the following considerations: (a) certain aspectual alternations; (b) their behavior with respect to the syntactic tests from the previous chapters ; (c) the serial verb parameter from chapter six. These same three kinds of evidence will also form the basis for the proposal in the second half of this chapter that there has been a *re-analysis* of the so-called instrumental SVCs, in particular the first verb is underspecified for all features except tense and theta role assignment. I will argue that sentences of the kind in (2) are in fact cases of clausal IP complementation.

## 7.2 The Modal-aspectual Verb Construction

Descriptively, I propose to begin this section by examining two aspects of the sentences in (1) and then contrast these properties with resultative, consequential and covert coordinations. The two relevant aspects are the following;

(a) the possibility of putting INFL morphemes between the verbs and whether this correlates with a difference in meaning

(b) the nature of the syntactic restrictions on the first verb.

The need to examine both of these aspects follows from the observations in the previous chapters that there are syntactic as well as semantic restrictions that regulate the relations between the verbs in SVCs. For example, resultative SVCs express a single event and the second verb must be unaccusative, while both verbs in the consequential SVC must be transitive and they also express linked events that are existentially bound by a single operator head E. Therefore, it will be a useful introduction to the Modal-aspectual verb

construction just to come to terms with the nature of syntactic and semantic characteristics of this particular construction.

### 7.2.1 An Apparent Contrast with SVCs: Infl vs. E projection

In this section, I will show that there is a systematic correspondence between the covert presence of INFL and its overt manifestation in the Modal-aspectual verb construction. Furthermore, I will argue that this alternation correlates with either a realis or irrealis interpretation of the event denoted by the second verb. By comparison, I will also show that this sort of alternation in Mood is distinctly not a property of resultative and consequential SVCs nor covert coordinations. Consider the following:

- (3) a. Ozó hìá rrí èvbàré
   Ozo try eat food
   'Ozo made effort and ate the food.'
  - Ozó hìá yá rrí èvbàré
     Ozo try INFL eat food
     'Ozo tried to eat the food.'
- (4) a. Òzó miànmián kpó!ló ówá
   Ozo forget sweep stall
   'Ozo forgot and swept the stall.'
  - b. Özó miànmián yá kpólló ówá Ozo forget INFL sweep stall 'Ozo forgot to sweep the stall.'

Notice the contrast in meaning between the (a) and (b) sentences in (3) and (4). In the (a) sentences there is no morpheme between the first and second verb and only a realis reading is possible for the second verb. Thus, (3a) implies that Òzó ate the food and (4a) implies that he swept the stall.

By simply looking at only these (a) sentences and their interpretations, we get some insight as to why they have been assumed to be SVCs. It has been proposed that the event(s) denoted by the verbs in SVCs are always asserted (cf. Sebba 1987 *et al*) as shown, for example, by the resultative and consequential SVCs in (5):

- (5) a. Ozó kòkó àdésúwà mòsé
   Ozo raise Adesuwa be.beautiful
   'Ozo raised Adesuwa to be beautiful.'
  - b. Ózó lé èvbàré ré
     Ozo cook food eat
     'Ozo cooked the food and ate it.'

In the resultative SVC (5a), it is true that Ozo raised Adesuwa and she (indeed) became beautiful; in the consequential (5b) it is true that  $\dot{O}z\dot{o}$  cooked the food and that he also ate it. Therefore, by simply comparing the (a) sentences in the Modal-aspectual verb construction (3)-(4) with the serial verb construction (5) we may be inclined to assume that they are both SVCs, but this is contrary to fact.

One evidence for the difference between Modal-aspectual verb constructions and SVCs is based on a comparison between the (a) sentences with the (b) sentences in (3)-(4). Observe in these (b) sentences that there is an additional morpheme between the two verbs and the meaning of the sentences is affected. In particular, the second verb now has an irrealis reading. For example, in (3b) the interpretation is that  $\dot{O}zo$  tried to eat the food, (but there is no implication that he succeeded). Similarly, in (4b) the interpretation is that Ozo forgot to sweep the stall; we do not know that he did (probably he did not, in fact). Clearly, the difference between the (a) and (b) sentences in (3-4) should be attributed to the presence of the morpheme *yá* between the verbs in the latter, and it is this that affects the modal interpretation of the meaning of the second verb to become irrealis.

This contrast between the (a) and (b) sentences in (3)-(4) points to a striking contrast between the Modal-aspectual verb constructions (3-4) and resultative-consequential SVCs (5). It is not possible to have this  $y\dot{a}$  morpheme occur before the second verb in SVCs as shown in (6). (This time I will include covert coordinations in the paradigm just to show that the Modal-aspectual verb construction is different).

(6) a. \*Ôzó kòkó àdésúwà yá mòsé 'resultative SVC'
 Ozo raise Adesuwa INFL be.beautiful
 'Ozo raised Adesuwa to be beautiful.'

b.	*Òzó lé èvbàré yá ré	'consequential SVC'
	Ozo cook food INFL eat	•
	'Ozo cooked the food to eat it.'	

c. \*Òzó hìín èrhán yá kpàán ívìn 'covert coordination'
 Ozo climb tree INFL pluck coconut
 'Ozo climb the tree to pluck the coconut.'

As the sentences above show, resultative-consequential SVCs and covert coordinations are very different from Modal-aspectual verb constructions in that they consistently do not allow the INFL particle  $y\dot{a}$  to occur before the second verb as shown in (6).<sup>1</sup> Therefore, the impossibility of  $y\dot{a}$  before the second verb can be taken as one piece of evidence pointing to the difference between superficially similar serial verb constructions. The emerging questions at this point are, what exactly is the  $y\dot{a}$  morpheme that I have glossed as INFL? what does its presence tell us about the syntactic structure of the Modal-aspectual verb construction?

## 7.2.2. A Separate Infl before Second Verb

The morphology of the infinitive construction may betray its origins in another grammatical category. For example, in Igbo the infinitive displays nominal and verbal properties as illustrated by the example in (7) from Emenanjo (1981).

As shown in (7), the infinitive marker in Igbo is a prefix on the verb+object. The prefixal nature of the infinitive makes it look like a nominalizing prefix with the VO order (cf. Manfredi, 1997). Clearly, the Èdó yá is not like the Igbo infinitive in this respect. The nominalization of a VP (verb+object) in Èdó is achieved via vowel prefixation (recall the discussion of predicate clefts). However, notice that the Èdó yá is a separate word and it

<sup>&</sup>lt;sup>1</sup> This is consistent with the traditional definition of SVCs as involving two or more verbs that occur without any marker of subordination or coordination (cf. Bamgbose 1974, Awobuluyi 1973, Lord 1973, Stahlke 1970, 1974, Schachter 1974 etc.).

does not obey the morpheme structure constraints on Èdó nominals because it does not begin with a vowel prefix.<sup>2</sup> Therefore, in order to adequately analyze the  $y\dot{a}$  morpheme I will now examine some of its peculiar properties.

## 7.2.3 yá as Evidence for Embedded Clause.

One useful piece of evidence for the analysis of  $y\dot{a}$  comes from a restriction on its distribution: it occurs in (special) fixed position with different verbs. Let us, first, consider the distribution of  $y\dot{a}$  in a simple clause with modal verbs (8) and compare with the full Modal-aspectual verb construction (9):

- (8) a. \*Òzó yá hí!á Ozo INFL try
  - b. \*Òzó yá mián!mián Ozo INFL forget
  - c. \*Òzó yá rhi<u>ó</u> Ozo INFL wake up early
- (9) a. Òzó hìá yá kí!é èkhù Ozo try INFL open door 'Ozo tried to open the door.'
  - Òzó miànmián yá kí!é èkhù
     Ozo forget INFL open door
     'Ozo forgot to open the door.'
  - c. Òzó rhió yá kí!é <u>è</u>khù Ozo wake up early INFL open door 'Ozo woke up early to open the door.'

From the sentences in (8) we observe that the morpheme  $y\dot{a}$  cannot occur between the subject and verb in a simple clause with an aspectual verb. Since the position between the subject and verb is an INFL (Infl) position by definition, so, on the basis of the ungrammatical sentences in (8) we may conclude that  $y\dot{a}$  cannot occur in the INFL position

<sup>&</sup>lt;sup>2</sup> In fact, yá cannot even be nominalized and so could not become a noun by derivational morphology.

of a simple clause. In other words,  $y\dot{a}$  does not occur in a tensed clause with an aspectual verb.

However, we observe from the sentences in (9) that  $y\dot{a}$  can occur after the modal verb, before the second verb in the Modal-aspectual verb construction. Now, if we take the proposal that  $y\dot{a}$  does not occur in a tensed clause seriously we are led to conclude that the second verb in (9) is in a separate and non-tensed clause, an embedded clause. Consequently, if the second verb is in an embedded clause we draw a comparison with the fact that there is no overt subject argument present in these embedded clauses and similar infinitival or gerundive complements in English which must license null subject PRO given standard assumptions (Chomsky 1981, Williams 1980 etc.). Therefore, I tentatively conclude that  $y\dot{a}$  is the head of an embedded IP and so I propose that the Modal-aspectual verb construction involves clausal subordination as shown in (10) for (9a).<sup>3</sup>



<sup>&</sup>lt;sup>3</sup> Based on evidence from the distribution of tobére particle in section 7.3.3.1, I will argue that the PRO subject of the embedded clause is base-generated in the Spec of VoiceP, this is comparable to the VP internal subject analysis of Sportiche (1988). At this point I will simply assume that PRO stays in this position, I will come back to this issue in section 7.3.3.1.

I will now present a second evidence that confirms this analysis of  $y\dot{a}$  and this is based on its distribution in simple clauses with non-aspectual verbs. I will illustrate this point with both unergative and transitive verbs to show that  $y\dot{a}$  heads an embedded infinitival Infl even with these verbs. Consider the following:

- (11) a. Òzó yá ló!vbi<u>é</u>
   Ozo INFL sleep
   'Ozo went to lie down.'
  - Òzó yá kí!é <u>è</u>khù
     Ozo INFL open door
     'Ozo went to open the door.'

In the sentences in (11), notice that  $y\dot{a}$  occurs before the main verb in contrast to the sentences in (8). Therefore, (11) appears to contradict the generalization that  $y\dot{a}$  can only occur in embedded clauses. However, I will now show that what is actually happening in (11) is that  $y\dot{a}$  is a morphologically complex form in these cases. Consider the following contrast:

- (12) a. Òzó yá kí!é èkhù
   Ozo INFL open door
   'Ozo went to open the door'
  - Òzó yó yá kí!é ékhù
     Ozo go INFL open door
     'Ozo went to open the door'

(12a) is similar to (12b) in its interpretation. In both cases what we know is that Ozo went to do something, i.e., to open the door, but there is no implication that the door was actually opened. I propose that the semantic similarities between (12a) and (12b) are a reflection of the fact that there is an underlying structure for (12a) which has a null verb meaning 'go'; this brings about the sense of motion evident from the translations. More specifically, I propose that (12a) is derived from the sentence in (12b) that actually contains a preceding motion verb  $y\phi$  "go" and it is this verb that selects the infinitive complement headed by  $y\dot{a}$ . What we see on the surface in (12a) is the result of a phonological process of coalescence involving  $y\dot{a}$  and  $y\dot{a}$  that results in  $y\dot{a}$ .<sup>4</sup>

Two pieces of evidence support this analysis that  $y\dot{a}$  is morphologically complex before the verb in a seemingly simple clause as in (12). First, observe that  $y\dot{a}$  cannot occur in the matrix clause where  $y\dot{a}$  "go" is phonologically spelled-out. This contrast is shown in (13):

- (13) a. Òzó (\*yá) yó yá ló!vbiệ Ozo INFL go INFL sleep 'Ozo went to lie down.'
  - Òzó (\*yá) yó yá kí!é ékhù
     Ozo INFL go INFL open door
     'Ozo went to open the door.'

(13) shows that when the morphologically complex form  $y\dot{a}$  is separately realized by the combination of the motion verb  $y\dot{a}$  and the INFL particle  $y\dot{a}$ , then it is ungrammatical for the matrix INFL to also be filled by a separate  $y\dot{a}$  where the  $y\dot{a}$  before  $y\dot{a}$  is assumed to be in the matrix INFL. The impossibility of having  $y\dot{a}$  occur before the motion verb is of course consistent with the generalization in (10) that  $y\dot{a}$  can only occur in an embedded clause.

Second, the proposal that  $y\dot{a}$  is generated in an embedded Infl makes a testable prediction based on the analysis from chapter six in which negation morphemes are seen as the realization of tense, generated in the matrix T node. Thus, consider the sentences in (14):

(14) a. Òzó má yó Ozo not go 'Ozo did not go.'

<sup>&</sup>lt;sup>4</sup> Elision of [+ sonorant] sounds is a very common phonological phenomenon in Èdó (cf. Amayo 1976, Aikhionbare 1989 etc.). This is especially obvious in any word boundary context where nouns are being combined since all nouns must begin with a vowel and all words must end in a vowel. The most general rule for this sort of elision can be stated as; V(owel)1 + V(owel)2 = V2. It is fairly common to hear the form "yoa" in the speech of some Èdó speakers. In fact, this form is used in Omoregie (1983) and by other Edo authors. When used this way, the phonological process is less opaque and is indeed consistent with glide deletion rules, since glides are also [+sonorant].

- b. \*Òzó hìá má kí!é èkhù Ozo try not open door
- c. \*Òzó hìá má yá kí!é èkhù Ozo try not INFL open door
- d. Ôzó má hí!á yá kí!é <u>è</u>khù
   Ozo not try INFL open door
   'Ozo did not try to open the door.'

In (14a), we observe that the negation morpheme occurs before the verb in the Tense position. In (14b), the negation morpheme occurs before the second verb that I have assumed is dominated by a null headed IP and this sentence is ungrammatical. Similarly, in (14c) we observe that having the negation before an overtly filled Infl in the embedded clause is still ungrammatical. The ungrammaticality of these sentences confirms two previous proposals. First, the same co-occurrence restrictions hold for the Infl head before the second verb regardless of whether it is null or overt. This implies, for example, that the sentences in (3a) and (3b) have the same structure. Second, there is no Tense position before the second verb since negation cannot occur there. This implies that the embedded Infl is like an infinitive. These conclusions are verified by the grammatical sentence in (14d) where we observe that negation is generated in the Tense position in the matrix clause, while the embedded Infl is headed by ya. Consequently, I conclude that ya can only occur in the Infl position of an embedded clause.

I return now to the proposal that when  $y\dot{a}$  occurs in what seems to be the matrix Infl it is morphologically complex. (14c) shows that both negation and  $y\dot{a}$  morphemes cannot occur in the same clause, because they compete for the head of TP position. However, the two can occur in what looks like a simple matrix clause:

- (15) a. Ôzó má yá kí!é ékhù
   Ozo Neg INFL open door
   'Ozo did not go to open the door.'
  - Òzó má (yó) yá kí!é ékhù
     Ozo Neg go INFL open door
     'Ozo did not go to open the door.'

The fact that  $m\dot{a}$  can occur with  $y\dot{a}$  suggests that  $m\dot{a}$  is in the Tense position of a matrix clause, while  $y\dot{a}$  is in the Tense head of an embedded clause, with the phonologically elided motion (aspectual) verb  $y\dot{o}$  'go' coming between them.  $Y\dot{a}$  is then the INFL of the complement of this aspectual verb, as shown in (10). Therefore, (15a) implies an underlying structure made up of two distinct functional heads with a possible surface realization being (15b).

Returning to the contrasts in (3) and (4), I propose that an embedded INFL is always structurally present in the Modal-aspectual verb construction and this head position could be either a null morpheme with a realis interpretation or an overt morpheme (yá) with an irrealis interpretation. The emerging structure is illustrated in (16).



According to (16), a sentence in which the embedded INFL head is null would have the realis interpretation, while in the case where it is overt such a sentence would have an irrealis interpretation. I will now discuss two further consequences of this analysis.

First,  $y\dot{a}$  can be compared to another word (closed class) which is  $gh\dot{a}$ . Two types of  $gh\dot{a}$  must be distinguished, which are different in terms of distribution and meaning. Consider the following:

- (17) a. Òzó ghá sùén Ozo FUT begin 'Ozo will begin.'
  - b. #Ôzó sùén ghá má àkhé
    Ozo begin FUT mold pot
    '\* as Ozo begin that he will mold pot'
    'OK as Ozo began molding pot.'
  - c. Òzó sù<u>é</u>n ghá má àkhé Ozo begin IMP mold pot 'Ozo began moulding pots.'
  - d. #Òzó ghá má àkhé Ozo IMP mold pot '\* as Ozo molding pots' 'OK as Ozo will mold pots.'
  - e. Ôzó ghá sù<u>é</u>n ghá má àkhé Ozo FUT begin IMP mold pot 'Ozo will begin moulding pots.'

(17a) shows the distribution of future tense  $gh\dot{a}$  in a simple sentence with an aspectual verb and we observe that it occurs in the Tense position between the subject and the verb. Now, in (17b) we observe that it is ungrammatical for this future tense  $gh\dot{a}$  to occur in the embedded INFL position after the aspectual first verb and before the second verb. I take this as confirming evidence of the fact that there is no Tense position in the embedded clause on a par with that of the matrix clause. However, notice from (17c) that it is possible for a  $gh\dot{a}$  morpheme with an imperfective, rather than future tense, meaning to occur in the embedded INFL position. (17d) confirms that there are two different functional heads associated with the two  $gh\dot{a}s$ , because the  $gh\dot{a}$  with the imperfective meaning cannot occur in the Tense position of the matrix clause. This complementarity between the two  $gh\dot{a}s$ reflects the basic difference between the Tense head of the matrix clause where, for example, future tense  $gh\dot{a}$  is generated and the INFL head in the embedded clause where, for example, the imperfective aspect  $gh\dot{a}$  is generated. The two possibilities are seen most clearly in (17e) where each  $gh\dot{a}$  occurs as a separate head. Therefore, I conclude that irrealis  $y\dot{a}$  and imperfective  $gh\dot{a}$  and realis  $\phi$  all occur in a subordinate INFL position and they indicate the fact that the second verb is in an embedded clause with irrealis, imperfective, and realis mood, respectively. Since  $y\dot{a}$ ,  $gh\dot{a}$ and the null morpheme are all in the embedded INFL, this suggests that  $y\dot{a}$  heads a functional rather than lexical projection.<sup>5</sup>

The immediate consequence of this conclusion that gha and ya are functional heads is that, like other functional heads, we predict that the verbs which they govern will manifest the high tone copying effect discussed in section 2.7.2. This prediction is borne out by the data below.

- (18) a. Òzó sùén ghá dún!mwún èmà
   Ozo begin IMP pound yam
   'Ozo began pounding the yam'
  - Òzó hìá yá dún!mwún èmà
     Ozo try INFL pound yam
     'Ozo tried to pound the yam'

As we observe from (18), the verb following an overtly filled INFL head exhibits the hightone copying behavior, realized as a high-downstep-high tone sequence on the disyllabic verb. I will provide a more elaborate discussion of this general issue in section 7.3.1.

As a summary of the arguments presented in this section, I have argued that certain so-called serial verbs display properties which do not apply to either the resultativeconsequential SVCs nor covert coordinations. One such property is the presence of an INFL projection that dominates the embedded second verb. There are potentially three fillers of the lower INFL: a null head,  $y\dot{a}$ , or  $gh\dot{a}$ . The difference between an overt and covert head is based on the meanings they contribute to the structure;  $y\dot{a}$  implies that the action denoted by the embedded verb is irrealis, a null head implies that the action denoted

<sup>&</sup>lt;sup>5</sup> Notice that I am assuming at this point that (imperfective) ghá occurs in INFL. I am not confusing it with the iterative ghá from chapter two and so no contradiction is intended between the imperfective ghá in embedded INFL and iterative ghá in E. I will argue below that this difference is a direct consequence of the restructuring effect in modal-aspectual verb constructions that removes the E position, which is present in SVCs.

by the embedded verb is realis, while ghá implies that part of the event happened. This is summarized in (19).

- (19) meanings of the morphemes in embedded INFL
- (a)  $\sigma$  morpheme = all of the event happened
- (b) ghá morpheme= part of the event happened
- (c) yá morpheme= none of the event happened

## 7.2.4 Modal-aspectual Verb Restriction

In this section, I turn my attention to another characteristic feature of the Modalaspectual verb construction. A closer look at this construction reveals that only a limited class of verbs can be first verbs in this construction--roughly those modal/aspectual verbs which can take clausal or verbal complements. This fact is illustrated in (20): <sup>6</sup>

- (20) a. Òzó hìá rrí èvbàré
   Ozo try eat food
   'Ozo tried (to) and he did eat the food.'
  - b. Ozo rhió kpá!á Ozo wake up early leave 'Ozo woke up early and left.'
  - c. Ozo há kí!é èkhù
     Ozo start suddenly open door
     'Ozo got up suddenly and opened the door.'
  - d. Òzó miànmián só Ozo forget shouted 'Ozo forgot and (inadvertently) shouted.'
  - e. Òzó yèé lé ùw<u>ò</u>nmw<u>è</u>n Ozo remember cook soup 'Ozo remembered and made the soup.'

<sup>&</sup>lt;sup>6</sup> Based on the analysis of the embedded INFL in the previous section, there is an underlying prediction that all of the sentences in (20) would have a close paraphrase in which  $y\dot{a}$  occurs before the second verb with a corresponding aspect-mood change. This prediction is correct and I will assume that the data and interpretation can be understood in the light of the discussion so far.

- f. Òzó yó tí!é èbé
   Ozo went read book
   'Ozo went and he read [his] book.'
- g. Ozo bá khú!<u>é</u> Ozo pretend bathe 'Ozo pretended and bathe.'

The class of first verbs found in (20) can be contrasted with the verbs found in resultativeconsequential SVCs and covert coordinations. Recall the fact the second verb of the resultative SVC must be unaccusative while both verbs must be transitive in the consequential SVC, and the covert coordination is essentially unrestricted. The Modalaspectual verb construction is different from all these other constructions because the syntactic and/or semantic restrictions apply only to the first verb. As evident from the data in (20), the first verb of the Modal-aspectual verb construction typically takes a clausal complement. Thus, the aspectual verb can have a subcategorization frame as in (21)

(21) subcategorization of aspectual verb

[ \_ IP]

In other words, while the first verb of the resultative and consequential SVCs must license an object, the first verb of the Modal-aspectual construction does not have to do so, rather it mainly selects for a clausal complement. Furthermore, the aspectual verb is potentially capable of assigning external theta role; this is assumed to be realized via the Voice head in my analysis.

### 7.3 Restructuring in Modal-aspectual Verb Construction

This section focuses on the challenge of providing a systematic account for the observed differences between Modal-aspectual verb constructions and resultativeconsequential SVCs, as well covert coordinations. These differences that were discussed in the previous section include; (a) the variation in mood associated with the embedded INFL head

(b) the aspectual verb restriction on the first verb which requires that it does not select for an object NP, but rather a clausal complement.

The syntactic account of these two properties and other salient features of the Modal-aspectual verb construction will be carried out in two stages. First, I will provide evidence for a control and restructuring analysis (cf. Rizzi 1978). Second, I will show that all of the syntactic tests that were used to argue for the distinction between resultative and consequential SVCs, as well as covert coordinations, also consistently pick out and reinforce a restructuring analysis of the Modal-aspectual verb construction.

### 7.3.1. Evidence for Infinitival Complement

The evidence that can be used to argue for an infinitival complement in the Modalaspectual verb construction is based on the observation that the second verb in this construction does not have the normal tone pattern like those found on the second verb in resultative-consequential SVCs or covert coordinations.

As background illustration of how tense is realized by tones, consider the following sentences with disyllabic verbs:

(22)	а.	Òzó dùnmwùn ìyán Ozo pound yam 'Ozo pounds yam (habitual) or Ozo	'habitual or progressive tenses' is pounding yam (present progressive)'
	b.	Òzó dùnmwún ìyán Ozo pound yam 'Ozo pounded yam' (past)'	'past tense'
	c.	Òzó ghá dún!mwún ìyán Ozo Fut. pound yam 'Ozo will pound yam (future) or is p	'future tense' ounding yam (progressive)'

Stewart (1996) claims that Tense in Èdó is mostly realized supra-segmentally on verbs, however it is important to note that, now, the basic tense tones show up varied depending on the verb's syllable structure. Thus, on most disyllabic verbs habitual or progressive tense is marked by two low tones (22a), while past tense is marked by a sequence of lowhigh tones (22b). Future tense is, however, realized by a morpheme with inherent high tone and this triggers a high tone spreading onto the verb because  $gh\dot{a}$  is in a functional head position (22c). Generalizing from this, we predict that the tones on a verb dominated by a functional head that is occupied by a morpheme will always have a copy of that morpheme's high tone. I will now show that this is correct when the verb is dominated by the embedded INFL that is headed by either  $gh\dot{a}$  (Imperfective) or  $y\dot{a} / \emptyset$  (infinitive). Consider the following.

(23)	а.	Òzó gbé úpkù gu <u>ò</u> gh <u>ó</u> Ozo hit cup break 'Ozo broke the cup'	'resultative SVC'
	b.	Òzó d <u>é</u> ìyán dùnmwún Ozo buy yam pound 'Ozo bought [the] yam and pounded	'consequential SVC' it'
	c.	Òzó gb <u>òó</u> ìyán kpàán ívìn Ozo plant yam pluck coconut 'Ozo planted yams and plucked coco	'covert coordination'
	d.	Òzó hìá dún!mwún ìyán Ozo try pound yam 'Ozo tried and pounded [the] yams'	'Modal-aspectual construction'

What we observe from the sentences in (23) is that in all cases of true SVCs in the simple past (23a-b), CC (23c), the second verb bears the normal low-high tone sequence just as in simple one-verb sentences. In contrast, in the Modal-aspectual verb construction (23d) the tone pattern on the second verb is high-downstep-high. This difference can be taken as confirming evidence for the proposal that the second verb in the Modal-aspectual verb construction is dominated by an INFL head. This head happens to be  $\emptyset$  segmentally, but triggers the same tones as  $gh\dot{a}$  and  $y\dot{a}$ . Furthermore, on the basis of the contrast in (23) I propose that the tonal difference on the second verb in the Modal-aspectual verb construction is a reflection of the fact that it is not in a finite clause. This proposal is based on the standard definition of a finite clauses as tensed clauses. I should also point out the

fact that the tones on the second verb never changes even when the first verb has a different tense, and this implies that the second verb is not in a tensed clause (see example (24) below).

It is clear that the status of the second verb in the Modal-aspectual verb construction is not on a par with second verb in either the resultative-consequential SVCs or covert coordination because of the difference in tones. According to my analysis, this difference in tones comes from the fact that there is a projection of INFL that dominates only the second verb and this head triggers high tone copying. This analysis is consistent with the proposal in Wurmbrand (1997) that Restructuring Infinitives (RIs) do not have an internal tense specification; thus, it is expected that the complement of a Restructuring verb (RV) can never be a finite clause. This kind of fact that is based on the manner of tonal (tense) Inflection on the second verb brings together the attributes of the head of the lower IP. The parallel that I am establishing here is between the general fact that matrix clauses cannot be non-finite verb and the fact that ya is syntactically constrained to occur only in embedded contexts. My claim, therefore, is that the rather odd tonal properties on the embedded verb should be attributed to the presence of the INFL head dominating the verb and this prevents it from having its own tense specification. Therefore, I conclude that the second verb is in a non-finite clause.

A closely related fact is that  $y\dot{a}$  or  $gh\dot{a}$  do not vary tonally for tense like verbs normally do, neither does the verb in the embedded clause. Consider the following:

(24)	a.	Òzó há lóvb!i <u>é</u> Ozo start suddenly sleep 'Ozo jumped suddenly to sleep.'	'past'
	b.	Òzó hà lóvb!i <u>é</u> Ozo start suddenly sleep 'Ozo jumps suddenly to sleep.'	'habitual'
	c.	Òzó há (yá) lóbv!i <u>é</u> Ozo start suddenly INFL sleep 'Ozo jumped suddenly to sleep.'	'past'

### d. \*Òzó hà yà lòvbi<u>è</u> 'habitual' Ozo start suddenly INFL sleep

According to the data in (24), the INFL head is tenseless and the embedded verb is an infinitive that cannot inflect for habitual or present tense. Thus, in (24a-c) observe that the same pattern of tones occur on the second verb, irrespective of the fact that the verb in the matrix clause (first verb) is in the past tense in (24a) and in habitual tense (24b), regardless of whether INFL head is filled by  $y\dot{a}$  or realis  $\phi$  in (24c). Moreover, it is ungrammatical to change the tone on  $y\dot{a}$  or on the verb that it governs (24d). This implies that there is no tense clash between the INFL head and the verb that it governs, consistent with the special pattern of tonal inflection on the embedded verb in the Modal-aspectual verb construction.<sup>7</sup>

Consequently, I propose that the contrast in the tones on the second verb between resultative-consequential SVCs and Modal-aspectual verb construction is evidence for the latter containing an infinitive clause. This then sets the stage for restructuring.

# 7.3.2. Evidence for Restructuring

A variety of the world's languages have the phenomenon of *restructuring* (cf. Rizzi 1978), and this involves the re-analysis of certain infinitives by means of the deletion or pruning of the S and S' nodes. In order words, restructuring involves clause unification between an infinitival embedded clause and the matrix clause. Much of the work on restructuring has been done in the Romance languages (e.g. Rizzi (1978), Burzio (1981), Zubizarreta (1982) etc.) and Germanic languages (e.g. Evers (1975, 1982), Haegeman & van Riemsdijk (1986). Based on these works, it is a fairly standard assumption that there is a distinction between restructuring infinitives (RIs) and non-restructuring infinitives (NRIs). NRIs are said to act more or less like finite clauses in that they represent an

267

<sup>&</sup>lt;sup>7</sup> One consequence of this analysis of the Modal-aspectual verb construction is the prediction that it would be different from covert coordinations with respect to verb movement to Tense. This can be tested in two ways, when the head of INFL is filled by yá we predict that verb movement in the embedded clause will be blocked, and even more profound would be the claim that verb movement in the lower clause is predicted to be impossible because there is a minus Tense head that cannot attract the verb. I will come back to these issues in section 7.3.2.1 and argue that this is one more evidence for restructuring.

independent clausal domain for properties like clitic movement, passive, and scrambling, while RIs seem to lack this clausal character in many respects. In particular, RIs do not show clause-boundedness effects for similar processes (cf. Wurmbrand, 1997).

In Kwa languages, there is only one widely-known reference to restructuring and this has been proposed to account for infinitive verb raising phenomena within embedded clauses in Abe (Tellier 1986).<sup>8</sup> Although, the usual tests for restructuring such as clitic climbing, long object preposing (Romance), long distance scrambling, and super-passive (German)) cannot be replicated in Èdó or Abe (Tellier 1986), there are, however, three arguments which point to a restructuring analysis for the Modal-aspectual verb construction.

### 7.3.2.1 Verb Raising and Object Cleft.

In arguing for restructuring in Romance and Germanic languages, one often-used test is Clitic Climbing. For example, in Italian, clitic (or 'weak') pronouns sometimes attach to the verb of which they are a complement in underlying structure (25a) and at other times to the verb of a higher clause (25b) (cf. Napoli, 1981 and others);

According to the data in (25), the clitic can remain attached to the embedded verb, or it may climb and cliticize onto the matrix verb. This instance in which the clitic climbs onto the verb of the matrix clause is taken as evidence that restructuring of the embedded infinitive has occurred causing the structure to act like a simple clause. This is taken as a fairly standard diagnostic in both Romance and Germanic languages. Essentially, restructuring has the effect of unifying the matrix and embedded clause so that the clitic is able to see the

<sup>&</sup>lt;sup>8</sup> An SOV language spoken in Cote dé Voire

entire (unified) clause as a single constituent. The standard view of this sort of object clitic movement (climbing) is that clitic movement is obligatory, but optional restructuring determines how it will apply, i.e., it can either remain with the lower verb or attach (climb up) to the higher verb, as illustrated in (25).

In Édó, I will show that there is an abstract manifestation of the same thing; although there is no clitic climbing per se. The relevant data has to do with the clefting of the object of the embedded verb in the general context of verb raising to Tense in the Modal-aspectual verb construction. Note first of all that the very fact that verb raising to Tense is sometimes possible in this construction shows that it is different from true SVCs, where V-raising is always blocked (see chapter six). Here now is a description of the basic facts of verb movement to Tense in the Modal-aspectual verb construction:

(26)	а.	Òzó gèlé miànmián yá dé ìyán Ozo truly forget INFL buy yam 'Ozo truly forgot to buy yam.'
	b.	*Òzó gèlé miànmián´!r <u>é</u> n yá d <u>é</u> ìyán Ozo truly forget INFL buy yam
	c.	*Òzó miànmián!rén gé!lé yá dé ìyán Ozo forget truly INFL buy yam
	d.	ìyán <u>ò</u> ré Òzó miànmián!r <u>é</u> n gé!lé yá d <u>é</u> yam Foc. Ozo forget truly INFL buy 'It is yam that Ozo truly forgot to buy .'

(26a) illustrates the ordering of I-type adverb with respect to the modal verb; it occurs to the left of the verb, as expected since the I-type adverb is generated as a left adjunct to the head of EP. (26b) shows that the modal verb cannot bear the past perfective suffix in its base-generated position. This too is expected, since -rV is in T position and obligatorily attracts the verb. Thus, overt verb movement is triggered, such that the verb raises to check a feature of Tense. This observation is particularly worth noting because in the previous discussion of verb movement in chapter six I used only transitive verbs. However, it is quite possible for the past perfective suffix to show up on intransitives as in (27) (alluded to in chapter six):

- (27) a. Òzó giégié sàán Ozo quickly jump 'Ozo quickly jumped.'
  - Òzó sá!ánrén giégié
     Ozo jump-rV quickly
     'Ozo has jumped quickly.'
  - \*Òzó giégié sá!ánrén
     Ozo quickly jump-rV
     'Ozo quickly has jumped.'

(27) shows that an intransitive verb can move past the adverb to Tense to support the -rV suffix (27b) but it can never bear this same Inflection in its unmoved position (27c). The point I am making here is that the aspectual verb resembles an intransitive verb because it does not have an internal object NP. Therefore, it seems that it should be able to undergo verb movement to Tense, because the issue of Case-licensing and object movement do not arise with neither aspectual nor intransitive verbs.

What this simple characterization of the facts predicts is that it should be possible in the Modal-aspectual verb construction for the first modal/aspectual verb alone to move to Tense without the clefting of an object NP since it does not subcategorize for an object NP to begin with, on a par with intransitives (27). However, (26c) shows that this prediction is incorrect as we see that it is ungrammatical for the modal verb to move past the adverb to Tense. In order to clearly illustrate the relevance of the ungrammaticality of (26c) let us compare it with a similar sentence in which there is a complementizer and tensed verb, rather than infinitival ya, after the modal verb and see if verb movement is possible. This is illustrated in (28):

(28) a. Òzó gèlé miànmián wèć úyì dè ìyán Ozo truly forget that Uyi buy yam 'Ozo truly forgot that Uyi buys yams.'
- \*Òzó gèlé miànmiánrén wèé úyì dé ìyán
   Ozo truly forget-rV that Uyi buy yam
   'Ozo truly forgot that Uyi bought yams.'
- Ozó mián!miánrén gèlé wèé úyì dé ìyán
   Ozo forget-rV truly that Uyi buy yam
   Ozo forgot truly that Uyi bought yams.'

Contrast between (26) and (28) provides the first solid evidence of a restructuring account of the Modal-aspectual verb construction. (28a) illustrates the fact that the I-type adverb exhibits the same ordering relation with the aspectual verb. In addition, the presence of the complementizer in (28) also implies that the second verb is in a tensed clause and this can be independently verified by the fact that the embedded verb in (28a) has the low tone which marks habitual tense. In (28b) we repeat the previous observation that the presence of the -rV suffix on the verb requires that the verb move to Tense, and this has not happened in this case because the adverb precedes the verb; therefore, the sentence is ungrammatical. The striking contrast comes from (28c) where we observe that the aspectual verb can undergo verb movement when the complement clause is tensed, thus providing a minimal contrast with (26c). I propose that this contrast is one evidence that restructuring has occurred in the embedded infinitive with ya in the Modal-aspectual verb construction but not when there is a complementizer introducing a tensed clause.

Another striking fact about the Modal-aspectual verb construction with respect to verb movement is that when the direct object of the embedded verb is clefted as in (26d), we observe that the past-perfective suffix can occur on the matrix verb. Since the presence of the -rV suffix can only mean that the modal verb has moved to Tense, it means therefore that the direct object of the embedded verb acts as the object of the matrix verb, in the way that it facilitates verb movement. I propose that this interesting fact whereby the direct object of the embedded verb comes to act syntactically as the direct object of the matrix modal verb is compatible only with restructuring explanation. In this respect, object cleft from the Modal-aspectual verb construction is like Clitic Climbing in the Romance and Germanic languages because in both cases an otherwise objectless verb comes to have what seems to function as a syntactic object of the matrix clause. The significance of the analysis of (26d) is based on the observations about verb-raising in section 6.4.2 where I showed that a transitive verb in a simple clause raises to Tense position if and only if its object is clefted, as shown in (29):

- (29) a. Òzó gbé éb<u>ó</u>lù Ozo hit ball 'Ozo kicked the ball.'
  - b. \*Òzó gbé!ré éb<u>ó</u>lù
     Ozo hit -rV ball
     'Ozo has kicked the ball.'
  - c. éb<u>ó</u>lù <u>ò</u>ré Òzó gbé!ré
     ball Foc Ozo hit -rV
     'Its the ball that Ozo has kicked.'

Therefore, the fact that the modal verb can raise to Tense only when the object of the verb in the embedded clause is clefted (26d) shows that restructuring has applied. Furthermore, (26d) provides a different kind of evidence for the analysis of restructuring in the Modalaspectual verb construction. This evidence points to a possible parametric difference between restructuring in Èdó and the Romance/Germanic languages. Whereas restructuring (hence Clitic Climbing) is said to be optional in Romance and Germanic languages, (26c) shows that object cleft prior to verb raising in these restructuring contexts is obligatory. This would explain the ungrammaticality of (26c) where the object is not clefted and the verb raises past the adverb.

Therefore, I conclude that we are able to derive restructuring effects like Clitic Climbing based on the interaction between verb raising to Tense and object cleft. In fact, the pattern of facts in (26c) and (26d) along with the contrast from the tensed clause in (28) are only compatible with a restructuring account of the Modal-aspectual verb construction.

### 7.3.2.2. Predicate Clefts

This section provides another piece of evidence that supports my proposal for a restructuring analysis of the Modal-aspectual verb construction, based on event structure. My basic claim is that the clause unification involved in restructuring also implies that event identification (unification) has taken place. Confirmation of this comes from predicate clefts. I will argue that the two verbs involved in the Modal-aspectual verb construction combine to express a single event. This claim is modeled after some works on Romance/Germanic restructuring (cf. Rosen (1990), Napoli (1982), Wurmbrand (1997), etc.), where it has been proposed that the event structure of restructuring verbs (RVs) is underspecified and that RVs similar to auxiliary or light verbs form a single event structure with the infinitive . Thus, the prediction is that predicate clefts from the Modal-aspectual verb construction will exhibit the same properties as single event resultative SVC. Recall the fact from chapter three that neither verb can be clefted in a single event resultative SVC, although predicate clefts of the verbs in the two-event consequential SVC or covert coordinations are possible.

As a preliminary step to the discussion of predicate clefts in the Modal-aspectual verb construction, I will first illustrate the behavior of some modal verbs in simple clauses. This is illustrated in (30):

- (30) a. Òzó hìà Ozo try 'Ozo tries.'
  - b. ùhiámwèn òré Òzó hìà nom-try-nom Foc. Ozo try
     'Ozo is really managing, not that he is strong or well enough.'
  - c. Ózó miànmián Ozo forget 'Ozo forgot.'
  - d. ùmián!miánmwèn òré Òzó miá!nmián nom-forget-nom Foc. Ozo forget
     'It is forgetting that Ozo did, not that he did it intentionally.'

As the data above shows, modal verbs like other verbs in the language can undergo predicate clefts for contrastive focus. According to my analysis, the grammaticality of the predicate clefts in (30) is derived from the idea that the cognate event argument of the modal verb, which is base-generated in the internal complement position of the VP, moves through Specifier of EP for checking on its way to Spec, FP/CP. I assume that there is only one such Specifier of EP position for each functional head. This raising of the event argument is also accompanied by the covert raising of the modal verb which is attracted by the functional head E, thus creating the relevant Spec-head configuration between the event argument and the verb to check the event-role. Thereafter, due to the [+Foc] feature on the event argument, it is attracted overtly by  $\partial r \dot{e}$  which is in the head of Focus Phrase (FP) to its Specifier position to check the focus feature and this gives us the surface order. This is illustrated for (30a) by the simple representation in (31) (VoiceP is omitted).



On the basis of this summary of predicate cleft licensing in simple clauses, let us consider the following sentences which illustrate predicate clefts from the Modal-aspectual verb construction:

- (32) a. Òzó hìá kó!kô ìkù
   Ozo try gather dirt
   'Ozo tried and gathered the dirt [together].'
  - b. \*ùhiámw<u>èn ò</u>ré Òzó hí!á kó!kó ìkù nom-try-nom Foc. Ozo try gather dirt
  - c. \*ùkókómw<u>èn</u> <u>d</u>ré Özó hí!á kó!kó ìkù nom-gather-nom Foc. Ozo try gather dirt
- (33) a. Òzó miànmián yá lé èvbàré Ozo forget INFL cook food 'Ozo forgot to cook the food.'
  - b. \*ùmián!miánmwèn dré Özó miá!nmián yá lé èvbàré nom-forget-nom Foc. Ozo forget INFL cook food
  - c. \*ùlémwèn <u>ò</u>ré Òzó miá!nmián yá lé èvbàré nom-cook-nom Foc. Ozo forget INFL cook food
- (34) a. Òzó miànmián wèć úyì lé èvbàré Ozo forget COMP Uyi cook food 'Ozo forgot that Uyi cooked the food.'
  - b. ùmián!miánmwèn òré Ôzó miá!nmián wèé úyì lé èvbàré nom-forget-nom Foc. Ozo forget COMP Uyi cook food
     'It's forgetting that Ozo forgot that uyi cooked the food, not intended act.'
  - c. ?ùlémwèn òré Òzó miá!nmián wèé úyì lé èvbàré nom-cook-nom Foc. Ozo forget COMP Uyi cook food 'It's cooking that Ozo forgot that Uyi did to the food, not selling.'

According to the sentences in (32) and (33), it is ungrammatical to have a predicate cleft of either of the verbs from the Modal-aspectual verb construction. This is so even though we know that each of the verbs can independently undergo predicate cleft (30) (although I have only shown this for modal verbs). In contrast, when there are two tensed clauses as in (34) the same verb can now undergo predicate cleft. Consequently, the ungrammaticality of predicate clefts of the verbs in the Modal-aspectual verb construction receive a consistent explanation if we assume that the event structure of the modal/aspectual verb is underspecified in this construction, and like an auxiliary or light verb it forms a single event with the embedded clause. This would imply that there is only a single projection of E(vent)Phrase for the two verbs which I assume, like in the single-event resultative SVC,

to be generated above the two verbs. Thus, (32a) would have a representation as in (35) (only the projections below TP are shown, I omit the lower VoiceP for simplicity).



In (35), I propose to represent the assumption that the event structure of the modal/aspectual verb is underspecified, thus forming a single event with the verb in the embedded clause, based on the requirement for the cognate event argument to move through the Specifier of EP on its way to Spec, FP/CP. As will become clear shortly, I propose that a single EP projection dominates both verbs, therefore, there is no intermediate EP projection for the embedded infinitive and this sets the Modal-aspectual verb construction apart from consequential SVC and covert coordination.

As before, I assume that multiple Specifiers are not allowed for functional projections, EP in this case. Therefore, the ungrammaticality of predicates clefts in the Modal-aspectual verb construction comes from the fact that both verbs are attracted at LF by the functional head E since there is no intervening node to block the LF raising of the embedded verb. However, there is only a single Specifier for the EP, so there is room for

the cognate event argument of only one of the verbs. Whichever is chosen, it fails to check with the two verbs in E at LF. Thus, the predicate cleft of either verb in the modal-aspectual construction is ungrammatical as shown in (32) and(33). On the contrary, when the second verb is in a tensed clause like (34), I assume that there is an EP projection dominating the verb and so predicate cleft will be properly licensed and the sentence is grammatical. On the basis of this contrast, I conclude that predicate clefts present evidence in favor of a restructuring analysis of the Modal-aspectual verb construction where clause unification between the matrix and embedded verbs also implies event structure unification.

Independent confirmation of my proposal that predicate clefts can serve as a useful diagnostic for restructuring effects come from related arguments in Tellier (1986) concerning Abe, where predicate cleft is used to show that otherwise separate event structures associated with verbs in two clauses merge into a single event structure. In particular, Tellier's claim is that two restructured verbs form a constituent at S-structure and, therefore, the lower clause containing the second verb is no longer able to undergo predicate cleft; rather both verbs must cleft as one syntactic unit. Consider the relevant data from Abe (Kwa) in Tellier (1986).

(36) a. kO ni E [Api yaya orovi kO ni] V2 V1 start catch Foc. intend snake start catch 'Api intended to START CATCHING snakes'
b. \*ni hOhO O [Yapi yaya orovi ni hOhO] V1 V2 catch learn Foc. intend snake catch learn 'Yapi intended to LEARN (HOW) TO CATCH snakes'

What the contrast in (36) is intended to illustrate is the effect of restructuring on different verbal combinations, depending on whether the second verb has undergone Verb-Inversion (36a)--which is taken to be evidence that restructuring has applied--or remains in situ (36b) (non-restructuring). In (36a) where restructuring has taken place the two verbs can undergo predicate cleft together, whereas in (36b) without restructuring (Verb-

Inversion), the two infinitive verbs do not form a constituent at S-structure and so it is ungrammatical for them to be clefted together.

These facts from Abe can be recast in a different way based on my proposed structural account of predicate cleft and restructuring. Accordingly, I propose that in (36a) there is a single projection of EP and like resultative SVCs in Yoruba (or Igbo resultative V-V compounds), both the event arguments of the two verbs undergo a coalescence and form a compound at LF; this is what allows predicate cleft under the relevant Spec-head configuration at LF. In the ungrammatical sentence (36b), there are separate projections of EP and this inhibits the predicate cleft of both verbs because the event argument of one of them will not be properly checked in the Specifier-head relation of a single EP. Therefore, predicate cleft facts from Abe are proof that clausal unification also involves event structure unification as well. Where restructuring has not taken place like in (36b) there is no evidence for event unification and the predicate cleft of both verbs together is ungrammatical. Note that examples like (36a) are ungrammatical in Èdó because of a surface morphological constraint against forming V-V compounds; compare the discussion of Èdó vs. Yoruba in section 3.5.1.

Now I must turn to the lack of an EP projection before the second verb which produces a rather unusual selection in the structure in (35) where IP immediately dominates VP without an intervening EP projection (see Travis, forthcoming). In fact, this is my specific claim for restructuring, i.e., that when clause unification occurs event structure unification also takes place, and this involves a lack of an EP projection within the embedded clause.

The empirical evidence to support this proposal comes from two related areas. First, consider the ungrammaticality of Modal-aspectual verb construction when an aspectual verb which selects for imperfective gha takes an indefinite object as in (37). #Òzó sùén ghá gbé èmá!tón
Ozo begin IMP. hit metal
'\* as Ozo began hitting metal repeatedly'
'OK as Ozo began hitting the metal'

(37)

(37) is intended to show the incompatibility of an iterative reading with an aspectual verb which selects for an IP that is progressive imperfective. Thus, it is ungrammatical to interpret the event iteratively as Ozo began hitting the metal repeatedly, rather than a single unfinished event which is Ozo began hitting the metal. This contrast of interpretation implies that an iteration meaning is absent in the Modal-aspectual verb construction like (37). Thus, I conclude that there is no EP projection before the second verb to host the iterative meaning of gha, rather there is an INFL head where gha (imperfective) is generated.

The second kind of data that supports the claim that there is no EP before the second verb in the Modal-aspectual verb construction comes from the observation that both INFL yá and iterative ghá cannot co-occur in the Modal-aspectual verb construction. This is illustrated in (38):

- (38)\*Ozó miànmián yá ghá lé ìyán a. INFL Iter. cook yam Ozo forget b. \*Ózó vèé vá ghá dệ ùkpòn Ozo remember INFL Iter buy cloth Ózó miànmián vá c. lé ìyán Ozo forget INFL cook vam 'Ozo forgot to cook yam.' d. Ózó yèé dé ùkpòn yá
  - Ozo remember INFL buy cloth 'Ozo remembered to a buy dress.'

Based on the grammaticality contrast between the sentences in (38), we observe that irrealis  $y\dot{a}$  and iterative *ghá* cannot occur together (38a,b), while similar sentences without the iterative morpheme are perfectly grammatical (38c,d). Thus, the ungrammaticality of (38a,b) implies that there is no position in the structure for the iterative morpheme, and this

supports my claim that there is restructuring of the EP projection before the second verb in the Modal-aspectual verb construction.

# 7.3.2.3 The Class of Restructuring Verbs (RVs)

The final evidence which supports a restructuring analysis for the Modal-aspectual verb construction is based on a cross-linguistic similarity amongst these verbs of the class of known restructuring verbs (RVs) in other languages. Wurmbrand (1997) notes that although the class of RVs shows some variation across languages displaying restructuring effects, there is, a discernible core of RVs which is illustrated in (39a-c) with some typical RVs in Italian, German, and Dutch;

(39) a. Italian:
 (39) a. modals, andare ('go'), cominciare ('begin'), continuare ('continue'), osare ('dare'), riuscire ('succeed'), sapere ('know'), venire ('come'), easy-adjectives.

German:

- b. modals, versuchen ('try'), beginnen ('begin'), gelingen ('succeed'), fortfahren ('continue'), wagen ('dare'), vergessen ('forget'), easyadjectives.
  - Dutch:
- c. modals, beginnen ('begin'), dreigen ('threaten'), durven ('dare'), helpen (help), leren ('learn, teach'), menen ('think, believe, mean'), proberen ('try'), trachten ('try), wagen ('dare'), weigeren ('refuse').

In a striking testimony to universal grammar (UG), it turns out that many of these verbs that are typical RVs in Romance and Germanic languages are similar to or the same as the modal, aspectual and motion verbs in Èdó that exhibit the restriction in (21). They constitute the class of verbs that can occur as the first verb in the Modal-aspectual verb construction as exemplified by the sentences in (20).

I will illustrate this comparison between Indo-European RVs and Edó modalaspectual verbs with three verbs from the list in (39) that were not included in (20). I will show that they also exhibit the properties associated with the modal-aspectual verb construction, albeit in a different way. Consider the following:

- (40) a. Òzó sùén ghá yó èsùkú Ozo begin INFL attend school 'Ozo began attending school.'
  - Òzó yó yá kí!é <u>é</u>khù
     Ozo go INFL open door
     'ozo went to open the door.'
  - C. Òzó rènrén yá lé èvbàré
     Ozo know INFL cook food
     'Ozo knew to cook the food.'

The verbs in (40) display strong selectional properties which require that the head of embedded INFL must be filled by an overt morpheme  $gh\dot{a}$  (40a) or  $y\dot{a}$  (40b,c). What is striking about these sentences is the fact that they do not have the alternation with a realis version of the same sentence in which the INFL head is null. This fact is illustrated by the ungrammaticality of the following sentences:

(41)	а.	*Òzó sù <u>é</u> n yó èsùkú Ozo begin attend school
	b.	*Òzó yó kìé <u>é</u> khù Ozo go open door
	c.	*Òzó rènrén lé èvbàré Ozo know cook food

Note that the examples in (40) do not fit under the traditional definition of SVCs because they are indeed linked together by markers of subordination like  $y\dot{a}$  and  $gh\dot{a}$ . Nevertheless, their syntactic behavior is substantially the same as other modal-aspectual verb constructions

Consequently, I propose that the fact that the list of aspectual verbs that can occur as the first verb in the Modal-aspectual verb construction is a subset of the RVs in Romance and Germanic (39) should be taken as evidence of a universal class of RVs. Thus, it is natural that the modal-aspectual verbs in Èdó should be analyzed in terms of restructuring. The aspectual verbs in Èdó are listed in (42) which includes some verbs that do not easily translate into English or other Indo-European languages.

 sùén ('begin, continue'), hìá ('try, manage'), yó ('go'), rènrén ('know'), miànmián ('forget'), rhió (wake up early), há (get up in a start/shock), bá (walk in a tip-toe/act stealthily), yèé ('remember') <sup>9</sup>

The fundamental point arising from this observation about the class of RVs is the fact that these modal-aspectual verbs in Èdó based on the general subcategorization in (21) do not take object NPs in addition to the infinitival complements. This point is relevant because it has been suggested that subject control verbs with objects cannot restructure in Romance (Zushi 1995) and that object control verbs do not allow clitic climbing (Kayne 1989:248-49, 250-51). Thus, the subcategorization of these verbs in (42) is consistent with a restructuring analysis of the Modal-aspectual verb construction.

Therefore, I conclude this section by reiterating my basic claim that the Aspectual verb construction involves restructuring phenomena on a par with Romance and Germanic languages. This is in spite of the fact that object cleft seems to be obligatory in the Aspectual verb construction for the purpose of verb raising unlike the optional Clitic Climbing in the Romance and Germanic restructuring to which it is being compared.

# 7.3.3 Additional Syntactic Tests for Restructured Clause

This section continues the general project of showing that a restructuring account is consistent with the behavior of the verbs in the Modal-aspectual verb construction, making them different from SVCs or CCs. I will now turn my attention to filling out the details of the structure that I have proposed as in (43) (only the relevant projections below EP are shown).

<sup>&</sup>lt;sup>9</sup> For some unknown reason a verb like 'want' which is a typical RV in any language does not exhibit the properties associated with aspectual verbs. Thus, this argument based on the similarity between RVs and aspectual verbs is suggestive but would require more in-depth study. I will leave this open for further research whether this is a lexical idiosyncrasy or part of some other generalization.



I will argue that several of the same tests which were used to establish the structures of resultative SVCs, consequential SVCs, and CCs from the previous chapters also consistently support the structure in (43).

# 7.3.3.1 There are Two Subject Positions

In chapter two, I showed that there is only a single subject position in the resultative and consequential SVCs, while there are two positions in CCs. In this section, I will argue based on evidence from the distribution of the adverbial particle *tobóre* (it/her/him-self) that there is a subject position inside the VP of the embedded complement clause of the modalaspectual verb construction. The leading idea here, also, is based on the assumption that tobóre can either right-adjoin to an overt NP or a trace of a moved NP. Consider the following sentences:

(44) a. Ozój miànmián [PROj yá lé èvbàré Ozo forget INFL cook food 'Ozo forgot to cook the food.' 283

- b. \*Òzój miànmián [tòb<u>ó</u>rèj yá lé èvbàré Ozo forget himself INFL cook food 'Ozo forgot to cook the food by himself.'
- c. Ozó mianmián [PROj yá tòb<u>ó</u>rèj lé èvbaré] Ozo forget INFL himself cook food 'Ozoj forgot [PROj to cook the food by himself.'
- (45) a. Ozój hìá [PROj [ſø dún!mwún ìyán Ozo try pound yam 'Ozo tried and pounded the yam.'
  - b. Ôzó hìá [PROj [ſ' tôb<u>ó</u>rèj dún!mwún ìyán]]
     Ozo try himself pound yam
     'Ozoj tried and [PROj pounded the yam by himself.'

Two sets of data have been presented in (44-45) which show the same distributional behavior of the particle when the INFL head is filled (44) and when it is null (45). In (44a) for example, we observe that the hypothesized PRO subject of the embedded infinitival clause is controlled by the matrix subject. This is consistent with the interpretation of the sentence whereby Ozo is the subject of the actions denoted by the two verbs. The evidence for this internal subject position comes from the contrast between (44b) and (44c) where we observe the distribution of the  $t\partial b \dot{g} r \dot{e}$  particle. In (44b) the particle occurs in the Spec of IP before the infinitive particle  $y\dot{a}$  and this sentence is ungrammatical. I propose that the reason why (44b) is ungrammatical is because the particle can only either right-adjoin to an overt NP or to the trace of the moved NP. (44c) shows the correct distribution of the particle where it occurs below the infinitive  $y\dot{a}$  and before the verb. I take this acceptable word order to be evidence of the fact that PRO subject is actually base-generated in the Spec of VoiceP below IP to which the particle right-adjoins.<sup>10</sup>

(i)  $\dot{O}z \dot{o} d \dot{e} i y \dot{a} n l \dot{e} prok t \dot{o} b \dot{o} r \dot{e}_k$ 

284

<sup>&</sup>lt;sup>10</sup> Now I can directly comment on the Spec position of the infinitive IP. The evidence from the distribution of tobore indicates that PRO is generated below IP (Spec of VoiceP). The question is, does PRO move to Spec of IP afterwards as assumed in Sportiche (1988)? Alternatively PRO could move to check null case. (cf. Chomsky 1993, 1995) However, on the basis of the distribution of the particle in the consequential SVC where it attaches to *pro* in a sentence like (i);

it is assumed that *pro* does not move (cf. Baker and Stewart 1997b), thus I make this generalization concerning PRO; that it does not move to Spec of IP in the modal-aspectual verb construction (cf. Baltin 1995). I will not go into any detail on this issue since it is not directly relevant to the point I want to make about restructuring and the fact that there is a subject position in the embedded clause of the modal-aspectual

These facts about the Modal-aspectual verb construction call to mind similar observations and analysis in Baltin (1995) and Sportiche (1988) concerning the distribution of the floated quantifier 'all' vis-a-vis an infinitival PRO in English. This is illustrated by the following:

- (46) a. \*They tried all to leave
  - b. They tried to all leave

Baltin (1995) and Sportiche (1988), although with some differences in the implementation of their analyses, both take the contrast in (46) to imply that subjects are always generated within lexical projections, so that the subject at S-structure 'They' is actually generated within the VP. Thus, like floated quantifiers in English the  $t\partial b \dot{o} r \dot{e}$  particle presents evidence that there is an internal subject, PRO, that is generated within the extended projection of VP (VoiceP) of the embedded clause.

In the light of the foregoing, I propose the structural representation in (47) that accounts for the fact that the PRO subject originates in a low position consistent with the claim that there are two subjects; an overt subject in the matrix clause which binds a null PRO in the embedded IP that is consistent with certain assumptions regarding obligatory control in infinitival clauses (cf. Williams 1980).<sup>11</sup>

- (i) \*Òzó miànmián úyì lé èvbàré
- Ozo forgot Uyi cook food

verb construction. Further evidence that this position must be PRO is based on the fact that it is ungrammatical to have an overt non-coreferential subject in the place of PRO. For example, the sentence in (ii) is ungrammatical.

I will leave these issues open for future research.

<sup>&</sup>lt;sup>11</sup> I have assumed a simple representation for the embedded VP just for ease of representing the specific point being made here, although the structure could very well be more detailed both in terms of representing the object argument as well as the event arguments, which are not relevant in this case.



#### 7.3.3.2 I-type Adverb before First Verb

We have seen from the discussion on V-to-I movement in section 6.2 that the I-type adverb occurs before the verb in the matrix clause and I have proposed that this kind of adverb is licensed as a left-adjunct to the head of EP. In this section, I will show that while this is true for the first verb in the Modal-aspectual verb construction, however, the consequence of restructuring whereby there is no EP projection before the second verb creates a situation in which the I-type adverb is now licensed as a left-adjunct to VP2. Therefore, I will argue that this licensing contrast between the two clauses with respect to Itype adverbs solves an apparent problem concerning their distribution in the embedded clause of Modal-aspectual verb construction, due to restructuring. Furthermore, this contrast in I-type adverb licensing provides a larger context of differentiating the Modalaspectual verb construction from resultative SVCs, consequential SVCs, and CCs in terms of the contrast between E and Infl.

286

Therefore, this section is intended to show that there is no contrast between resultative-consequential SVCs and Modal-aspectual verb construction based on the distribution and licensing of the I-type adverb when they occur before the first verb. This confirms the assumption that in all these constructions there is an EP position before the first verb. Consider the following:

- (48) a. Òzó giègiệ hìà (yá) dún!mwún ìyán Ozo quickly try INFL pound yam 'Ozo quickly tries (to) pound the yam.'
  - b. Òzó gi<u>é!gié</u> hìá (yá) dún!mwún ìyán Ozo quickly try INFL pound yam 'Ozo quickly tried (to) pound the yam.'
- (49) a. Òzó giègiè fiàn ágá kànmwàn Ozo quickly cut chair be-short 'Ozo quickly cuts the chair short.'
  - b. Ozó gi<u>é</u>!gi<u>é</u> fián ágá kànmwán Ozo quickly cut chair be-short 'Ozo quickly cut the chair short.'
- (50) a. Ozó giègiè dè èvbàré rè
   Ozo quickly buy food eat
   'Ozo quickly buys [the] food and he eats it (quickly).'
  - Òzó gi<u>é</u>!gi<u>é</u> d<u>é</u> èvbàré ré
     Ozo quickly buy food eat
     'Ozo quickly bought [the] food and he ate it (quickly).'

(48) shows an I-type adverb before the first verb in the modal-Aspectual verb construction, while (49)-(50) show the same thing for the resultative SVC and consequential SVC respectively. Observe that in (48)- (50), the I-type adverb occurs in the same position before the verb and also the fact that in the (a) sentences the adverbs all show the same low tone Inflection for habitual-present tense, while in the (b) sentences the adverbs all have the same high tone past tense Inflection. This consistency implies that the I-type adverb is generated in the same position in both the Modal-aspectual verb construction and resultative

and consequential SVCs. In short, they all have an EP projection before the first verb which the I-type adverb left-adjoins to its head.<sup>12</sup>

### 7.3.3.3 Iterative vs. Imperfective

This section shows that there is a contrast between consequential SVCs and modalaspectual verb construction with respect to the licensing of I-type adverbs in the position before the second verb. I will argue that this contrast is a reflection of the structural difference within the internal projections of both constructions: consequential SVCs have an internal E projection, while Modal-aspectual verb constructions only have an internal projection of INFL causing I-type adverbs to adjoin to the left of VP itself.

In order to present a minimal contrast between I-type adverb licensing in consequential SVCs and Modal-aspectual verb constructions, I propose to use differences in meaning of the morpheme  $gh\dot{a}$  as test for the presence of either E or Infl projection before the second verb. This is based on the fact that the  $gh\dot{a}$  morpheme can either have an iterative meaning, in which case I assume that it is in E, or a progressive meaning which is taken to imply that it is in Infl. This contrast is illustrated by the following sentences:

- (51) a. Òzó hìá ghá dún!mwún ìyán nà Ozo try PRG pound yam this 'Ozo tried pounding this yam'
  - Ózó dé ìyán ghá dún!mwún
     Ozo buy yam ITE pound
     'Ozo bought [the] yams and pounded them repeatedly'

The two sentences in (51) illustrate very important points about the sort of aspect/mood interpretations that are compatible with either SVC or the Modal-aspectual verb construction. *A priori*, we could assume that iteration and progressive are subcases of the general semantic class of imperfective as shown in (52) (cf. Comrie, 1976).

<sup>&</sup>lt;sup>12</sup> I do not attempt to provide examples with ghá before the adverb since the presence of ghá in INFL will cause tone copying across the clause.



I propose that these two kinds of imperfective are actually disjoint in the structures of consequential SVCs and Modal-aspectual verb constructions. In the Modal-aspectual verb construction (51a)  $gh\dot{a}$  can only have a progressive (continuing action) meaning because the only functional head available is Infl, as in (53a). In contrast, in the consequential SVC (51b)  $gh\dot{a}$  can only have the iteration meaning because only E is structurally available for lexical insertion, as in (53b). Consequently, I propose that this meaning difference reflects a difference in the nature of heads that are compatible with each kind of construction.

(53a) the lower functional head in Modal-aspectual verb construction



The position of I-type adverbs before the second verb provides syntactic evidence that confirms this structural difference illustrated in (53). Compare the following:

			290
	C.	Ôzó hìà ghá gi <u>é</u> !gié dún!mwún ìyán Ozo try PROG quickly pound yam 'Ozo is trying quickly pounding the yam'	√INFL + Adv. order
	d.	Òzó hìá ghá gi <u>é</u> !gi <u>é</u> dún!mwún ìyán Ozo try PROG quickly pound yam 'Ozo tried quickly pounding the yam'	11
(55)	а.	*Òzó d <u>è</u> ìyán ghá gi <u>ègi</u> è dún!mwún Ozo buy yam ITE quickly pound	*E + Adv order
	b.	*Òzó d <u>é</u> ìyán ghá gi <u>égié</u> dún!mwún Ozo buy yam ITE quickly pound	U
	c.	Òzó d <u>è</u> ìyán gi <u>ègiè</u> ghá dún!mwún Ozo buy yam quickly ITE pound 'Ozo buys [the] yam and he quickly pounds it repea	√Adv + E order atedly'
	d.	Òzó d <u>é</u> ìyán gi <u>égié</u> ghá dún!mwún Ozo buy yam quickly ITE pound 'Ozo bought [the] yam and he pounded it quickly re	" peatedly'

200

The sentences in (54) and (55) illustrate a significant difference between the Modalaspectual verb construction and the consequential SVC in terms of the nature of inner Infl head and the licensing of adverbs. This difference comes from word order of I-type adverb and the overt functional head. Observe from (54a,b) that the order in which the I-type adverb comes before the functional head  $gh\dot{a}$  is ungrammatical in the Modal-aspectual verb construction. This implies that the I-type adverb cannot left adjoin to Infl. This is in sharp contrast with the consequential SVC (55c,d) where the order of I-type adverb before  $gh\dot{a}$  is grammatical, implying that this  $gh\dot{a}$  is an instance of E, to which the I-adverbs can left adjoin.

I propose that this contrast follows from the structural difference between the two constructions. Thus, while there is an E in the consequential SVC which is consistent with the licensing of I-type adverbs and the distribution and interpretation of iterative gha, in the restructuring context of the Modal-aspectual verb construction there is only an INFL head present; and EP is "pruned" and so it requires a different kind of I-type adverb licensing. This is evident from the fact that the order of INFL + adv produces grammatical results as in (54c,d) which I assume implies that the I-type adverb adjoins to the left of VP, possibly

as a preverb. This is in sharp contrast to the order between the adverb and the functional head in the consequential SVC, where it is ungrammatical for the adverb to occur after ghá (55a,b). This contrast is therefore another primary evidence that there is restructuring in the Modal-aspectual verb construction. This VP-adjunction of the I-type adverb in restructuring context of the Modal-aspectual verb construction is illustrated in (56).



On the basis of the structure of the Modal-aspectual verb construction and the facts from the distribution of I-type adverbs, I propose that although the Modal-aspectual verb construction expresses a single event like the resultative SVC, the second verb can be modified by an adverb. Based on my assumptions about adverbs as predicates of events (cf. Parsons 1990) it follows that there is some sense in which the second verb has an event and this allows the I-type adverb to occur. Nevertheless, the overall nature of the event denoted in the Modal-aspectual verb construction is blurred by restructuring, as evident from the rather unexpected distribution of the I-type adverb.

#### 7.3.3.4 N-type adverbs as VP Modifier

In this section, I will provide evidence based on the distribution of N-type adverb showing the phrasal status of the verbs in the Modal-aspectual verb construction. This will again support the proposal that there is a single event like in the resultative SVC. The primary assumption that underlies this analysis is that N-type adverbs only modify phrasal categories like VP by right adjoining to them, but not lexical category such as the verb. This will reveal the essential difference between non-phrasal nature of the first verb without its clausal complement, as opposed to the second verb which clearly projects a verb phrase. The background comparison is with similar distribution of N-type adverbs in resultative SVCs. Consider the following:

(57)	а.	*Ôzó hìá <u>ègìégìé</u> (yá) dún!mwún ìyán Ozo try quickly INFL pound yam
	b.	Òzó hìá (yá) dún!mwún ìyán <u>ègìégìé</u> Ozo try INFL pound yam quickly 'Ozo tried to pound the yam quickly'
(58)	а.	*Òzó kòkó àdésúwà <u>ègìégìé</u> mòsé Ozo raise Adesuwa quickly be.beautiful
	Ъ.	Òzó kòkó àdésúwà mòsé <u>ègiégié</u> Ozo raise Adesuwa be beautiful quickly 'Ozo raised Adesuwa to be beautiful quickly'
(59)	а.	Òzó dé èvbàré <u>èg</u> ìégìé ré Ozo buy food quickly eat 'Ozo bought [the] food quickly and he ate it'
	h	Òzó dé èvhàré ré ègiégié

b. Ozo de evbare re<u>egiegie</u> Ozo buy food eat quickly 'Ozo bought [the] food and he ate it quickly'

In (57a), we observe that the N-type adverb cannot occur in the position after the first verb. One important control for (57a) is to see whether an N-type adverb can occur after the Aspectual verb in isolation. This is possible, as shown in (60):

(60) a. Òzó hìá <u>ègiégié</u> Ozo try quickly 'Ozo tried quickly' Òzó sù<u>én ègìégìé</u>
 Ozo begin quickly
 'Ozo began quickly'

Based on the grammaticality of the N-type adverb modification with aspectual verbs in (60), I conclude that the ungrammaticality of (57a) is based on the fact it is impossible to have a separate modification of the first verb since there is no VP projection independent of its clausal complement (IP) to which the adverb can adjoin. The ungrammaticality of (57a) mirrors a similar fact in the resultative SVC, as shown in (58a). The distribution of the N-type adverb is consistent with the structure that I have proposed for the Modal-aspectual verb construction (56) modified appropriately in (61).



In fact, we can justify the account for why the N-type adverb cannot occur between the verb and its IP complement which is attributed to restructuring by contrasting (57a) with similar sentence in which there is a CP before the second verb. This is shown in (62).

The grammaticality of (62) where an N-type adverb can occur after the first verb and modify it provides another striking evidence based on the contrast between restructuring and non-restructuring contexts that there is restructuring in the modal-aspectual verb construction.

# 7.3.3.5 Locative Prepositional Phrase

This section provides evidence which shows that the first verb of the Modalaspectual verb construction forms a syntactic unit with its clausal complement because of restructuring. Like N-type adverbs, locative PPs are assumed to be phrasal modifiers of the VP and so I predict similar ungrammaticality when it appears between the first verb and the IP. Consider the following:

- (63) a. Ozó miànmián vbè òwá Ozo forget at home 'Ozo forgot at home'
  - b. \*Òzó miànmián vbè òwá (yá) dún!mwún ìyán Ozo forget at home INFL pound yam
  - c. Òzó miànmián (yá) dún!mwún ìyán vbè òwá Ozo forget INFL pound yam at home 'Ozo forgot to pound the yam at home'
  - d. Òzó miànmián vbè òwá w<u>èé</u> ìr<u>è</u>n dùnmwún ìyán Ozo forget at home that he pound yam 'Ozo forgot at home that he (Ozo) pounded yams'

The sentences in (63) illustrate an interesting fact about the Modal-aspectual verb construction. This is based on the analysis of locative PPs as a VP modifier, from which we observe that it exhibits the same structural constraints as that of the N-type adverb. (63a) shows that the locative PP can occur with an aspectual verb in a simple clause, however, in the modal-aspectual verb construction (63b) it is ungrammatical for the locative PP to immediately follow the verb. This implies therefore, that the first verb is not by itself a VP, rather it forms a VP with its clausal complement. Thus, when the locative PP occur

after the second verb (63c) the sentence is grammatical and I propose that this is the case because the first verb forms a tight syntactic unit with the clausal complement due to restructuring. The crucial contrast is between (63b) and (63d) where there is a CP in the latter and the locative PP like the adjunct N-type adverb can occur after the first verb, and I take this to be evidence for obligatory restructuring since a CP extraposition allows adjuncts to occur between the verb and its CP complement but not with restructuring infinitival IP. This contrast can only mean one thing; there is restructuring of the infinitival clause in the modal-aspectual verb construction.

On the basis of the foregoing discussion, I conclude that the Modal-aspectual verb construction involves restructuring. However, unlike the Romance and Germanic restructuring we observe that restructuring is obligatory in Èdó Modal-aspectual verb construction while it is optional in Romance and Germanic languages. I believe there is a possible parameter here that can be stated to account for this difference between Èdó and Romance/Germanic languages, but I will leave this open for future research.

### 7.4 On Instrumental Constructions

In this section, I examine another kind of re-analysis concerning the underspecification of features internal to the verb rather than features associated with the verbal projection (7.2 above). The construction that I am concerned with here is instrumental constructions which have been assumed to be SVCs in a variety of African languages (cf. Awoyale 1988, Baker 1989, 1991, Lefebvre 1991, Li 1991, Sebba 1987, Déchaine 1993, Collins 1997, etc.), as well as Japanese (Nishiyama 1995) a language which is not usually classified as a SVC language.

Instrumental constructions appear on the surface to contain two verbs, each with its own object, and there is some semantic grounds to interpret the relations between the verbs and their arguments as involving argument sharing. There are two general kinds in Èdó and these are illustrated in the sentences in (64):<sup>13</sup>

- (64) a. Òzó yá áb<u>é</u> fián èmió!wó (=2) Ozo use knife cut meat 'Ozo cut the meat with the knife'
  - b. Òzó rhié úghánmwàn ghu<u>ó</u>!gh<u>ó</u> ówá Ozo take axe break stall 'Ozo broke the stall with an axe'

What we observe on the surface is a sequence of two verbs, each with its object complement but sharing the same subject. Furthermore, observe that the object of the first verb is an instrument with respect to the action named by the second verb. There is, however, a difference between the two kinds of instrumental constructions with respect to 'lexical content' and syntactic properties: the first verb in (64a) is underspecified for a lot of features since it cannot do any of the following: occur as a main verb in a simple clause, undergo predicate cleft, be modified by an N-type adverb, undergo verb movement. In contrast, the opposite is true for both verbs in (64b).<sup>14</sup> (64b) may either have a conjunction structure or a complement structure, but the more interesting case that fits the pattern of discussion introduced by the modal-aspectual verb construction is (64a), where on a closer look we will find that yet another so-called 'SVC' exhibits properties that are different from true SVC. Therefore, I will focus on the analysis of (64a) drawing illustration from the other kind of instrumental construction when the need arises (see section 7.4.3.2).

<sup>&</sup>lt;sup>13</sup> One very common trend in the syntactic analyses of the so-called instrumental SVC is the idea that there is some kind of asymmetry in the projections of the two verbs involved in the construction. For example, in Sranan where the object of the first verb can be extracted, speakers show variation in the extractability of the object of the second verb (cf. Jansen et al. 1978, Li 1991). Similarly, in Igbo and Yoruba it has been observed that there are asymmetries between the two verbs and their objects with respect to object extraction and predicate clefts (cf. Déchaine 1993, Manfredi and Laniran 1988, Awoyale 1988 etc.)

<sup>&</sup>lt;sup>14</sup>Both constructions, however, share some common properties: there is a null or overt Infl head between the verbs and predictably the tones on the second verb in both cases are of the high-down-step-high pattern, the distribution of tob<u>ó</u>re shows that there are two subject positions (one below the IP and before the second verb). These properties distinguish them from resultative and consequential SVCs. Thus, I assume that both types of instrumental constructions involve control. They differ in the ways that I have outlined in the text (see below for more discussion of these properties based on the kind in (64a)).

There are three distinct approaches that can be identified in the previous analyses that treats instrumental constructions as SVCs. First, the lexical approach (cf. Lefebvre 1991) where, for example, they are treated as causatives that are derived by some mechanism of 'conflation'. Such a 'conflation' process, according to Larson (1991), may take place at the Lexical Conceptual Structure (LCS). Alternatively, Li (1991) argues that the 'conflation' applies to 'prelexical representations' which allow substantial aspects of prelexical structure to remain thematically intact but 'invisible' while still exercising syntactic effects. Second, there is the syntactic approach in which instrumental constructions are considered to be object sharing SVCs where the second verb is allowed to assign a secondary theta role such as instrument to the object of the first verb (Baker 1989). As stated in chapter two, here also object sharing can be analyzed as either true internal object sharing under a co-headed VP in the sense of Baker (1989), or as control of an empty category, pro that is generated in the Specifier of a separate VP headed by the second verb (Collins 1997). Third, there is the bivalent projection analysis in Déchaine (1993) where although each verb forms a separate VP, one of them counts as the head of the whole construction in some sense (cf. Manfredi 1991).

In light of the foregoing, the goal of this section is to contribute to the study of socalled instrumental SVC based on data from the Èdó language. My basic claims are that the so-called instrumental SVC is not true SVC at all; rather it involves control, i.e., subject control, and that the first verb is a defective verb somewhat like a functor predicate or light verb (cf. Ritter and Rosen, 1993, Rosen 1990) which has no semantic content but can assign a theta role; it combines with a second verb that is fully lexical and transitive to express a single event. This is sketched in (65) (only the relevant projections are given).



My analysis is presented in two parts. The first part will provide arguments that support my proposal that the so-called instrumental SVC involves subject Control. The second part applies various syntactic arguments from previous chapters that support my analysis.

# 7.4.1. Motivating Subject Control

This section will show that the instrumental construction is like the Modal-aspectual verb construction in the respect that there is an Infl head before the second verb. However, there can be no restructuring in the instrumental construction since the first verb obligatorily selects for an object in contrast with the first verb in the Modal-aspectual verb construction with restructuring. The two constructions also differ in that there is no meaning alternation between a null Infl (which has a realis reading) and an overt ya, which should have an (irrealis) reading. In the instrumental construction, the presence of INFL is simply

298

associated with an infinitive clause. I will now present three arguments that support this proposal.

#### 7.4..2 INFL before the Second Verb

In this section, I will demonstrate that there is a special tonal pattern on the second verb in the instrumental construction which is like the one on the second verb in the Modalaspectual verb construction. I will argue that this tonal inflection is due to the presence of an Infl head before the second verb. Thus, the instrumental construction contrasts sharply with SVCs and CCs with respect to tense inflection, thus indicating a difference in clause structure. Consider the following sentences in which the second verb is disyllabic:

(66)	а.	Òzó yá áb <u>é</u> vá!l <u>ó</u> èmió!wó Ozo use knife slice meat 'Ozo used the knife to slice the meat'	'past'
	b.	Ìs <u>òkè</u> n yà òyìyà sá!l <u>ó</u> ètó Isoken use comb comb hair 'Isoken uses the comb to comb her hair'	'habitual'

Observe that the tones on the second verb remain invariant in spite of the tense changes on the first verb; it is always the same sequence of high-downstep-high tones. This data contrasts in this respect with resultative and consequential SVCs, as shown in (67):

(67)	a.	Òzó fián ìrùnmwùn kànmwán (*kán!mwán) Ozo cut grass be.short 'Ozo cut the grass short'
	b.	Òzó lé ìyán dùnmwún (*dún!mwún) Ozo cook yam pound

Ozo cook yam pound 'Ozo cooked the yam and pounded it'

(67) shows that there is a clear contrast between SVCs and the instrumental constructions in terms of the tones on the second verb. In fact, it is ungrammatical to have the high downstep-high tone sequence on the second verb in SVCs (this is in accordance with the tone matching condition from chapter six). What is even more striking is the fact that the sentences in (66) cannot be made into covert coordinations where each verb would be within a different projection of Tense and have its normal, simple clause inflection. This is illustrated in (68).

(68)	а.	*Òzó yá áb <u>é</u> vàl <u>ó</u> èmió!wó Ozo use knife slice meat	'past'
	b.	*Ìs <u>òkè</u> n yà òyìyà sàl <u>ò</u> ètó Isoken use comb comb hair	'habitual'

Thus, we conclude that instrumental constructions must be different from true SVCs in at least two ways; (a) tense matching does not apply to the second verb, (b) the barrier to tense matching cannot be a finite Tense head or else covert coordination would be possible.

I propose that what is responsible for the tone difference on the second verb is the projection of an Infl head which introduces an infinitive clause. This head blocks tone matching from spreading onto the second verb and accounts for why covert coordination is not possible. Further evidence for positing the Infl head before the second verb comes from the following sentences, which freely alternate with those in (66).

(69)	a.	Òzó yá áb <u>é</u> yá vá!l <u>ó</u> èmió!wó Ozo use knife INFL slice meat 'Ozo used the knife to slice the meat'	'past'
	b.	Ìsòkèn yà òyìyà yá sá!ló ètó Isoken use comb INFL comb hair 'Isoken uses the comb to comb her hair'	'habitual'

These sentences show that the irrealis Infl particle ya can occur before the second verb-which again shows that these sentences are not on a par with the resultative or consequential SVCs. Following the analysis of similar facts involving the Modal-aspectual verb construction, I conclude that there is a structural Infl head that dominates the second verb in the instrumental construction and this head may be null or overt.<sup>15</sup> This is represented in the structure in (70) for a sentence like (69a).<sup>16</sup>



The structural difference between instrumental and modal-aspectual constructions in terms of functional heads before the second verb can be seen in the distribution of  $gh\dot{a}$  (iterative) and  $y\dot{a}$  (irrealis) in the embedded INFL. Compare (71) and (72):

(71) a.		Òzó miànmián yá lé èvbàré Ozo forget INFL cook food 'Ozo forgot to cook the food'	√INFL head	
	b.	*Òzó miànmián ghá lé èvbàré Ozo forget ITER cook food 'Ozo forgot and cooked the food repeatedly'	*E head	

<sup>&</sup>lt;sup>15</sup> There is, however, a difference between Modal-aspectual verb construction and instrumental constructions even though their structures both contain an embedded projection of Infl. This comes from the observation that the null vs. overt head alternation in the instrumental construction does not seem to express the realis/irrealis contrast that we observed in Modal-aspectual verb construction; rather embedded clauses of the instrumental construction only have realis readings. I suspect that this may be due to the semantics of the verb 'yá' which freely translates as 'use'.

<sup>&</sup>lt;sup>16</sup> I have assumed a simple V NP for the projection of VP2 in order to simplify the presentation of this structure.

	c.	*Òzó miànmián yá ghá lé èvbàré Ozo forget INFL ITER cook food	*INFL plus E heads
(72) a b c	а.	Òzó yá áb <u>é</u> yá fián émió!wó Ozo use knife INFL cut meat 'Ozo used the knife to cut the meat'	√INFL head
	Ъ.	Òzó yá ábé ghá fián émió!wó Ozo use knife ITER cut meat 'Ozo used the knife to cut the meat repeatedly'	√E head
	c.	Òzó yá ábý yá ghá fián émió!wó Ozo use knife INFL ITER cut meat 'Ozo used the knife to cut the meat repeatedly'	√INFL plus E heads

It is impossible for iterative gha, which is generated in the E head, to appear in the Modalaspectual verb construction (71b), (71b), whereas ya that is generated in INFL head is possible (71a). This confirms the proposal that there is no E head before the second verb, because of restructuring. This pattern sharply contrasts with the distribution of similar morphemes in the instrumental construction, which does not involve restructuring. In (72a), we observe that ya can occur before the second verb, and (72b) shows that gha can also occur before the second verb. These sentences suggest, therefore, that there are two separate heads, Infl and E, before the second verb. We can confirm this from (72c) where we observe that the iterative and infinitive morphemes can co-occur. This proves that the instrumental construction.

In conclusion, both Aspectual and instrumental constructions differ from true SVCs in that they have an embedded Infl projection. However, the instrumental construction is different from the Modal-aspectual verb construction in that it has an EP projection before the second verb unlike in the Modal-aspectual verb construction.

#### 7.4.3 PRO Subject in Embedded Clause

The representation of the instrumental construction in (71) assumes that there is a PRO subject in the embedded IP that is controlled by the overt subject in the matrix clause. In this section, I will examine two kinds of data which support this proposal.

# 7.4.3.1 tòb<u>ó</u>rè particle

Given the discussion of  $t \partial b \underline{\phi} r \dot{e}$  in the Modal-aspectual verb construction, we are able to predict its distribution in the instrumental construction where there is also an embedded INFL. Thus, if the previous assumption from section 7.2 about a PRO subject in the embedded IP is correct, then we predict that the particle will show up in the Specifier of VoiceP in the embedded clause. Consider the following:

(73)	a.	[Òzó <sub>i</sub> yá àdésúwàk [PR	ROj/*k (yá)	khuán	íghó]]	
		Ozo <sup>°</sup> use Adesuwa	INFL	reap	money	
		'Ozo used Adesuwa to mal	ke money'	-	-	

- b. [Òzój yá àdésúwàk [PROj/\*k (yá) tòbórèj/\*k khuán íghó]]
   Ozo use Adesuwa INFL himself reap money
   'Ozo used Adesuwa to make money'
- c. [Òzój yá àdésúwàk [PROj/\*k (yá) khuán íghó tòb<u>ó</u>rèj/\*k]]
   Ozo use Adesuwa INFL reap money himself
   'Ozo used Adesuwa to make money'

Observe that the same pattern of distribution with  $t\partial b \underline{o} r \dot{e}$  as in the modal-aspectual verb construction occurs in the instrumental constructions (73). The interpretation of (73a) is that the action denoted by the second verb is performed by the matrix subject Ozo, not by the object of the matrix verb, Adesuwa. Therefore, the null PRO subject of the embedded clause must be controlled by Ozo. In (73b), we observe that  $t\partial b \underline{o} r \dot{e}$  occurs after the Infl head, before the verb and is still interpreted as modifying the matrix subject. I assume that

this distribution of  $t \partial b \underline{o} r \dot{e}$  provides evidence that PRO is generated low in the clause (in Specifier of VoiceP).<sup>17</sup>

The fact that the object of the matrix verb (Adesuwa) cannot be coreferent with  $t \partial b \dot{Q} r \dot{e}$  can be taken as evidence that the object NP (instrument) has not moved from anywhere within the lower IP. This data in (76b) constitutes evidence against previous analyses of the instrumental construction in which object sharing is invoked or in which an empty category is said to be controlled by the overt object (Baker 1989, Collins 1997 etc.).

Consequently, the coreference facts from the distribution of  $t \partial b \underline{o} r \dot{e}$  can be seen as evidence that there is a PRO subject NP that is generated in the Specifier of VoiceP in the embedded clause, consistent with the structure in (74) where only the relevant projections are shown.



Therefore, I conclude that the instrumental construction is only compatible with a subject control analysis, not with an object sharing or empty category analysis that would unify them with SVCs, as has been previously assumed in the literature (cf. Law and Veenstra 1992, Carstens 1988, etc.).

<sup>&</sup>lt;sup>17</sup> As with the modal-aspectual verb construction, I will not go into any details as to whether PRO moves to Spec of IP, but simply assume based on the distribution of  $t\partial b dr e$  that it is base-generated low, in the Specifier of VoiceP.

### 7.4.3.2 Purposive Clauses

This section examines a comparison between the instrumental construction and one form of Purpose clause in English (cf. Chomsky, 1977, Jones 1991 etc.) which provides indirect support for the proposal that there is a PRO subject in the embedded INFL in the Èdó instrumental construction. I propose that English purpose clause in which the first verb is 'use' (see Jones 1991) is like the instrumental constructions in Èdó in involving clausal complementation. The sentence in (75) illustrates what is usually classified as a kind of Purpose clause in English.

(75) John used the knife to cut the meat

In particular, Jones (1991) argues that sentences like this exhibit some of the diagnostics for a purposive clause.

(a) Purpose clauses are adjuncts. Observe that in (75), 'to cut the meat' is quite freely ommisible, John used the knife being a well formed sentence. In this respect, it is like a purpose clause adjunct.

(b) The phrase *in order* can occur before the infinitive as in (76).

(76) John used the knife in order to cut the meat

Nevertheless, in spite of these two properties, I would like to claim that (75) is not exactly a Purposive construction; rather it is an instrumental construction somewhat like the one found in Èdó.

There are two reasons for making such a claim. First, observe the minor fact that (75) has a realis reading which asserts (or presupposes) that John actually did cut the meat. This interpretation is rather different from that of most purpose clauses. For example, contrast (75) with the clear case of purposive clause in (77).

(77) They brought John to talk to Mary (Jones=15a)

Notice that unlike (75), (77) is ambiguous: it could be that John actually spoke with Mary, or it could be that they brought John along for this purpose, but it was not achieved for one reason or another. I propose that this difference between (75) and (77) arise from the inherent lexical meaning of <u>use</u> (when one uses something, it entails that the action of the verb in the complement occurs). Given this, I propose that (75) is different from (77) in that the former is an instrumental construction while the latter is a purposive adjunct, contrary to Jones (1991) who treats both as Subject Purposive Clauses (SPCs).

The second and major reason to reject a unified analysis for both (75) and (77) comes from the difference between them with respect to wh-extraction of the object of the embedded verb. This contrast is illustrated in (78)

(78) a. what did John use the knife to cut?

b. \*Who did they bring John to talk to

(78a) illustrates the extraction of the object of the embedded verb from the instrumental construction with <u>use</u> as first verb, while (78b) shows the same facts for an object in a clear purposive clause. They differ in grammaticality. In particular, the fact that it is possible to extract the object from the embedded clause in the <u>use</u> sentence in (78a) shows that it is not a purpose clause, given that purpose clauses are islands (Chomsky 1977). The expected island effect is clearly present in (78b), where we observe that it is ungrammatical to extract the object from a true purpose clause.

What is striking about this data is that the same contrast can be illustrated with respect to wh-extraction between the instrumental construction and a purpose clause in Èdó:<sup>18</sup>

<sup>&</sup>lt;sup>18</sup> I illustrate the purpose clause with the other kind of instrumental construction because yá does not select for a CP complement (I suppose this is part of its defective paradigm). This switch of first verbs is not a weakness of the argument being presented here since I assume that this other kind of instrumental construction with a fully lexical first verb is also compatible with a control (complementation) analysis as well. In fact, in (80) we are dealing with another construction and so the contrast holds.
- (79) a. Òzó yá ábé yá vá!ló émió!wó Ozo use knife INFL cut meat 'Ozo used the knife to cut the meat'
  - b. émió!wó <u>à</u>ré Òzó yá áb<u>é</u> yá vá!l<u>ó</u> meat Foc. Ozo use knife INFL cut 'It is the meat that Ozo used the knife to cut'
- (80) a. Ôzó rhiế ábế nề ố yá vá!lố émiố!wô Ozo take knife that he INFL cut meat 'Ozo used the knife in order to cut the meat'
  - b. \*émió!wó <u>d</u>ré Özó rhié áb<u>é</u> nè <u>ó</u> yá vá!l<u>ó</u> meat Foc. Ozo take knife that he INFL cut

It is quite possible for the object of the verb in the embedded clause of the instrumental construction to be wh-moved as shown in (79b), while a comparable extraction from a real purposive clause results in an ungrammatical sentence as shown in (80b). Consequently, based on the parallel with English I conclude that instrumental constructions are different from purposive clauses. Furthermore, since English allows a version of the instrumental construction like Èdó, but English is not a serializing language, therefore we have comparative evidence that instrumental constructions are not SVCs in any theoretical sense.

## 7.4.4 On the Syntax of 'use'-type Instrumental Construction

In section 7.4.2 (in particular, example (66)), we saw that  $y\dot{a}$  can inflect for tense by varying its tones like verbs. However, in this section I will argue that it is 'underspecified' in several respects to be made clear by examining the other salient properties of this first verb with respect to the syntactic tests from previous chapters.<sup>19</sup>

- (i) Özó yá ábé yá fián émió!wó 'past tense high tone on yá'
  Ozo use knife INFL cut meat
- (ii) Özó yà ábé yá fián émió!wó 'habitual tense low tone on yá'
  Ozo use knife INFL cut meat

<sup>&</sup>lt;sup>19</sup> We can clearly demonstrate the fact that yá is a verb in spite of its 'defective' nature based on the fact that it behaves like a verb in the language in terms of tonal inflection for tense

Thus, we observe that the lexical  $y\dot{a}$  is different from the one in the head of Infl in terms of tense tone inflection.

As a necessary background, consider the difference between the <u>use</u> verb and the <u>take</u> verb in (81) both of which are said to come from a closed class of verbs (Lefebvre 1991):

- (81) a. \*Òzó yá áb<u>é</u> Ozo take knife
  - Òzó rhié áb<u>é</u>
    Ozo take knife
    'Ozo took the knife'

The contrast above shows that  $y\dot{a}$  cannot occur in a simple clause, while *rhié* can. Thus,  $y\dot{a}$  is some kind of defective verb (perhaps a functor predicate in the terminology of Ritter and Rosen 1993) while *rhié* is fully lexical. I propose that this is a significant contrast which we can use to explain the other rather different properties of the verb  $y\dot{a}$ .

This contrast in (81) seems to mirror the distinction in English between a functor predicate such as *have* and a real causative verb like *make* proposed in Ritter and Rosen (R&R) (1993). This is illustrated in (82):

(82) a. The teacher didn't make Bill write the article, but he did it anyway.

b. \*The teacher didn't have Bill write the article, but he did it anyway.

When *make* is negated as in (82a) the causing event has not taken place. However, the writing event may still occur, as indicated by the fact that it is possible to add the <u>but</u> clause, indicating that the <u>writing</u> took place without coercion. The functor predicate *have* works differently: in (82b) <u>have</u> cannot be negated when the embedded complement is presupposed to have happened. R&R (1993) propose that this difference with respect to negation in (82) is because <u>make</u> and <u>write</u> denote two distinct events in (82a), while the ungrammaticality of the <u>but</u> clause in (82b) comes from the fact that <u>have</u> and the embedded verb <u>write</u> denote a single event. Thus, the writing is automatically negated along with the causation.

When this distinction is applied to the verbs in (81), it seems that (81a) in which the defective verb  $y\dot{a}$  occurs is comparable with the general semantic behavior of the functor predicate <u>have</u>. Interestingly enough, this verb also shows different behavior with respect to negation and presupposition (cf. Manfredi 1993 and papers in that volume). Consider the following:

- (83) a. Òzó má yá áb<u>é</u> yá fián èmió!wó,
  Ozo neg use knife INFL cut meat
  'Ozo did not use the knife to cut the meat'
  - b. \*Òzó má yá ábé yá fián èmió!wó, sòkpán ò yé rú érè
    Ozo neg use knife INFL cut meat, but he still do it

In (83a) we observe that we can negate the two verbs of the instrumental construction. However, in (83b) we observe that one cannot negate a part of the instrumental construction (such as the first verb) and the second verb is presupposed. I conclude that (83b) is evidence in support of the proposal that the first verb takes an obligatory IP complement containing the second verb, and so it cannot be negated apart from its complement. This may be taken as an indication that both verbs may express a single event. The relation between the verb and its complement is diagrammed in (84).



It is predicted from (84) that the obligatory complement relation between the IP and the first verb cannot be violated by word-level or phrasal level categories, nor can the first verb act independently from its complement with respect to predicate cleft and others. These issues and related ones are discussed in the following sections.

#### 7.4.4.1 I-type Adverbs

This section explores the distribution of I-type adverbs in the instrumental construction, based on which I will argue that there are two structural E positions. Consider first the position before the first verb:

(85)	а.	Òzó giégié yá ábé (yá) fián èmió!wó Ozo quickly use knife INFL cut meat 'Ozo quickly used the knife to cut the meat'

Ózó gi<u>égié</u> yá ùkéké (yá) gb<u>é</u>n èbé
 Ozo quickly use pen INFL write book
 'Ozo quickly used the pen to write the letter'

(85) shows that the I-type adverb can occur before the first verb in the instrumental construction. This is consistent with the generalization in chapter two that there is a fixed order of functional projections before the first verb in all main clauses. In terms of interpretation, the I-type adverb in this position modifies the actions denoted by both the first and second verbs, i.e., the cutting is quick as well as the using (85a), and the writing is quick as well as the using (85b).

Based on Parsons (1990), I propose that the interpretation of the I-type adverb in the position before the first verb in (85) reflects the event structure of the construction: there is a single (macro) event. I will come back to how this event relation between the two verbs is represented in section (7.4.4.5)

Next, let us turn our attention to the position before the second verb to examine the internal structure of the embedded clause. Consider the following:

(86) a. Òzó yá ábé (\*giégié) (yá) giégié fián èmió!wó Ozo use knife quickly INFL quickly cut meat 'Ozo used the knife to quickly cut the meat' Òzó yá ùkéké (\*giégié) (yá) giégié gbén èbé
 Ozo use pen quickly INFL quickly write book
 'Ozo used the pen to quickly write the letter'

In (86), we observe that the embedded clause shares a resemblance with the matrix clause in terms of the ordering of functional heads. This is based on the fact that the I-type adverb cannot occur before the Infl particle  $y\dot{a}$ ; this shows that the licensing of this kind of adverb is uniform across all constructions except those affected by restructuring ( the Modalaspectual verb construction). In (86), the I-type adverb modifies the second verb only. Based on the analysis that I have proposed, this implies that there is an E position before the second verb. This structure of the embedded clause is shown in (87).



#### 7.4.4.2 N-type Adverb

This section is intended to provide evidence in support of the structures given in (84) and (87) for the two clauses in the instrumental construction. This is based on the distribution of N-type adverb, which is assumed to be a VP modifier that adjoins to the right of a VP. The primary goal here is to reveal the difference between the first and second verbs in terms of their phrasal status. Consider the following:

(88) a. \*Òzó yá ábé ègiégié (yá) fián èmió!wó Ozo use knife quickly INFL cut meat 'Ozo used the knife quickly to cut the meat' b. \*Ôzó yá ùkéké <u>ègìégìé</u> (yá) gbén èbé
 Ozo use pen quickly INFL write book
 'Ozo used the pen quickly to write the letter'

In (88), we observe that an N-type adverb cannot occur after the V1-NP sequence. Based on my assumptions about N-type adverb licensing, the ungrammaticality of the sentences in (88) must come from the fact that the V1-NP sequence does not constitute a VP to which the N-type adverb can right-adjoin. In addition, the ungrammaticality of (88) is also consistent with the view that the first verb does not denote an event by itself that is distinct from the second verb; they both form one large event expressed by the first verb and the verb contained within the IP complement. This is consistent with the structure in (84).

Predictably, the N-type adverb can occur after the second verb as shown in (89).

- (89) a. Ozó yá ábé (yá) fián èmió!wó ègiégié Ozo use knife INFL cut meat quickly 'Ozo used the knife to cut the meat quickly'
  - Òzó yá ùkéké (yá) gbén èbé ègìégìé
    Ozo use pen INFL write book quickly
    'Ozo used the pen to write the letter quickly'

The interpretation of the N-type adverb in the sentence final position provides another indication that the two verbs in the instrumental construction express a single event. In (89), both verbs are modified, just as when the I-type adverb occurs in the position before the first verb. Structurally, this implies that the N-type adverb adjoins to the right of the VP in the matrix clause and modify both the first and second verbs. This is illustrated in (90).



# 7.4.4.3 Locative Prepositional Phrase

This section provides evidence which confirms the conclusions based on the distribution of N-type adverb. Like N-type adverb, locative PPs can only attach to a VP; thus, we predict similar distribution for these phrases as we saw with N-type adverbs. Consider the following:

- (91) a. Òzó yá áb<u>é</u> (\*vbè òwá) (yá) fián èmió!wó (√vbè òwá) Ozo use knife at home INFL cut meat at home 'Ozo used the knife to cut the meat at home'
  - b. Ôzó yá ùkéké (\*vbè òwá) (yá) gbén èbé (√vbè òwá)
    Ozo use pen at home INFL write book at home
    'Ozo used the pen to write the letter at home'

In (91), we observe that the locative PP cannot occur after V1-NP1. This implies two things: there is no VP projection made up of the V1 and NP1; it confirms the proposal that there is a close affinity between the first verb and its complement which cannot be disrupted by placing syntactic material between them. However, the locative PP can occur in the sentence final position, after the VP2, where it has the meaning that the actions denoted by both verbs were done in the same location. This interpretation is compatible with the proposal that this kind of instrumental construction expresses a single event and, therefore, the action of the first verb cannot have a separate location from the action of the second verb. Rather, the two verbs must combine as one macro event, which can then be modified by the locative PP in the sentence final position.

# 7.4.4.5 Predicate Clefts

This section elaborates the nature of the single event expressed by the two verbs in the instrumental construction based on their behavior in predicate clefts. Consider the following sentences:

(92) a. \*ùyámw<u>èn ò</u>ré Òzó yá áb<u>é</u> yá fián èmió!wó nom-use-nom Foc. Ozo use knife INFL cut meat

- b. ùfiánmwèn òré Òzó yá ábé yá fián èmió!wó nom-cut-nom Foc. Ozo use knife INFL cut meat
   'It is cutting that Ozo used the knife to cut the meat, not for slicing'
- (93) a. \*ùyámw<u>èn ò</u>ré Òzó yá ùkéké (yá) gb<u>é</u>n èbé nom-use-nom Foc. Ozo use pen INFL write book
  - b. ùgbénmwèn òré Òzó yá ùkéké (yá) gbén èbé nom-write-nom Foc. Ozo use pen INFL write book
     'Its writing that Ozo used the pen to write the letter, not for something else '

According to the data in (92) and (93) there is a sharp contrast between the two verbs in the instrumental construction with respect to predicate cleft; ya' cannot be clefted while the verb in the IP complement can. Observe from the translations of the grammatical sentences in (92b) and (93b) that the using part denoted by the first verb is included in the contrastive focus of the clefted predicate: not that it was used for slicing (92b), not that it was used for something else (93b). My analysis of this contrast is as follows: the defective verb ya' is a light verb in the sense of Parsons (1990) which does not introduce a new event quantification, and so cannot undergo predicate cleft; indeed there is no nominal form  $\dot{u}yamwen$ . However, unlike the single event resultative SVC the second verb in this construction can undergo predicate cleft. I take this as evidence that the second verb is enough to define the large event based on a event chain between the E in the embedded clause and that in the matrix clause. This is realized via the indexing between the E heads and the verbs that they dominate. This is illustrated in (94) where only the relevant projections are represented.



Based on the event chain in (94), we are able to provide an account for the fact that the predicate cleft of the second verb provides a contrastive focus which includes the meaning of the first verb. The contrastive focus of the two verbs comes from the cognate event argument of the second verb moving to the Specifier of the EP in the embedded clause whose head attracts the verb covertly. The checking relation that holds between these two under Spec-head also applies to the higher E and its verb via a process of 'percolation' derived from the event chain. Therefore, the single event expressed by the two verbs is licensed as one.

This analysis of predicate cleft provides a consistent explanation for the distribution of adverbs and locative PPs in terms of picking out a single event and an obligatory selectional relation between the matrix and the embedded clause. In addition, it also illustrates the difference between single event true SVCs (resultative) and the single event non-SVC (instrumental) with respect to the properties of the verbs and the nature of E heads.

#### 7.4.4.6 Verb Raising

In the last section, I argued that there is a difference between single event instrumental construction and single event resultative SVC based on differences in predicate cleft. In this section, I will further illustrate this distinction based on the analysis of verb movement to Infl which appears on the surface to be ungrammatical with either verb in the two constructions. However, I will argue that verb raising is ungrammatical in the instrumental construction for different reasons. Thus, another aspect of the defective paradigm of ya is based on the fact that it does not undergo verb raising. In addition, I will also show that the failure of V-raising by the second verb comes from the inability of a [-tense] Infl to attract. Consider the following :

- (95) a. ábé òré Òzó gié!gié yá -- fián èmió!wó knife Foc. Ozo quickly use cut meat 'Its a knife that Ozo quickly used to cut the meat'
  - \*ábé òré Òzó yá!ré gié!gié (yá) fián èmió!wó knife Foc. Ozo use+rV quickly INFL cut meat
  - c. \*èmió!wó <u>d</u>ré Özó yá áb<u>é</u> fiá!nrén gi<u>é</u>!gi<u>é</u> meat Foc. Ozo use knife cut+rV quickly 'It is meat that Ozo used the knife to cut quickly'
- (96) a. ùkéké òré Òzó gié!gié yá (yá) gbén èbé pen Foc. Ozo quickly use INFL write book 'Its a pen that Ozo quickly used to write the letter'
  - b. \*ùkéké <u>d</u>ré Özó yá!ré gi<u>é</u>!gi<u>é</u> (yá) gb<u>é</u>n èbé pen Foc. Ozo use+rV quickly INFL write book
  - c. \*èbé <u>ò</u>ré Òzó yá ùkéké (yá) gb<u>é</u>n!r<u>é</u>n gi<u>é</u>!gi<u>é</u> book Foc. Ozo use pen INFL write+rV quickly

(96a) and (97a) show that it is possible to extract the object of ya and in this respect it behaves as a proper governor like other verbs. However, in the (b) sentences we observe that it is ungrammatical for ya to occur after the I-type adverb and bear the past perfective rV suffix inflection. This is taken as evidence that ya cannot raise to Infl. This is unexpected in a complementation structure (97b) based on the ATTRACT analysis of verb movement (97a):<sup>20</sup>

(97a) ATTRACT

X attracts Y only if Y could check a feature of X, and all Z such that Z could check a feature of X, Y asymmetrically c-commands Z.



Clearly, the closest thing that could check a feature of Tense is the first verb ya, and we know independently that ya can inflect for tense, however as (95b) and (96b) show it cannot raise to support the -rV. Since instrumental constructions are different from true SVCs, and based on the fact that semantically light verbs like auxiliaries are better raisers in French (and possibly English), I propose that the failure of ya to raise to Tense is part of its defective paradigm along the same lines as the analysis of predicate cleft.

<sup>&</sup>lt;sup>20</sup> For ease of illustration, I have base-generated the I-adverb in the head of EP rather than as an adjunct to the head. Under the analysis that I have assumed, generating the I-adverb as head poses a problem for head-movement when V-raising occurs. Thus, this discussion assumes that the I-adverb does not block V-raising since it is an adjunct rather than an actual head.

Turning now to (95c) and (96c), we observe that it is ungrammatical for the second verb to occur after the I-type adverb and bear the -rV suffix. Based on a comparison with (86), the structure of the embedded clause is given in (98):



Under the analysis of verb movement that is based on ATTRACT (97a), the closest thing that can check the relevant feature of Infl is the verb, but as the (c) sentences show, this is not possible. Consequently, I take the failure of the second verb to occur after the I-type adverb and bear the -rV suffix as evidence that it has failed to raise to INFL; this is based on the fact that the -rV cannot be generated in the Infl of the embedded clause (which must be  $\emptyset$  or  $y\acute{a}$ ) and so the ATTRACT analysis holds in a different sort of way here because there is no relevant feature of Infl to be checked by the verb. Therefore, we observe that verb raising to Infl by either of the two verbs in the instrumental construction is ruled out by different interpretations of ATTRACT.

# 7.5 Conclusion

The most obvious point made in this chapter is that some previously analyzed SVCs turn out under careful scrutiny to involve two separate clauses: a main clause and an embedded infinitival clause, which makes them very distinct from true SVCs that have been analyzed as single clauses. It was argued, however, in the case of the modal-aspectual verb construction that these two clauses have been affected by obligatory restructuring that results in what acts like a single clause. In addition, it was argued that one kind of instrumental construction manifests signs of re-analysis where the first verb is underspecified in several ways: it cannot be clefted since it is a light verb that does not introduce a new event quantification, it fails to undergo verb raising which is part of its defective paradigm. However, single event instrumental constructions are typically different from single event resultative SVCs in terms of what licenses predicate cleft and verb movement to Infl; in the former this was derived from the properties of the obligatory infinitival IP complementation structure, while in the latter it was derived from the properties of the co-headed VP structure.

# Chapter Eight SVC Consequences

# 8.1 Introduction

The preceding chapters have been filled with many detailed analyses of verb sequencing constructions with data mainly from Èdó, which was shown to accurately represent a cross-section of languages. The basic goals of the thesis have been the following: (a) to provide clear and systematic tests based on solid empirical evidence that could distinguish the various kinds of SVCs from one another as well as from other constructions that appear on the surface to be sequences of verbs, (b) to establish and formulate concretely the parameter that allows SVCs in some languages but not others. These will be discussed in turn.

#### 8.2 On Resultative vs. Consequential SVCs, and Covert Coordinations

One of the themes of this dissertation has been to define the notion of SVCs--in which two or more verbs with their complements occur in a single clause without any form of coordination or subordination, sharing the same Tense/Aspect, subject, and object. These characteristics were offered as the general definition of SVCs based on observations about interpretation and verb sequencing in section 1.3, as summarized in (1).

(1) In a serial verb construction, the verbs must share external and internal arguments.

(1) provides argument sharing constraints on SVCs which rule out word order patterns such as [NP V1 V2 NP2], where V1 and V2 do not form a compound. It also rules out argument sharing with the object of a preposition (Baker and Stewart 1997b and Collins 1997). In addition, it implies that surface [NP V1 NP1 V2 NP2] is ambiguous between conjunction interpretation and SVC; it can only be the latter if the constraint in (1) hold.

One other implication of (1) is that the surface order of [NP V1 V2] can only be a SVC if the second verb is unaccusative, as in (2):

(2) Òzó khián kpàá Èdó Ozo walk leave 'Ozo left by walking'

Thus, it was derived from (1) that one difference between resultative and consequential SVCs is that the second verb is typically unaccusative in the former while only sequences of transitive verbs are allowed in the latter, as summarized in (3):

(3) Resultative SVCs are those in which the second verb is typically an unaccusative.Consequential SVCs are those in which there are sequences of two transitive verbs.

(3) provides the basis for distinguishing true SVCs from covert coordinations (CCs), which are characterized by two verbs where each has its own overt (thematic) object. Thus, covert coordinations do not involve internal argument sharing and so they are not SVCs. The significance of this is that, although CCs have been recognized in the literature on SVCs (Déchaine 1986, 1993, Manfredi 1991, Baker 1989, 1991, Collins 1997, Campbell 1989 etc.), they have never been carefully distinguished from true SVCs (cf. Stewart 1996)-- a sort of dumping ground for things that did not fit into the mainstream analysis of what are assumed to be SVCs. The distinction between true SVCs and CCs provides useful materials for experiments with sentence processing because speakers parse true SVCs as a single prosodic unit, while CCs are processed as two separate prosodic units that are conjoined.

On the basis of (1), it was proposed, therefore, that there are two kinds of SVCs: <u>resultative</u> and <u>consequential</u>. The intuition behind these labels echoes the observations from the earliest analysis of SVCs (Christaller 1875); resultatives are made up from two verbs in which the first is 'principal', the second is supplemental (a complement), and both

form a VP, while consequentials are those in which two or more verbs express successive precedence-consequence actions. Thus, it was argued in Chapter two that resultatives are constrained to just two verbs which together express an accomplishment being made up of a process-activity first verb and a state or result second verb. Consequential SVCs are less constrained so long as they involve sequences of transitive verbs that can be parsed as one prosodic unit.

The issue that arises from (1) is what is the exact mechanism for the licensing of argument sharing with object and subject in true SVCs? This thesis shows that there is a single functional head, E, that existentially quantifies over the two verbs and this contrasts very sharply with CCs which involve separate quantification with two separate E heads. The event or events expressed by the verbs in SVCs and CCs were shown to derive from the nature of this functional head E. Thus, the single head E in resultative SVCs combines the 'sub-atomic' events of the two verbs which are licensed as one single event. The verbs in consequential SVCs express 'atomic' events which are combined by a single E head to form a complex 'macro event'. Within this 'macro event', the event denoted by the second verb can be independently quantified over by a lower E operator, unlike in the resultative SVC, but this lower E operator is referentially dependent on a higher E which binds the closest event (that of the first verb) and the one expressed by the second verb. Thus, the syntactic generalization from this is that resultative SVCs have a co-headed VP structure with a single E, while consequential SVCs have an adjunction structure with two Es, where the lower E is in the c-command domain of a higher E.

One consequence of this analysis based on the nature of the functional head E is that it correlates with that of subjects. It was shown that in true SVCs there is a single subject (Agent) which sets about the plan of action expressed by the verbs. For example, an ungrammatical reading of a sentence like 'Òzó dé èbé tié' (John bought the book and read it) is 'John went to the store to buy a book for Bill and then after buying it he changed his mind and read the book instead'. The sentence can only have a reading in which the subject must have set out to achieve a plan involving both actions and does them. Thus, subjects in true SVCs are analyzed as being introduced by a single Voice head which hosts the external theta role (CAUSE) feature of the verbs. This contrasts with CCs where there are indications of two subject positions and hence two subjects (one of which is deleted under identity for pragmatic reasons).

On object sharing in SVCs, it was argued that resultative SVCs are characterized by true internal argument sharing which states that there is a single object that is shared by both verbs and there is no empty category as previously claimed in Collins (1997). The concept of true internal argument sharing is consistent with the structure in Baker (1989) but not his generalization since it turns out, in fact, that there is no object sharing, in the sense of Baker, in consequential SVCs. Rather, object sharing involves an empty category. This is essentially the claim made in Collins (1997), which is the same conclusion reached in Baker and Stewart (1997b).

Theoretically, my analysis provides a way to remove the lack of generalization from these other analyses based on the empirically-motivated distinction between resultatives and consequentials. Of immediate relevance here is that we are able to account for the constraints on the second verb in true SVCs: there is a true internal object sharing in the resultative because the unaccusative second verb lacks accusative Case (Burzio 1986) and so cannot assign Case to a null *pro* (Rizzi 1986). On the other hand, an empty category, *pro*, is involved in object sharing in the consequential SVC because the transitive second verb can Case-license the null *pro*. This account is lost under an approach that unifies both constructions as in Baker (1989) and Collins (1997).

Yet another consequence that can be derived from the nature of E heads in true SVCs and CCs is the licensing of predicate clefts which provides further support for the distinction between resultative and consequential SVCs. Predicate clefts are analyzed as being derived from cognate objects which are non-thematic objects (event argument). Thus, based on an account of predicate cleft licensing which involves Spec-head matching in EP at LF, we are able to provide an elegant account for the failure of either of the two verbs to cleft in resultative SVCs, but not in the consequential SVCs, or CCs. It follows that if resultatives have a single E head and multiple Specifiers are not allowed, there cannot be Spec-head matching since one of the cognate event arguments of the raised verbs will not be properly licensed. However, in consequential SVCs where there are two E heads, the predicate cleft of the cognate object of either of the two verbs will be properly licensed in the relevant EP at LF. This is also true for CCs.

However, apart from providing a basis for the distinction between resultative and consequential SVCs, Chapter three discusses some of the cross-linguistic ramifications of predicate clefts on the basis of evidence from Èdó. For example, the Èdó data shows that the clefted copy is a nominal and an XP rather than an X<sup>O</sup> as proposed in Koopman (1984). In addition, the proposal that predicate cleft is derived from cognate objects presents a new way to view the *Se* morpheme in Haitian (Lumsden and Lefebvre 1990, DeGraff 1993, Manfredi 1993 etc.) since the language is said to lack cognate objects (Lefebvre, p.c.). Finally, the discussion of predicate clefts provides a new outlook for the analysis of the reduplicative prefix in Yoruba which I claim involve partial nominalization unlike the complete nominalization in Èdó. The consequence of this is that while predicate cleft asymmetry can be shown clearly between resultative SVCs and consequential SVCs in Èdó, there are cases of overlap in Yoruba which although are not at variance with my analysis but they subsequently could get more refined on the basis of the distinction between resultative and consequential SVCs and the analysis thereof in this thesis.

The discussion of double object constructions in this thesis is a new discovery in the literature on SVCs, and it also clearly illustrates the distinction between resultative and consequential SVCs. The basic fact is that DOCs can occur in consequential but not in resultative SVCs. Again, the contrast is based on differences in lexical and functional structure of the clause. Resultatives by definition are accomplishments which imply a single inner complement position for the delimiter. It was shown that both the V-bar dominating the second verb and the second object of the DOC compete for this inner complement position and so DOCs or double delimiters are not allowed in resultative constructions in general (cf. Hoekstra 1992, Tenny 1987 etc.) On the other hand, consequential SVCs do not have rigid aspectual properties. Consequently, DOCs are allowed since there is no competition for the inner complement position. The theoretical consequence of this is that it shows that analyses such as Baker (1989, 1991) make the wrong claim. For example, Baker would predict that if DOCs are possible the second object should be in the projection of the lower V-bar, but this is contrary to fact because both objects are within the projection of the first verb only and this creates a real split between the verbs unlike the co-headed VP structure assumed by Baker.

A general consequence of the analysis of DOCs is that only the object (theme) can be shared in SVCs and not the derived object. Thus, this provides the window of opportunity to assess the distinction between underlying and derived objects. This distinction and the analysis of DOCs interacts very nicely with the licensing of null *pro* in the consequential SVC. It was argued that the underlying (direct) object is generated in the Specifier of VP, while the derived (indirect) object occupies the Specifier of AspP (Travis 1991), and so based on an ECP account (cf. Koopman and Sportiche 1982), it follows that only the (theme) direct object will be shared but not the indirect object (cf. Collins 1997, Baker and Stewart 1997b). This analysis was also shown to have implications for DOCs in English, Heavy NP shift, as well as locative alternations where there are curious properties associated with moving the derived object. According to my analysis, this stems from an ECP violation, which can be observed in English because it does not have the resumptive pronoun strategy like Èdó or Yoruba.

#### 8.3 On the Serial Verb Construction Parameter

The distinction between resultative and consequential SVCs was shown to be crosslinguistically relevant in that a language such as Igbo which is a neighbor (genetically and

325

geographically) to Èdó and Yoruba has more pronounced V-V compounds and less of what is often controversially classified as SVCs. This puzzle was solved in this thesis by showing that Èdó resultative SVCs consistently show up as (resultative) V-V compounds in Igbo. This difference between SVCs and V-V compounds was derived from the difference in V-to-Infl movement in the two languages: in Igbo V-V compounds, the verbs must obligatorily raise to check a feature of Infl but not in Èdó SVCs. This difference in verb raising comes from the ATTRACT condition given in (4):

#### (4) <u>ATTRACT</u>

X attracts Y only if Y could check a feature of X, and all Z such that Z could check a feature of X, Y asymmetrically c-commands Z

It turns out in Edó that verbs in SVCs cannot bear any morphological inflection which is generalized as the bare stem condition (BSC) stated in (5):

# (5) The Bare Stem Condition (BSC) in SVCs

No verb in the serial verb construction can bear morphological tense inflection.

What Igbo does with (4) and (5) that is not allowed in Edó is that there is obligatory Vincorporation thus creating a single word which is then the closest thing to be attracted and this leads to V-V compounds. However, there is a PF filter that rules out morphological forms like V-V compounds in Edó and so both verbs are likely candidates for ATTRACT. Since there is mutual c-command (cf. Aoun and Sportiche 1983), both verbs fail to be attracted and this results in SVCs. Thus, SVCs occur in languages where a functional head bearing a tense feature fails to attract (being licensed in situ by a Tense copying rule). The SVC parameter is formalized in (6):

#### (6) <u>V-raising serial verb parameter</u>

A verb serializing language is one in which Tense (or other Infl type categories) does not need to be checked.

This analysis extends to resultative V-V compounds in Chinese where it was shown that resultative notions translate as V-V compounds but not putative SVCs, like in Igbo. Thus, based on the fact that the two verbs must bear the aspect maker <u>-le</u>, it was proposed that verb raising to the functional head bearing <u>-le</u> is obligatory, on a par with Igbo, and this results in surface V-V compounds.

Furthermore, the serial verb parameter also accounts for why English does not have SVCs. The discussion focused on the AP vs. VP contrast based on striking similarities between AP and VP resultatives, with both occurring in Èdó while VP resultatives are ungrammatical in English (hence not an SVC language). This contrast is also partially based on the fact that a consequential SVC, with its intricate interpretation and structure, lacks any real analogue in English contra Larson (1991).

Now, in the AP resultative in English and Èdó, Tense sees only the verb because the adjective cannot check a feature of Infl and so there is only a Y and this gets attracted at the relevant level in the two languages. Since it is standardly assumed that Verb raising takes place at LF in English, the question is why can't it have a VP resultative. The answer is that it is because the resulting structure with two verbs in the resultative construction is ruled out at LF by the same S-structure condition in Èdó. This is based on the fact that both verbs will mutually c-command each other and so none of them would count as the closest thing to be attracted and so the derivation would crash. Thus, English cannot have SVCs even at LF. Based on Pollock (1989), French would be like Igbo where Tense has strong V-feature which can be checked by the verb, and, therefore by assuming that French does not allow Igbo type V-V compounds (even at LF), it follows that verbs must raise obligatorily in French due to the strength of Tense and so Verb Serialization also fails to occur.

#### 8.3.1 True SVCs vs. other Surface Verb Constructions

The other way to illustrate the serial verb parameter is based on the analysis of other constructions which have been previously analyzed as SVCs. Now, this is where English and French are typically different since there are clear indicators of clausal embedding. In many of these SVC languages there is very little morphology to show clausal embedding; however, the serial verb parameter illustrates the difference clearly. The basic assumption is that true SVCs have a single Tense projection while clausal coordinations have two. It turns out that either of the verb in CCs can undergo Verb raising thus confirming that they are not SVCs. Furthermore, it was shown that what were previously classified as manner SVCs (Oyelaran 1982, George 1975, 1976) are constructions involving two clauses that have undergone restructuring and even then the Verb raising parameter still applies. This discussion has important empirical and theoretical consequences. On the empirical level, the contrast between modal-aspectual constructions serves to illustrate the definition of true SVCs as we observe point by point differences in the syntactic tests that were developed in this thesis.

On a theoretical level, the idea of a restructuring analysis for so-called SVCs (Modal-aspectual verb construction) is interesting and the fact that restructuring is obligatory in the Modal-aspectual verb construction compared with the Romance and Germanic cases where it is assumed to be optional opens up a new area of comparative research. Finally, the thesis recognizes two kinds of instrumental constructions and shows that they also involve two separate clauses and so are not true SVCs as has been previously claimed. One particular kind was focused on, and it was shown that the first verb is a defective verb that obligatorily takes an IP complement. The interesting consequence of this analysis is that it provides an avenue for making the cross-linguistic proposal for Èdó and English about the use-type of instrumental construction: it is not only different from SVCs but also must be distinguished from a purpose clause. The conclusion from this is that,

328

since no one has ever analyzed the English sentence 'John used the knife to cut the meat' as being a SVC, then instrumental constructions cannot be SVC.

# 8.4 Conclusion

This thesis has shown that the core of SVCs is much more restricted than has been generally assumed; resultative and consequential. It is hoped that the cross-linguistic success illustrating the empirical and theoretical results from this thesis will launch a new approach to the analysis of SVCs.

#### References

- Agheyisi, Rebecca. 1986a. Verb serialization and lexical re-analysis: The case of compound verbs in Edó. Studies in African Linguistics Volume 17, #13
- Agheyisi, Rebecca. 1986b. Edó-English Dictionary. Ethiope publishers. Benin City
- Agheyisi, Rebecca. 1990. A grammar of Èdó. ms.
- Aikhionbare, Osayaba Matt. 1988. The verbal suffixes in Edó. Afrika und Ubersee, Band 71, 205-228
- Aikhionbare, Osayaba Matt. 1989. Defining the domain of nasality in Èdó. Studies in African Linguistics, Vol. 20 #3, 301-315
- Amayo, Airen Moses. 1975. The structure of verbal construction in Edó (Bini). Journal of West African Languages, 10 #1, 5-27.
- Amayo, Airen Moses. 1976. A generative phonology of Èdó (Bini). University of Ibadan doctoral dissertation, Ibadan.
- Ameka, Felix. 1992. Focus constructions in Ewe and Akan: A comparative perspective. MIT Working Papers in Linguistics, 17: 1-26. ed. by Chris Collins and Victor Manfredi.
- Ansre, Gilbert. 1966. The verbid-A caveat to "serial verbs." Journal of West African Languages, 3, #1: 29-32
- Aoun, Joseph. and Dominique Sportiche. 1983. On the formal theory of government. Linguistic Review 2, 211-36.
- Arnold, Mark. D. 1998. Indirect grammatical pressure driving language change: Evidence from the loss of V-raising in English. paper presented at Berkeley Linguistics Society, U.C. Berkeley, February 1998
- Awobuluyi, Oladele. 1969. Splitting verbs in Yorùba. Actes du 8<sup>e</sup> congrès de la société linguistique de l'Afrique Occidental 1: 151-64. Université d'Abidjan.
- Awobuluyi, Oladele. 1973. The modifying serial construction: A critique. Studies in African linguistics 4: 87-111.
- Awoyale, Yiwola. 1987. Perspectives on verb serialization. In Niger-Congo Syntax and Semantics 1, ed. Victor Manfredi, 3-36. Papers from the April 1987 Workshop, Boston University.
- Awoyale, Yiwola. 1988. Complex predicates and verb serialization. Lexicon Project Working Papers 28. MITWPL, Dept. of Linguistics and Philosophy MIT, Cambridge, Mass.
- Bailey, Beryl. Loftman. 1966. Jamaican Creole syntax. London, Cambridge University Press.

- Baker, Mark. 1988. Incorporation: a theory of grammatical function changing. Chicago: University of Chicago Press.
- Baker, Mark. 1989. Object sharing and projection in serial verb construction. Linguistic Inquiry 20, #4: 513-553.
- Baker, Mark. 1991. On the relation of serialization to verb extensions. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. Claire Lefebvre, 79-102. Amsterdam: John Benjamins.
- Baker, Mark. 1992. Why unaccusative verbs cannot dative-shift. NELS, University of Massuchusetts-Amherst, University of Ottawa. 1992.
- Baker, Mark. 1996a. The polysynthesis parameter . New York: Oxford University Press.
- Baker, Mark. 1996b. Unaccusativity and the adjective/verbs distinction: English and Mohawk evidence. ESCOL eds. Anthony Green and Virginia Motapanyane, p 12-35, Cornell University, Ithaca: CLC Publications.
- Baker, Mark. 1997a. Thematic roles and syntactic structure. In *Elements of grammar*, ed. by Liliane Haegerman, 73-137. Dordrecht: Kluwer.
- Baker, Mark. 1997b. On the distinction between adjectives and verbs. ms. McGill University.
- Baker, Mark, and Osamuyimen T. Stewart. 1997a. Unaccusativity and the Adjective/Verb Distinction: Edó Evidence. In *Proceedings of NELS* 27, 33-47 GLSA, McGill University.
- Baker, Mark, and Osamuyimen T. Stewart. 1997b. Empty objects: serialization, donkey anaphora, and extraction in Edó. ms. McGill University. Presented at Cornell University November 6, 1997.
- Baltin, Mark. 1995. Floating quantifiers, PRO, and predication. Linguistic Inquiry 26 #2 p,199-248.
- Barss, Andrew, and Howard Lasnik. 1986. A note on anaphora and double objects. Linguistic inquiry 17, 347-354.
- Bamgbose, Ayo. 1972. ed. The Yorùba verb phrase . Ibadan: Ibadan University Press.
- Bamgbose, Ayo. 1973. The modifying serial construction: a reply. Studies in African Linguistics 4, #2, 207-217.
- Bamgbose, Ayo. 1974. On Serial Verbs and Verbal Status. Journal of West African Languages 9, 17-48.
- Bamgbose, Ayo. 1986. Negation and serial verb construction types in Yoruba, In Current Approaches to African Linguistics, ed. Dimmendaal, 31-40.
- Bennett, P.C. and J.P. Sterk. 1977. South-central Niger-Congo: a reclassification. Studies in African Linguistics 8, 241-73.

- Bendix, Edward. 1972. Serial verbs in the Caribbean and West Africa: their semantic analysis in Papiamento. Hunter College of the City University of New York, mimeo.
- Boadi, L.A. 1968. Notes on the comparative construction in Twi. In Papers from the 5th Conference on African Linguistics.
- Bobaljik, Jonathan. David. 1997. One implication of PF-adjacency: Two kinds of "LF" movement. ms. McGill/Harvard.
- Bodomo, Adams. 1993. Complex predicates and event structure: An intergrated Analysis of Serial Verb Constructions in the Mabia Languages of West Africa. In University of Trondheim Working papers in Linguistics 20. Dept. of Linguistics, University of Trondheim, Norway.
- Bolinger, D. 1971. The phrasal verb in English. Cambridge, Mass.: Harvard University Press.
- Borer, Hagit. 1983. Parametric Syntax. Dordrecht. Foris
- Bowers, John. 1993. The syntax of predication. Linguistic inquiry 24, 591-656
- Burzio, Luigi. 1981. Intransitive verbs and Italian auxiliaries. MIT doctoral dissertation, Cambridge, MA.
- Burzio, Luigi. 1986. Italian syntax: A government-binding approach. Dordrecht: Reidel
- Campbell, Richard. 1989. The grammatical structure of verbal predicates. Ph.D dissertation, UCLA, Los Angeles, California.
- Campbell, Richard. 1992. Serial verbs and Unaccusativity. In MIT Working papers in Linguistics; Proceedings of the Kwa comparative syntax workshop 17, 41-52, ed. by Chris Collins and Victor Manfredi
- Campbell, Richard. 1996. Serial verbs and shared arguments. The Linguistic Review 13, 83-118
- Carrier, J. and J.H. Randall. 1992. The argument structure and syntactic structure of resultatives. *Linguistic Inquiry*, 23, 173-234.
- Carstens, Vicki. 1988. Serial verbs in Yorùba. Paper presented at the 2nd workshop in Niger-Congo syntax and semantics. MIT, Cambridge, Mass.
- Choe, Hyun-Sook. 1988. Restructuring parameters and complex predicates: A transformational approach. MIT doctoral dissertation
- Chomsky, Noam. 1957. Syntactic structures . The Hague: Mouton
- Chomsky, Noam. 1965. Aspects for the theory of syntax . Cambridge, Mass., MIT Press
- Chomsky, Noam. 1977. On wh-movement. In *Formal syntax*, ed. by Peter Culicover, T.Wasow, and Adrian Akmajian, 71-132. New York: Academic Press.

Chomsky, Noam. 1981. Lectures on government and binding. Dordrecht: Foris.

- Chomsky, Noam. 1986. Knowledge of language: Its nature, origin, and use . New York: Praeger.
- Chomsky, Noam. 1993. A minimalist program for linguistic theory. The view from Building 20, ed. by Kenneth Hale and Samuel Jay Keyser, 1-52. Cambridge, Mass.: MIT Press.

Chomsky, Noam. 1995. The Minimalist Program . Cambridge, Mass.: MIT Press.

- Chomsky, Noam, and Howard Lasnik. 1993. The theory of principles and parameters. In Syntax: An international handbook of contemporary research, ed. by Joachim Jacobs, Amin von Stechow, Wolfgang Sternefeld, and Theo Vennemann, 506-569. Berlin: Walter de Gruyter.
- Christaller, Rev. J.G 1875. A grammar of the Asante and Fante language called Tshi. Basel. Republished 1964 by Gregg, Ridgewood.
- Cinque, Guglielimo. 1997. Adverbs and functional heads: a cross-linguistic perspective. New York: Oxford
- Clements, Nick. G.1984. Binding domains in Kikuyu. Studies in the Linguistic Sciences 14, 37-56.
- Collins, Chris. 1994. The factive construction in Kwa. Travaux de recherche sur le créole haitien. No. 23, 31-65.
- Collins, Chris. 1995. Serial verb constructions and the theory of multiple feature checking. ms., Cornell University, Ithaca, N.Y.
- Collins, Chris. 1997. Argument sharing in serial verb construction. *Linguistic inquiry*, 28: 461-497.
- Collins, Chris, and Hoskuldur Thráinsson. 1993. Object shift in double object constructions and the theory of Case. ms. Harvard/University of Iceland.
- Comrie, Bernard. 1976. Aspect. Cambridge University Press.
- Cormack, Annabel and Smith Neil. 1994. Serial verbs. UCL Working papers in Linguistics . 6, 63-88.
- Craig, C. and Kenneth Hale. 1988. Verb sequencing constructions in Maccro-Chibchan. paper presented at 2nd Niger-Congo Syntax and Semantics Workshop, MIT, Cambridge, Massachusetts, April 1988.
- Davidson, David. 1967. The logical form of action sentences. In *The logic of decision and action*, ed. by N. Rescher. Pittsburgh.
- Déchaine, Rose-Marie. 1986. Opérations sur les structures d'argument: Le cas des constructions sérielles en haitien. M.A thesis, Université du Québec à Montréal.
- Déchaine, Rose-Marie. 1992. Inflection in Ìgbo and Yorùba. MIT Working Papers in Linguistics, 17: 95-120. ed. by Chris Collins and Victor Manfredi.

- Déchaine, Rose-Marie.1993. Predicates Across Categories: Towards a Category-Neutral Syntax. Ph.D Dissertation, University of Massachusetts, Amherst.
- DeGraff, Michel Anne Frederic. 1993. Predicate-movement, quantification, events, and properties. In Niger-Congo syntax and semantics No 6. papers from the Workshop at Boston University, ed. by Victor Manfredi and Karl Reynolds. April 1993.
- Dekydtspotter, Laurent.P. 1992. The syntax of predicate clefts. MIT Working Papers in Linguistics, 17: 121-138. ed. by Chris Collins and Victor Manfredi.
- Den Dikken, Marcel. 1995. Particles . Oxford. Oxford University Press.
- Egharevba, Jacob. 1968. A short history of Benin . Ibadan: Ibadan University Press.
- Ejele, Philomena. 1986. Transitivity, Tense and Aspect in Ésán (Ishan). University College London doctoral dissertation, London.
- Ekundayo, S.A & F.N. Akinnaso. 1983. Yoruba Serial Verb String Commutability Constraints. *Lingua* 60, 115-133.
- Elugbe, Ben Ohi. 1976. Noun class Vestiges in Degema. Afrika und Ubersee, Band LIX, 224-33.
- Elugbe, Ben.Ohi and Kay.Williamson.1977. Reconstructing Nasals in Proto Benue-Kwa. In Linguistic Studies offered to Joseph Greenberg, (ed) A. Julliard, 339-63. Anna Libri, Saratoga, California.
- Elugbe, Ben. Ohi. 1979. Some tentative historical inferences from comparative Edoid studies. Kiabara 2.2, 82-101
- Eménanjo, Emmanuel Nolue. 1981. Auxiliaries in Ìgbo syntax. Ph.D dissertation, University of Ibadan. Bloomington, Ind.: Indiana University Linguistics Club 1984.
- Evers, Arnold. 1975. The transformational cycle of Dutch and German. Ph.D dissertation, University of Utrecht.
- Evers, Arnold. 1982. Restructuring versus reindexing. Utrecht Working papers in Linguistics 11, 1-36
- Ézikèójiaku, P.A. 1979. Tense or aspect in Igbo: a preliminary study. Kiabàra [Port Harcourt] 2.2: 113-126
- Foley, William A. and Mike Olson. 1985. Clausehood and verb serialization. In Grammar inside and outside the clause, ed. Johanna Nichols and Anthony Woodbury, 17-60, Cambridge University Press.
- George, Isaac Madugu. 1975. Typology of verb serialization. Journal of West African Languages 9, 78-97.
- George, Isaac Madugu. 1976. Verb serialization and lexical decomposition," Studies in African Linguistics, supplement 6, 63-72

- Georgopoulos, Carol. 1985. Variables in Palauan syntax. Natural Language and Linguistic Theory 3, 59-94
- Georgopoulos, Carol. 1991. Syntactic variables: resumptive pronouns and A'-binding in Palauan. Dordrecht, Kluwer Academic publishers.
- Givón, Talmy. 1991. Some substantive issues concerning verb serialization: Grammatical vs. cognitive packaging. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. Claire Lefebvre, 169-183. Amsterdam: John Benjamins.
- Greenberg, Joseph. 1963. The languages of Africa. International Journal of African Languages 11:1, 45-61
- Grimshaw, Jane. 1990. Argument structure . Cambridge, MA.: MIT Press.
- Gruber, Jeffrey. 1965. Studies in lexical relations. MIT doctoral dissertation, Cambridge
- Gruber, Jeffrey. 1992a. Thematic Configurational Constraints in Serial Verb Constructions. Paper presented at ACAL, Michigan
- Gruber, Jeffrey. 1992b. Proper Argument Projection in Igbo and Yoruba. MITWPL 17, 139-64.
- Gruber, Jeffrey, and Chris Collins. 1996. Argument projection, thematic configurationality and Case theory. In *Projections and interface conditions: Essays on Modularity*, ed. by Anne-Marie Di Sciullo. Oxford: Oxford University Press.
- Haegeman, Liliane, and Henk van Riemsdijk. 1986. Verb projection raising, scope, and the typology of rules affecting verbs. *Linguistic inquiry* 17, 417-466.
- Haik, Isabelle, Hilda Koopman, and Dominique Sportiche. 1985. 'INFL en mooré et le liage dans le système A'. In Rapport de recherches du groupe de linguistique africaniste, anée 1985-1986, Montréal.
- Haik, Isabelle. 1990. Anaphoric, pronominal and referential Infl. Natural Language and Linguistic Theory 8, 347-374.
- Hale, Kenneth. 1991. Misumalpan verb sequencing constructions. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. Claire Lefebvre, 1-36. Amsterdam: John Benjamins.
- Hale, Kenneth, and Samuel Jay Keyser. 1987a. A view from the middle. Lexicon project Working papers No 10. Cambridge, Mass.: MIT Center for Cognitive Science.
- Hale, Kenneth, and Samuel Jay Keyser. 1988. Explaining and constraining the English middle. Ms. MIT, Cambridge, MA
- Heim, I. 1982. The semantics of definite and indefinite noun phrases. UMass at Amherst doctoral dissertation.

Higginbotham, James. 1985. On semantics. Linguistic inquiry 16, 547-593.

Hoekstra, Teun. 1992. Aspect and theta theory. *Thematic structure and its role in grammar*, ed. by I.M Roca, 145-174. Berlin: Foris.

- Hutchinson, J.P. 1989. Copy nominals and the interpretation of predicate clefts in Haitian. Paper presented at the 3rd Niger-Congo syntax and semantics Workshop. MIT, I.A.P., 23 January.
- Hyman, Larry. 1996. The syntax of body parts in Haya. The grammar of inalienability, ed. H Chappel and W. McGregor. 870-890, Berlin: Mouton de Gruyter.
- Hyman, Larry.M. 1971. Consecutivization in Fe?fe?. Journal of African Languages 10: 29-43
- Igbafe, Phillip. 1979. Benin under British administration: The impact of colonial rule on an African Kingdom 1897-1938, Longman Group Limited, London
- Ihionu, Peter. 1992. Verb compounding in Igbo: An overview. MIT Working Papers in Linguistics, 17: 165-182. ed. by Chris Collins and Victor Manfredi.
- Jackendoff, Ray. 1990. Semantic Structures. MIT Press, Cambridge, MA.
- Jaeggli, Osvaldo, and Nina Hyams. 1993. On the independence and interdependence of syntactic and morphological properties: English aspectual *come* and *go*. *Natural Language and Linguistic Theory* 11, 313-346.
- Jansen, Bert, Hilda Koopman, and Pieter Muysken. 1978. Serial verbs in the Creole Languages. University of Amsterdam Instituut voor Algemene Taalwetenshap, Amsterdam Creole Studies II
- Jones, Charles. 1991. Purpose clauses . Dordrecht: Kluwer Academic Press.
- Joseph, Brian, D. and Arnold M. Zwicky. 1990. (eds) When verb collide: Papers from the 1990 Ohio State mini-conference on serial verbs. Department of Linguistics, Ohio State University, Columbus, Ohio.
- Kayne, Richard. 1984. Connectedness and binary branching. Dordrecht: Foris.
- Kayne, Richard. 1989. Null subjects and Clitic Climbing. In The null subject parameter. eds Osvaldo Jaeggli and Ken Safir. Dordrecht: Academic Press.
- Kayne, Richard. 1991. Romance clitics, verb movement, and PRO. Linguistic inquiry 22, 647-686.
- Kayne, Richard. 1994. The antisymmetry of syntax .Cambridge, Mass.: MIT Press.
- Kratzer, Angelika. 1996. Severing the external argument from its verb. *Phrase structure* and the lexicon. ed. by Johan Rooryck and Laurie Zaring, 109-138. Dordrecht: Kluwer
- Koizumi, Masatoshi. 1993. Object agreement phrases and the split VP hypothesis. MIT Working papers in linguistics 18, 99-148. Dept. of Linguistics and Philosophy, MIT, Cambridge, Mass.
- Koopman, Hilda. 1984. The syntax of verbs: From verb movement in Kru languages to Universal grammar. Dordrecht: Foris.

- Koopman, Hilda. 1992. On the absence of Case chains in Bambara. Natural Language and Linguistic Theory 10, 555-594.
- Koopman, Hilda, and Dominique Sportiche. 1982. Variables and the bijection principle. The Linguistic Review 2 (2), 139-160.
- Koopman, Hilda, and Dominique Sportiche. 1986. A note on long extraction in Vata and the ECP. Natural Language and Linguistic Theory 4, 357-474.
- Larson, Richard. 1988. On the Double Object Construction Linguistic Inquiry 19, 335-392
- Larson, Richard. 1991. Some issues in verb serialization. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. Claire Lefebvre, 185-210. Amsterdam: John Benjamins.
- Larson, Richard and Claire Lefebvre. 1991. Predicate clefting in Haitian Creole. Proceedings of NELS 21 ed. by Tim Sherer, 247-262. GLSA
- Laenzlinger, C. 1994. Principles for a formal account of adverb syntax. In Gen GenP 1,# 2, 47-76.
- Law, Paul, and Tonjes Veenstra. 1992. On empty operators in serial verb constructions. In MIT Working papers in linguistics 17: proceedings of the Kwa comparative syntax workshop, 183-203. Dept. of Linguistics and Philosophy, MIT, Cambridge, Mass.
- Lee, Kong-Bae. 1976. Notes on pronouns, reflexives, and pronominalization. Ehak Yunkwu (linguistic Research) 12.2.
- Lee, S. 1993. The syntax of serialization in Korean. Japanese/Korean Linguistics 2, ed. Patricia Clancy. CLSI, Stanford, CA, 447-463
- Lefebvre, Claire. 1990. On the interpretation of predicate cleft. paper presented at the Conference on Focus and Grammatical Relations in Creole languages, University of Chicago. May 1990.
- Lefebvre, Claire. 1991. Take serial verb constructions in Fon. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. by Claire Lefebvre, 37-78. Amsterdam: John Benjamins.
- Lefebvre, Claire. 1994. On spelling out E. In *Travaux de recherche sur le créole haitien No* 23 : 1-29. Université du Québec à Montréal.
- Levin, Beth, and Malka Rappaport-Hovav. 1995. Unaccusativity: At the syntax-semantics interface. Cambridge, Mass.: MIT Press.
- Li, Charles, and Sandra Thompson. 1973. Serial verb constructions in Mandarin Chinese: Subordination or coordination? CLS comparative syntax festival.
- Li, Yafei. 1990. On V-V compounds in Chinese. Natural Language and Linguistic Theory 8, 177-207.

- Li, Yafei. 1991. On Deriving Serial Verb Constructions. In Serial verbs: Grammatical, comparative, and cognitive approaches, ed. by Claire Lefebvre, 103-136. Amsterdam: John Benjamins.
- Lord, Carol. 1973. Serial Verbs in Transition. Studies in African Linguistics 4, 269-295.
- Lord, Carol. 1975. Igbo verb compounds and the lexicon. Studies in African Linguistics 6 .1: 23-48
- Lord, Carol. 1995. Historical change in serial verb constructions. Amsterdam: John Benjamins.
- Lu, John H-T. 1977. Resultative verb compound versus directional verb compounds in Mandarin. Journal of Chinese Linguistics, 5: 2, 276-313
- Lumsden, John, and Claire Lefebvre. 1990. Predicate cleft constructions and why they aren't what you might think they are. *Linguistics* 28.4:761-782.
- Manfredi, Victor. 1991. Ágbò and Éhugbò: Ìgbo linguistic consciousness, its origins and limits. Ph.D dissertation. Harvard University.
- Manfredi, Victor. 1993. Verb focus in the typology of Kwa/Kru and Haitien. In Focus and grammatical relations, ed. by Francis Byrne and D.Winford, 3-51. Amsterdam: John Benjamins.
- Manfredi, Victor. 1997. Aspectual shift in Benue-Kwa. In Objects in Benue-Kwa, ed. by Rose-Marie Déchaine and Victor Manfredi, 87-122, The Hague: Graphics Academic Press.
- Manfredi, Victor, and Yetunde Laniran. 1988. Extractions from Yorùba serializations. 2nd Niger-Congo syntax and semantics Workshop, MIT lexicon project. April 11.
- Massam, Diane. 1990. Cognate objects as thematic objects. Canadian Journal of Linguistics 35.2: 161-190.
- May, R. 1985. Logical Form : Its structure and derivation . Cambridge, MA, MIT Press.
- Melzian, Hans. 1937. Concise Dictionary of the Bini Language of Southern Nigeria. Kegan Paul, London.
- Mufwene, Salikoko. 1990. Serialization and subordination in Gullah: Towards a definition of serialization. In When verbs collide: papers from the (1990) Ohio miniconference on serial verbs. eds. Brian D.Joseph and Arnold M. Zwicky, 91-108, The Ohio State University, Columbus, Ohio.
- Muysken, Pieter. 1988. Parameters for Serial Verbs. ms., University of Amsterdam, paper presented at the 1st Niger-Congo Syntax and Semantic workshop, Boston University.
- Napoli, Donna. 1981. Semantic interpretation vs. lexical governance: Clitic climbing in Italian. Language 57, 841-887.
- Nishiyama, Kunio. 1995. Serialization across languages. paper presented at 26th Annual Conference on African Linguistics, Cornell University, Ithaca.

- Omoregie, Osaren S.B. 1983. Èdó grammar for Schools and Colleges. Lagos: Thomas Nelson Limited.
- Omoruyi, Thomas. 1986a. Adjectives and adjectivization processes in Èdó. Studies in African Linguistics 17, 283-301.
- Omoruyi, Thomas. 1986b. Pronominalization and reflexivization in Edó. Afrika und Ubersee, Band 69/1, 81-93
- Omoruyi, Thomas. 1986c. Pluralization strategies in Edó. Journal of West African Languages 16.2, 61-75
- Omoruyi, Thomas. 1987. The concept of definiteness in Èdó. Afrika und Ubersee, Band 70, 211-223
- Omoruyi, Thomas. 1988. On the formation of questions in Edó. Journal of African Languages, 10, 19-31
- Omoruyi, Thomas. 1989. Focus and question formation in Edó. Studies in African Linguistics, Vol. 20, #3, 279-299
- Omoruyi, Thomas. 1991. Tense, aspect and modality in Edó. Afrika und Ubersee, Band 74/1, 1-19
- Oyelaran, Olasope. 1982. On the Scope of the Serial Verb Construction in Yoruba. Studies in African Linguistics 13, 109-146.
- Pandharipande, Rajeshwari.1990. Serial verbs in Marathi. In When verbs collide: papers from the (1990) Ohio mini-conference on serial verbs. eds. Brian D.Joseph and Arnold M. Zwicky, 178-199, The Ohio State University, Columbus, Ohio.
- Parsons, Terence. 1990. Events in the semantics of English: A study in subatomic semantics. Cambridge, Mass.: MIT Press.
- Pesetsky, David. 1989. Language-particular processes and the earliness principle. ms. MIT.
- Pi, Chia-Yi. Tony and Osamuyimen. T. Stewart. 1998. Micro events in two serial verb constructions. (to appear) In *Proceedings of Semantics and Linguistics Theory* (SALT VIII), Cornell, CLC publications.
- Piou, Nanie. 1982. Le clivage du prédicat. In syntax de l'haitien, ed. by Claire Lefebvre, Hélèn Magloire-Holly, and Nanie Piou, 122-151. Ann Arbor, Michigan: Karoma Press
- Pollock, Jean-Yves. 1989. Verb movement, universal grammar, and the structure of IP. Linguistic inquiry 20, 365-424.

Pustejovsky, J. 1991. The syntax of event structure. Cognition 41, 47-81.

Rappaport, Malka, and Beth Levin. 1985. A case study of lexical analysis: the locative alternation. ms., MIT.

Rapoport, Tova.R. 1990. Stage and adjunct predicates. ms. MIT.

- Rapoport, Tova.R. 1991. Adjunct-predicate licensing and d-structure. In Syntax and Semantics: Perspectives on phrase structure . 25, 159-187. Academic Press.
- Reinhart, Tanya. 1976. The syntactic domain of anaphora. Ph.D dissertation, MIT, Cambrdige, Mass.
- Rizzi, Luigi. 1978. A restructuring rule in Italian syntax. In Samuel Jay Keyser ed. Recent Transformational studies in European languages. Linguistic Inquiry Monograph 3, 113-159
- Rizzi, Luigi. 1982. Issues in Italian syntax. Dordrecht: Foris.
- Rizzi, Luigi. 1986. Null objects in Italian and the theory of pro . Linguistic inquiry 17, 501-557.
- Roberts, Ian. 1985a. Agreement parameters and the development of English modal auxiliaries. Natural Language and Linguistic Theory 3, 21-58.
- Roberts, Ian. 1985b. Serial verbs and government and binding theory. Studies in African Linguistics, supplement 9, 262-268
- Roberts, Ian. 1993. Verbs and diachronic syntax: a comparative history of English and French. Boston: Kluwer.
- Ross, John. 1967. Constraints on variables in syntax. Ph.D dissertation, MIT. Cambridge, Mass. published in 1986 as *Infinite syntax*. Norwood, N.J.: Ablex.
- Rosen, Sara Thomas. 1990. Restructuring verbs are light verbs. Proceedings of WCCFL 9, 477-491
- Rothstein, Susan. 1983. The syntactic forms of predication. Ph.D dissertation, MIT. Cambridge, Mass.
- Ritter, Betsy, and Sara Rosen. 1993. Deriving causation. Natural Language and Linguistic Theory, 11 #3, 519-555.
- Saah, Kofi and Ejike, B. Eze. 1997. Double objects in Akan and Igbo. In Objects in Benue-Kwa, ed. by Rose-Marie Déchaine and Victor Manfredi, 139-151, The Hague: Ghraphics Academic Press.
- Sebba, Mark. 1987. The syntax of serial verbs . Amsterdam: John Benjamins.
- Schachter, Paul. 1974a. A non-transformational account of serial verbs," Studies in African Linguistics, supplement 5, 253-270
- Schachter, Paul. 1974b. Serial verbs as verbs: A reply to a reply. Studies in African Linguistics supplement 5, 278-282.
- Schaefer, Ron. 1997. On the properties of Emai possessor promotion. paper presented at External possession conference, University of Oregon, 7-9 September.

- Schiller, E. 1991. An autolexical account of subordinating serial verb constructions. University of Chicago dissertation.
- Simpson, Jane. 1983. Resultatives. *Papers in Lexical-Functional Grammar*, ed. by Lori Lori Levin, Malka Rappaport, and Anne Zaenen, 143-157. Bloomington, Ind.: Indiana University Linguistics Club.
- Sportiche, Dominique. 1988 A theory of floating quantifiers and its corollaries for constituent structure. *Linguistic inquiry* 19, 425-450.
- Stahlke, H. 1970. Serial verbs. Studies in African Linguistics 1:1, 60-99
- Stahlke, H. 1974. Serial verbs as adverbs: A reply to Paul Schachter. Studies in African Linguistics supplement 5, 271-277.
- Stewart, John. M. 1963. Some restrictions on objects in Twi. Journal of African Languages 2, #2, 145-149
- Stewart, Osamuyimen. T. 1997. Object agreement and the serial verb construction: Some minimalist considerations. In *Objects in Benue-Kwa*, ed. by Rose-Marie Déchaine and Victor Manfredi, 153-168 The Hague: Graphics Academic Press.
- Stewart, Osamuyimen. T. 1996. Adverb placement and the structure of the serial verb construction. *Proceedings of NELS 26* 409-423, MIT/Harvard GLSA.
- Stewart, Osamuyimen. T. 1992. An assessment of a centenary of Edó linguistic evolution. Paper presented at the 1st International Conference on Edó (Benin) Studies: University of Benin, Benin City, March 20-23 1992.
- Thompson, Sandra A. 1973. Resultative verb compounds in Mandarin Chinese: A case for lexical rules. Language 49, 361-379.
- Tellier, Christine. 1986. Restructuring and complement order in Abe infinitives. In Current approaches to African linguistics, ed. by David Odden, 369-381. Dordrecht: Foris.
- Tenny, Carol. 1987. Grammaticalizing aspect and affectedness. Ph.D dissertation, MIT. Cambridge: Mass.
- Tenny, Carol. 1994. Aspectual roles and the syntax-semantics interface. Dordrecht: Kluwer
- Travis, Lisa deMena. 1984. Parameters and effects of word order variation. Ph.D dissertation, MIT, Cambridge, Mass.
- Travis, Lisa deMena. 1988. 'The syntax of Adverbs' McGill Working Papers in Linguistics: Special Issue on Comparative Germanic Syntax
- Travis, Lisa deMena. 1991. Derived Objects, Inner Aspect, and the Structure of VP. Paper presented at NELS 22, University of Delaware
- Travis, Lisa deMena. 1990. Parameters of Phrase Structure. In Alternative conceptions of phrase structure . ed. by M.Baltin and A.Kroch, Chicago: University of Chicago

- Travis, Lisa deMena. 1994. Event phrase and a theory of functional categories. Proceedings of CLA, 559-570.
- Travis, Lisa deMena. forthcoming. Inner aspect.. Dordrecht: Kluwer
- Tuller, L. 1985. Tense features and operators in Hausa. In Rapport de recherches du groupe de linguistique africaniste, année 1985-1986, Montréal, 493-516
- Uesaka, Miwako. 1996. The *te-i-ru* construction in Japanese: interaction between aspect and syntax. MA thesis, McGill University
- United Bible Societies. 1980. *İyèn nòmàá óghé èmwán hìá, ílé nó!gbón*. The New Testament in Èdó. The Bible Society of Nigeria, Lagos
- Welmers, William E. 1973. African Language Structure. Berkeley: University of California Press.
- Wescott, Roger. W. 1963. A Bini Grammar, Part 3: Leximics . Research Program in African Languages, New Haven College, West Haven, Conn.
- Westermann, Diedrich. 1930. A study of the Ewe language. London, Oxford University Press
- Williams, Edwin. 1980. Predication. Linguistic inquiry 11, 203-238
- Williams, Wayne R. 1976. Linguistic change in the syntax and semantics of sierra Leone Krio. Indiana University D.Phil, Xerox # 77-1948.
- Williamson, Kay. 1963. The syntax of verbs of motion in Ijo. Journal of African Languages . 2.2, 150-154.
- Williamson, Kay. 1965. A grammar of the Kolokuma dialect of Ijo. West African Language Monographs 2: Cambridge University press.
- Wingerd, J. 1977. Serial verb in Haitian creole. In Language and Linguistic Problems in Africa. (eds) P.F.A. Kotey and H. Der-Housikian. Hornbeam Press Inc.
- Wu, Ching-Huei Teresa. Forthcoming. Serial verb constructions and verbal compounding. paper presented at the conference On the formal way to Chinese languages at U.C. Irvine. December 1997.
- Wurmbrand, Susi. 1997. Restructuring Infinitives. ConSOLE 5, 277-292.
- Yang, Dong-Whee. 1976. Korean Negation Revisted. Ene (Language) 1.1
- Zushi, Mihoko. 1995. Long distance dependencies. Doctoral dissertation, McGill University.